## Tel Aviv University

Raymond and Beverly Sackler Faculty of Exact Sciences

The Blavatnik School of Computer Science

# Semantic Investigation of Proof Systems for Non-classical Logics

Thesis submitted for the degree of Doctor of Philosophy

by

# Ori Lahav

This work was carried out under the supervision of **Prof. Arnon Avron** 

Submitted to the Senate of Tel Aviv University February 2014

## Acknowledgements

First and foremost, I would like to express my sincere gratitude to my advisor Prof. Arnon Avron for his continuous support during this research. Not only that I have greatly benefited from his profound expertise and vast knowledge, but he also increased my enthusiasm and curiosity by constantly challenging me with intriguing questions.

I thank my collaborators Dr. Anna Zamansky, Prof. Agata Ciabattoni, Prof. Matthias Baaz, Yoni Zohar, and Lara Spendier. Working with them was tremendously enriching and encouraging, as well as a great delight.

I am also grateful to Professors Nachum Dershowitz, Alexander Rabinovich, and Mooly Sagiv for many fruitful discussions and valuable feedbacks.

### Abstract

Gentzen-type sequent calculi and their natural extensions (such as many-sided sequent and hypersequent calculi) provide suitable proof-theoretic frameworks for a huge variety of logics, starting from classical logic and intuitionistic logic, and including modal logics, substructural logics, many-valued logics, fuzzy logics, and paraconsistent logics. In many important cases they suggest an "algorithmic presentation" of a logic, which is particularly useful in practical applications of it, as well as for studying its properties. Thus in the last decades Gentzen-type calculi frequently arise for handling and introducing new non-classical logics. Each such calculus requires a soundness and completeness theorem with respect to its corresponding logic, and its proof-theoretic properties should be verified. Traditionally, this is done each time from scratch. In many cases the fundamental theorem of *cut-elimination* is proved. This implies the redundancy of the well-known cut rule, something which usually ensures the usefulness of the calculus. Another desirable property of Gentzen calculi is *analyticity*, namely the fact that proofs may consist only of syntactic material contained in the sequent to be proved. Often it is an immediate corollary of cut-elimination, but in various cases cut-elimination fails, and the calculus can still be shown to be analytic. This calls for an investigation of Gentzen-type calculi as mathematical objects in their own right.

This thesis aims at such a systematic general investigation of a wide variety of sequent and hypersequent calculi for many logics of different natures. Our main contribution is a *semantic* analysis of several general families of propositional Gentzen-type sequent and hypersequent calculi, that consists of the following:

- 1. We provide a uniform (possibly non-deterministic) semantic characterization for each calculus in the families we study. This has the form of general and modular soundness and completeness results that establish strong connections between the syntactic ingredients of a given Gentzen calculus and semantic restrictions on the corresponding set of models. The semantics provides a complementary view on Gentzen calculi, and, as we show, for certain general families of calculi it is also effective, naturally inducing a semantic decision procedure for the corresponding calculi.
- 2. We apply this semantic presentation (refining and extending it, when needed) for investigating crucial proof-theoretic properties of the calculi we study. This includes general notions of cut-admissibility, analyticity, and axiom-expansion. Indeed, an

illuminating contribution of a semantic study of proof systems is the ability to provide semantic proofs (or refutations) of syntactic properties. Even when a traditional syntactic proof exists, in many cases the semantic proofs are much simpler and easier to check. Thus we characterize these properties from a semantic point of view, providing general tools that can be applied in semantic proofs of these properties. In some of the families we study, this naturally leads to simple and decidable exact criteria for important proof-theoretic properties.

In addition, to demonstrate the applicability of our ideas and methods beyond the propositional level, we consider two particular hypersequent calculi for first-order and second-order Gödel logic. By extending the semantic analysis of propositional hypersequent calculi, we prove that these two calculi are indeed sound and complete for first-order and second-order Gödel logic (respectively), and that they enjoy cut-admissibility. In the case of the calculus for first-order Gödel logic this provides a semantic alternative account for a known result (proven syntactically in other works). In contrast, to the best of our knowledge cut-elimination was not proved before for the calculus for second-order Gödel logic.

# Contents

1	Intr	oducti	on	1
2	Pur	e Sequ	ient Calculi	10
	2.1	Prelin	ninaries	11
	2.2	Pure (	Calculi	12
	2.3	Proof-	Theoretic Properties	18
		2.3.1	Analyticity	18
		2.3.2	Cut-Admissibility	20
		2.3.3	Axiom-Expansion	22
3	Sen	nantics	for Pure Sequent Calculi	24
	3.1	The S	emantic Framework	24
		3.1.1	Partial Valuations and Semantic Analyticity	27
	3.2	Semar	ntics for Pure Sequent Calculi	29
	3.3	Chara	cterization of Proof-Theoretic Properties	36
		3.3.1	Strong Analyticity	36
		3.3.2	Strong Cut-Admissibility	39
		3.3.3	Axiom-Expansion	42
	3.4	Sound	ness and Completeness Proofs	43
4	Car	onical	Calculi	47
	4.1	Canon	nical Calculi	48
	4.2	Partia	l Non-deterministic Matrices	50
		4.2.1	Effectiveness	53
	4.3	PNma	trices for Canonical Calculi	55
	4.4	Chara	cterization of Proof-Theoretic Properties	60
5	Qua	asi-can	onical Calculi	64
	5.1	Quasi-	-canonical Calculi	65
	5.2	From	Quasi-canonical to Canonical Calculi	66

CONTENTS v

6	Non-pure Sequent Calculi								
	6.1	Basic Calculi	73						
		6.1.1 Proof-Theoretic Properties	78						
	6.2	Kripke-style Semantics for Basic Calculi	80						
		6.2.1 Examples	87						
	6.3	Characterization of Proof-Theoretic Properties	93						
	6.4	Soundness and Completeness Proofs	100						
7	Car	Canonical Gödel Hypersequent Calculi							
	7.1	Preliminaries	105						
	7.2	Canonical Gödel Calculi	106						
	7.3	Many-Valued Semantics	112						
	7.4	Applications of the Semantics	118						
	7.5	Soundness and Completeness Proofs	124						
8	Cal	Calculus for First-Order Gödel Logic							
	8.1	Preliminaries	134						
	8.2	Standard First-Order Gödel Logic	137						
	8.3	The Hypersequent Calculus <b>HIF</b>	140						
	8.4	Soundness	141						
	8.5	Complete Non-deterministic Semantics	143						
	8.6	Completeness for the Ordinary Semantics	154						
9	Calculus for Second-Order Gödel Logic								
	9.1	Preliminaries	158						
	9.2	Henkin-style Second-Order Gödel Logic	160						
	9.3	The Hypersequent Calculus ${\bf HIF^2}$	163						
	9.4	Soundness	166						
	9.5	Complete Non-deterministic Semantics	168						
	9.6	Completeness for the Ordinary Semantics	175						
10	Sun	nmary and Further Work	178						
Bi	bliog	graphy	181						

# Chapter 1

# Introduction

Ever since the introduction of sequent calculi for classical and intuitionistic logic by Gentzen [56], sequent calculi have been widely applied in the fields of proof theory, mathematical logic, and automated deduction. These systems and their natural extensions (such as many-sided sequent and hypersequent calculi) provide suitable proof-theoretic frameworks for a huge variety of non-classical logics, including modal logics [96], substructural logics [55], many-valued logics [61], fuzzy logics [76], and paraconsistent logics [13]. In many important cases they suggest an "algorithmic presentation" of a logic, which is particularly useful in practical applications of it, as well as for studying its properties, such as decidability (for propositional logics), consistency, interpolation, the Herbrand theorem (for first-order logics) and others. Thus in the last decades Gentzen-type calculi frequently arise for handling and introducing new non-classical logics. Each such calculus requires a soundness and completeness theorem with respect to its corresponding logic, and its proof-theoretic properties should be verified. Traditionally, this is done each time from scratch. In many cases the fundamental theorem of *cut-elimination* is proved. This implies the redundancy of the well-known cut rule, something which usually ensures the usefulness of the calculus. Another desirable property of Gentzen calculi is analyticity, namely the fact that proofs may consist only of syntactic material contained in the sequent to be proved. Often it is an immediate corollary of cut-elimination, but in various cases cut-elimination fails, and the calculus can still be shown to be analytic.

This thesis aims at a systematic investigation of Gentzen-type systems as mathematical objects in their own right. We study a wide variety of sequent and hypersequent calculi for many logics of different natures. Our main contribution is a *semantic* analysis of several general families of propositional Gentzen-type sequent and hypersequent calculi, that, generally speaking, consists of the following:

1. We provide a uniform and general semantic characterization for each system in the

families we study. Thus each calculus G corresponds to a certain set of semantic structures  $\mathcal{V}_{\mathbf{G}}$ ; and the consequence relation induced by  $\mathcal{V}_{\mathbf{G}}$  (using an appropriate notion of when a semantic structure in  $\mathcal{V}_{\mathbf{G}}$  is a model of a given sequent or hypersequent) is shown to be identical to  $\vdash_{\mathbf{G}}$ , the provability relation of G. For each family of calculi, we present a general uniform method for extracting the set  $\mathcal{V}_{\mathbf{G}}$  for a given system G in this family. In many important cases the soundness and completeness of some known Gentzen-type system with respect to its usual semantics is then obtained as a particular instance of the proposed general method. The semantics provides a complementary view on Gentzen systems. In addition, we identify certain families of systems for which the obtained semantics is also effective, i.e. it naturally induces a semantic decision procedure for the calculi in that family. Thus we derive new general decidability results for large families of propositional Gentzen-type systems.

2. We apply this semantic presentation of calculi (and extend and refine it, when needed) for investigating crucial proof-theoretic properties of the systems we study. This includes general notions of cut-admissibility, analyticity, and axiom-expansion. Indeed, an illuminating contribution of a semantic study of proof systems is the ability to provide semantic proofs (or refutations) of syntactic properties. Even when a traditional syntactic proof exists, in many cases the semantic proofs are much simpler and easier to check. Thus we characterize these properties from a semantic point of view, providing general tools that can be applied in semantic proofs of these properties. In some of the families we study, these characterizations naturally lead to simple and decidable exact criteria for the aforementioned proof-theoretic properties.

Our investigation is carried out in the following five families of propositional *fully-structural* Gentzen-type systems (i.e., systems that include all the usual structural rules: exchange, contraction, and weakening):

Pure Sequent Calculi. These are sequent calculi, whose derivation rules do not enforce any limitation on the context formulas. In addition to usual two-sided sequent calculi, we include here also calculi that employ one-sided sequents or many-sided ones. This family of calculi provides a suitable proof-theoretic framework for several important propositional logics, including classical logic, many well-studied many-valued logics, and various paraconsistent logics. In the definition of this family, we do not assume any predefined set of cut rules or identity axioms, and thus handle any possible combination of these rules.

Canonical Calculi. This is a subfamily of pure sequent calculi, in which each logical rule introduces exactly one logical connective, where all formulas in the premises of a rule are immediate subformulas of the formula introduced in its conclusion. Such "well-behaved" logical rules (called: canonical rules), have a philosophical motivation: they naturally serve a guiding principle in the philosophy of logic, due to Gentzen [56], according to which the meanings of the connectives are determined by their derivation rules. Like in the more general case of pure sequent calculi, we again include here many-sided sequent systems with arbitrary combinations of cut rules and identity axioms. Since this family of calculi is a subfamily of pure sequent calculi, all results concerning the semantics of pure sequent calculi and the semantic characterizations of their proof-theoretic properties can be applied for canonical calculi as well. However, we show that for this more restricted family of calculi we are always able to obtain simple and effective semantics, as well as decidable characterizations of their proof-theoretic properties.

Quasi-canonical Calculi. This is another subfamily of pure sequent calculi, that extends the family of canonical calculi. Here we allow also logical rules in which unary connectives precede the connective to be introduced in conclusions of logical rules (allowing, e.g., the introduction of a formula of the form  $\neg(\varphi_1 \land \varphi_2)$ ), as well as the formulas in the premises. Calculi of this family are particularly useful for many-valued logics (e.g. for the relevance logic of first degree entailment [1]) and paraconsistent logics (see, e.g., [13]). Our investigation of these calculi is not direct: instead of studying the semantics of quasi-canonical calculi, we show how to translate each quasi-canonical calculus to a canonical equivalent one, and then exploit the results concerning canonical calculi.

Basic Calculi. These are multiple-conclusion two-sided sequent calculi whose derivation rules may allow certain restrictions and manipulations on the context formulas (and for that reason they are not pure sequent calculi). Various sequent calculi that seem to have completely different natures belong to this family. Thus it includes all standard sequent calculi for modal logics, as well as the usual multiple-conclusion systems for intuitionistic logic, its dual, and bi-intuitionistic logic.

Canonical Hypersequent Calculi. We import the ideas behind canonical sequent calculi to hypersequent calculi, and define a general structure of a canonical hypersequential logical rule. Here there are many options concerning the additional hypersequential structural rules. To demonstrate our methods, we choose to study single-conclusion canonical hypersequent calculi that are based on the *communication rule*. The prototype example here is the hypersequent calculus for propositional

Gödel logic (see [22]), and thus, we call the calculi of this family canonical Gödel hypersequent calculi. In particular, it is possible to introduce in these calculi new non-deterministic connectives and add them to Gödel logic. Note that hypersequent calculi are now the main proof-theoretic framework for fuzzy logics [76], but Gödel logic is the only fundamental fuzzy logic that has a fully-structural hypersequent calculus (and thus falls in the scope of this work).

While the aforementioned families of calculi are all propositional, the ideas and methods for them are applicable for first-order and higher-order calculi as well. We demonstrate this in two specific calculi: the hypersequent calculus **HIF** for standard first-order Gödel logic [30], and its extension, that we call **HIF**<sup>2</sup>, for Henkin-style second-order Gödel logic. In particular, by extending the semantic methods developed for the family of (propositional) canonical Gödel hypersequent calculi, we are able to prove that the cut rule is admissible in **HIF** and **HIF**<sup>2</sup>. In the case of **HIF** this provides a semantic alternative account for the fact that **HIF** admits cut-elimination (proven syntactically in [30, 22]). In contrast, to the best of our knowledge cut-elimination was not proved for **HIF**<sup>2</sup> before.

A crucial feature of a systematic procedure relating proof systems and semantics should be its modularity - the correspondence between semantics and proof systems should be based on local equivalences between semantic ingredients (requirements from the semantic structures) and their syntactic counterparts (derivation rules). Such a correspondence can allow, e.g., to predict the semantic impact of employing the same rule in different proof systems, or to provide an appropriate rule for a given semantic condition added to different logics. In particular, all semantic characterizations of cutadmissibility in each of the families of calculi listed above are based on identifying the semantic impact of the cut rule(s), and comparing the semantics of the calculi with and without the cut rule(s). These tasks are of course impossible when the proof system and its semantics are considered as a whole, and there is no possibility to separate between the different semantic effects of each particular rule. The major key to have this modularity, as well to provide semantics to every calculus in the families that we study, is the use of non-deterministic semantics. Thus, following [17, 21], we relax the principle of truth-functionality, and allow cases in which the truth value of a compound formula is not uniquely determined by the truth values of its subformulas. By allowing non truth-functional semantic structures, we are able to separately analyze the semantic effect of each component of the syntactic machinery (each derivation rule, and in fact also each ingredient of a rule). The full semantics of the calculus is then obtained by joining the semantic effects of all of its components. For this matter, we develop several frameworks of non-deterministic semantics:

Many-Valued Systems. These provide a semantic framework for specifying sets of valuations – functions assigning truth values to formulas of a given propositional language. Each many-valued system includes a set of semantic conditions, that can be easily read off the derivation rules of a pure sequent calculi, and used to restrict its corresponding set of valuations (e.g. "If  $\varphi_1$  has some truth value  $u_1$ , and  $\neg \varphi_1$  has some truth value  $u_2$ , then  $\neg(\varphi_1 \land \varphi_1)$  should have the truth value  $u_3$ "). This framework generalizes the "bivaluation semantics" [34, 40], many-valued matrices [93, 61], and non-deterministic many-valued matrices [17, 21], and is used here to provide semantics for pure sequent calculi.

Partial Non-deterministic Matrices. These form a special case of many-valued systems that serve as a simpler semantic framework for canonical and quasi-canonical calculi. Thus in partial non-deterministic matrices, the semantic conditions for specifying restrictions on valuation functions can be arranged in generalized truth tables. Usual logical matrices are particular instances, while non-determinism is introduced as done in non-deterministic matrices (see [17, 21]), by possibly allowing several options in some entries of the truth tables (thus the value of  $\diamond(p_1, \ldots, p_n)$  is restricted, but not uniquely determined, by the values of  $p_1, \ldots, p_n$ ). However, to handle arbitrary canonical and quasi-canonical calculi we had to slightly extend the framework of non-deterministic matrices by allowing also the option of having an empty set of options in the entries of the truth tables (which intuitively mean that certain combinations of truth values are disallowed).

Non-deterministic Kripke Valuations. For basic sequent systems, we introduce a generalization of Kripke-style semantics for modal and intuitionistic logic, that we call Kripke valuations. As Kripke models, these semantic structures employ a set of possible worlds and accessibility relations, and certain conditions connect the truth value assigned to a formula in each world w with values assigned to other formulas in the worlds accessible from w.

We show that Kripke valuations that are based on three or four truth values can be used in semantic characterizaions of basic sequent systems with restricted cut rule and/or identity axiom (as needed e.g for characterizing cut-admissibility).

Non-deterministic Gödel Valuations. For canonical Gödel hypersequent calculi, we introduce Gödel valuations. These consist of some linearly ordered set of truth values, and a function assigning a pair of truth values from this set to each formula of a given propositional language. Intuitively, the first element in the pair of truth values assigned to some formula  $\varphi$  is used for occurrences of  $\varphi$  on the left sides

of sequents, while the second element in the pair is used for occurrences of  $\varphi$  on the right sides. We show that the cut rule and the identity axiom "connect" these two elements: if they are both available for some formula  $\varphi$ , then the two elements in the pair of truth values of  $\varphi$  must be equal. In addition, the two values assigned to each compound formula of the form  $\diamond(\varphi_1, \dots, \varphi_n)$  must lie within certain intervals whose edges are computed from the values assigned to  $\varphi_1, \dots, \varphi_n$ . The usual algebraic semantics of Gödel logic is a particular instance, in which all of these pairs and intervals are degenerate, and thus the value of  $\diamond(\varphi_1, \dots, \varphi_n)$  is uniquely determined by the values of  $\varphi_1, \dots, \varphi_n$ . In turn, we provide a general construction of the functions for computing these intervals for  $\diamond$ -formulas given some (canonical) rules for introducing each connective  $\diamond$ .

### Outline

The structure of this thesis is as follows. Chapter 2 is devoted to precise definitions of pure sequent calculi and their proof-theoretic properties, as well as some basic consequences of these properties. Chapter 3 introduces the semantic framework of many-valued systems, and provides a method to obtain a many-valued system for any given pure sequent calculus. Based on this semantics, in Section 3.3 we present necessary and sufficient semantic conditions for analyticity, cut-admissibility and axiom-expansion in pure calculi.

Chapter 4 discusses canonical sequent calculi which are defined as pure sequent calculi with additional restrictions on the structure of the logical introduction rules. In turn, in Section 4.2 we present the corresponding (effective) semantic framework of partial non-deterministic matrices, as a special restricted instance of many-valued systems. Based on the results of Chapter 3, we then show that canonical sequent calculi can be characterized by partial non-deterministic matrices, and that the aforementioned proof-theoretic properties can be easily checked using this alternative semantic presentation.

In Chapter 5 we introduce *quasi-canonical sequent calculi*, and show that each such calculus can be translated into an equivalent canonical one. In certain important cases, this translation may be used to obtain a characteristic partial non-deterministic matrix for a given quasi-canonical calculus.

In Chapter 6 we go beyond the scope of pure sequent calculi by introducing basic sequent calculi, in which derivation rules may include limitations on the context formulas used in their applications. Then we show that each basic calculus induces a set of generalized Kripke valuations for which it is strongly sound and complete. In Section 6.3 we derive characterization of proof-theoretic properties of basic calculi based on this Kripke semantics. Their nature is similar to the corresponding characterizations from Section 3.3. We demonstrate their applicability in various examples, including sequent

calculi for modal logics and a sequent calculus for bi-intuitionistic logic.

In Chapter 7 we define and study hypersequent Gödel calculi from a similar angle. The semantics in this chapter is based on Gödel valuations, that generalize the usual many-valued semantics of propositional Gödel logic.

Chapters 8 and 9 are of a completely different nature, as each of them is devoted to one particular calculus for one particular logic. Chapter 8 discusses the hypersequent calculus **HIF** for first-order Gödel logic, and provides a semantic proof for cut-admissibility in this calculus. Chapter 9 introduces an extension of **HIF** with usual rules for second-order quantifiers, called **HIF**<sup>2</sup>. We show that **HIF**<sup>2</sup> is sound and complete for second-order Gödel logic, and that it enjoys cut-admissibility. Note that the fact that **HIF** enjoys cut-admissibility actually follows from the fact that **HIF**<sup>2</sup> does. Nevertheless, as a preparation and for the convenience of the reader, we provide first a full account for **HIF**, that is relatively easier to follow than the one for **HIF**<sup>2</sup>.

Finally, in Chapter 10 we conclude with a discussion of some directions for further research.

#### Some Related Works

Usually, the study of Gentzen-type systems is tailored to a specific logic or family of logics. Several notable exceptions include the following:

- [34] studies a general family of sequent systems, and shows that (possibly non-truth functional) bivaluation semantics can be read off the sequent rules for any given system in this family. This work is close to what we do in Chapter 3. However, the sequent systems studied in [34] are just a particular subset of the pure sequent calculi that we study here, as they all employ the usual cut rules and identity axioms. In addition, [34] does not study at all the effectiveness of this semantic framework, as well as semantic characterizations of syntactic properties of the studied calculi. Therefore besides a new look on the sequent calculus, the semantics proposed in [34] does not seem to have much practical or proof-theoretic applications.
- The introduction and first semantic investigation of canonical sequent calculi were done in [17]. That work considered only two-sided sequent calculi with arbitrary canonical rules and the usual cut rule and identity axiom. It was shown that each such calculus can be characterized by a non-deterministic matrix (Nmatrix). That Nmatrix can in turn be used to check whether the calculus is analytic and whether it enjoys cut-admissibility. Later, in [19] that work was extended to many-sided canonical sequent calculi (see also [21]). Our study of canonical calculi in Chapter 4 considers more general family of systems, with arbitrary set of "primitive rules"

(these include the cut rules and the identity axioms). In fact, the main results for previously studied canonical calculi (as the characterization of analyticity and cut-admissibility) can be easily obtained from the more general theorems given in Chapter 4.

- [26] studies from a semantic point of view another general family of sequent systems, which is a proper subfamily of our canonical sequent calculi. The semantic framework employed there is truth-functional (based on usual logical many-valued matrices), and thus many sequent calculi cannot be semantically characterized in this framework. By allowing non-deterministic semantics we are able to cover much more general family of calculi, and it can be shown that for the calculi studied in [26] we practically obtain the same (deterministic) semantics.
- A variety of works studies the connection between syntax and semantics in sequent and hypersequent substructural calculi, with a focus on developing semantic and algebraic conditions for cut-admissibility in such systems (e.g. [44],[32],[91]). In this thesis we only consider fully structural Gentzen-type calculi, but nevertheless, some (obviously, not all) of the calculi in the families that we study fall in the scope of these works, and their semantic criteria for cut-admissibility are applicable in these cases. However, the semantic frameworks used in these works (particularly, phase semantics) is significantly more abstract and complex than the semantic frameworks that we employ.

At this point it should be noted that the idea of using non-deterministic semantics for proving cut-admissibility of a sequent system has a very long history. Indeed, in the quest to verify *Takeuti's conjecture* [89] (that was open for several years) regarding cut-admissibility in the calculus for second-order classical logic, Schütte developed a three-valued non-deterministic semantics for the cut-free fragment of this calculus [85]. This provided a semantic equivalent to Takeuti's conjecture, that was verified by Tait a few years later [87], when it was shown that it is possible to extract a usual (two-valued) counter-model from every three-valued non-deterministic Schütte's counter-model. As a simple consequence, one obtains that if there is no cut-free proof of a certain sequent, then there is no proof at all (see also [58]). Basically, our semantic characterizations of cut-admissibility, as well as the cut-admissibility proofs in Chapters 7 to 9, are based on a similar (generalized) approach.

<sup>&</sup>lt;sup>1</sup>More precisely, Takeuti's conjecture concerned full type-theory, namely, the completeness of the cut-free sequent calculus that includes rules for quantifiers of any finite arity. However, the proof for second-order fragment was the main breakthrough. Note that the usual syntactic arguments to prove cut-elimination dramatically fail when it comes to higher-order logic.

Finally, besides the aforementioned works on canonical calculi, we are not aware of any works aiming to study analyticity of general Gentzen-type systems, *regardless* of cut-admissibility. In many cases our criteria of analyticity turn out to be much simpler than those of cut-admissibility.

### Publications Related to this Dissertation

Most of the contributions described in this thesis have first appeared in other publications. They are roughly divided as follows:

- Chapters 4-5: [68], [70], [28], [29].
- Chapter 6: [15], [73].
- Chapter 7: [71], [69].
- Chapter 8: [16], [72].

The material in Chapters 2,3 and 9 was not published before.

More details about the connections between these publications and this thesis will be given in the beginning of each chapter.

# Chapter 2

# Pure Sequent Calculi

In this chapter we introduce the family of pure sequent calculi. These will be the object of a semantic investigation in the next chapter. Roughly speaking, pure sequent calculi are propositional fully-structural sequent calculi (sequent calculi that include all the usual structural rules: exchange, contraction, and weakening), whose derivation rules do not enforce any limitation on the side formulas (following [5], the adjective pure stands for this requirement). This family of calculi provides a suitable proof-theoretic framework for several important propositional logics, including classical logic, important many-valued logics, and various paraconsistent logics. Our scope is broader than what is usually considered as a sequent system:

- We consider *many-sided* sequents, rather than just ordinary *two-sided* ones. This allows us to naturally capture a large family of many-valued logics (see, e.g., [67]).
- We do not presuppose that all systems include identity axioms or cut rules of a given form. This will play a major role in the semantic characterizations of proof-theoretic properties of these systems (e.g., we will be able to compare the semantics of a given system with cut, and the semantics of the same system without cut).

This chapter is organized as follows. We start by defining the notion of a propositional logic in Section 2.1. Then, we precisely formulate the framework of pure sequent calculi, and the logics they induce (Section 2.2). In Section 2.3 we introduce some fundamental proof-theoretic properties of pure sequent calculi that we will study later from a *semantic* perspective.

#### Publications Related to this Chapter

The material in this chapter was not published before.

2.1. Preliminaries 11

## 2.1 Preliminaries

**Definition 2.1.1.** A propositional language  $\mathcal{L}$  consists of a countably infinite set of variables  $at_{\mathcal{L}} = \{p_1, p_2, ...\}$  (whose elements are called atomic formulas), and a finite set  $\diamondsuit_{\mathcal{L}}$  of propositional connectives. Each  $\diamondsuit \in \diamondsuit_{\mathcal{L}}$  has a fixed finite arity  $ar(\diamondsuit) \ge 0$ . The set of all n-ary connectives of  $\mathcal{L}$  (for  $n \ge 0$ ) is denoted by  $\diamondsuit_{\mathcal{L}}^n$ .

Note that propositional constants are considered as nullary connectives.

**Notation 2.1.2.** We shall specify proportional languages by a set of connectives, and indicate their arities in superscripts. For example,  $\{\neg^1, \wedge^2\}$  denotes a language with two connectives: a unary one denoted by  $\neg$ , and a binary one denoted by  $\wedge$ .

Given a propositional language  $\mathcal{L}$ ,  $\mathcal{L}$ -formulas are constructed as usual. We usually use  $\varphi$ ,  $\psi$  as metavariables for  $\mathcal{L}$ -formulas,  $\Gamma$ ,  $\Delta$  for finite sets of  $\mathcal{L}$ -formulas, and  $\mathcal{T}$ ,  $\mathcal{F}$ ,  $\mathcal{C}$  for (possibly infinite) sets of  $\mathcal{L}$ -formulas. Henceforth,  $\mathcal{L}$  stands for an arbitrary propositional language. We shall usually identify the set of  $\mathcal{L}$ -formulas with  $\mathcal{L}$  itself, e.g. when writing " $\varphi \in \mathcal{L}$ " instead of " $\varphi$  is an  $\mathcal{L}$ -formula".

**Definition 2.1.3.** An  $\mathcal{L}$ -substitution is a function  $\sigma: at_{\mathcal{L}} \to \mathcal{L}$ . It is recursively extended to  $\mathcal{L}$ , by  $\sigma(\diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})) = \diamond(\sigma(\varphi_1), \dots, \sigma(\varphi_{ar(\diamond)}))$  for every  $\diamond \in \diamondsuit_{\mathcal{L}}$ .

We follow [21] in taking the following definitions of Tarskian consequence relations and Tarskian propositional logics:

**Definition 2.1.4.** A relation  $\Vdash$  between sets of  $\mathcal{L}$ -formulas and  $\mathcal{L}$ -formulas is:

Reflexive: if  $\mathcal{T} \Vdash \varphi$  whenever  $\varphi \in \mathcal{T}$ .

*Monotone*: if  $\mathcal{T}' \Vdash \varphi$  whenever  $\mathcal{T} \Vdash \varphi$  and  $\mathcal{T} \subseteq \mathcal{T}'$ .

Transitive: if  $\mathcal{T}, \mathcal{T}' \Vdash \varphi$  whenever  $\mathcal{T} \Vdash \psi$  and  $\mathcal{T}', \psi \Vdash \varphi$ .

Structural: if  $\sigma(\mathcal{T}) \Vdash \sigma(\varphi)$  for every  $\mathcal{L}$ -substitution  $\sigma$  whenever  $\mathcal{T} \Vdash \varphi$ .

**Definition 2.1.5.** A relation between sets of  $\mathcal{L}$ -formulas and  $\mathcal{L}$ -formulas which is reflexive, monotone and transitive is called a *Tarskian consequence relation (tcr)* for  $\mathcal{L}$ . A (*Tarskian propositional*) logic is a pair  $\langle \mathcal{L}, \Vdash \rangle$ , where  $\mathcal{L}$  is a propositional language, and  $\Vdash$  is a structural ter for  $\mathcal{L}$ .

**Definition 2.1.6.** A logic  $\langle \mathcal{L}, \Vdash \rangle$  is *finitary* if  $\Gamma \Vdash \varphi$  for some finite  $\Gamma \subseteq \mathcal{T}$  whenever  $\mathcal{T} \Vdash \varphi$ .

The most important (and popular) propositional logic is of-course classical logic. Its language is  $\{\neg^1, \wedge^2, \vee^2, \supset^2\}$ , and it is denoted below by  $\mathcal{L}_{cl}$ . The well-known ter of classical logic will be denoted by  $\Vdash_{cl}$  (see Example 2.2.20).

 $<sup>^{1}</sup>$ We use the symbol  $\vdash$  to relate sets of formulas and formulas. The usual symbol  $\vdash$  will be used to denote derivability of a sequent from a set of sequents.

## 2.2 Pure Calculi

Usual sequent systems are two-sided, and sequents are often written as expressions of the forms  $\varphi_1, \ldots, \varphi_n \Rightarrow \psi_1, \ldots, \psi_m$ . Dealing with many sided-sequents, we find it convenient to use a (finite) set of labels, for specifying the position(s) inside the sequent in which a certain formula occurs. Thus, in what follows  $\mathcal{L}$  denotes a finite non-empty set of labels. We usually use  $\mathbf{x}$  as a metavariable for a label in  $\mathcal{L}$ , and  $\mathbf{X}$  for sets of such labels. Sequents are defined as follows:

**Definition 2.2.1.** An  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula is an ordered pair  $\langle \mathbf{x}, \varphi \rangle$ , denoted by  $\mathbf{x}$ : $\varphi$ , where  $\mathbf{x} \in \mathcal{L}$  and  $\varphi \in \mathcal{L}$ . An  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent is a finite set of  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formulas.

We usually use  $\alpha, \beta$  as metavariables for labelled formulas, and s, c for sequents. Substitutions are extended to labelled formulas, sequents, sets of sequents, etc. in the obvious way. In particular,  $\sigma(\emptyset) = \emptyset$ .

**Notation 2.2.2.** For  $X \subseteq \mathcal{L}$  and  $\varphi \in \mathcal{L}$ , the expression  $(X:\varphi)$  denotes the sequent  $\{x:\varphi \mid x \in X\}$ .

Notation 2.2.3. Usual two-sided sequents can be seen as  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -sequents, where  $\mathcal{L}_2$  denotes the set of labels  $\{f, t\}$ . The labels f, t denote the "left side" and the "right side" respectively. The more usual notation  $\varphi_1, \ldots, \varphi_n \Rightarrow \psi_1, \ldots, \psi_m$  is interpreted as  $\{f:\varphi_1, \ldots, f:\varphi_n, t:\psi_1, \ldots, t:\psi_m\}$ .

The use of f and t at this point is just a matter of tradition, as the labels should not be confused with truth values! Only in certain specific (important) cases, the truth values employed in the semantic characterization presented in Chapter 3 have one-to-one correspondence with the set  $\mathcal{L}$  of labels.

Remark 2.2.4. For our purposes, we find it most convenient to define sequents using sets. In particular, the  $\langle \mathcal{L}, \pounds_2 \rangle$ -sequents  $\{f:p_1, t:p_2\}$ ,  $\{t:p_2, f:p_1\}$ ,  $\{f:p_1, f:p_1, t:p_2\}$  are all the same object. This immediately entails that the exchange rule, the contraction rule and the expansion rule (the converse of contraction) are all built-in in all sequent calculi that we study. To have a fully-structural system, we should only further require the presence of the weakening rules (one weakening rule for each label, as defined below).

Next, we define the form of derivation rules that are allowed in pure sequent systems.

**Definition 2.2.5.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule is a pair of the form  $\mathcal{S}/s$ , where  $\mathcal{S}$  is a finite set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and s is a single  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent. The elements of  $\mathcal{S}$  are called the premises of the rule, and s is called the conclusion of the rule. To improve readability, we

2.2. Pure Calculi 13

usually drop the set braces of the set of premises. An application of a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $s_1, \ldots, s_n/s$  is any inference step of the following form:

$$\frac{\sigma(s_1) \cup c_1 \dots \sigma(s_n) \cup c_n}{\sigma(s) \cup c_1 \cup \dots \cup c_n}$$

where  $\sigma$  is an  $\mathcal{L}$ -substitution, and  $c_i$  is an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent for every  $1 \leq i \leq n$ . The sequents  $\sigma(s_1) \cup c_1, \ldots, \sigma(s_n) \cup c_n$  are called the *premises* of the application, while the sequent  $\sigma(s) \cup c_1 \cup \ldots \cup c_n$  is called the *conclusion* of the application. In addition, the sequents  $c_1, \ldots, c_n$  are called the *context sequents* (of the application).

Note that the propositional variables of the "object language"  $\mathcal{L}$  are also employed in the formulation of the rules. In particular, meta-variables (which are usually used to represent derivation rules by schemes) are not used. Roughly speaking, applications of some rule are obtained by applying a substitution on the premises  $s_1, \ldots, s_n$  and the conclusion s of the rule, and freely adding *context* formulas.

**Example 2.2.6.** Suppose that  $\mathcal{L}$  contains the binary "implication" connective  $\supset$ . The following pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules are usually used for introducing this connective:

$$\{t:p_1\}, \{f:p_2\}/\{f:p_1\supset p_2\}$$
 and  $\{f:p_1, t:p_2\}/\{t:p_1\supset p_2\}.$ 

Their applications have (respectively) the forms:

$$\frac{\{\mathbf{t}:\varphi_1\} \cup c_1 \quad \{\mathbf{f}:\varphi_2\} \cup c_2}{\{\mathbf{f}:\varphi_1 \supset \varphi_2\} \cup c_1 \cup c_2} \quad \text{and} \quad \frac{\{\mathbf{f}:\varphi_1, \mathbf{t}:\varphi_2\} \cup c}{\{\mathbf{t}:\varphi_1 \supset \varphi_2\} \cup c}$$

In [62], a different implication connective is used, whose introduction rules can be easily formulated as pure rules. The f-rule (whose conclusion is  $\{f:p_1 \supset p_2\}$ ) is the same rule as above, but the t-rule has the form:  $\{t:p_2\}/\{t:p_1 \supset p_2\}$ . Its applications have the form:

$$\frac{\{\mathsf{t}{:}\varphi_2\}\cup c}{\{\mathsf{t}{:}\varphi_1\supset\varphi_2\}\cup c}$$

**Example 2.2.7.** Suppose that  $\mathcal{L}$  contains the binary "implication" connective  $\supset$ , and let  $\mathcal{L}_3 = \{f, i, t\}$ . The following pure  $\langle \mathcal{L}, \mathcal{L}_3 \rangle$ -rules are used for introducing  $\supset$  in the calculus for three-valued Łukasiewicz's logic presented in [97]:

$$\{\mathtt{t}: p_1\}, \{\mathtt{f}: p_2\}/\{\mathtt{f}: p_1\supset p_2\}$$
 
$$(\{\mathtt{i},\mathtt{t}\}: p_1), \{\mathtt{i}: p_1,\mathtt{i}: p_2\}, \{\mathtt{t}: p_1,\mathtt{f}: p_2\}/\{\mathtt{i}: p_1\supset p_2\}$$
 
$$(\{\mathtt{f},\mathtt{i}\}: p_1)\cup \{\mathtt{t}: p_2\}, \{\mathtt{f}: p_1\}\cup (\{\mathtt{i},\mathtt{t}\}: p_2)/\{\mathtt{t}: p_1\supset p_2\}$$

**Example 2.2.8.** The following rule scheme appears in a sequent system from [12] for da Costa's paraconsistent logic  $C_1$ :

$$\frac{\Gamma \Rightarrow \varphi, \Delta \quad \Gamma \Rightarrow \neg \varphi, \Delta}{\Gamma, \neg(\varphi \land \neg \varphi) \Rightarrow \Delta}$$

This rule scheme can be formulated as the following pure  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}} \rangle$ -rule (where  $\neg \in \diamondsuit^1_{\mathcal{L}}$  and  $\wedge \in \diamondsuit^2_{\mathcal{L}}$ ):

$$\{t:p_1\}, \{t:\neg p_1\}/\{f:\neg(p_1 \land \neg p_1)\}.$$

Convention 2.2.9. Obviously the names of the variables in pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules are immaterial (e.g.,  $\{t:p_2\}/\{t:p_1 \supset p_2\}$  is completely equivalent to  $\{t:p_3\}/\{t:p_5 \supset p_3\}$ ). To avoid further technical complications, we assume that a unique representative is chosen from every equivalence class of rules in some reasonable way, and only these representatives are considered as pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules. For example, when only one variable is involved in a rule (as in Example 2.2.8), we may suppose that this variable is  $p_1$ .

A special family of pure rules is the family of *primitive* rules. These rules are used to perform simple manipulations on the labels, and they do not mention any specific connective of the language  $\mathcal{L}$ . Formally they are defined as follows:

Notation 2.2.10. Given an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\alpha$ , we denote by  $frm[\alpha]$  the  $\mathcal{L}$ -formula appearing in  $\alpha$ . frm is extended to sets of  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formulas, sets of sets of  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formulas, etc. in the obvious way.

**Definition 2.2.11.** A primitive  $\mathcal{L}$ -rule is any pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent rule  $\mathcal{S}/s$  such that  $frm[\mathcal{S} \cup \{s\}] = \{p_1\}.$ 

By definition, all primitive  $\mathcal{L}$ -rules have the form  $(X_1:p_1), \ldots, (X_n:p_1)/(X:p_1)$  for some  $X_1, \ldots, X_n, X \subseteq \mathcal{L}$ . An application of the primitive  $\mathcal{L}$ -rule  $(X_1:p_1), \ldots, (X_n:p_1)/(X:p_1)$  is any inference steps of the following form:

$$\frac{(\mathbf{X}_1:\varphi)\cup c_1 \dots (\mathbf{X}_n:\varphi)\cup c_n}{(\mathbf{X}:\varphi)\cup c_1\cup \dots \cup c_n}$$

where  $\varphi$  is an  $\mathcal{L}$ -formula, and  $c_i$  is a  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent for every  $1 \leq i \leq n$ .

**Example 2.2.12.** Let  $\mathcal{L} = \{f, i, t\}$ , and consider the primitive  $\mathcal{L}$ -rule

$$\{\mathtt{f}\!:\!p_1\}, \{\mathtt{i}\!:\!p_1\}/(\{\mathtt{i},\mathtt{t}\}\!:\!p_1).$$

This rule allows to infer  $(\{i, t\}: \varphi) \cup c_1 \cup c_2$  from  $\{f: \varphi\} \cup c_1$  and  $\{i: \varphi\} \cup c_2$  for every two  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents  $c_1, c_2$  and  $\mathcal{L}$ -formula  $\varphi$ .

The following primitive rules are usually present in Gentzen-type systems:

Weakening Rules For each  $\mathbf{x} \in \mathcal{L}$ , the weakening rule  $(\mathbf{x}:weak)$  is the primitive  $\mathcal{L}$ -rule  $\{\emptyset\}/\{\mathbf{x}:p_1\}$ . Its applications have the form:

$$\frac{c}{\{\mathbf{x} : \varphi\} \cup c}$$

2.2. Pure Calculi 15

Cut Rules These are primitive  $\mathcal{L}$ -rules of the form  $(X_1:p_1), \dots, (X_n:p_1)/\emptyset$  for non-empty  $X_1, \dots, X_n$ . An application of a cut rule of this form has the form:

$$\frac{(\mathbf{X}_1:\varphi)\cup c_1 \quad \dots \quad (\mathbf{X}_n:\varphi)\cup c_n}{c_1\cup \dots \cup c_n}$$

 $\varphi$  is called the *cut-formula* of the application. Two-sided sequent systems usually employ  $\{f:p_1\}, \{t:p_1\}/\emptyset$  as the only cut rule. We denote this rule by (cut).

**Identity Axioms** These are primitive £-rules of the form  $\emptyset/(X:p_1)$  for non-empty X. An application of an identity axiom of this form has the form:

$$(\mathtt{X}:\varphi)$$

 $\varphi$  is called the *id-formula* of the application. Note that applications of identity axioms do not include context formulas. However, when the weakening rules are available for every  $\mathbf{x} \in \mathcal{L}$ , it is possible to derive  $(\mathbf{X}:\varphi) \cup c$  from  $(\mathbf{X}:\varphi)$  for every sequent c. Two-sided sequent systems usually employ  $\emptyset/\{\mathbf{f}:p_1,\mathbf{t}:p_1\}$  as the only identity axiom. We denote this rule by (id).

Remark 2.2.13. Note that there are several useful options for cut rules and identity axioms when  $|\mathcal{L}| > 2$ . For example, the systems in [27] have a cut rule  $\{\mathbf{x}:p_1\}, \{\mathbf{y}:p_1\}/\emptyset$  for every  $\mathbf{x} \neq \mathbf{y}$  in  $\mathcal{L}$ , and  $\emptyset/(\mathcal{L}:p_1)$  is their only identity axiom; while the systems in [24] employ one cut rule of the form  $\{\{\mathbf{x}:p_1\} \mid \mathbf{x} \in \mathcal{L}\}/\emptyset$ , and an identity axiom  $\emptyset/(\{\mathbf{x},\mathbf{y}\}:p_1)$  for every  $\mathbf{x} \neq \mathbf{y}$  in  $\mathcal{L}$ . Other useful combinations arise when quasi-canonical systems are translated into canonical ones (see Chapter 5).

Next, we define the family of pure sequent calculi. In addition to the structural rules of contraction, exchange and expansion that are implicit in our calculi, we also require that pure sequent calculi contain all weakening rules. Thus we refer to these systems as fully-structural.

**Definition 2.2.14.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus consists of a finite set of pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules, that includes  $(\mathbf{x}:weak)$  for every  $\mathbf{x} \in \mathcal{L}$ . A proof in a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$  of an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s from a set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents (called assumptions) is a finite list<sup>2</sup> of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents ending with s, such that every sequent in the list is either an element of  $\mathcal{S}$ , or a conclusion of some application of some rule of  $\mathbf{G}$ , provided that all premises of this application appear before. We write  $\mathcal{S} \vdash_{\mathbf{G}} s$  to denote the existence of such a proof.

Convention 2.2.15. Henceforth, we assume that **G** does not include the (trivial) pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $\emptyset/\emptyset$ .

<sup>&</sup>lt;sup>2</sup>Similarly, one can use finite trees or DAGs.

**Notation 2.2.16.** Given a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, we denote by  $P_{\mathbf{G}}$  and  $R_{\mathbf{G}}$  the set of primitive  $\mathcal{L}$ -rules of  $\mathbf{G}$  except for the weakening rules, and the set of non-primitive rules of  $\mathbf{G}$  (respectively).

The following simple observations will be useful in the sequel.

**Proposition 2.2.17.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus.

- 1. If  $S \cup \{s\} \vdash_{\mathbf{G}} s'$ , then  $S \cup \{s \cup c\} \vdash_{\mathbf{G}} s' \cup c$  for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent c.
- 2. If  $S \vdash_{\mathbf{G}} s$  then  $\sigma(S) \vdash_{\mathbf{G}} \sigma(s)$  for every  $\mathcal{L}$ -substitution  $\sigma$ .

Recall that sequent calculi are a tool to characterize logics. As defined below, each pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus naturally induces  $2^{|\mathcal{L}|}$  logics, each of which is based on some subset of  $\mathcal{L}$ .

**Definition 2.2.18.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, and  $X \subseteq \mathcal{L}$  a set of labels. The tcr induced by **G** and X, denoted by  $\Vdash_{\mathbf{G}}^{\mathbf{X}}$ , is the relation between sets of  $\mathcal{L}$ -formulas and  $\mathcal{L}$ -formulas defined by:  $\mathcal{T} \Vdash_{\mathbf{G}}^{\mathbf{X}} \varphi$  iff  $\{(X:\psi) \mid \psi \in \mathcal{T}\} \vdash_{\mathbf{G}} (X:\varphi)$ .

It is easy to verify that for every **G** and X,  $\Vdash_{\mathbf{G}}^{X}$  is indeed a tcr (see Definition 2.1.4). In fact, we have the following:

**Proposition 2.2.19.** For every **G** and **X** as above,  $\langle \mathcal{L}, \Vdash_{\mathbf{G}}^{\mathbf{X}} \rangle$  is a finitary logic.

*Proof.* The fact that  $\Vdash_{\mathbf{G}}^{\mathbf{X}}$  is structural directly follows from Proposition 2.2.17. The fact that it is finitary follows from the definitions.

**Example 2.2.20.** The most important sequent calculus is the fundamental Gentzen's system LK for classical logic [56]. Its propositional fragment can be straightforwardly presented as a pure  $\langle \mathcal{L}, \pounds_2 \rangle$ -calculus, which we denote by  $\mathbf{LK}$ . The rules of  $\mathbf{LK}$  are presented in Figure 2.1. The consequence relation  $\Vdash_{cl}$  of propositional classical logic is equal to  $\Vdash^{\{t\}}_{\mathbf{LK}}$  – the logic induced by  $\mathbf{LK}$  and the set  $\{t\}$ .

Remark 2.2.21. In the case of LK, there is another natural way to define the induced logic:  $\mathcal{T} \Vdash_{cl} \varphi$  iff  $\vdash_{\mathbf{LK}} \{\mathbf{f}:\psi \mid \psi \in \Gamma\} \cup \{\mathbf{t}:\varphi\}$  for some finite  $\Gamma \subseteq \mathcal{T}$ . It is easy to see that in LK, and actually in every pure  $\langle \mathcal{L}, \pounds_2 \rangle$ -calculus  $\mathbf{G}$  such that  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ , we have that  $\mathcal{T} \Vdash_{\mathbf{G}}^{\{\mathbf{t}\}} \varphi$  (according to Definition 2.2.18) iff  $\vdash_{\mathbf{G}} \{\mathbf{f}:\psi \mid \psi \in \Gamma\} \cup \{\mathbf{t}:\varphi\}$  for some finite  $\Gamma \subseteq \mathcal{T}$ . Therefore, the two alternatives to define  $\Vdash_{cl}$  are equivalent. However, the formulation we gave in Definition 2.2.18 is more general, as it ensures that we obtain a logic for *every* pure calculus with arbitrary primitive rules.

2.2. Pure Calculi 17

```
(\mathbf{f}:weak) = \{\emptyset\}/\{\mathbf{f}:p_1\}
                                                                (t:weak) \{\emptyset\}/\{t:p_1\}
                \{f:p_1\}, \{t:p_1\}/\emptyset
                                                                                 \emptyset/\{f:p_1,t:p_1\}
(cut)
                                                                (id)
                                                                (t:¬)
(t:∧)
                \{t:p_1\}/\{f:\neg p_1\}
(f:¬)
                                                                                 \{f:p_1\}/\{t:\neg p_1\}
                \{f:p_1,f:p_2\}/\{f:p_1 \wedge p_2\}
                                                                                \{t:p_1\}, \{t:p_2\}/\{t:p_1 \land p_2\}
(f: \land)
                                                                (t:\lor) \{t:p_1,t:p_2\}/\{t:p_1\lor p_2\}
               \{f:p_1\}, \{f:p_2\}/\{f:p_1\vee p_2\}
(f: \lor)
                                                                           \{f:p_1,t:p_2\}/\{t:p_1\supset p_2\}
                \{t:p_1\}, \{f:p_2\}/\{f:p_1\supset p_2\}
(f:\supset)
                                                                (\mathsf{t}:\supset)
```

Figure 2.1: The pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculus **LK** 

In addition to **LK**, the family of *canonical calculi*, that was defined and studied in [17], falls under the definition of pure calculi. Many other previously studied useful sequent calculi can be naturally presented as pure calculi. This includes all calculi for paraconsistent logics from [12], all labelled calculi for finite valued logics from [26], and the signed calculi from [20].

**Example 2.2.22.** In [12] a pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculus for da Costa's historical paraconsistent logic  $\mathbf{C_1}$  was introduced.<sup>3</sup> This calculus, denoted here by  $\mathbf{G_{C_1}}$ , is obtained from  $\mathbf{LK}$  by discarding the rule  $(\mathbf{f}:\neg)$  and adding the following rules:

```
\begin{array}{ll} (\mathbf{f}:\neg\neg) & \{\mathbf{f}:p_1\}/\{\mathbf{f}:\neg\neg p_1\} \\ (\mathbf{f}:\neg\wedge^1) & \{\mathbf{t}:p_1\}, \{\mathbf{t}:\neg p_1\}/\{\mathbf{f}:\neg(p_1\wedge\neg p_1)\} \\ (\mathbf{f}:\neg\wedge^2) & \{\mathbf{f}:\neg p_1\}, \{\mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\wedge p_2)\} \\ (\mathbf{f}:\neg\vee^1) & \{\mathbf{f}:\neg p_1\}, \{\mathbf{f}:p_2, \mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\vee p_2)\} \\ (\mathbf{f}:\neg\vee^2) & \{\mathbf{f}:p_1, \mathbf{f}:\neg p_1\}, \{\mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\vee p_2)\} \\ (\mathbf{f}:\neg\supset^1) & \{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2, \mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\supset p_2)\} \\ (\mathbf{f}:\neg\supset^2) & \{\mathbf{f}:p_1, \mathbf{f}:\neg p_1\}, \{\mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\supset p_2)\} \end{array}
```

Remark 2.2.23. One can choose to define sequents using *lists* (as in the original work of Gentzen) or *multisets*, and explicitly include contraction and exchange in the definition of a pure sequent calculus. Obviously, this would not affect the derivability relation  $\vdash_{\mathbf{G}}$ . In fact, for all aspects of proof systems studied in this thesis (semantics, cut-admissibility, analyticity, etc.) this choice is immaterial, since any result in one formulation trivially holds in the other. Of course, this might not be the case when studying other properties (like e.g. in [49]). Similarly, we formulated the applications of rules as *multiplicative* (context-independent) rather than additive (context-sharing) applications (see [57],[92]). Clearly, in the presence of all structural rules, the multiplicative version and the additive one are interderivable. Again, this decision does not affect any property we discuss below.

<sup>&</sup>lt;sup>3</sup>Here and henceforth, when we say that a two-sided calculus **G** is a calculus for a logic **L**, we mean that  $\Vdash_{\mathbf{G}}^{\{\mathbf{t}\}}$  is equal to the consequence relation of the logic **L**.

## 2.3 Proof-Theoretic Properties

In this section we define several important proof-theoretic properties of pure calculi. In the next chapter we will provide a semantic counterpart for each of these properties.

### 2.3.1 Analyticity

Analyticity is a crucial property of fully-structural propositional proof systems, as it usually implies its decidability and consistency (the fact that the empty sequent is not derivable). Roughly speaking, a sequent calculus is analytic if whenever a sequent s is provable in it, then s can be proven using only the syntactic material available inside s. Now, there is more than one way to precisely define the "material available within some sequent". Usually, it is taken to consist of all subformulas occurring in the sequent, and then analyticity amounts to the global subformula property (i.e., if there exists a proof of a sequent s, then there exists a proof of s using only its subformulas). However, it is also possible (and sometimes necessary, see, e.g., Example 3.3.4) to consider analyticity properties that are based on different relations defining the "material available within sequents". While these substitutes might be weaker than the global subformula property, they still suffice to imply the consistency and the decidability of a proof system. Next we define a generalized analyticity property, based on an arbitrary safe partial order.

**Definition 2.3.1.** Let  $\leq$  be a partial order on  $\mathcal{L}$ . For every formula  $\varphi$ , we denote by  $\downarrow^{\leq}[\varphi]$  the set  $\{\psi \in \mathcal{L} \mid \psi \leq \varphi\}$ . This notation is extended to sets of formulas, sequents, and sets of sequents in the natural way:  $\downarrow^{\leq}[\mathcal{T}] = \bigcup_{\varphi \in \mathcal{T}} \downarrow^{\leq}[\varphi]$  for a set  $\mathcal{T}$  of formulas;  $\downarrow^{\leq}[s] = \downarrow^{\leq}[frm[s]]$  for a sequent s; and  $\downarrow^{\leq}[\mathcal{S}] = \bigcup_{s \in \mathcal{S}} \downarrow^{\leq}[s]$  for a set  $\mathcal{S}$  of sequents.  $\leq$  is called safe if  $\downarrow^{\leq}[\varphi]$  is finite for every  $\varphi \in \mathcal{L}$ , and  $\lambda \varphi \in \mathcal{L}. \downarrow^{\leq}[\varphi]$  is computable.

Henceforth,  $\leq$  denotes an arbitrary safe partial order on  $\mathcal{L}$ . A particularly important one is the subformula relation (here we mean the reflexive-transitive closure of the direct subformula relation). For this relation we employ the following notation:

**Notation 2.3.2.** We denote by sub the subformula relation between formulas. In the case of sub, we simply write  $sub[\cdot]$  instead of  $\downarrow^{sub}[\cdot]$ .

**Definition 2.3.3.** Given a set  $\mathcal{F}$  of  $\mathcal{L}$ -formulas, a formula  $\varphi$  is called an  $\mathcal{F}$ -formula if  $\varphi \in \mathcal{F}$ . In turn, an  $\mathcal{L}$ -labelled  $\mathcal{F}$ -formula is an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}$ : $\varphi$  with  $\varphi \in \mathcal{F}$ ; and an  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent is an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent consisting only of  $\mathcal{L}$ -labelled  $\mathcal{F}$ -formulas.

**Notation 2.3.4.** For a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , we write  $\mathcal{S} \vdash_{\mathbf{G}}^{\mathcal{F}} s$  if there is a proof in  $\mathbf{G}$  of s from  $\mathcal{S}$  consisting only of  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequents.

**Definition 2.3.5.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is  $\leq$ -analytic if for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s:  $\vdash_{\mathbf{G}} s$  implies that  $\vdash_{\mathbf{G}}^{\downarrow \leq [s]} s$ .

The above notion of analyticity considers only proof from empty set of assumptions (speaking only about the theorems of the system). A strong version is defined as follows:

**Definition 2.3.6.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is  $strongly \leq -analytic$  if for every set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s: \mathcal{S} \vdash_{\mathbf{G}} s$  implies that  $\mathcal{S} \vdash_{\mathbf{G}}^{\downarrow \leq [\mathcal{S} \cup \{s\}]} s$ .

In the next chapters we focus only on strong  $\leq$ -analyticity. Obviously,  $\leq$ -analyticity follows from strong  $\leq$ -analyticity (take  $\mathcal{S} = \emptyset$ ). Next, we show that in the simple (and most common) case of two-sided calculi that include (cut) and (id) these two properties are actually equivalent. The main idea of this proof appeared already [6], where it was proved that cut-admissibility implies strong cut-admissibility (see definition below) for the specific case of  $\mathbf{L}\mathbf{K}$ .

**Theorem 2.3.7.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus, that includes (cut) and (id). If **G** is  $\leq$ -analytic then it is strongly  $\leq$ -analytic.

Proof. Suppose that  $\mathbf{G}$  is  $\leq$ -analytic. We show that  $\mathcal{S} \vdash_{\mathbf{G}} s$  implies  $\mathcal{S} \vdash_{\mathbf{G}}^{\downarrow \leq |\mathcal{S}|} s$ . Clearly, it suffices to prove this for finite  $\mathcal{S}$  (otherwise, take a finite subset  $\mathcal{S}^*$  of  $\mathcal{S}$  such that  $\mathcal{S}^* \vdash_{\mathbf{G}} s$ ). We use induction on the number of  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -sequents in  $\mathcal{S}$ . The case that  $\mathcal{S}$  is empty follows from our assumption. Suppose the claim holds when the number of sequents in  $\mathcal{S}$  is n, and let  $\mathcal{S}' = \{s_0, \dots, s_n\}$  be a set of n+1  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -sequents, such that  $\mathcal{S}' \vdash_{\mathbf{G}} s$ . Proposition 2.2.17 implies that  $\{s_0 \cup c, s_1, \dots, s_n\} \vdash_{\mathbf{G}} s \cup c$ , for every  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -sequent c. In particular, for every  $\mathcal{L}_2$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}: \varphi \in s_0$ ,  $\{s_0 \cup \{\overline{\mathbf{x}}: \varphi\}, s_1, \dots, s_n\} \vdash_{\mathbf{G}} s \cup \{\overline{\mathbf{x}}: \varphi\}$ , where  $\overline{\mathbf{f}} = \mathbf{t}$  and  $\overline{\mathbf{t}} = \mathbf{f}$ . Now, for every  $\mathbf{x}: \varphi \in s_0$ , the sequent  $s_0 \cup \{\overline{\mathbf{x}}: \varphi\}$  is derivable in  $\mathbf{G}$  using only (id) and weakenings, and therefore we have  $\{s_1, \dots, s_n\} \vdash_{\mathbf{G}} s \cup \{\overline{\mathbf{x}}: \varphi\}$ . By the induction hypothesis we obtain that  $\{s_1, \dots, s_n\} \vdash_{\mathbf{G}} s \cup \{\overline{\mathbf{x}}: \varphi\}$  for every  $\mathcal{L}_2$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}: \varphi \in s_0$ . The sequent s can then be inferred from these sequents and  $s_0$  by  $|s_0|$  applications of (cut) without introducing any formulas outside  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ .

The following are three major consequences of (strong)  $\leq$ -analyticity.

**Proposition 2.3.8** (Consistency). Let **G** be pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, which is  $\leq$ -analytic for some safe partial order  $\leq$ . Then,  $\not\vdash_{\mathbf{G}} \emptyset$ .

*Proof.* Assume that  $\vdash_{\mathbf{G}} \emptyset$ . Since  $\mathbf{G}$  is  $\leq$ -analytic, there exists a proof of the empty sequent using no formulas at all. The only way to have this is using the rule  $\emptyset/\emptyset$ , which was disallowed in pure systems (Convention 2.2.15).

**Proposition 2.3.9** (Conservativity). Let  $\mathcal{L}_1$  and  $\mathcal{L}_2$  be two propositional languages, such that  $\mathcal{L}_2$  is an extension of  $\mathcal{L}_1$  by some set of connectives. Let  $\mathbf{G_1}$  and  $\mathbf{G_2}$  be a pure  $\langle \mathcal{L}_1, \mathcal{L} \rangle$ -calculus and a pure  $\langle \mathcal{L}_2, \mathcal{L} \rangle$ -calculus (respectively). Assume that  $\mathbf{G_2}$  is obtained from  $\mathbf{G_1}$  by adding to the latter rules involving connectives in  $\mathcal{L}_2 \setminus \mathcal{L}_1$  (i.e., at least one connective in  $\mathcal{L}_2 \setminus \mathcal{L}_1$  appears in each rule in  $\mathbf{G_2} \setminus \mathbf{G_1}$ ). Let  $\leq$  be a safe partial order on  $\mathcal{L}_2$ , such that  $\mathcal{L}_1$  is closed under  $\leq$  (i.e.,  $\downarrow^{\leq} [\mathcal{L}_1] = \mathcal{L}_1$ ). If  $\mathbf{G_2}$  is strongly  $\leq$ -analytic, then  $\mathbf{G_2}$  is a conservative extension of  $\mathbf{G_1}$  (i.e., if  $frm[\mathcal{S} \cup \{s\}] \subseteq \mathcal{L}_1$  then  $\mathcal{S} \vdash_{\mathbf{G_1}} s$  iff  $\mathcal{S} \vdash_{\mathbf{G_2}} s$ ).

*Proof.* Obviously,  $\mathcal{S} \vdash_{\mathbf{G_1}} s$  implies  $\mathcal{S} \vdash_{\mathbf{G_2}} s$ . For the converse, assume that  $\mathcal{S} \vdash_{\mathbf{G_2}} s$ . Since  $\mathbf{G_2}$  is strongly  $\leq$ -analytic, there exists a proof in  $\mathbf{G_2}$  of s from  $\mathcal{S}$  consisting of  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ -formulas only. Since  $frm[\mathcal{S} \cup \{s\}] \subseteq \mathcal{L}_1$ , and  $\mathcal{L}_1$  is closed under  $\leq$ , this is also a proof in  $\mathbf{G_1}$ , and so  $\mathcal{S} \vdash_{\mathbf{G_1}} s$ .

**Proposition 2.3.10** (Decidability). Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. Suppose that **G** is strongly  $\leq$ -analytic for some safe partial order  $\leq$ . Then, given a finite set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, it is decidable whether  $\mathcal{S} \vdash_{\mathbf{G}} s$  or not.

Proof. Exhaustive proof-search is possible. Since **G** is strongly  $\leq$ -analytic,  $\mathcal{S} \vdash_{\mathbf{G}} s$  iff there exists a proof in **G** of s from  $\mathcal{S}$  consisting of  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ -sequents only. Since  $\leq$  is safe, one can construct the (finite) set  $\mathcal{S}'$  of all  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ -sequents. Clearly,  $\mathcal{S} \vdash_{\mathbf{G}} s$  iff there exists a proof in **G** of s from  $\mathcal{S}$  of length less than or equal to  $|\mathcal{S}'|$ , consisting only of sequents from  $\mathcal{S}'$ . Thus one can construct all possible candidates. By definition **G** is finite, and hence it is possible to check whether a certain candidate is indeed a proof in **G** of s from  $\mathcal{S}$ .

## 2.3.2 Cut-Admissibility

Usual two-sided sequent calculi include the rule (cut), which is very problematic from a proof-search perspective. The admissibility of (cut) (i.e. the fact that for every sequent  $s, \vdash_{\mathbf{G}} s$  implies that there is a cut-free proof in  $\mathbf{G}$  of s) is then desirable. However, forbidding all applications of cut rules seems to be too strong while dealing with arbitrary pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculi. Indeed, consider applications of a cut rule in which the cut-formula occurs inside the context sequents (i.e. inferring a sequent of the form  $c_1 \cup ... \cup c_n$  from the sequents  $(\mathbf{X}_1:\varphi) \cup c_1, ..., (\mathbf{X}_n:\varphi) \cup c_n$ , where  $\varphi \in frm[c_1 \cup ... \cup c_n]$ ). Such applications are not harmful for proof-search, as every formula in the conclusion of the application also occurs (as is) in one of its premises. These considerations lead to the following formulation of cut admissibility in pure calculi:

**Definition 2.3.11.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys *cut-admissibility* if for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s:  $\vdash_{\mathbf{G}} s$  implies that there exists proof P of s in **G** such that the cut-formula of every application of a cut rule in P occurs in one of the context sequents of that application.

As for analyticity, this notion of cut-admissibility refers only to proofs without assumptions (i.e., proofs from  $\emptyset$ ). Obviously, we cannot expect full cut-admissibility when the set of assumptions is not empty (in usual calculi, the only way to derive the empty sequent from  $\{f:p_1\}$  and  $\{t:p_1\}$  is using (cut)). Thus we consider the property called strong cut-admissibility in [6], which is formulated as follows in our framework:

**Definition 2.3.12.** A pure calculus G enjoys  $strong\ cut-admissibility$  if  $S \vdash_G s$  implies that there exists a proof P of s from S in G such that the cut-formula of every application of a cut rule in P occurs either in one of the context sequents of that application or in frm[S].

Obviously, cut-admissibility follows from strong cut-admissibility (take  $S = \emptyset$ ).

Equivalent definition of cut-admissibility and strong cut-admissibility are obtained by considering an enrichment of G with non-cut rules, so that all applications of the cut rules in which the cut-formula occurs in the context can be replaced by applications of the new rules. This is done as follows:

**Definition 2.3.13.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus.  $s(\mathbf{G})$  denotes the pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus obtained by augmenting **G** with the primitive  $\mathcal{L}$ -rules  $(X_1:p_1), \dots, (X_n:p_1)/\{x:p_1\}$  for every cut rule  $(X_1:p_1), \dots, (X_n:p_1)/\emptyset$  of **G** and  $\mathbf{x} \in \mathcal{L}$  such that  $\{\mathbf{x}\} \notin \{X_1, \dots, X_n\}$ .

**Example 2.3.14.** Let  $\mathcal{L} = \{f, i, t\}$  and suppose that is  $\{f:p_1\}, \{i:p_1\}/\emptyset$  is the only cut rule of **G**.  $s(\mathbf{G})$  is obtained by adding to **G** the primitive rule  $\{f:p_1\}, \{i:p_1\}/\{t:p_1\}$ .

Note that for a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus **G** whose only cut rule is (cut), we have that  $s(\mathbf{G}) = \mathbf{G}$ .

**Proposition 2.3.15.**  $\vdash_{s(\mathbf{G})} = \vdash_{\mathbf{G}}$  for every pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ .

*Proof.* All applications of the new rules in  $s(\mathbf{G})$  can be simulated in  $\mathbf{G}$  by applications of the corresponding cut rule, followed by an application of a weakening rule.

**Notation 2.3.16.** Given a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , we denote by  $\mathbf{G}_{cf}$  the calculus obtained from  $\mathbf{G}$  by discarding all cut rules of  $\mathbf{G}$ . In particular,  $s(\mathbf{G})_{cf}$  is the calculus obtained from  $\mathbf{G}$  by replacing every cut rule of the form  $(\mathbf{X}_1:p_1), \ldots, (\mathbf{X}_n:p_1)/\emptyset$  with all rules of the form  $(\mathbf{X}_1:p_1), \ldots, (\mathbf{X}_n:p_1)/\{\mathbf{x}:p_1\}$  such that  $\mathbf{x} \in \mathcal{L}$  and  $\{\mathbf{x}\} \notin \{\mathbf{X}_1, \ldots, \mathbf{X}_n\}$ .

**Notation 2.3.17.** Given a set  $\mathcal{C} \subseteq \mathcal{L}$ , a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , a set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, we write  $\mathcal{S} \vdash_{\mathbf{G}}^{cuts:\mathcal{C}} s$  if there exists a proof in  $\mathbf{G}$  of s from  $\mathcal{S}$  in which the cut-formula of every application of a cut rule is an element of  $\mathcal{C}$ .

**Proposition 2.3.18.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus.

- 1. **G** enjoys cut-admissibility iff  $\vdash_{s(\mathbf{G})_{cf}} s$  whenever  $\vdash_{s(\mathbf{G})} s$ .
- 2. **G** enjoys strong cut-admissibility iff  $\mathcal{S} \vdash_{s(\mathbf{G})}^{cuts:frm[\mathcal{S}]} s$  whenever  $\mathcal{S} \vdash_{s(\mathbf{G})} s$ .

*Proof.* Note that every application of a cut rule in  $\mathbf{G}$  in which the cut-formula occurs in the context sequents can be simulated in  $s(\mathbf{G})_{cf}$  (by using its new primitive rules or weakening). Similarly, every application of a new primitive  $\mathcal{L}$ -rule in  $s(\mathbf{G})_{cf}$  can be simulated in  $\mathbf{G}$  by applying weakening and the corresponding cut rule where the cut-formula occurs in the context sequents. The claims then follow from the definitions.  $\square$ 

It follows that a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus G whose only cut rule is (cut) (and thus s(G) = G) enjoys cut-admissibility iff  $\vdash_{\mathbf{G}} \subseteq \vdash_{\mathbf{G}_{cf}}$ . Such a calculus enjoys strong cut-admissibility iff  $\mathcal{S} \vdash_{\mathbf{G}} s$  implies that there exists a proof of s from  $\mathcal{S}$  in G such that the cut-formula of every application of a cut rule is an element of  $frm[\mathcal{S}]$ . Hence Definition 2.3.11 and Definition 2.3.12 indeed generalize the known notions for ordinary two-sided sequent calculi. In addition, in ordinary two-sided calculi that include (cut) and (id) cut-admissibility is equivalent to strong cut-admissibility (like in the case of analyticity, see Theorem 2.3.7).

**Theorem 2.3.19.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus, that includes (cut) and (id). If **G** enjoys cut-admissibility then **G** enjoys strong cut-admissibility.

*Proof.* Similar to the proof of Theorem 2.3.7.

## 2.3.3 Axiom-Expansion

Another property which is often studied in two-sided sequent calculi (that include (id)) is the property of axiom-expansion [44]. This property means that non-atomic applications of (id) (deriving sequents of the form  $\{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}$  where  $\varphi$  is not atomic) are redundant.<sup>4</sup> In our broader context it can be formulated as follows:

Notation 2.3.20. Given a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ ,  $\mathbf{G}_{if}$  denotes the pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus obtained from  $\mathbf{G}$  by discarding all identity axioms of  $\mathbf{G}$ .

 $<sup>^4</sup>$ The term "axiom-expansion" is commonly used, but it is somewhat unfortunate. In fact, this property concerns the reducibility of arbitrary axioms to atomic ones.

**Definition 2.3.21.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys *axiom-expansion* if

$$\{(\mathbf{X}:p)\mid\emptyset/(\mathbf{X}:p_1)\in\mathsf{P}_{\mathbf{G}},p\in\mathit{at}_{\mathcal{L}}\}\vdash_{\mathbf{G}_{if}}(\mathbf{Y}:\varphi)$$

for every  $\emptyset/(Y:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $\varphi \in \mathcal{L}$ .

It is easy to see that we have the following:

**Proposition 2.3.22.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys axiom-expansion if

$$\{(\mathbf{X}:p_i) \mid \emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}, 1 \leq i \leq ar(\diamond)\} \vdash_{\mathbf{G}_{if}} (\mathbf{Y}: \diamond (p_1, \dots, p_{ar(\diamond)}))$$

for every  $\emptyset/(Y:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $\diamond \in \diamondsuit_{\mathcal{L}}$ .

*Proof.* A simple inductive argument (using Proposition 2.2.17) suffices.

Thus, following [44], we define this property for a given connective as follows:

**Definition 2.3.23.** A connective  $\diamond \in \diamond_{\mathcal{L}}$  admits axiom-expansion in a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$  if  $\{(X:p_i) \mid \emptyset/(X:p_1) \in \mathsf{P}_{\mathbf{G}}, 1 \leq i \leq ar(\diamond)\} \vdash_{\mathbf{G}_{if}} (Y:\diamond(p_1, \ldots, p_{ar(\diamond)}))$  for every  $\emptyset/(Y:p_1) \in \mathsf{P}_{\mathbf{G}}$ .

**Remark 2.3.24.** Unlike [44], we do not require that there exists a *cut-free* proof of  $(Y: \diamond (p_1, \ldots, p_{ar(\diamond)}))$ .

Note that a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys axiom-expansion iff each connective of  $\mathcal{L}$  admits axiom-expansion in **G**.

**Example 2.3.25.** It is easy to see that each connective of  $\mathcal{L}_{cl}$  admits axiom-expansion in **LK**, and thus **LK** enjoys axiom-expansion.

# Chapter 3

# Semantics for Pure Sequent Calculi

In this chapter we introduce a method for providing (possibly non-deterministic) many-valued semantics for any given pure calculus. The semantics is then used to characterize the proof-theoretic properties of pure calculi that were introduced in the previous chapter. It should be noted that while dealing with the general framework of pure calculi, one cannot expect to obtain effective semantics in all cases. Indeed, the proposed semantics is quite close to the input proof system. However, it provides a complementary presentation of pure calculi, sheds light on their syntactic properties, reveals deep useful connections between semantics and proof theory, and turns out to be useful for proving these properties in particular examples. In addition, applying the tools of this chapter for narrower families of pure calculi (as done in the next chapter for canonical calculi) leads to effective semantics and decidable criteria for proof-theoretic properties.<sup>1</sup>

#### Publications Related to this Chapter

The material in this chapter was not published before.

## 3.1 The Semantic Framework

The proposed semantic framework is a denotational semantics, based on *valuations*, which are simply functions whose domain is  $\mathcal{L}$ :

<sup>&</sup>lt;sup>1</sup>This inherent limitation of an investigation of such general frameworks was summarized by J.Y. Béziau in [34] as follows:

<sup>&</sup>quot;What is involved in this paper is mainly general abstract nonsense. The main difficulty of our results is rather conceptual. But what we show is that when this general abstract nonsense is rightly organized we can get meaningful results with a lot of powerful applications."

**Definition 3.1.1.** An  $\mathcal{L}$ -valuation is a function v from  $\mathcal{L}$  to some set  $\mathcal{U}$  of truth values. We denote by  $Im_v$  the image of v.

Note that we do not specify what are truth values, and in fact anything can serve as a truth value. To define restrictions on  $\mathcal{L}$ -valuations we introduce  $\mathcal{L}$ -semantic conditions.

#### **Definition 3.1.2.** Let $\mathcal{U}$ be some set of truth values.

- 1. An  $\mathcal{L}$ -semantic disjunction over  $\mathcal{U}$  is a finite set of pairs  $\langle \varphi, u \rangle$ , denoted by  $\varphi \doteq u$ , where  $u \in \mathcal{U}$  and  $\varphi \in \mathcal{L}$ .  $\mathcal{L}$ -substitutions are naturally extended to  $\mathcal{L}$ -semantic disjunctions by  $\sigma(I) = \{\sigma(\varphi) \doteq u \mid \varphi \doteq u \in I\}$ .
- 2. An  $\mathcal{L}$ -semantic condition over  $\mathcal{U}$  is a pair  $\langle \mathcal{I}, I \rangle$ , denoted by  $\mathcal{I} \Rightarrow I$ , where  $\mathcal{I}$  is a finite set of  $\mathcal{L}$ -semantic disjunctions over  $\mathcal{U}$ , and I is a single  $\mathcal{L}$ -semantic disjunction over  $\mathcal{U}$ .
- 3. An  $\mathcal{L}$ -valuation v with  $Im_v \subseteq \mathcal{U}$  satisfies:
  - (a) an  $\mathcal{L}$ -semantic disjunction I over  $\mathcal{U}$  if  $\varphi = v(\varphi) \in I$  for some  $\varphi \in \mathcal{L}$ .
  - (b) an  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I$  over  $\mathcal{U}$  if for every  $\mathcal{L}$ -substitution  $\sigma$ , v satisfies  $\sigma(I)$  whenever it satisfies  $\sigma(I')$  for every  $I' \in \mathcal{I}$ .
  - (c) a set  $\Lambda$  of  $\mathcal{L}$ -semantic conditions over  $\mathcal{U}$  if it satisfies every  $\mathcal{I} \Rightarrow I \in \Lambda$ .

We write  $v \models X$  to denote that v satisfies X, where X is either an  $\mathcal{L}$ -semantic disjunction, an  $\mathcal{L}$ -semantic condition, or a set of  $\mathcal{L}$ -semantic conditions.<sup>2</sup>

**Example 3.1.3.** Suppose that  $\diamond, \triangleright \in \diamondsuit^2_{\mathcal{L}}$ , and let  $\mathcal{U} = \{u_1, u_2\}$ . Consider the semantic disjunctions:  $I_1 = \{p_1 \doteqdot u_1, p_2 \doteqdot u_2\}$ , and  $I_2 = \{p_1 \diamond p_2 \doteqdot u_1, p_2 \triangleright p_1 \doteqdot u_1\}$ . Let v be an  $\mathcal{L}$ -valuation with  $Im_v \subseteq \mathcal{U}$ . Then,  $v \models I_1$  iff  $v(p_1) = u_1$  or  $v(p_2) = u_2$ .  $v \models I_2$  iff  $v(p_1 \diamond p_2) = u_1$  or  $v(p_2 \triangleright p_1) = u_1$ . v satisfies the  $\mathcal{L}$ -semantic condition  $\{I_1\} \Rightarrow I_2$  iff for every  $\varphi_1, \varphi_2 \in \mathcal{L}$ : if  $v(\varphi_1) = u_1$  or  $v(\varphi_2) = u_2$ , then  $v(\varphi_1 \diamond \varphi_2) = u_1$  or  $v(\varphi_2 \triangleright \varphi_1) = u_1$ .

**Example 3.1.4.** Obviously, restrictions arising from "truth tables" can be represented as semantic conditions. For example, to capture the classical truth table of implication, we use the following conditions over  $\{f,t\}$ :  $\{\{p_1 \doteq t\}, \{p_2 \doteq f\}\} \Rightarrow \{p_1 \supset p_2 \doteq f\}$  and  $\{\{p_1 \doteq f, p_2 \doteq t\}\} \Rightarrow \{p_1 \supset p_2 \doteq t\}$ . An  $\mathcal{L}$ -valuation satisfies these two  $\mathcal{L}$ -semantic conditions iff it respects the usual truth table of  $\supset$ .

To obtain semantic characterizations of logics we introduce a class of structures called many-valued systems, that generalizes the usual notion of a many-values matrix (see, e.g., [93]), by allowing arbitrary semantic conditions.

<sup>&</sup>lt;sup>2</sup>Note that  $v \models \emptyset$  is ambiguous: it holds for the empty set of  $\mathcal{L}$ -semantic conditions, and does not hold for the empty  $\mathcal{L}$ -semantic disjunction.

### **Definition 3.1.5.** A many-valued system M for $\mathcal{L}$ consists of:

- 1. A set  $\mathcal{V}_{\mathbf{M}}$  of truth values.
- 2. A subset  $\mathcal{U}_{\mathbf{M}} \subseteq \mathcal{V}_{\mathbf{M}}$  of legal truth values.
- 3. A subset  $\mathcal{D}_{\mathbf{M}} \subseteq \mathcal{V}_{\mathbf{M}}$  of designated truth values.
- 4. A set  $\Lambda_{\mathbf{M}}$  of  $\mathcal{L}$ -semantic conditions over  $\mathcal{V}_{\mathbf{M}}$ .

 $\mathbf{M}$  is called *finite* if so are  $\mathcal{V}_{\mathbf{M}}$  and  $\Lambda_{\mathbf{M}}$ .

### **Definition 3.1.6.** Let M be a many-valued system for $\mathcal{L}$ .

- 1. An  $\mathcal{L}$ -valuation v is called  $\mathbf{M}$ -legal if  $Im_v \subseteq \mathcal{U}_{\mathbf{M}}$  and  $v \models \Lambda_{\mathbf{M}}$ .
- 2. An  $\mathcal{L}$ -valuation v is said to be a *model* (with respect to  $\mathbf{M}$ ) of:
  - (a) an  $\mathcal{L}$ -formula  $\varphi$ , written  $v \models^{\mathbf{M}} \varphi$ , if  $v(\varphi) \in \mathcal{D}_{\mathbf{M}}$ .
  - (b) a set  $\mathcal{T}$  of  $\mathcal{L}$ -formulas, written  $v \models^{\mathbf{M}} \mathcal{T}$ , if  $v \models^{\mathbf{M}} \varphi$  for every  $\varphi \in \mathcal{T}$ .
- 3. An  $\mathcal{L}$ -formula  $\varphi$  follows from a set  $\mathcal{T}$  of  $\mathcal{L}$ -formulas with respect to  $\mathbf{M}$  (denoted by:  $\mathcal{T} \Vdash_{\mathbf{M}} \varphi$ ) if for every  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation  $v: v \models^{\mathbf{M}} \varphi$  whenever  $v \models^{\mathbf{M}} \mathcal{T}$ .

**Proposition 3.1.7.** For every many-valued system M for  $\mathcal{L}$ ,  $\langle \mathcal{L}, \Vdash_{\mathbf{M}} \rangle$  is a logic.

*Proof.* Easily follows from the definitions. To prove that  $\Vdash_{\mathbf{M}}$  is structural, note that if v is an  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation, then so is  $v \circ \sigma$  for every  $\mathcal{L}$ -substitution  $\sigma$ .

**Example 3.1.8.** Classical logic  $\langle \mathcal{L}_{cl}, \Vdash_{cl} \rangle$  is obtained by taking a many-valued system  $\mathbf{M}_{cl}$  with  $\mathcal{V}_{\mathbf{M}_{cl}} = \{f, t\}$ ,  $\mathcal{D}_{\mathbf{M}_{cl}} = \{t\}$ , and  $\Lambda_{\mathbf{M}_{cl}}$  consist of the semantic conditions over  $\{f, t\}$  that correspond to the classical truth tables (e.g. as in Example 3.1.4).

**Example 3.1.9.** Many-valued systems generalize the notion of a logical many-valued matrix [93, 61]. Thus any many-valued logic that is defined by such a matrix is captured in this general framework. In addition, non-deterministic many-valued matrices [17, 21] can be easily presented as particular cases of many-valued systems. This will be discussed in Chapter 4.

**Example 3.1.10.** The framework of *bivaluations* [34, 40] corresponds to *two*-valued systems (that is, many-valued systems with  $|\mathcal{V}_{\mathbf{M}}| = |\mathcal{U}_{\mathbf{M}}| = 2$  and  $|\mathcal{D}_{\mathbf{M}}| = 1$ . In addition, the dyadic semantics of [39] is also a subclass of two-valued many-valued systems.

In many cases in the sequel, we need a many-valued system M just to specify a set of (M-legal) valuations, rather than to define a logic. The set  $\mathcal{D}_{M}$  of designated truth values is redundant in these cases, and can be discarded. The obtained structures will be called many-valued pre-systems:

**Definition 3.1.11.** A many-valued pre-system M for  $\mathcal{L}$  is defined exactly like a many-valued system (Definition 3.1.5), except that we exclude the set  $\mathcal{D}_{M}$  of designated truth values.

Remark 3.1.12. Evidently, it suffices to consider many-valued systems with  $\mathcal{U}_{\mathbf{M}} = \mathcal{V}_{\mathbf{M}}$ . Indeed, given a many-valued system  $\mathbf{M}$ , we can always define a many-valued system  $\mathbf{M}'$  by  $\mathcal{V}_{\mathbf{M}'} = \mathcal{U}_{\mathbf{M}}$ ,  $\mathcal{D}_{\mathbf{M}'} = \mathcal{D}_{\mathbf{M}} \cap \mathcal{U}_{\mathbf{M}}$ , and  $\Lambda_{\mathbf{M}'}$  is obtained from  $\Lambda_{\mathbf{M}}$  by discarding all occurrences of pairs  $\varphi \doteqdot u$  with  $u \notin \mathcal{U}_{\mathbf{M}}$  from the semantic conditions. Clearly,  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuations are exactly  $\mathbf{M}'$ -legal  $\mathcal{L}$ -valuations, and we also have  $\Vdash_{\mathbf{M}} = \Vdash_{\mathbf{M}'}$ . However, we find the distinction between  $\mathcal{U}_{\mathbf{M}}$  and  $\mathcal{V}_{\mathbf{M}}$  technically convenient, as it allows us to change the set of legal truth values in many-valued (pre-) systems, without changing any of its other components. This is mainly beneficial for the modularity of the constructions below.

### 3.1.1 Partial Valuations and Semantic Analyticity

An important attractive property that we would like a semantic framework to have is effectiveness, namely the fact that it can be used to provide a semantic decision procedure for the logics it induces. The framework of many-valued systems is too wide to have this property in general. In this section we identify a sufficient condition for the the effectiveness of a many-valued system. This condition will also play a main role below for characterizing  $\leq$ -analyticity in pure sequent calculi.

Generally speaking, the naive approach to check whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  for a many-valued system  $\mathbf{M}$  (given a finite set of formulas  $\Gamma$  and a formula  $\varphi$ ) would be to consider one by one all possible  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuations, and return "true" iff none of them is a *counter-model* – a model of  $\Gamma$  but not of  $\varphi$  (with respect to  $\mathbf{M}$ ). Obviously, this cannot serve as a decision procedure since there are infinitely many  $\mathcal{L}$ -valuations to check, and each of them is infinite. Thus, as is usually done in decision procedures based on denotational semantic frameworks, one has to consider partial valuations defined only on the syntactic material included in  $\Gamma$  and  $\varphi$ . This, however, requires that the existence of a countermodel in the form of a partial valuation always indicates the existence of an (infinite) full counter-model. Obviously, this requirement holds when every partial valuation can be extended to a full one. Next, we define partial valuations, and precisely formulate these observations.

**Definition 3.1.13.** A partial  $\mathcal{L}$ -valuation is a function v from some set  $Dom_v \subseteq \mathcal{L}$  to some set  $\mathcal{U}$  of truth values. We denote by  $Im_v$  the image of v.

The previous notions for  $\mathcal{L}$ -valuations are adapted to partial  $\mathcal{L}$ -valuations as follows:

### **Definition 3.1.14.** A partial $\mathcal{L}$ -valuation v satisfies:

- 1. an  $\mathcal{L}$ -semantic disjunction I if  $\varphi = v(\varphi) \in I$  for some  $\varphi \in Dom_v$ .
- 2. an  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I$  if for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\sigma(\varphi) \in Dom_v$  and for every  $\varphi$  that occurs in  $\mathcal{I} \Rightarrow I$ , v satisfies  $\sigma(I)$  whenever it satisfies  $\sigma(I')$  for every  $I' \in \mathcal{I}$ .
- 3. a set  $\Lambda$  of  $\mathcal{L}$ -semantic conditions if it satisfies every  $\mathcal{I} \Rightarrow I \in \Lambda$ .

We write  $v \models X$  to denote that v satisfies X, where X is either an  $\mathcal{L}$ -semantic disjunction, an  $\mathcal{L}$ -semantic condition, or a set of  $\mathcal{L}$ -semantic conditions.

Given a many-valued pre-system  $\mathbf{M}$  for  $\mathcal{L}$ ,  $\mathbf{M}$ -legal partial  $\mathcal{L}$ -valuations are defined exactly as  $\mathbf{M}$ -legal (full)  $\mathcal{L}$ -valuations (i.e.  $Im_v \subseteq \mathcal{U}_{\mathbf{M}}$  and  $v \models \Lambda_{\mathbf{M}}$ ). Note that Definitions 3.1.13 and 3.1.14 generalize the corresponding notions defined above for (full)  $\mathcal{L}$ -valuations. Indeed, by taking  $Dom_v = \mathcal{L}$ , we obtain exactly the definitions for  $\mathcal{L}$ -valuations.

**Definition 3.1.15.** Let v and v' be two partial  $\mathcal{L}$ -valuations. We say that v' extends v if  $Dom_v \subseteq Dom_{v'}$  and  $v'(\varphi) = v(\varphi)$  for every  $\varphi \in Dom_v$ .

**Definition 3.1.16.** Let  $\leq$  be a partial order on  $\mathcal{L}$ . A many-valued (pre-) system  $\mathbf{M}$  is called  $\leq$ -analytic if any  $\mathbf{M}$ -legal partial  $\mathcal{L}$ -valuation whose domain is finite and closed under  $\leq$  can be extended to an  $\mathbf{M}$ -legal (full)  $\mathcal{L}$ -valuation.

**Example 3.1.17.** Revisiting the many-valued system  $\mathbf{M}_{cl}$  from Example 3.1.8 for classical logic, we note that this system is sub-analytic. Indeed,  $\mathbf{M}_{cl}$ -legal partial  $\mathcal{L}_{cl}$ -valuation, whose domain is finite and closed under  $\leq$ , are usual classical partial valuations which can be obviously extended to full classical valuations (i.e.  $\mathbf{M}_{cl}$ -legal  $\mathcal{L}_{cl}$ -valuations).

Note that we use the same term " $\leq$ -analytic" in two different contexts. When referring to many-valued (pre-) systems as  $\leq$ -analytic we mean the semantic extension property defined above (the term "analyticity" was used to describe a similar property in previous works, see e.g. [21]). On the other hand, we call a pure sequent system  $\leq$ -analytic if it satisfies the syntactic property given in Definition 2.3.5. In Theorem 3.3.3 below we establish a correspondence between these two notions of analyticity.

Next, we prove that  $\leq$ -analyticity (for a safe relation  $\leq$ ) suffices for the effectiveness of a given many-valued system.

**Theorem 3.1.18.** Let  $\mathbf{M}$  be a finite many-valued system for  $\mathcal{L}$ . Suppose that  $\mathbf{M}$  is  $\leq$ -analytic for some safe partial order  $\leq$  on  $\mathcal{L}$ . Given a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  or not.

Proof. Since  $\leq$  is safe,  $\downarrow^{\leq}[\Gamma \cup \{\varphi\}]$  is finite. Thus, to decide whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  one can enumerate all partial  $\mathcal{L}$ -valuations v with  $Dom_v = \downarrow^{\leq}[\Gamma \cup \{\varphi\}]$  and  $Im_v = \mathcal{U}_{\mathbf{M}}$ , and check if one of them satisfies the following three conditions: (1) v is  $\mathbf{M}$ -legal; (2)  $v \models^{\mathbf{M}} \Gamma$ ; and (3)  $v \not\models^{\mathbf{M}} \varphi$ . Each of these conditions is obviously decidable for a given v. We claim that  $\Gamma \Vdash_{\mathbf{M}} \varphi$  iff such a function is not found. To see this, note that if  $\Gamma \not\Vdash_{\mathbf{M}} \varphi$ , then by definition there exists an  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation v' such that  $v' \models^{\mathbf{M}} \Gamma$  but  $v' \not\models^{\mathbf{M}} \varphi$ . Its restriction to  $\downarrow^{\leq}[\Gamma \cup \{\varphi\}]$  is a function  $v: \downarrow^{\leq}[\Gamma \cup \{\varphi\}] \to \mathcal{U}_{\mathbf{M}}$  satisfying the conditions above. On the other hand, if there exists such a function v, then since  $\mathbf{M}$  is  $\leq$ -analytic, v can be extended to an  $\mathbf{M}$ -legal (full)  $\mathcal{L}$ -valuation v'. Clearly,  $v' \models^{\mathbf{M}} \Gamma$  but  $v' \not\models^{\mathbf{M}} \varphi$ . Consequently,  $\Gamma \not\Vdash_{\mathbf{M}} \varphi$  in this case.

Examining the proof above, we are able to provide a slightly weaker requirement:

**Theorem 3.1.19.** Let  $\mathbf{M}$  be a finite many-valued system for  $\mathcal{L}$ , and  $\leq$  a safe partial order on  $\mathcal{L}$ . Suppose that given an  $\mathbf{M}$ -legal partial  $\mathcal{L}$ -valuation v, whose domain is finite and closed under  $\leq$ , it is decidable whether v can be extended to an  $\mathbf{M}$ -legal (full)  $\mathcal{L}$ -valuation or not. Then, given a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  or not.

*Proof.* The proof goes as the proof of Theorem 3.1.18, with the addition of a forth condition: (4) v can be extended to an M-legal  $\mathcal{L}$ -valuation.

In Chapters 4 and 5 we will use this theorem to prove the decidability of a large family of logics induced by many-valued systems of a certain restricted form (of which many-valued matrices and their non-deterministic counterparts are particular instances). The decidability of important subfamilies of pure sequent calculi will be obtained as a consequence.

Remark 3.1.20. In the literature of non-deterministic matrices (see, e.g., [9]) effectiveness is usually identified with (semantic) analyticity. However, the observations above show that this property is not a necessary condition for decidability. To guarantee the latter, instead of requiring that *all* partial valuations are extendable, it is sufficient to have an algorithm that establishes which of them are.

## 3.2 Semantics for Pure Sequent Calculi

In this section we show that the logics induced by pure calculi can be semantically characterised by finite many-valued systems. Thus, our goal is to construct a finite many-valued system  $\mathbf{M}_{\mathbf{G}}^{\mathbf{X}}$  for a given pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$  and a set of labels  $\mathbf{X} \subseteq \mathcal{L}$ , for which we would have  $\Vdash_{\mathbf{G}}^{\mathbf{X}} = \Vdash_{\mathbf{M}_{\mathbf{G}}^{\mathbf{X}}}$ . To obtain this, we begin with a construction of a

many-valued pre-system  $\mathbf{M}_{\mathbf{G}}$  for a given pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , together with a definition of when an  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation v is a model of an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s. Then we show that  $\mathbf{M}_{\mathbf{G}}$  induces the same consequence relation between sequents that is induced by the calculus  $\mathbf{G}$ . Our construction is modular: each syntactic ingredient corresponds to a certain semantic component, and the semantics of the whole calculus is obtained by joining all semantic components. We start with precise definitions of each component.

**Truth Values** Intuitively, the truth value assigned to an  $\mathcal{L}$ -formula  $\varphi$  should carry enough information to determine for which labels  $\mathbf{x} \in \mathcal{L}$  the  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x} : \varphi$  is "true". In general, there can be  $2^{|\mathcal{L}|}$  options for that. Thus we take the truth values in the many-valued system for a given pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$  to be the subsets of  $\mathcal{L}$  (that is,  $\mathcal{V}_{\mathbf{M}_{\mathbf{G}}} = 2^{\mathcal{L}}$ ), and use the following definition:

**Definition 3.2.1.** An  $\mathcal{L}$ -valuation v with  $Im_v \subseteq 2^{\mathcal{L}}$  is said to be a *model* of an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}:\varphi$ , written  $v \models \mathbf{x}:\varphi$ , if  $\mathbf{x} \in v(\varphi)$ .

Note that the last definition concerns an  $\mathcal{L}$ -valuation v with  $Im_v \subseteq 2^{\mathcal{L}}$  regardless of a many-valued system. Clearly, given such an  $\mathcal{L}$ -valuation v and a formula  $\varphi$ ,  $v(\varphi) = \{\mathbf{x} \in \mathcal{L} \mid v \models \mathbf{x}:\varphi\}$ . Thus if we have a many-valued system  $\mathbf{M}$  for  $\mathcal{L}$  with  $\mathcal{V}_{\mathbf{M}} = 2^{\mathcal{L}}$ , then  $v \models^{\mathbf{M}} \varphi$  (according to Definition 3.1.6) iff the set  $\{\mathbf{x} \in \mathcal{L} \mid v \models \mathbf{x}:\varphi\}$  is in  $\mathcal{D}_{\mathbf{M}}$ . In turn, sequents are intuitively interpreted as disjunctions of labelled formulas, and sets of sequents (that constitute the sets of assumptions) are conjunctions of sequents.

**Definition 3.2.2.** An  $\mathcal{L}$ -valuation v with  $Im_v \subseteq 2^{\mathcal{L}}$  is said to be a *model* of:

- 1. an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, written  $v \models s$ , if  $v \models \alpha$  for some  $\alpha \in s$ .
- 2. a set S of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, written  $v \models S$ , if  $v \models s$  for every  $s \in S$ .

These notions naturally lead to the following definition of the consequence relation between sequents induced by a many-valued (pre-) system with  $\mathcal{V}_{\mathbf{M}} = 2^{\pounds}$ :

**Definition 3.2.3.** Let  $\mathbf{M}$  be a many-valued (pre-) system for  $\mathcal{L}$  with  $\mathcal{V}_{\mathbf{M}} = 2^{\mathcal{L}}$ . An  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s follows from a set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents with respect to  $\mathbf{M}$  (denoted by:  $\mathcal{S} \vdash_{\mathbf{M}} s$ ) if every  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation which is a model of  $\mathcal{S}$  is also a model of s.

**Remark 3.2.4.** The general framework of many-valued (pre-) systems presented above allows anything to serve as a truth value. However, a semantic consequence relation  $\vdash_{\mathbf{M}}$  between sequents is defined here only for many-valued (pre-) systems whose truth values consist of sets of labels ( $\mathcal{V}_{\mathbf{M}} = 2^{\pounds}$ ).

The soundness of the weakening rules directly follows from the definitions:

**Proposition 3.2.5.**  $\{s\} \vdash_{\mathbf{M}} s \cup \{\alpha\}$  for every many-valued (pre-) system  $\mathbf{M}$  for  $\mathcal{L}$  with  $\mathcal{V}_{\mathbf{M}} = 2^{\mathcal{L}}$ ,  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, and  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formulas  $\alpha$ .

**Legal Truth Values** While  $\mathcal{V}_{\mathbf{M}_{\mathbf{G}}}$  is taken to be all subsets of  $\mathcal{L}$ , the set  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  of legal truth values (those that are actually allowed to be used in  $\mathbf{M}_{\mathbf{G}}$ -legal valuations) is determined according to the primitive rules of  $\mathbf{G}$ . Indeed, each primitive rule of  $\mathbf{G}$  forbid some of the  $2^{|\mathcal{L}|}$  options. For example, say that  $\mathbf{f}, \mathbf{i} \in \mathcal{L}$ , and consider the primitive rule  $\{\mathbf{f}:p_1\}/\{\mathbf{i}:p_1\}$ . Semantically, this rule means that an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula of the form  $\mathbf{i}:\varphi$  should be "true" whenever  $\mathbf{f}:\varphi$  is "true". This semantic requirement can be easily reflected by disallowing truth values that include the label  $\mathbf{f}$  but not  $\mathbf{i}$ . We will denote by  $\mathcal{L}(r)$  the set of subsets of  $\mathcal{L}$  that are not forbidden by the primitive rule r. Formally,  $\mathcal{L}(r)$  is defined as follows:

**Definition 3.2.6.** Let  $r = (X_1:p_1), \dots, (X_n:p_1)/(X:p_1)$  be a primitive  $\mathcal{L}$ -rule. Then:

$$\pounds(r) = \{ \mathbf{Y} \subseteq \pounds \mid \mathbf{X}_i \cap \mathbf{Y} = \emptyset \text{ for some } 1 \leq i \leq n \text{ or } \mathbf{X} \cap \mathbf{Y} \neq \emptyset \}.$$

This definition is naturally extended to sets R of primitive  $\mathcal{L}$ -rules by:  $\mathcal{L}(R) = \bigcap_{r \in R} \mathcal{L}(r)$ .

**Example 3.2.7.** Let  $\mathcal{L} = \{ f, i, t \}$ . For a primitive  $\mathcal{L}$ -rule  $r = (\{ f, t \} : p_1), \{ i : p_1 \} / \{ t : p_1 \}, \{ f, i \}$ .

**Example 3.2.8.** For a cut rule  $r = (X_1:p_1), \dots, (X_n:p_1)/\emptyset$ ,

$$\pounds(r) = \{ \mathbf{Y} \subseteq \pounds \mid \mathbf{X}_i \cap \mathbf{Y} = \emptyset \text{ for some } 1 \le i \le n \}.$$

For an identity axiom  $r = \emptyset/(X:p_1)$ ,

$$\pounds(r) = \{ \mathbf{Y} \subset \pounds \mid \mathbf{X} \cap \mathbf{Y} \neq \emptyset \}.$$

In particular,  $\mathcal{L}(\emptyset/(\mathcal{L}:p_1)) = 2^{\mathcal{L}} \setminus \{\emptyset\}$ . Note that if a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** has at least one cut rule then  $\mathcal{L} \not\in \mathcal{L}(\mathsf{P}_{\mathbf{G}})$ , and similarly, if **G** has at least one identity axiom then  $\emptyset \not\in \mathcal{L}(\mathsf{P}_{\mathbf{G}})$ .

For a given pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , we will take  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \mathcal{L}(\mathsf{P}_{\mathbf{G}})$  (recall that  $\mathsf{P}_{\mathbf{G}}$  denotes the set of primitive rules of  $\mathbf{G}$ ).

**Example 3.2.9.** For a pure  $\langle \mathcal{L}, \pounds_2 \rangle$ -calculus **G** with  $\mathsf{P}_{\mathbf{G}} \subseteq \{(cut), (id)\}$  we have:

$$\pounds(\mathsf{P}_{\mathbf{G}}) = \begin{cases} \{\{\mathbf{f}\}, \{\mathbf{t}\}\} & \mathsf{P}_{\mathbf{G}} = \{(cut), (id)\} \\ \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}\} & \mathsf{P}_{\mathbf{G}} = \{(cut)\} \\ \{\{\mathbf{f}\}, \{\mathbf{t}\}, \{\mathbf{f}, \mathbf{t}\}\} & \mathsf{P}_{\mathbf{G}} = \{(id)\} \\ \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}, \{\mathbf{f}, \mathbf{t}\}\} & \mathsf{P}_{\mathbf{G}} = \emptyset \end{cases}$$

Thus for an ordinary pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus with  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$  we get a two-valued semantics; for pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus with  $\mathsf{P}_{\mathbf{G}} = \{(id)\}$  or  $\mathsf{P}_{\mathbf{G}} = \{(cut)\}$  we get a three-valued semantics; and if  $\mathsf{P}_{\mathbf{G}} = \emptyset$  we obtain a four-valued semantics.

**Semantic Conditions** The semantic conditions in  $\Lambda_{\mathbf{M}_{\mathbf{G}}}$  are straightforwardly derived from the rules in  $\mathsf{R}_{\mathbf{G}}$  according to the next definitions (recall that  $\mathsf{R}_{\mathbf{G}}$  denotes the set of non-primitive rules of  $\mathbf{G}$ ):

**Definition 3.2.10.**  $I(\cdot)$ , the  $\mathcal{L}$ -semantic disjunction over  $2^{\mathcal{L}}$  induced by:

- 1. an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}:\varphi$ , is defined by  $I(\mathbf{x}:\varphi) = \{\varphi \neq \mathbf{X} \mid \{\mathbf{x}\} \subseteq \mathbf{X} \subseteq \mathcal{L}\}.$
- 2. an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, is defined by  $I(s) = \bigcup_{\alpha \in s} I(\alpha)$ .

**Definition 3.2.11.** The  $\mathcal{L}$ -semantic condition over  $2^{\mathcal{L}}$  induced by a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $\mathcal{S}/s$ , denoted by  $\mathbf{Sem}(\mathcal{S}/s)$ , is defined by:

$$\mathbf{Sem}(\mathcal{S}/s) = \{ I(s') \mid s' \in \mathcal{S} \} \Rightarrow I(s).$$

This definition is extended to sets R of pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules in the obvious way:

$$\mathbf{Sem}(R) = {\mathbf{Sem}(r) \mid r \in R}.$$

**Example 3.2.12.** Suppose that  $\supset \in \diamondsuit^2_{\mathcal{L}}$ , and consider the usual  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules for  $\supset$ :  $(\mathfrak{f}: \supset) = \{\mathfrak{t}: p_1\}, \{\mathfrak{f}: p_2\}/\{\mathfrak{f}: p_1 \supset p_2\}$  and  $(\mathfrak{t}: \supset) = \{\mathfrak{f}: p_1, \mathfrak{t}: p_2\}/\{\mathfrak{t}: p_1 \supset p_2\}$  (see Example 2.2.6). Then:

$$\mathbf{Sem}((\mathbf{f}: \supset)) = \{ \{p_1 \doteqdot \{\mathbf{t}\}, p_1 \doteqdot \{\mathbf{f}, \mathbf{t}\}\}, \{p_2 \doteqdot \{\mathbf{f}\}, p_2 \doteqdot \{\mathbf{f}, \mathbf{t}\}\} \} \Rightarrow \{p_1 \supset p_2 \doteqdot \{\mathbf{f}\}, p_1 \supset p_2 \doteqdot \{\mathbf{f}, \mathbf{t}\} \}.$$

$$\mathbf{Sem}((\mathtt{t}:\supset)) = \{ \{p_1 \doteqdot \{\mathtt{f}\}, p_1 \doteqdot \{\mathtt{f},\mathtt{t}\}, p_2 \doteqdot \{\mathtt{t}\}, p_2 \doteqdot \{\mathtt{f},\mathtt{t}\}\} \} \Rightarrow \{p_1 \supset p_2 \doteqdot \{\mathtt{t}\}, p_1 \supset p_2 \doteqdot \{\mathtt{f},\mathtt{t}\} \}.$$

Note that an  $\mathcal{L}$ -valuation v (with  $Im_v \subseteq 2^{\mathcal{L}}$ ) satisfies these two semantic  $\mathcal{L}$ -conditions iff for every  $\varphi_1, \varphi_2 \in \mathcal{L}$ : (1) if  $\mathbf{t} \in v(\varphi_1)$  and  $\mathbf{f} \in v(\varphi_2)$ , then  $\mathbf{f} \in v(\varphi_1) \supset \varphi_2$ ; and (2) if  $\mathbf{f} \in v(\varphi_1)$  or  $\mathbf{t} \in v(\varphi_2)$ , then  $\mathbf{t} \in v(\varphi_1) \supset \varphi_2$ .

**Example 3.2.13.** Suppose that  $\wedge \in \diamondsuit^2_{\mathcal{L}}$  and  $\neg \in \diamondsuit^1_{\mathcal{L}}$ . For the  $\langle \mathcal{L}, \pounds_2 \rangle$ -rule from Example 2.2.8,  $r = \{\mathsf{t}: p_1\}, \{\mathsf{t}: \neg p_1\}/\{\mathsf{f}: \neg (p_1 \wedge \neg p_1)\}, \mathbf{Sem}(r)$  is

$$\{ \{p_1 \doteqdot \{\mathtt{t}\}, p_1 \doteqdot \{\mathtt{f}, \mathtt{t}\} \}, \{ \neg p_1 \doteqdot \{\mathtt{t}\}, \neg p_1 \doteqdot \{\mathtt{f}, \mathtt{t}\} \} \} \Rightarrow$$
$$\{ \neg (p_1 \land \neg p_1) \doteqdot \{\mathtt{f}\}, \neg (p_1 \land \neg p_1) \doteqdot \{\mathtt{f}, \mathtt{t}\} \}.$$

Note that an  $\mathcal{L}$ -valuation v (with  $Im_v \subseteq 2^{\ell}$ ) satisfies this semantic  $\mathcal{L}$ -conditios iff for every  $\varphi \in \mathcal{L}$ : if  $\mathbf{t} \in v(\varphi)$  and  $\mathbf{t} \in v(\neg \varphi)$ , then  $\mathbf{f} \in v(\varphi \land \neg \varphi)$ ).

To conclude,  $\mathbf{M}_{\mathbf{G}}$  is defined as follows:

**Definition 3.2.14.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. The many-valued pre-system  $\mathbf{M}_{\mathbf{G}}$  is defined by:  $\mathcal{V}_{\mathbf{M}_{\mathbf{G}}} = 2^{\mathcal{L}}$ ,  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \mathcal{L}(\mathsf{P}_{\mathbf{G}})$ , and  $\Lambda_{\mathbf{M}_{\mathbf{G}}} = \mathbf{Sem}(\mathsf{R}_{\mathbf{G}})$ .

The following theorem establishes the connection between pure calculi and their corresponding many-valued pre-systems. Its proof is given in Section 3.4.

**Theorem 3.2.15.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. Then  $\vdash_{\mathbf{G}} = \vdash_{\mathbf{M}_{\mathbf{G}}}$ . In other words: there exists a proof in **G** of an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s from a set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, iff every  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation which is a model of  $\mathcal{S}$  is also a model of s.

Corollary 3.2.16. Let G be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. and  $\leq$  a safe partial order on  $\mathcal{L}$ . Suppose that given an  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation v, whose domain is finite and closed under  $\leq$ , it is decidable whether v can be extended to an  $\mathbf{M}_{\mathbf{G}}$ -legal (full)  $\mathcal{L}$ -valuation or not. Then, given a finite set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and a single  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, it is decidable whether  $\mathcal{S} \vdash_{\mathbf{G}} s$  or not.

Proof. Construct  $\mathbf{M}_{\mathbf{G}}$  according to the definitions above. Then, enumerate all partial  $\mathcal{L}$ -valuations  $v: \downarrow^{\leq} [\mathcal{S} \cup \{s\}] \to \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ , and check if one of them is  $\mathbf{M}_{\mathbf{G}}$ -legal, can be extended to an  $\mathbf{M}_{\mathbf{G}}$ -legal (full)  $\mathcal{L}$ -valuation, and is model of  $\mathcal{S}$  but not of s. As in Theorem 3.1.19, we have that  $\mathcal{S} \vdash_{\mathbf{M}_{\mathbf{G}}} s$  iff such a partial  $\mathcal{L}$ -valuation is not found.

Now, to obtain a many-valued system for the logic induced by G and a set X of labels, one should take the designated truth values to be subsets of  $\mathcal{L}$  that contain at least one label from X:

**Definition 3.2.17.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, and let  $X \subseteq \mathcal{L}$ . The many-valued system  $\mathbf{M}_{\mathbf{G}}^{\mathbf{X}}$  is obtained by augmenting the many-valued pre-system  $\mathbf{M}_{\mathbf{G}}$  with  $\mathcal{D}_{\mathbf{M}_{\mathbf{G}}^{\mathbf{X}}} = \{X' \subseteq \mathcal{L} \mid X' \cap X \neq \emptyset\}.$ 

Corollary 3.2.18. Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, and  $X \subseteq \mathcal{L}$  a set of labels. Then  $\Vdash_{\mathbf{G}}^{X} = \Vdash_{\mathbf{M}_{\mathbf{G}}^{X}}$ .

Proof. Note that an  $\mathbf{M}_{\mathbf{G}}$ -legal valuation v is a model of a sequent of the form  $(\mathtt{X}:\varphi)$  iff it is a model of  $\varphi$  with respect to  $\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}$ . Therefore, the claim is an immediate corollary of Theorem 3.2.15. For the convenience of the reader, we prove one direction with all details: Suppose that  $\mathcal{T} \Vdash_{\mathbf{G}}^{\mathtt{X}} \varphi$ . Then, by Definition 2.2.18,  $\{(\mathtt{X}:\psi) \mid \psi \in \mathcal{T}\} \vdash_{\mathbf{G}} (\mathtt{X}:\varphi)$ . By Theorem 3.2.15, we have that every  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation which is a model of  $\{(\mathtt{X}:\psi) \mid \psi \in \mathcal{T}\}$ , is also a model of  $\{(\mathtt{X}:\varphi)$ . We prove that  $\mathcal{T} \Vdash_{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}} \varphi$ . Thus, by Definition 3.1.6, we should show that for every  $\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}$ -legal  $\mathcal{L}$ -valuation  $v: v \models^{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}} \varphi$  whenever  $v \models^{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}} \mathcal{T}$ . Let v be an  $\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}$ -legal  $\mathcal{L}$ -valuation such that  $v \models^{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}} \mathcal{T}$ . By definition,  $v(\psi) \in \mathcal{D}_{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}}$  for every  $\psi \in \mathcal{T}$ . Now, the definition of  $\mathcal{D}_{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}}$  entails that  $v(\psi) \cap \mathtt{X} \neq \emptyset$  for every  $\psi \in \mathcal{T}$ . It follows, according to Definitions 3.2.1 and 3.2.2, that  $v \models (\mathtt{X}:\psi)$  for every  $\psi \in \mathcal{T}$ . Consequently,  $v \models \{(\mathtt{X}:\psi) \mid \psi \in \mathcal{T}\}$ , and so  $v \models (\mathtt{X}:\varphi)$ . Thus  $v(\varphi) \cap \mathtt{X} \neq \emptyset$ , and so  $v(\varphi) \in \mathcal{D}_{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}}$ . It follows that  $v \models^{\mathbf{M}_{\mathbf{G}}^{\mathtt{X}}} \varphi$ .

**Example 3.2.19.** As a particular instance we obtain the soundness and completeness of **LK** and  $\{t\}$  for classical logic. Indeed, consider the pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_2 \rangle$ -calculus **LK** from Example 2.2.20. By Corollary 3.2.18, we have  $\Vdash_{\mathbf{LK}}^{\{t\}} = \Vdash_{\mathbf{M}_{\mathbf{LK}}^{\{t\}}}$ . It is straightforward to verify that  $\mathbf{M}_{\mathbf{LK}}$ -legal valuations are practically classical two-valued valuations. Indeed, since **LK** includes both (cut) and (id), we have  $\mathcal{U}_{\mathbf{M}_{\mathbf{LK}}} = \{\{f\}, \{t\}\}$ . The semantic  $\mathcal{L}_{cl}$ -conditions arising from the non-primitive rules of **LK** provide the usual definition of truth

values of compound formulas in classical logic (the rules for each connective enforce its usual truth table, see, e.g., Example 3.2.12).

Example 3.2.20. By applying Corollary 3.2.18 to the pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_{2} \rangle$ -calculus  $\mathbf{G}_{\mathbf{C}_{1}}$  from Example 2.2.22 and  $\mathbf{X} = \{\mathbf{t}\}$ , we obtain effective semantics for da Costa's paraconsistent logic  $\mathbf{C}_{1}$ . The many-valued system  $\mathbf{M} = \mathbf{M}_{\mathbf{G}_{\mathbf{C}_{1}}}^{\{\mathbf{t}\}}$  is given by:  $\mathcal{V}_{\mathbf{M}} = \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}, \{\mathbf{f}, \mathbf{t}\}\};$   $\mathcal{U}_{\mathbf{M}} = \{\{\mathbf{f}\}, \{\mathbf{t}\}\};$  and  $\Lambda_{\mathbf{M}}$  consists of one  $\mathcal{L}_{cl}$ -semantic condition over  $\mathcal{V}_{\mathbf{M}}$  for each non-primitive rule of  $\mathbf{G}_{\mathbf{C}_{1}}$ . For example, for the rule  $(\mathbf{f}:\neg \wedge^{1})$  we have the semantic condition given in Example 3.2.13. Similarly,  $(\mathbf{f}:\neg \neg)$  and  $(\mathbf{f}:\neg \wedge^{2})$  yield the following  $\mathcal{L}_{cl}$ -semantic conditions:

$$\mathbf{Sem}((\mathbf{f}:\neg\neg)) = \{ \{p_1 \doteqdot \{\mathbf{f}\}, p_1 \doteqdot \{\mathbf{f}, \mathbf{t}\}\} \} \Rightarrow \{\neg\neg p_1 \doteqdot \{\mathbf{f}\}, \neg\neg p_1 \doteqdot \{\mathbf{f}, \mathbf{t}\}\} \}$$

$$\mathbf{Sem}((\mathbf{f}:\neg\wedge^2)) = \{ \{\neg p_1 \doteqdot \{\mathbf{f}\}, \neg p_1 \doteqdot \{\mathbf{f}, \mathbf{t}\}\}, \{\neg p_2 \doteqdot \{\mathbf{f}\}, \neg p_2 \doteqdot \{\mathbf{f}, \mathbf{t}\}\} \} \Rightarrow$$

$$\{\neg (p_1 \land p_2) \doteqdot \{\mathbf{f}\}, \neg (p_1 \land p_2) \doteqdot \{\mathbf{f}, \mathbf{t}\}\}$$

It is easy to see that the conditions in  $\Lambda_{\mathbf{M}}$  (including the ones mentioned above) dictate the following requirements from  $\mathbf{M}$ -legal  $\mathcal{L}_{cl}$ -valuations v for every formula  $\psi$ :<sup>3</sup>

```
(t:¬) If \psi = \neg \varphi and \mathbf{f} \in v(\varphi), then \mathbf{t} \in v(\psi).
```

$$(f:\wedge)$$
 If  $\psi = \varphi_1 \wedge \varphi_2$  and  $(f \in v(\varphi_1))$  or  $f \in v(\varphi_2)$ , then  $f \in v(\psi)$ .

(t:
$$\wedge$$
) If  $\psi = \varphi_1 \wedge \varphi_2$ ,  $t \in v(\varphi_1)$  and  $t \in v(\varphi_2)$ , then  $t \in v(\psi)$ .

(f:
$$\forall$$
) If  $\psi = \varphi_1 \vee \varphi_2$ ,  $\mathbf{f} \in v(\varphi_1)$  and  $\mathbf{f} \in v(\varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

(t:
$$\vee$$
) If  $\psi = \varphi_1 \vee \varphi_2$  and ( $\mathsf{t} \in v(\varphi_1)$  or  $\mathsf{t} \in v(\varphi_2)$ ), then  $\mathsf{t} \in v(\psi)$ .

(f: 
$$\supset$$
) If  $\psi = \varphi_1 \supset \varphi_2$ ,  $\mathbf{t} \in v(\varphi_1)$  and  $\mathbf{f} \in v(\varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

(t: 
$$\supset$$
) If  $\psi = \varphi_1 \supset \varphi_2$  and ( $\mathbf{f} \in v(\varphi_1)$  or  $\mathbf{t} \in v(\varphi_2)$ ), then  $\mathbf{t} \in v(\psi)$ .

(f:¬¬) If  $\psi = \neg \neg \varphi$  and  $\mathbf{f} \in v(\varphi)$ , then  $\mathbf{f} \in v(\psi)$ .

$$(\mathbf{f}:\neg \wedge^1)$$
 If  $\psi = \neg(\varphi \wedge \neg \varphi)$ ,  $\mathbf{t} \in v(\varphi)$  and  $\mathbf{t} \in v(\neg \varphi)$ , then  $\mathbf{f} \in v(\psi)$ .

$$(\mathbf{f}:\neg \wedge^2)$$
 If  $\psi = \neg(\varphi_1 \wedge \varphi_2)$ ,  $\mathbf{f} \in v(\neg \varphi_1)$  and  $\mathbf{f} \in v(\neg \varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

$$(\mathbf{f}: \neg \vee^1)$$
 If  $\psi = \neg(\varphi_1 \vee \varphi_2)$ ,  $\mathbf{f} \in v(\neg \varphi_1)$  and  $(\mathbf{f} \in v(\varphi_2))$  or  $\mathbf{f} \in v(\neg \varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

$$(\mathbf{f}: \neg \vee^2)$$
 If  $\psi = \neg(\varphi_1 \vee \varphi_2)$ ,  $(\mathbf{f} \in v(\varphi_1) \text{ or } \mathbf{f} \in v(\neg \varphi_1))$  and  $\mathbf{f} \in v(\neg \varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

$$(\mathbf{f}:\neg\supset^1)$$
 If  $\psi=\neg(\varphi_1\supset\varphi_2)$ ,  $\mathbf{f}\in v(\varphi_1)$  and  $(\mathbf{f}\in v(\varphi_2))$  or  $\mathbf{f}\in v(\neg\varphi_2)$ , then  $\mathbf{f}\in v(\psi)$ .

$$(\mathbf{f}:\neg \supset^2)$$
 If  $\psi = \neg(\varphi_1 \supset \varphi_2)$ ,  $(\mathbf{f} \in v(\varphi_1) \text{ or } \mathbf{f} \in v(\neg \varphi_1))$  and  $\mathbf{f} \in v(\neg \varphi_2)$ , then  $\mathbf{f} \in v(\psi)$ .

It is easy to verify that these requirements correspond exactly to the conditions on  $C_1$ -bivaluations described in [34].

Now, **M** is not *sub*-analytic (see Definition 3.1.16). Indeed, consider the partial  $\mathcal{L}_{cl}$ -valuation v with  $Dom_v = \{p_1, p_2, \neg p_1, \neg p_2, \neg \neg p_1, \neg p_1, \neg p_2, \neg (\neg p_1 \wedge \neg p_2)\}$ , and:

<sup>&</sup>lt;sup>3</sup>Note that since  $\mathcal{U}_{\mathbf{M}} = \{\{\mathbf{f}\}, \{\mathbf{t}\}\}\$ , we can write  $v(\varphi) = \{\mathbf{x}\}$  instead of  $\mathbf{x} \in v(\varphi)$  (for  $\mathbf{x} \in \{\mathbf{f}, \mathbf{t}\}$ ). We prefer the latter since we will reuse this list of conditions in Example 3.3.10 where  $\{\mathbf{f}, \mathbf{t}\}$  is also included in  $\mathcal{U}_{\mathbf{M}}$ .

- $v(p_1) = v(p_2) = v(\neg \neg p_1) = \{f\},\$
- $v(\neg p_1) = v(\neg p_2) = v(\neg p_1 \land \neg p_2) = v(\neg(\neg p_1 \land \neg p_2)) = \{\mathsf{t}\}.$

 $Dom_v$  is finite and closed under subformulas, and v is M-legal. Now, assume for a contradiction that there is an M-legal (full)  $\mathcal{L}$ -valuation v' that extends v. ( $\mathbf{f}:\neg\neg$ ) enforces that  $v'(\neg\neg p_2) = \{\mathbf{f}\}$ . On the other hand, if  $v'(\neg\neg p_2) = \{\mathbf{f}\}$ , then ( $\mathbf{f}:\neg\wedge^2$ ) enforces that  $v'(\neg(\neg p_1 \land \neg p_2)) = \{\mathbf{f}\}$ . But,  $v'(\neg(\neg p_1 \land \neg p_2)) = v(\neg(\neg p_1 \land \neg p_2)) = \{\mathbf{t}\}$ .

Nevertheless, it can be shown that  $\mathbf{M}$  is nsub-analytic, where nsub denotes the transitive closure of  $sub \cup \{\langle \neg \varphi_i, \neg (\varphi_1 \diamond \varphi_2) \rangle \mid \varphi_1, \varphi_2 \in \mathcal{L}, \diamond \in \{\land, \lor, \supset\}, i = 1, 2\}$ . Indeed, let v be an  $\mathbf{M}$ -legal partial  $\mathcal{L}_{cl}$ -valuation, whose domain is finite and closed under nsub. We construct an  $\mathbf{M}$ -legal (full)  $\mathcal{L}_{cl}$ -valuation v' that extends v. Let  $\psi_1, \psi_2, ...$  be an enumeration of all  $\mathcal{L}_{cl}$  formulas such that:  $i \leq j$  whenever  $\langle \psi_i, \psi_j \rangle \in nsub$ . We recursively construct v'. Let  $i \geq 1$ , and suppose that  $v'(\psi_j)$  was defined for every j < i.  $v'(\psi_i)$  is defined as follows. First, if  $\psi_i \in Dom_v$  then  $v'(\psi_i) = v(\psi_i)$ . Otherwise, if  $\psi_i$  is an atomic formula  $v'(\psi_i) = \{\mathbf{f}\}$  (say). Otherwise,  $\psi_i$  is a compound formula and then  $v'(\psi_i)$  is set to be either  $\{\mathbf{f}\}$  or  $\{\mathbf{t}\}$  based on "classical logic reasoning" using the subformulas of  $\psi_i$  (for example, if  $\psi_i = \neg \psi_j$  then  $v'(\psi_i) = \{\mathbf{f}\}$  if  $v'(\psi_j) = \{\mathbf{t}\}$ , and otherwise  $v'(\psi_i) = \{\mathbf{t}\}$ ). Obviously, v' extends v. It remains to show that v' is  $\mathbf{M}$ -legal. For that we prove by induction on i that all the properties above hold for v' and  $\psi = \psi_i$ . Suppose they hold for  $\psi_j$  for every j < i. We do here several cases (the others are similar):

- (t:¬) Suppose that  $\psi = \neg \varphi$  and  $v'(\varphi) = \{f\}$ . If  $\psi \in Dom_v$ , then  $\varphi \in Dom_v$  as well, and  $v'(\psi) = \{t\}$  follows since v is M-legal. Otherwise,  $v'(\psi) = \{t\}$  as well, but this time because of the classical truth tables.
- (f:¬¬) Suppose that  $\psi = \neg \neg \varphi$  and  $v'(\varphi) = \{f\}$ . If  $\psi \in Dom_v$ , then  $\varphi \in Dom_v$  as well, and  $v'(\psi) = \{f\}$  follows since v is M-legal. Otherwise,  $v'(\neg \varphi) = \{t\}$  (by the induction hypothesis since  $\neg \varphi = \psi_j$  for some j < i, and the condition (t:¬)), and thus  $v'(\psi) = \{f\}$  according to the classical truth tables.
- (f: $\neg \wedge^1$ ) Suppose that  $\psi = \neg(\varphi \wedge \neg \varphi)$ ,  $v'(\varphi) = \{t\}$  and  $v'(\neg \varphi) = \{t\}$ . If  $\psi \in Dom_v$ , then  $\varphi, \neg \varphi \in Dom_v$  as well, and  $v'(\psi) = \{f\}$  follows since v is M-legal. Otherwise,  $v'(\varphi \wedge \neg \varphi) = \{t\}$  (by the induction hypothesis since  $\varphi \wedge \neg \varphi = \psi_j$  for some j < i, and the condition  $(t:\wedge)$ ), and so  $v'(\psi) = \{f\}$  according to the classical truth tables.
- (f: $\neg \wedge^2$ ) Suppose that  $\psi = \neg(\varphi_1 \wedge \varphi_2)$ ,  $v'(\neg \varphi_1) = \{f\}$  and  $v'(\neg \varphi_2) = \{f\}$ . If  $\psi \in Dom_v$ , then  $\neg \varphi_1, \neg \varphi_2 \in Dom_v$  as well, and  $v'(\psi) = \{f\}$  follows since v is M-legal. Otherwise,  $v'(\varphi_1) = v'(\varphi_2) = \{t\}$  according to the classical truth tables (by the induction hypothesis since  $\neg \varphi_1 = \psi_{j_1}$  and  $\neg \varphi_2 = \psi_{j_2}$  for some  $j_1, j_2 < i$ , and the condition (t: $\neg$ )). Thus  $v'(\varphi_1 \wedge \varphi_2) = \{t\}$  (by the induction hypothesis since  $\varphi_1 \wedge \varphi_2 = \psi_j$  for some j < i, and the condition (t: $\wedge$ )), and so  $v'(\psi) = \{f\}$  according to the classical truth tables.

Note that nsub is safe, and thus it follows that  $\mathbf{M}$  provides an effective semantics for the logic da Costa's paraconsistent logic  $\mathbf{C_1}$ . We note that a semantic decision for this logic was included in [47]. While its formulation is completely different than ours, the procedure in [47] is based on similar ideas. In particular, a notion equivalent to nsub-analyticity plays a major role there as well.

## 3.3 Characterization of Proof-Theoretic Properties

In this section we use the above general soundness and completeness theorem (and provide some extensions of it) for deriving semantic characterizations of the proof-theoretic properties of pure calculi discussed in Section 2.3.

#### 3.3.1 Strong Analyticity

Analyticity for a given calculus is traditionally obtained as a corollary of cut-admissibility (this was the case in the seminal work of Gentzen [56]). Indeed, if all rules in a pure calculus system (except for (cut)) admit the local subformula property (i.e., the premises of each rule consist only of subformulas of the formulas its conclusion), then cut-admissibility implies sub-analyticity.<sup>4</sup> However, there are many cases in which a calculus does not enjoy cut-admissibility, and it is analytic nevertheless. Thus we provide a semantic characterization of strong analyticity which is independent of cut-admissibility. To do so, we need to identify semantics for proofs in which only some formulas may appear. This can be easily done by considering partial valuations (see Definition 3.1.13), whose domain consists of all formulas that may be used in proofs.

First, Definitions 3.2.1 and 3.2.2 are adapted to partial  $\mathcal{L}$ -valuations as follows:

**Definition 3.3.1.** A partial  $\mathcal{L}$ -valuation v with  $Im_v \subseteq 2^{\mathcal{L}}$  is said to be a *model* of:

- 1. an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}:\varphi$  if  $\varphi \in Dom_v$  and  $\mathbf{x} \in v(\varphi)$ .
- 2. an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s if s is a  $\langle Dom_v, \mathcal{L} \rangle$ -sequent and v is a model of some  $\alpha \in s$ .
- 3. a set S of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents if v is a model of every  $\langle Dom_v, \mathcal{L} \rangle$ -sequent  $s \in S$ .

We write  $v \models X$  to denote that v is a model of X, where X is either an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula, an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent, or a set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents.

Generally, we agree with the first observation (though, we believe that other notions of analyticity based on different relation than "subformula" have similar importance). However, we aim to show that in many cases cut elimination is not necessarily "the most convenient" technique.

<sup>&</sup>lt;sup>4</sup> Quoting Ono [79]:

<sup>&</sup>quot;The most important proof-theoretic property is the subformula property, and the most convenient way of showing the subformula property is to show the cut elimination theorem."

Obviously, Definitions 3.2.1 and 3.2.2 are obtained from Definition 3.3.1 by taking  $Dom_v = \mathcal{L}$ . Note also that in Item 2 a partial  $\mathcal{L}$ -valuation v can only be a model of sequents consisting solely of formulas in  $Dom_v$ . Nevertheless, it can be a model of a set of sequents containing formulas which are not in  $Dom_v$ , because only  $\langle Dom_v, \mathcal{L} \rangle$ -sequents are considered in Item 3.

Now, the following theorem strengthens Theorem 3.2.15, by showing that proofs that consist only of formulas from a set  $\mathcal{F}$  precisely correspond to the semantics given by partial valuations whose domain is  $\mathcal{F}$ .

**Theorem 3.3.2.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus,  $\mathcal{F}$  a set of  $\mathcal{L}$ -formulas,  $\mathcal{S}$  a set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and s an  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent. Then,  $\mathcal{S} \vdash_{\mathbf{G}}^{\mathcal{F}} s$  (i.e. there exists a proof in **G** of s from  $\mathcal{S}$  consisting only of  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequents, see Notation 2.3.4) iff for every  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation v with  $Dom_v = \mathcal{F}$ : if v is a model of  $\mathcal{S}$  then it is also a model of s.

The proof is given in Section 3.4. We can now establish the connection between the (syntactic) strong  $\leq$ -analyticity of  $\mathbf{G}$  and the (semantic)  $\leq$ -analyticity of  $\mathbf{M}_{\mathbf{G}}$  (see Definition 3.1.16).

**Theorem 3.3.3.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is strongly  $\leq$ -analytic iff  $\mathbf{M}_{\mathbf{G}}$  is  $\leq$ -analytic.

*Proof.* ( $\Rightarrow$ ) Suppose that  $\mathbf{M}_{\mathbf{G}}$  is not  $\leq$ -analytic. Let v be an  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation whose domain is finite and closed under  $\leq$ , but there does not exist an  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation that extends v. Let  $\mathcal{S}$  and s be the set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and the  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent defined by:

 $\mathcal{S} = \{ \{ \mathbf{x} : \varphi \} \mid \varphi \in Dom_v, \mathbf{x} \in v(\varphi) \}, \qquad s = \{ \mathbf{x} : \varphi \mid \varphi \in Dom_v, \mathbf{x} \not\in v(\varphi) \}.$ 

Then, by definition  $v \models \mathcal{S}$  and  $v \not\models s$ . By Theorem 3.3.2 we have  $\mathcal{S} \not\vdash_{\mathbf{G}}^{Dom_v} s$ . We show that  $\mathcal{S} \vdash_{\mathbf{G}} s$ . Since  $Dom_v$  is closed under  $\leq$ ,  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}] = Dom_v$ , and it would follow that  $\mathbf{G}$  is not strongly  $\leq$ -analytic. Let v' be an  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation. Our assumption entails that v' does not extend v. Therefore there is some  $\varphi \in Dom_v$  such that  $v'(\varphi) \neq v(\varphi)$ . Thus, at least one of the following holds: (i) there is some  $\mathbf{x} \in \mathcal{L}$ , such that  $\mathbf{x} \in v'(\varphi)$  and  $\mathbf{x} \not\in v(\varphi)$ ; (ii) there is some  $\mathbf{x} \in \mathcal{L}$ , such that  $\mathbf{x} \in v(\varphi)$  and  $\mathbf{x} \not\in v'(\varphi)$ . If (i) holds, then  $v' \models s$ . If (ii) holds, then  $v' \not\models s$ . It follows that v' is either a model of s, or not a model of s. Consequently, every  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation which is a model of s is also a model of s. Hence Theorem 3.2.15 implies that  $s \mapsto_{\mathbf{G}} s$ .

 $(\Leftarrow)$  Suppose that  $\mathbf{M}_{\mathbf{G}}$  is  $\leq$ -analytic. We show that  $\mathbf{G}$  is strongly  $\leq$ -analytic. Let  $\mathcal{S} \cup \{s\}$  be a set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents. Suppose that  $\mathcal{S} \not\vdash_{\mathbf{G}}^{\mathcal{F}} s$  for  $\mathcal{F} = \downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ . We show that  $\mathcal{S} \not\vdash_{\mathbf{G}} s$ . It suffices to show that for every finite subset  $\mathcal{S}' \subseteq \mathcal{S}$ ,  $\mathcal{S}' \not\vdash_{\mathbf{G}} s$ . Let  $\mathcal{S}' \subseteq \mathcal{S}$  be a finite subset. Obviously,  $\mathcal{S}' \not\vdash_{\mathbf{G}}^{\mathcal{F}'} s$  for  $\mathcal{F}' = \downarrow^{\leq} [\mathcal{S}' \cup \{s\}]$ . By Theorem 3.3.2 (note that s is an  $\langle \mathcal{F}', \mathcal{L} \rangle$ -sequent), there is some  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation v with

 $Dom_v = \mathcal{F}'$ , which is a model of  $\mathcal{S}'$  but not of s.  $\mathcal{F}'$  is finite and closed under  $\leq$ , and thus our assumption entails that there exists an  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuation v' that extends v. By definition, v' is a model of some  $\langle Dom_v, \pounds \rangle$ -sequent s' iff v is a model of s'. It follows that v' is a model of  $\mathcal{S}'$  but not of s. Hence Theorem 3.2.15 implies that  $\mathcal{S}' \not\vdash_{\mathbf{G}} s$ .  $\square$ 

Example 3.3.4. Consider again the pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_{2} \rangle$ -calculus  $\mathbf{G}_{\mathbf{C}_{1}}$  from Example 2.2.22. Following Example 3.2.20, the many-valued pre-system  $\mathbf{M}_{\mathbf{G}_{\mathbf{C}_{1}}}$  is not sub-analytic. Thus, by Theorem 3.3.3,  $\mathbf{G}_{\mathbf{C}_{1}}$  is not strongly sub-analytic. However, in the same example we showed that  $\mathbf{M}_{\mathbf{G}_{\mathbf{C}_{1}}}$  is nsub-analytic for the (safe) partial order nsub defined there. Therefore Theorem 3.3.3 implies that  $\mathbf{G}_{\mathbf{C}_{1}}$  is strongly nsub-analytic. Note that we prove below that  $\mathbf{G}_{\mathbf{C}_{1}}$  enjoys strong cut-admissibility (using another semantic characterization, see Example 3.3.10). The fact that it is strongly nsub-analytic follows from this proof too since all rules of  $\mathbf{G}_{\mathbf{C}_{1}}$  except for (cut) are closed under nsub (that is, for each formula  $\varphi$  in a premise of a rule, there is some  $\psi$  in its conclusion such that  $\langle \varphi, \psi \rangle \in nsub$ ).

**Example 3.3.5.** Suppose that  $\mathcal{L}$  consists of one binary connective  $\mathbb{X}$ . Let  $\mathbf{G}$  be a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus, with  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ , and the following pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules:

Then partial  $\mathcal{L}$ -valuations (whose domain is closed under subformulas) are  $\mathbf{M}_{\mathbf{G}}$ -legal iff they satisfy the following conditions:

- $v(\varphi) \in \{\{f\}, \{t\}\}\}$  for every  $\varphi \in Dom_v$ .
- If  $\varphi_1 \bowtie \varphi_2 \in Dom_v$  for some  $\varphi_1, \varphi_2 \in \mathcal{L}$ , then:
  - If  $v(\varphi_1) = v(\varphi_2) = \{f\}$  then  $v(\varphi_1 \bowtie \varphi_2) = \{f\}$ .
  - If  $v(\varphi_1) = v(\varphi_2) = \{t\}$  then  $v(\varphi_1 \bowtie \varphi_2) = \{t\}$ .
  - If  $\varphi_2 \bowtie \varphi_1 \in Dom_v$  then  $v(\varphi_1 \bowtie \varphi_2) = v(\varphi_2 \bowtie \varphi_1)$ .

To see the reason for the last condition, note that

$$\mathbf{Sem}((sym)) = \emptyset \Rightarrow \{p_1 \otimes p_2 \doteqdot \{\mathtt{f}\}, p_1 \otimes p_2 \doteqdot \{\mathtt{f},\mathtt{t}\}, p_2 \otimes p_1 \doteqdot \{\mathtt{t}\}, p_2 \otimes p_1 \doteqdot \{\mathtt{f},\mathtt{t}\}\}.$$

A partial  $\mathcal{L}$ -valuation v with  $Im_v \subseteq \{\{f\}, \{t\}\}$  satisfies  $\mathbf{Sem}((sym))$  if for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\{\sigma(p_1 \otimes p_2), \sigma(p_2 \otimes p_1)\} \subseteq Dom_v$ , we have that  $\mathbf{f} \in v(\sigma(p_1 \otimes p_2))$  or  $\mathbf{t} \in v(\sigma(p_2 \otimes p_1))$ . Equivalently, v satisfies  $\mathbf{Sem}((sym))$  if for every  $\mathcal{L}$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $\{\varphi_1 \otimes \varphi_2, \varphi_2 \otimes \varphi_1\} \subseteq Dom_v$ , we have that  $v(\varphi_1 \otimes \varphi_2) = \{\mathbf{f}\}$  or  $v(\varphi_2 \otimes \varphi_1) = \{\mathbf{f}\}$ . By "switching the roles of  $\varphi_1$  and  $\varphi_2$ ", we obtain that v satisfies  $\mathbf{Sem}((sym))$  if for every  $\mathcal{L}$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $\{\varphi_1 \otimes \varphi_2, \varphi_2 \otimes \varphi_1\} \subseteq Dom_v$ , we have that  $\{v(\varphi_1 \otimes \varphi_2) = \{\mathbf{f}\}\}$  or  $v(\varphi_2 \otimes \varphi_1) = \{\mathbf{f}\}$  or  $v(\varphi_1 \otimes \varphi_2) = \{\mathbf{f}\}$ . The last condition above is equivalent to this requirement.

Using Theorem 3.3.3, it easily follows that **G** is strongly sub-analytic. Roughly speaking, given an  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation v (whose domain is closed under subformulas), we can recursively extend v to an  $\mathcal{L}$ -valuation by setting  $v(\varphi_1 \bowtie \varphi_2) = v(\varphi_1)$  if  $v(\varphi_1) = v(\varphi_2)$ ; otherwise  $v(\varphi_1 \bowtie \varphi_2) = v(\varphi_2 \bowtie \varphi_1)$  if  $v(\varphi_2 \bowtie \varphi_1)$  was defined before; and  $v(\varphi_1 \bowtie \varphi_2) = \{\mathbf{f}\}$  (say) otherwise.

### 3.3.2 Strong Cut-Admissibility

To obtain a simple semantic characterization of strong cut-admissibility, we slightly extended Theorem 3.2.15 by: (a) considering "extended sequents" that may be infinite; and (b) restricting the truth values of certain formulas (those on which cut is allowed) to a certain subset of  $\mathcal{V}_{\mathbf{M}}$ .

**Definition 3.3.6.** An extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent is a (possibly infinite) set of  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formulas. An  $\mathcal{L}$ -valuation v with  $Im_v \subseteq 2^{\mathcal{L}}$  is said to be a model of an extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu$ , written  $v \models \mu$ , if  $v \models \alpha$  for some  $\alpha \in \mu$ .

**Definition 3.3.7.** Let  $\mathbf{M}$  be a many-valued pre-system for  $\mathcal{L}$ ,  $\mathcal{U} \subseteq \mathcal{V}_{\mathbf{M}}$  a set of truth values, and  $\mathcal{C}$  a set of  $\mathcal{L}$ -formulas. An  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation v is called  $\langle \mathcal{U}, \mathcal{C} \rangle$ -restricted  $v(\varphi) \in \mathcal{U}$  for every  $\varphi \in \mathcal{C}$ .

**Theorem 3.3.8.** The following are equivalent for every pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G**, set  $\mathcal{C}$  of  $\mathcal{L}$ -formulas, set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu$ :

- $\mathcal{S} \vdash_{\mathbf{G}}^{cuts:\mathcal{C}} s$  for some  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s \subseteq \mu$  (recall that  $\mathcal{S} \vdash_{\mathbf{G}}^{cuts:\mathcal{C}} s$  denotes that there exists a proof in  $\mathbf{G}$  of s from  $\mathcal{S}$  in which the cut-formula of every application of a cut rule is an element of  $\mathcal{C}$ ).
- Every  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal  $\mathcal{L}$ -valuation which is a model of  $\mathcal{S}$  is also a model of  $\mu$  (recall that  $\mathbf{G}_{cf}$  is the calculus obtained from  $\mathbf{G}$  by discarding all cut rules).

The proof is given in Section 3.4. Using this soundness and completeness theorem, we obtain the following semantic characterization of strong cut-admissibility in pure calculi.

**Theorem 3.3.9.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys strong cut-admissibility iff for every  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v, there exists an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v' such that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ .<sup>5</sup>

Frecall that for a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus **G** whose only cut rule is (cut), we have that  $s(\mathbf{G}) = \mathbf{G}$  (see Definition 2.3.13). In this case  $\mathbf{M}_{s(\mathbf{G})_{cf}}$  can be replaced by  $\mathbf{M}_{\mathbf{G}_{cf}}$ .

Before proving this theorem, we demonstrate its usefulness in the particular case of  $\mathbf{G}_{\mathbf{C}_1}$ .

**Example 3.3.10.** Consider again the pure  $\langle \mathcal{L}_{cl}, \mathcal{L}_{2} \rangle$ -calculus  $\mathbf{G}_{\mathbf{C}_{1}}$  from Examples 2.2.22 and 3.3.4. In [12] it was shown that  $G_{C_1}$  enjoys cut-admissibility (and thus, by Theorem 2.3.19 it enjoys strong cut-admissibility as well). We use Theorem 3.3.9 to show that this fact can be obtained by our semantic criterion above. The only cut rule of  $G_{C_1}$  is (cut), and thus  $s(G_{C_1}) = G_{C_1}$ . The many-valued pre-system  $M = M_{G_{C_1}cf}$ has  $\mathcal{U}_{\mathbf{M}} = \{\{f\}, \{f, t\}\}\$ . The conditions in  $\Lambda_{\mathbf{M}} = \mathbf{Sem}(\mathsf{R}_{\mathbf{G}_{\mathbf{C}_1}})$  are those described in Example 3.2.20. Now, let v be an M-legal  $\mathcal{L}_{cl}$ -valuation. We construct an M-legal  $\mathcal{L}_{cl}$ -valuation v' that satisfies the condition in Theorem 3.3.9. Let  $\psi_1, \psi_2, \dots$  be an enumeration of all  $\mathcal{L}_{cl}$  formulas such that  $i \leq j$  whenever  $\langle \psi_i, \psi_j \rangle \in nsub$  (see Example 3.2.20 for the definition of nsub). We recursively construct v'. Let  $i \geq 1$ , and suppose that  $v'(\psi_j)$ was defined for every j < i.  $v'(\psi_i)$  is defined as follows. First, if  $v(\psi_i) \neq \{f, t\}$ , then  $v'(\psi_i) = v(\psi_i)$ . Otherwise, if  $\psi_i$  is an atomic formula  $v'(\psi_i) = \{f\}$  (say). Otherwise,  $\psi_i$ is a compound formula and then  $v'(\psi_i)$  is set to be either  $\{f\}$  or  $\{t\}$  based on "classical logic reasoning" using the subformulas of  $\psi_i$  (for example, if  $\psi_i = \neg \psi_j$  then  $v'(\psi_i) = \{f\}$ if  $v'(\psi_i) = \{t\}$ , and otherwise  $v'(\psi_i) = \{t\}$ ). Obviously,  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \{f, t\}$ . It remains to show that v' is M-legal. For that we prove by induction on i that all the requirements of  $\Lambda_{\mathbf{M}}$  (listed in Example 3.2.20) hold for v' and  $\psi = \psi_i$ . Suppose they hold for  $\psi_j$  for every j < i.

- (t:¬) Suppose that  $\psi = \neg \varphi$  and  $\mathbf{f} \in v'(\varphi)$  (i.e.  $v'(\varphi) = \{\mathbf{f}\}$ ). Then  $\mathbf{f} \in v(\varphi)$ , and since v is M-legal,  $v(\psi)$  is either  $\{\mathbf{t}\}$  or  $\{\mathbf{f},\mathbf{t}\}$ . In the first, case we have  $v'(\psi) = v(\psi) = \{\mathbf{t}\}$ . In the latter,  $v'(\psi) = \{\mathbf{t}\}$  as well, but this time because of the classical truth tables.
- (f:¬¬) Suppose that  $\psi = \neg \neg \varphi$  and  $f \in v'(\varphi)$  (i.e.  $v'(\varphi) = \{f\}$ ). Then  $f \in v(\varphi)$ , and since v is M-legal,  $v(\psi)$  is either  $\{f\}$  or  $\{f,t\}$ . In the first, case we have  $v'(\psi) = v(\psi) = \{f\}$ . In the latter, we have  $v'(\neg \varphi) = \{t\}$  (by the induction hypothesis since  $\neg \varphi = \psi_j$  for some j < i, and the condition  $(t:\neg)$ ), and thus  $v'(\psi) = \{f\}$  according to the classical truth tables.
- (f: $\neg \wedge^1$ ) Suppose that  $\psi = \neg(\varphi \wedge \neg \varphi)$ ,  $\mathbf{t} \in v'(\varphi)$  and  $\mathbf{t} \in v'(\neg \varphi)$  (i.e.  $v'(\varphi) = \{\mathbf{t}\}$  and  $v'(\neg \varphi) = \{\mathbf{t}\}$ ). Then  $\mathbf{t} \in v(\varphi)$  and  $\mathbf{t} \in v(\neg \varphi)$ . Since v is M-legal,  $v(\psi)$  is either  $\{\mathbf{f}\}$  or  $\{\mathbf{f},\mathbf{t}\}$ . In the first, case we have  $v'(\psi) = v(\psi) = \{\mathbf{f}\}$ . In the latter,  $v'(\varphi \wedge \neg \varphi) = \{\mathbf{t}\}$  (by the induction hypothesis since  $\varphi \wedge \neg \varphi = \psi_j$  for some j < i, and the condition  $(\mathbf{t}:\wedge)$ ), and so  $v'(\psi) = \{\mathbf{f}\}$  according to the classical truth tables.
- $(\mathbf{f}:\neg \wedge^2)$  Suppose that  $\psi = \neg(\varphi_1 \wedge \varphi_2)$ ,  $\mathbf{f} \in v'(\neg \varphi_1)$  and  $\mathbf{f} \in v'(\neg \varphi_2)$  (i.e.  $v'(\neg \varphi_1) = \{\mathbf{f}\}$  and  $v'(\neg \varphi_2) = \{\mathbf{f}\}$ ). Then  $\mathbf{f} \in v(\neg \varphi_1)$  and  $\mathbf{f} \in v(\neg \varphi_2)$ . Since v is M-legal,  $v(\psi)$

is either {f} or {f,t}. In the first, case we have  $v'(\psi) = v(\psi) = \{f\}$ . In the latter,  $v'(\varphi_1) = \{t\}$  and  $v'(\varphi_2) = \{t\}$  according to the classical truth tables (by the induction hypothesis since  $\neg \varphi_1 = \psi_{j_1}$  and  $\neg \varphi_2 = \psi_{j_2}$  for some  $j_1, j_2 < i$ , and the condition (t:\nabla)). Thus  $v'(\varphi_1 \wedge \varphi_2) = \{t\}$  (by the induction hypothesis since  $\varphi_1 \wedge \varphi_2 = \psi_j$  for some j < i, and the condition (t:\nabla)), and so  $v'(\psi) = \{f\}$  according to the classical truth tables.

(f:¬∨¹) Suppose that  $\psi = \neg(\varphi_1 \lor \varphi_2)$ ,  $\mathbf{f} \in v'(\neg \varphi_1)$  (i.e.  $v'(\neg \varphi_1) = \{\mathbf{f}\}$ ) and  $(\mathbf{f} \in v'(\varphi_2))$  or  $\mathbf{f} \in v'(\neg \varphi_2)$ ). Then  $\mathbf{f} \in v(\neg \varphi_1)$  and  $(\mathbf{f} \in v(\varphi_2))$  or  $\mathbf{f} \in v(\neg \varphi_2)$ . Since v is M-legal,  $v(\psi)$  is either  $\{\mathbf{f}\}$  or  $\{\mathbf{f},\mathbf{t}\}$ . In the first, case we have  $v'(\psi) = v(\psi) = \{\mathbf{f}\}$ . In the latter,  $v'(\varphi_1) = \{\mathbf{t}\}$  (by the induction hypothesis since  $\neg \varphi_1 = \psi_j$  for some j < i, and the condition  $(\mathbf{t}:\neg)$ ). Thus  $v'(\varphi_1 \lor \varphi_2) = \{\mathbf{t}\}$  (by the induction hypothesis since  $\varphi_1 \lor \varphi_2 = \psi_j$  for some j < i, and the condition  $(\mathbf{t}:\lor)$ ), and so  $v'(\psi) = \{\mathbf{f}\}$  according to the classical truth tables.

The other cases are similar. It follows that  $\mathbf{G}_{\mathbf{C_1}}$  enjoys strong cut-admissibility.

To prove Theorem 3.3.9, we use the following lemma.

**Lemma 3.3.11.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus.

- 1.  $\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}} = \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}} \cup \{\pounds\}.$
- 2. An  $\mathcal{L}$ -valuation v is  $\mathbf{M}_{s(\mathbf{G})}$ -legal iff it is  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal and  $v'(\varphi) \neq \mathcal{L}$  for every  $\varphi \in \mathcal{L}$ .

Proof. 2 directly follows from 1 since the only difference between  $\mathbf{M}_{s(\mathbf{G})}$  and  $\mathbf{M}_{s(\mathbf{G})_{cf}}$  is in the set of legal truth values. We prove 1. Since  $\mathsf{P}_{s(\mathbf{G})_{cf}}\subseteq \mathsf{P}_{s(\mathbf{G})}$ , we obviously have  $\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}\subseteq \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$ . In addition, since there are no cut rules in  $s(\mathbf{G})_{cf}$ ,  $\pounds\in\pounds(r)$  for every  $r\in\mathsf{P}_{s(\mathbf{G})_{cf}}$ , and thus  $\pounds\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$ . Now, let  $\mathtt{X}\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$ , and suppose that  $\mathtt{X}\not\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}$ . We show that  $\mathtt{X}=\pounds$ . By definition,  $\mathtt{X}\subseteq\pounds$ . To show that  $\pounds\subseteq\mathtt{X}$ , let  $\mathtt{x}\in\pounds$ . Since  $\mathtt{X}\not\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}$ ,  $\mathtt{X}\not\in\pounds(r)$  for some  $r\in\mathsf{P}_{s(\mathbf{G})}$ . The fact that  $\mathtt{X}\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$  entails that r must be a cut rule, namely  $r=(\mathtt{X}_1:p_1),\ldots,(\mathtt{X}_n:p_1)/\emptyset$  for some  $\mathtt{X}_1,\ldots,\mathtt{X}_n\subseteq\pounds$ . Since  $\mathtt{X}\not\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}$ , we have  $\mathtt{X}_i\cap\mathtt{X}\not=\emptyset$  for every  $1\le i\le n$ . Now, if  $\{\mathtt{x}\}\in\{\mathtt{X}_1,\ldots,\mathtt{X}_n\}$ , then we have that  $\mathtt{x}\in\mathtt{X}$  and we are done. Otherwise, by definition,  $s(\mathbf{G})_{cf}$  includes the primitive  $\pounds$ -rule  $r_{\mathtt{x}}=(\mathtt{X}_1:p_1),\ldots,(\mathtt{X}_n:p_1)/\{\mathtt{x}:p_1\}$ . The fact that  $\mathtt{X}\in\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$  then entails that  $\mathtt{X}\in\pounds(r_{\mathtt{x}})$ . It follows that  $\{\mathtt{x}\}\cap\mathtt{X}\ne\emptyset$ , and thus  $\mathtt{x}\in\mathtt{X}$ .

Proof of Theorem 3.3.9. ( $\Rightarrow$ ) Suppose that v is an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation, and there does not exist an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v' such that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ . Let  $\mathcal{S}$  and  $\mu$  be the set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and the extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent defined by:

$$\mathcal{S} = \{ \{\mathtt{x} : \varphi\} \mid \varphi \in \mathcal{L}, v(\varphi) \neq \pounds, \mathtt{x} \in v(\varphi) \} \quad \text{and} \quad \mu = \{\mathtt{x} : \varphi \mid \varphi \in \mathcal{L}, \mathtt{x} \not\in v(\varphi) \}.$$

Then, by definition  $v \models \mathcal{S}$  and  $v \not\models \mu$ . Now,  $frm[\mathcal{S}] \subseteq \{\varphi \in \mathcal{L} \mid v(\varphi) \neq \mathcal{L}\}$ , and thus v is  $\langle \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}, frm[\mathcal{S}] \rangle$ -restricted (using Lemma 3.3.11). By Theorem 3.3.8 we have  $\mathcal{S} \not\vdash_{s(\mathbf{G})}^{cuts:frm[\mathcal{S}]} s$  for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s \subseteq \mu$ . We show that  $\mathcal{S} \vdash_{s(\mathbf{G})} s$  for some  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s \subseteq \mu$ . By Proposition 2.3.18, it would then follow that  $\mathbf{G}$  does not enjoy cut-admissibility. Let v' be an  $\mathbf{M}_{s(\mathbf{G})}$ -legal  $\mathcal{L}$ -valuation. By Lemma 3.3.11, it is  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal and  $v'(\varphi) \neq \mathcal{L}$  for every  $\varphi \in \mathcal{L}$ . Thus our assumption entails that there is some  $\varphi \in \mathcal{L}$  such that  $v'(\varphi) \neq v(\varphi)$  and  $v(\varphi) \neq \mathcal{L}$ . Thus, at least one of the following holds: (i) there is some  $\mathbf{x} \in \mathcal{L}$ , such that  $\mathbf{x} \in v(\varphi)$  and  $\mathbf{x} \not\in v(\varphi)$  and  $\mathbf{x} \not\in v(\varphi)$ ; (ii) there is some  $\mathbf{x} \in \mathcal{L}$ , such that  $\mathbf{x} \in v(\varphi)$  and  $\mathbf{x} \not\in v(\varphi)$ . If (i) holds, then  $v' \models \mu$ . If (ii) holds, then  $v' \not\models \{\mathbf{x} : \varphi\}$ , and thus  $v' \not\models \mathcal{S}$ . It follows that v' is either a model of  $\varphi$ , or not a model of  $\varphi$ . Consequently, every  $\mathbf{M}_{s(\mathbf{G})}$ -legal  $\mathcal{L}$ -valuation which is a model of  $\varphi$  is also a model of  $\varphi$ . Theorem 3.3.8 implies that  $\mathcal{S} \vdash_{s(\mathbf{G})} s$  for some  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s \subseteq \mu$ .

( $\Leftarrow$ ) If **G** does not have any cut rule, then the claim is trivially true. Assume otherwise. Suppose that for every  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v, there exists an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v' such that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ . We prove that for every set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s,  $\mathcal{S} \vdash_{s(\mathbf{G})}^{cuts:frm[\mathcal{S}]} s$  whenever  $\mathcal{S} \vdash_{s(\mathbf{G})} s$ . By Proposition 2.3.18 it would follow that  $\mathbf{G}$  enjoys strong cut-admissibility. Suppose that  $\mathcal{S} \not\models_{s(\mathbf{G})}^{cuts:frm[\mathcal{S}]} s$ . Let  $\mathcal{C} = frm[\mathcal{S}]$ . By Theorem 3.3.8, there is a  $\langle \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v which is a model of  $\mathcal{S}$ , but not of s. Our assumption entails that there exists an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v' such that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ . Note that in particular, we must have  $v'(\varphi) \neq \mathcal{L}$  for every  $\varphi \in \mathcal{L}$ , and thus, by Lemma 3.3.11, v' is  $\mathbf{M}_{s(\mathbf{G})}$ -legal. Now, since  $\mathbf{G}$  has at least one cut rule,  $\mathcal{L} \notin \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}$ . Since v is  $\langle \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}}, \mathcal{C} \rangle$ -restricted,  $v(\varphi) \neq \mathcal{L}$  for every  $\varphi \in frm[\mathcal{S}]$ . Thus  $v'(\varphi) = v(\varphi)$  for every  $\varphi \in frm[\mathcal{S}]$ , and so  $v' \models \mathcal{S}$  as well. Since  $v'(\varphi) \subseteq v(\varphi)$  for every  $\varphi \in \mathcal{L}$ , we have  $v' \not\models s$ . By Theorem 3.2.15,  $\mathcal{S} \not\models_{s(\mathbf{G})} s$ .

## 3.3.3 Axiom-Expansion

Using the soundness and completeness theorem above (Theorem 3.2.15), we automatically obtain a semantic characterization of axiom-expansion for a given connective in a given pure calculus.

Corollary 3.3.12. A connective  $\diamond \in \diamond_{\mathcal{L}}$  admits axiom-expansion in a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ calculus  $\mathbf{G}$  iff for every  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal  $\mathcal{L}$ -valuation v: if  $\mathbf{X} \cap v(p_i) \neq \emptyset$  for every  $\mathbf{X} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $1 \leq i \leq ar(\diamond)$ , then  $\mathbf{X} \cap v(\diamond(p_1, \ldots, p_{ar(\diamond)})) \neq \emptyset$  for every  $\mathbf{X} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}$ .

*Proof.* By definition,  $\diamond$  admits axiom-expansion in **G** iff for every  $\emptyset/(Y:p_1) \in P_{\mathbf{G}}$ , we have  $\{(X:p_i) \mid \emptyset/(X:p_1) \in P_{\mathbf{G}}, 1 \leq i \leq ar(\diamond)\} \vdash_{\mathbf{G}_{if}} (Y:\diamond(p_1,\ldots,p_{ar(\diamond)}))$ . The given condition is

equivalent by Theorem 3.2.15.

For ordinary pure  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculi in which the only identity axiom is (id) we obtain that a connective  $\diamond \in \diamondsuit_{\mathcal{L}}$  admits axiom-expansion in  $\mathbf{G}$  iff for every  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal  $\mathcal{L}$ -valuation v: if  $v(p_i) \neq \emptyset$  for every  $1 \leq i \leq ar(\diamond)$  then  $v(\diamond(p_1, \ldots, p_{ar(\diamond)})) \neq \emptyset$ .

# 3.4 Soundness and Completeness Proofs

In this section we provide proofs of the three soundness and completeness theorems above (Theorems 3.2.15, 3.3.2 and 3.3.8). To avoid repetitions, we will prove one more general result, from which these theorems directly follow. We consider  $\langle \mathcal{U}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuations. These are defined as  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuations with the additional requirement that  $v(\varphi) \in \mathcal{U}$  for every  $\varphi \in \mathcal{C} \cap Dom_v$ . In addition, we say that a  $\langle \mathcal{U}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuation v is a model of an extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu$  if  $\mu$  is an extended  $\langle Dom_v, \mathcal{L} \rangle$ -sequent (that is, all  $\mathcal{L}$ -formulas that occur in  $\mu$  are in  $Dom_v$ ), and  $v \models \alpha$  for some  $\alpha \in \mu$ . The general soundness and completeness theorem is given by:

**Theorem 3.4.1.** Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus,  $\mathcal{F}, \mathcal{C}$  sets of  $\mathcal{L}$ -formulas,  $\mathcal{S}$  a set of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and  $\mu$  an extended  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent. Then the following are equivalent:

- 1. There exists a proof P in G of some  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s \subseteq \mu$  from  $\mathcal{S}$ , that consists only of  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequents, and the cut-formula of every application of a cut rule in P is an element of  $\mathcal{C}$ .
- 2. For every  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuation v with  $Dom_v = \mathcal{F}$ , it holds that if v is a model of  $\mathcal{S}$  then it is also a model of  $\mu$ .

Note that the availability of the weakening rules ensures that when  $\mu$  is finite (forming a usual  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent), then 1 can be equivalently written as: there exists a proof P in  $\mathbf{G}$  of  $\mu$  from  $\mathcal{S}$  that consists only of  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequents and the cut-formula of every application of a cut rule in P is an element of  $\mathcal{C}$ . Therefore:

- Theorem 3.3.2 follows by taking  $C = \mathcal{L}$  (since  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{L} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuations are exactly  $\mathbf{M}_{\mathbf{G}}$ -legal partial  $\mathcal{L}$ -valuations).
- Theorem 3.2.15 follows by taking  $\mathcal{F} = \mathcal{C} = \mathcal{L}$  (since  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{L} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuations with  $Dom_v = \mathcal{L}$  are exactly  $\mathbf{M}_{\mathbf{G}}$ -legal  $\mathcal{L}$ -valuations).

In addition, Theorem 3.3.8 follows by taking  $\mathcal{F} = \mathcal{L}$  (since  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuations with  $Dom_v = \mathcal{L}$  are exactly  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal  $\mathcal{L}$ -valuations).

The following simple proposition will be useful in the proof below:

**Proposition 3.4.2.** Let v be a partial  $\mathcal{L}$ -valuation with  $Im_v \subseteq 2^{\mathcal{L}}$ , s an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent, and  $\sigma$  an  $\mathcal{L}$ -substitution such that  $\sigma(s)$  is a  $\langle Dom_v, \mathcal{L} \rangle$ -sequent. Then,  $v \models \sigma(I(s))$  iff  $v \models \sigma(s)$  (see Definition 3.2.10).

*Proof.* Directly follows from the definitions (note that  $\sigma(I(s)) = I(\sigma(s))$ ).

Next, we prove Theorem 3.4.1.

## Soundness

Assume that 1 holds. Let v be an  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuation with  $Dom_v = \mathcal{F}$ . Suppose that  $v \models \mathcal{S}$ . Using induction on the length of P, we show that  $v \models s$  for every sequent s occurring in P. It follows that  $v \models \mu$ . This trivially holds for the sequents of  $\mathcal{S}$  (note that only  $\langle Dom_v, \mathcal{L} \rangle$ -sequents in  $\mathcal{S}$  can appear in P). We show that this property is also preserved by applications of the rules of  $\mathbf{G}$ . Suppose that  $s = \sigma(s') \cup c_1 \cup ... \cup c_n$  is derived from  $\sigma(s_1) \cup c_1, ..., \sigma(s_n) \cup c_n$  using a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $r = s_1, ..., s_n/s'$  of  $\mathbf{G}$ . Assume that  $v \models \sigma(s_i) \cup c_i$  for every  $1 \leq i \leq n$ . We show that  $v \models s$ . Since s occurs in s, s is a s-sequent. Thus, by definition it suffices to show that s-sequents assumption entails that s-sequents. Then we are done. Assume otherwise. Then our assumption entails that s-sequents for every s-sequents. Then our assumption entails that s-sequents. Then we are done. Assume otherwise. Then our assumption entails that s-sequents for every s-sequents. Then our assumption entails that s-sequents for every s-sequents. Then our assumption entails that s-sequents for every s-sequents.

- Suppose that r is a primitive  $\pounds$ -rule. Then,  $\sigma(s_i) = (X_i:\varphi)$  for  $1 \leq i \leq n$  and  $\sigma(s') = (X:\varphi)$  for some  $X_1, \ldots, X_n, X \subseteq \pounds$  and  $\varphi \in \pounds$ . For every  $1 \leq i \leq n$ , since  $v \models \sigma(s_i)$ , there exists some  $\mathbf{x} \in X_i$  such that  $\mathbf{x} \in v(\varphi)$ . In other words, for every  $1 \leq i \leq n$ ,  $X_i \cap v(\varphi) \neq \emptyset$ . Now, note that r cannot be a weakening rule. Indeed, if r were a weakening rule then n = 1 and  $X_1 = \emptyset$ , and this contradicts the fact that  $X_1 \cap v(\varphi) \neq \emptyset$ . In addition, r cannot be a cut rule (i.e. we must have  $\mathbf{X} \neq \emptyset$ ). Indeed, if r were a cut rule, then since the cut-formula of every application of a cut rule in P is an element of  $\mathcal{C}$ , we would have that  $\varphi \in \mathcal{C}$ . In this case, since v is  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted,  $v(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ , and in particular  $v(\varphi) \in \pounds(r)$ . It would then follow that  $\mathbf{X} \cap v(\varphi) \neq \emptyset$ , but this is not possible when  $\mathbf{X} = \emptyset$ . Therefore, we have that  $r \in \mathsf{P}_{\mathbf{G}_{cf}}$ . Since v is  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal,  $v(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{cf}}}$ . In particular,  $v(\varphi) \in \pounds(r)$ , and we obtain that  $\mathbf{X} \cap v(\varphi) \neq \emptyset$ . Thus  $\mathbf{x}:\varphi \in \sigma(s')$  for some  $\mathbf{x} \in v(\varphi)$ . It follows that  $v \models \sigma(s')$ .
- Suppose that r is not a primitive  $\mathcal{L}$ -rule. For every  $1 \leq i \leq n$ , since  $v \models \sigma(s_i)$ , we have that  $v \models \sigma(I(s_i))$  (using Proposition 3.4.2). Since v is  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal, v satisfies  $\mathbf{Sem}(r) = \{I(s_i) \mid 1 \leq i \leq n\} \Rightarrow I(s')$ . Since  $\sigma(\varphi) \in Dom_v$  for every  $\varphi$  that occurs in  $\mathbf{Sem}(r)$  (since  $\sigma(s_i)$  for  $1 \leq i \leq n$  and  $\sigma(s')$  occur in the proof P),

and  $v \models \sigma(I(s_i))$  for every  $1 \leq i \leq n$ , we have that  $v \models \sigma(I(s'))$  as well. By Proposition 3.4.2,  $v \models \sigma(s')$ .

#### Completeness

We prove that 2 implies 1. For this proof, call an extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu'$  provable if there is a proof P in G of some  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $s' \subseteq \mu'$  from S, that consists only of  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequents, and the cut-formula of every application of a cut rule in P is an element of C. Otherwise, say that  $\mu'$  is unprovable. Suppose that  $\mu$  is unprovable. We construct an  $\langle \mathcal{U}_{\mathbf{M}_{G}}, \mathcal{C} \rangle$ -restricted  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal partial  $\mathcal{L}$ -valuation v with  $Dom_{v} = \mathcal{F}$ , that is a model of S, but not of  $\mu$ . Call an extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu^{*}$  maximal if it satisfies the following conditions:

- 1.  $frm[\mu^*] \subseteq \mathcal{F}$ .
- 2.  $\mu^*$  is unprovable.
- 3. For every  $\mathcal{L}$ -labelled  $\mathcal{F}$ -formula  $\alpha \notin \mu^*$ ,  $\{\alpha\} \cup \mu^*$  is provable.

We first construct a maximal extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $\mu^*$ , such that  $\mu \subseteq \mu^*$ . Let  $\alpha_1, \alpha_2, ...$  be an enumeration of all  $\mathcal{L}$ -labelled  $\mathcal{F}$ -formulas. Recursively define an (infinite) sequence of extended  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents  $\mu_0, \mu_1, ...$ , as follows. Let  $\mu_0 = \mu$ . For  $k \geq 1$ , let  $\mu_k = \mu_{k-1}$  iff  $\{\alpha_k\} \cup \mu_{k-1}$  is provable. Otherwise, let  $\mu_k = \{\alpha_k\} \cup \mu_{k-1}$ . Finally, let  $\mu^* = \bigcup_{k \geq 0} \mu_k$ . It is easy to verify that  $\mu^*$  has all required properties.

Now, define a partial  $\mathcal{L}$ -valuation v by  $Dom_v = \mathcal{F}$  and  $v(\varphi) = \{ \mathbf{x} \in \mathcal{L} \mid \mathbf{x} : \varphi \notin \mu^* \}$  for every  $\varphi \in \mathcal{F}$ . Note that the following property holds:

(a) For every  $\langle \mathcal{F}, \pounds \rangle$ -sequent  $s, v \models s$  iff  $s \cup c$  is provable for some  $\langle \mathcal{F}, \pounds \rangle$ -sequent  $c \subseteq \mu^*$ .

Proof. Let s be an  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent. Suppose that there exists an  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent  $c \subseteq \mu$  such that  $s \cup c$  is provable. The maximality of  $\mu^*$  entails that  $s' \not\subseteq \mu^*$ . Thus  $\mathbf{x} : \varphi \not\in \mu^*$  for some  $\mathbf{x} : \varphi \in s'$ . The definition of v entails that  $\mathbf{x} \in v(\varphi)$ , and so  $v \models s'$ . For the converse, assume that  $v \models s'$ . Hence there exists some  $\mathbf{x} : \varphi \in s'$  such that  $\mathbf{x} \in v(\varphi)$ . By definition,  $\mathbf{x} : \varphi \not\in \mu^*$ . The maximality of  $\mu^*$  entails that there exists a sequent  $c \subseteq \mu^*$  such that  $\{\mathbf{x} : \varphi\} \cup c$  is provable. The availability of the weakening rules entails that  $s' \cup c$  is provable as well.

Next, we show that v is  $\mathbf{M}_{\mathbf{G}_{cf}}$ -legal. We first prove that  $v(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{cf}}}$  for every  $\varphi \in \mathcal{F}$ . Thus we prove that for every  $\varphi \in \mathcal{F}$ ,  $v(\varphi) \in \mathcal{L}(r)$  for every  $r \in \mathsf{P}_{\mathbf{G}_{cf}}$ . Let  $\varphi \in \mathcal{F}$ , and let  $r = (\mathsf{X}_1:p_1), \ldots, (\mathsf{X}_n:p_1)/(\mathsf{X}:p_1)$  be a primitive  $\mathcal{L}$ -rule in  $\mathsf{P}_{\mathbf{G}_{cf}}$ . To see that  $v(\varphi) \in \mathcal{L}(r)$ , we show that if  $\mathsf{X}_i \cap v(\varphi) \neq \emptyset$  for every  $1 \leq i \leq n$ , then  $\mathsf{X} \cap v(\varphi) \neq \emptyset$ .

Suppose that  $X_i \cap v(\varphi) \neq \emptyset$  for every  $1 \leq i \leq n$ . Hence  $v \models (X_i:\varphi)$ . (a) entails that for every  $1 \leq i \leq n$ , there exists some  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent  $c_i \subseteq \mu^*$  such that  $(X_i:\varphi) \cup c_i$  is provable. Using the rule r, we obtain that  $(X_i:\varphi) \cup c_1 \cup ... \cup c_n$  is provable. (a) again entails that  $v \models (X_i:\varphi)$ , and so  $X \cap v(\varphi) \neq \emptyset$ .

Now, we prove that  $v \models \Lambda_{\mathbf{M}_{\mathbf{G}_{cf}}}$ , that is:  $v \models \mathbf{Sem}(r)$  for every  $r \in \mathsf{R}_{\mathbf{G}}$  (by definition,  $\mathsf{R}_{\mathbf{G}} = \mathsf{R}_{\mathbf{G}_{cf}}$ ). Let  $r = s_1, \ldots, s_n/s$  be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule in  $\mathsf{R}_{\mathbf{G}}$ . By definition,  $\mathbf{Sem}(r) = \{I(s_i) \mid 1 \leq i \leq n\} \Rightarrow I(s)$ . Let  $\sigma$  be an  $\mathcal{L}$ -substitution, such that  $\sigma(\varphi) \in Dom_v = \mathcal{F}$  for every  $\varphi$  that occurs in  $\mathbf{Sem}(r)$ . Suppose that  $v \models \sigma(I(s_i))$  for every  $1 \leq i \leq n$ . Thus  $v \models \sigma(s_i)$  (Proposition 3.4.2) for every  $1 \leq i \leq n$ . (a) entails that for every  $1 \leq i \leq n$  there exists some  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent  $c_i \subseteq \mu^*$  such that  $\sigma(s_i) \cup c_i$  is provable. Now, using these proofs for  $1 \leq i \leq n$  and the rule r we obtain that  $\sigma(s) \cup c_1 \cup \ldots \cup c_n$  is provable. (a) entails that  $v \models \sigma(s)$ . By Proposition 3.4.2,  $v \models \sigma(I(s))$ .

Next, we show that v is  $\langle \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}, \mathcal{C} \rangle$ -restricted. Let  $\varphi \in \mathcal{C} \cap \mathcal{F}$ . To see that  $v(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  it suffices to show that  $v(\varphi) \in \mathcal{L}(r)$  for every cut rule  $r \in \mathsf{P}_{\mathbf{G}}$  (we already have  $v(\varphi) \in \mathcal{L}(r)$  for every other primitive rule r of  $\mathbf{G}$ ). Let  $(\mathsf{X}_1:p_1),\ldots,(\mathsf{X}_n:p_1)/\emptyset$  be a cut rule in  $\mathsf{P}_{\mathbf{G}}$ . To see that  $v(\varphi) \in \mathcal{L}(r)$ , we show that  $\mathsf{X}_i \cap v(\varphi) = \emptyset$  for some  $1 \leq i \leq n$ . Suppose otherwise. Then for every  $1 \leq i \leq n$   $v \models (\mathsf{X}_i:\varphi)$ , and (a) entails that there exists some  $\langle \mathcal{F}, \mathcal{L} \rangle$ -sequent  $c_i \subseteq \mu^*$  such that  $(\mathsf{X}_i:\varphi) \cup c_i$  is provable. Applying the rule r (with  $\varphi \in \mathcal{C}$  as the cut-formula), we obtain that  $c_1 \cup \ldots \cup c_n$  is provable. But since  $c_1 \cup \ldots \cup c_n \subseteq \mu^*$ , this contradicts the properties of  $\mu^*$ .

Finally, we show that  $v \models \mathcal{S}$  but  $v \not\models \mu$ . Let  $s \in \mathcal{S}$  be an  $\langle \mathcal{F}, \pounds \rangle$ -sequent. Obviously, s is provable, and by (a),  $v \models s$ . To see that  $v \not\models \mu$ , note that since  $\mu \subseteq \mu^*$ ,  $\mathbf{x} \not\in v(\varphi)$  for every  $\mathbf{x} : \varphi \in \mu$ .

# Chapter 4

# Canonical Calculi

In the previous chapter we established a strong connection between pure calculi and many-valued systems. However, there is a price to pay for the high generality. Evidently, the set of semantic conditions of the many-valued pre-system  $\mathbf{M}_{\mathbf{G}}$  for some calculus  $\mathbf{G}$  might be very complicated, which makes the search for a refuting valuation as complicated as proof-search in  $\mathbf{G}$ . In such cases the semantic criteria for cut-admissibility and analyticity might be useless. Thus it makes sense to study narrower families of sequent calculi, for which we can obtain *effective* semantics, as well as *simple and decidable* characterizations of their proof-theoretic properties.

A particular interesting family of this sort is the family of canonical systems, that was introduced in [17]. The idea behind canonical systems implicitly underlies a long tradition in the philosophy of logic, established by Gentzen in his seminal paper [56]. According to this tradition, the meaning of a connective  $\diamond$  is determined by the derivation rules which are associated with it. For that matter, one should have rules of some "ideal" type, in each of which  $\diamond$  is mentioned exactly once, and no other connective is involved. Formulating this idea, [17] introduced the notion of a "canonical (introduction) rule", and "canonical propositional Gentzen-type systems" were defined as two-sided sequent systems in which: (i) all logical rules are canonical rules; (ii) the usual cut rule, identity-axiom and all structural rules are included. The semantics of canonical systems was given using non-deterministic matrices (Nmatrices), a natural generalization of logical many-valued matrices. This revealed a remarkable triple correspondence in canonical systems between cut-admissibility, sub-analyticity, and the existence of a characteristic two-valued Nmatrix. Later the theory was generalized to many-sided sequent calculi that employ certain (fixed) cut rules and identity axioms [19, 21].

In this chapter we generalize previously studied canonical systems, and study a wide family of pure calculi employing canonical logical rules. Since all of them are pure sequent calculi, we utilize the general results of the previous chapter. We show that in this more restricted settings, the obtained many-valued systems are very simple, and they can be seen as a slight generalization the framework of Nmatrices from [17]. We call these many-valued systems partial non-deterministic matrices (PNmatrices). As Nmatrices, the main attractive property of (finite) PNmatrices is their effectiveness, as they always induce decision procedures for their underlying logics. In addition, we show that using the PNmatrix that characterizes a given canonical calculus  $\mathbf{G}$ , it is easy to decide whether  $\mathbf{G}$  admits strong sub-analyticity, strong cut-admissibility and axiom-expansion. In particular, our results show that strong sub-analyticity is equivalent to strong cut-admissibility in this family of calculi.

The structure of this chapter is as follows. First, we precisely define the family of canonical calculi that we study (which is a subfamily of pure calculi studied in the previous chapters). In Section 4.2 we present the framework of PNmatrices (which is a subfamily of many-valued systems introduced in the previous chapter) and show that it is effective. The connection between canonical calculi and PNmatrices is established and exemplified in Section 4.3. Finally, Section 4.4 provides simple decidable characterizations of proof-theoretic properties of canonical calculi based on their characteristic PNmatrices.

#### Publications Related to this Chapter

Most of the material in this chapter was included in [28, 29]. However, the notions and proofs in these papers consider canonical calculi and PNmatrices in their own right. Here we introduce them as particular cases of pure calculi and many-valued systems, and derive our results using the more general theorems from the previous chapters. In addition, specific cases of canonical two-sided calculi without cut and/or identity axiom were included in [68, 70].

### 4.1 Canonical Calculi

Canonical calculi are pure calculi that, in addition to the primitive rules, include only pure rules of a special well-behaved form, called *canonical rules*. Each canonical rule is associated with some connective  $\diamond$ , and  $\diamond(p_1, \ldots, p_{ar(\diamond)})$  is the only formula occurring in its conclusion. In turn, its premises are composed only from  $p_1, \ldots, p_{ar(\diamond)}$ . Formally, this is defined as follows:

**Definition 4.1.1.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule r is called canonical if its conclusion has the form  $\{X: \diamond (p_1, \ldots, p_{ar(\diamond)})\}$  for some non-empty  $X \subseteq \mathcal{L}$  and  $\diamond \in \diamond_{\mathcal{L}}$ , and its premises are all  $\langle \{p_1, \ldots, p_{ar(\diamond)}\}, \mathcal{L} \rangle$ -sequents. Canonical pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules will be also called  $canonical \langle \mathcal{L}, \mathcal{L} \rangle$ -rules, and in the case above we will say that r is a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $for \diamond$ .

Note that by definition canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules for a connective  $\diamond \in \Diamond_{\mathcal{L}}$  have the form:

$$(\mathbf{X}_1^1:p_1)\cup\ldots\cup(\mathbf{X}_1^{ar(\diamond)}:p_{ar(\diamond)}),\ldots,(\mathbf{X}_m^1:p_1)\cup\ldots\cup(\mathbf{X}_m^{ar(\diamond)}:p_{ar(\diamond)})/(\mathbf{X}:\diamond(p_1,\ldots,p_{ar(\diamond)})).$$
 for some  $m\geq 0$  and  $\mathbf{X}_1^1,\ldots,\mathbf{X}_1^{ar(\diamond)},\ldots,\mathbf{X}_m^1,\ldots,\mathbf{X}_m^{ar(\diamond)},\mathbf{X}\subseteq\pounds$  (where  $\mathbf{X}$  is not empty). An application of a rule of this form is any inference step of the following form:

$$\begin{array}{c} (\mathbf{X}_{1}^{1}:\varphi_{1}) \cup \ldots \cup (\mathbf{X}_{1}^{ar(\diamond)}:\varphi_{ar(\diamond)}) \cup c_{1} & \ldots & (\mathbf{X}_{m}^{1}:\varphi_{1}) \cup \ldots \cup (\mathbf{X}_{m}^{ar(\diamond)}:\varphi_{ar(\diamond)}) \cup c_{m} \\ \\ (\mathbf{X}: \diamond (\varphi_{1}, \ldots, \varphi_{ar(\diamond)})) \cup c_{1} \cup \ldots \cup c_{m} \end{array}$$

where  $\varphi_1, \dots, \varphi_{ar(\diamond)}$  are  $\mathcal{L}$ -formulas, and  $c_i$  is an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent for every  $1 \leq i \leq m$ .

**Example 4.1.2.** The pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules from Example 2.2.6  $\{t:p_1\}, \{f:p_2\}/\{f:p_1 \supset p_2\}, \{f:p_1,t:p_2\}/\{t:p_1 \supset p_2\}, \text{ and } \{t:p_2\}/\{t:p_1 \supset p_2\}, \text{ are all canonical. On the other hand, the rules } \{f:\neg p_1\}, \{f:\neg p_2\}/\{f:\neg (p_1 \land p_2)\} \text{ and } \{t:p_1\}, \{t:\neg p_1\}/\{f:\neg (p_1 \land \neg p_1)\} \text{ from Example 2.2.22 are not canonical.}$ 

**Example 4.1.3.** Note that we allow the formula in the conclusion to appear with more than one label. For example,  $\emptyset/\{\mathbf{f}:\star p_1,\mathbf{t}:\star p_1\}$  is a canonical  $\langle\{\star^1\},\{\mathbf{f},\mathbf{t},\mathbf{i}\}\rangle$ -rule. In fact, this rule is also a canonical  $\langle\{\star^1\},\pounds_2\rangle$ -rule, that may be useful when (id) is not available in general.

In turn, canonical calculi are defined as follows.

**Definition 4.1.4.** A canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, such that  $R_{\mathbf{G}}$  (the set of non-primitive rules of **G**) consists only of canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules.

Note that we do impose any limitations on the primitive rules, in particular allowing any set of cuts and identity axioms in canonical calculi. Hence the canonical calculi studied here are substantially more general than previously studied canonical systems:

- The canonical systems of [17] correspond to canonical  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculi whose primitive rules consist of (cut) and (id) (except for the weakening rules), and the conclusion of each non-primitive rule takes the form  $\{\mathbf{x}: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  for some  $\mathbf{x} \in \mathcal{L}_{\mathbf{2}}$  and  $\diamond \in \diamondsuit_{\mathcal{L}}$ . In particular, **LK** (see Example 2.2.20) is such a canonical  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculus.
- The signed canonical calculi of [19, 21] are similar to our canonical calculi, but they all employ  $\{x:p_1\}, \{y:p_1\}/\emptyset$  for every  $x \neq y$  as cut rules and  $\emptyset/(\pounds:p_1)$  as the only identity axiom.

In addition, the family of canonical calculi studied here includes all labelled calculi from [26] for many-valued logics.

Due to the special form of the rules of canonical calculi (which, except for cuts, enjoy the subformula property), in this family of calculi cut-admissibility guarantees sub-analyticity:

**Proposition 4.1.5.** If a canonical calculus enjoys (strong) cut-admissibility, then it is (strongly) sub-analytic.

Using the semantics, we show below that these two properties are actually equivalent (Corollary 4.4.4).

### 4.2 Partial Non-deterministic Matrices

In this section we introduce the semantic framework of partial non-deterministic matrices (PNmatrices), and show that the consequence relations induced by (finite) PNmatrices are always decidable. In the next section we show that this framework is indeed suitable for all canonical calculi.

PNmatrices are many-valued systems (see Definition 3.1.5) in which all semantic conditions have the following well-behaved form:

**Definition 4.2.1.** An  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I$  over  $\mathcal{U}$  is called *canonical* if there is some  $\diamond \in \diamondsuit_{\mathcal{L}}$ , such that the formulas occurring in  $\mathcal{I}$  are all in  $\{p_1, \dots, p_{ar(\diamond)}\}$ , and I has the form  $\{\diamond(p_1, \dots, p_{ar(\diamond)}) \doteq u_i \mid u_i \in \mathcal{U}'\}$  for some non-empty (finite)  $\mathcal{U}' \subseteq \mathcal{U}$ . In this case we also say that  $\mathcal{I} \Rightarrow I$  is a *canonical*  $\mathcal{L}$ -semantic condition for  $\diamond$ .

The semantic conditions in Example 3.2.12 are canonical  $\mathcal{L}$ -semantic conditions for  $\supset$ . On the other hand, the  $\mathcal{L}$ -semantic condition in Example 3.2.13 is not canonical.

**Example 4.2.2.** Suppose that  $\supset \in \diamondsuit^2_{\mathcal{L}}$ , and let  $\mathcal{U} = \{f, t\}$ . The following are canonical  $\mathcal{L}$ -semantic conditions for  $\supset$  (over  $\mathcal{U}$ ):

$$\{\{p_1 \doteqdot t\}, \{p_2 \doteqdot f\}\} \Rightarrow \{p_1 \supset p_2 \doteqdot f\}$$
$$\{\{p_2 \doteqdot t\}\} \Rightarrow \{p_1 \supset p_2 \doteqdot t\}$$

**Definition 4.2.3.** A *(pre-) partial non-deterministic matrix* ((pre-) *PNmatrix*, for short) for  $\mathcal{L}$  is a many-valued (pre-) system  $\mathbf{M}$  for  $\mathcal{L}$ , in which all  $\mathcal{L}$ -semantic conditions in  $\Lambda_{\mathbf{M}}$  are canonical.

Now, the semantic conditions in (pre-) PNmatrices can be easily presented using many-valued truth tables. However, usual deterministic truth tables would not suffice. Thus, as done in [17], we use non-deterministic truth tables, where there might be more than one option to determine the value of a compound formula according to the values of its immediate subformulas. In fact, we slightly extend this notion, by introducing partial non-deterministic truth tables, in which we also allow empty sets of options to appear. This extension will enable us to have "truth tables presentation" for all (pre-) PNmatrices.

**Definition 4.2.4.** Let  $\diamond \in \Diamond_{\mathcal{L}}$  and let  $\mathcal{U}$  be a set of truth vales.

- 1. A partial non-deterministic truth table (PNtable, for short) for  $\diamond$  over  $\mathcal{U}$  is a function from  $\mathcal{U}^{ar(\diamond)}$  to  $2^{\mathcal{U}}$ .
- 2. A partial  $\mathcal{L}$ -valuation v with  $Im_v \subseteq \mathcal{U}$  respects a PNtable  $\Xi$  for  $\diamond$  over  $\mathcal{U}$  if

$$v(\sigma(\diamond(p_1,\ldots,p_{ar(\diamond)}))) \in \Xi(v(\sigma(p_1)),\ldots,v(\sigma(p_{ar(\diamond)})))$$

for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\sigma(\{p_1, \dots, p_{ar(\diamond)}, \diamond(p_1, \dots, p_{ar(\diamond)})\}) \subseteq Dom_v$ .

**Example 4.2.5.** Let  $\mathcal{L} = \{\star^1\}$ . Consider the following PNtables for  $\star$  over  $\mathcal{U} = \{u_1, u_2\}$ :

The  $\mathcal{L}$ -valuation assigning  $u_2$  to all formulas respects  $\Xi_2$  and  $\Xi_3$ , but not  $\Xi_1$ . In addition, this is the only  $\mathcal{L}$ -valuation (whose image is contained in  $\mathcal{U}$ ) that respects  $\Xi_3$ .

Now, any set of canonical  $\mathcal{L}$ -semantic conditions for some connective naturally induces a PNtable.

**Example 4.2.6.** Suppose that  $\supset \in \diamondsuit^2_{\mathcal{L}}$ . Let  $\mathbf{M}$  be a pre-PNmatrix for  $\mathcal{L}$  such that  $\mathcal{V}_{\mathbf{M}} = \mathcal{U}_{\mathbf{M}} = \{f, t\}$ , and  $\Lambda_{\mathbf{M}}$  consists of the canonical  $\mathcal{L}$ -semantic conditions for  $\supset$  from Example 4.2.2. Consider the PNtable Xi for  $\supset$  over  $\{f, t\}$ :

$$\begin{array}{c|cccc}
\Xi & f & t \\
\hline
f & \{f,t\} & \{t\} \\
\hline
t & \{f\} & \{t\}
\end{array}$$

It is easy to verify that a partial  $\mathcal{L}$ -valuation v with  $Dom_v$  closed under subformulas and  $Im_v \subseteq \{f, t\}$  is M-legal iff it respects  $\Xi$ . Note that this PNtable corresponds to the semantics of "primal implication" from [62].

The general construction is given by:

**Definition 4.2.7.** Let **M** be a (pre-) PNmatrix. For every  $\diamond \in \diamond_{\mathcal{L}}$ ,  $\diamond_{\mathbf{M}}$  is the PNtable for  $\diamond$  over  $\mathcal{V}_{\mathbf{M}}$  defined as follows. For every  $u_1, \ldots, u_{ar(\diamond)} \in \mathcal{V}_{\mathbf{M}}$ ,  $\diamond_{\mathbf{M}}(u_1, \ldots, u_{ar(\diamond)})$  consists of all truth values  $u \in \mathcal{V}_{\mathbf{M}}$  such that for every canonical  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I \in \Lambda_{\mathbf{M}}$  for  $\diamond$ ,  $\diamond(p_1, \ldots, p_{ar(\diamond)}) \doteq u \in I$  whenever for every  $I' \in \mathcal{I}$  we have  $p_i \doteq u_i \in I'$  for some  $1 \leq i \leq ar(\diamond)$ .

**Example 4.2.8.** Let  $\mathcal{L} = \{ \star^1 \}$ . Consider the three pre-PNmatrices  $\mathbf{M}_1, \mathbf{M}_2$  and  $\mathbf{M}_3$  for  $\mathcal{L}$  defined by:  $\mathcal{V}_{\mathbf{M}_i} = \mathcal{U}_{\mathbf{M}_i} = \{u_1, u_2\}$  and:

<sup>&</sup>lt;sup>1</sup>We represent PNtables for binary connectives by two-dimensional tables. The lines range over the first argument, and the columns over the second one.

- 1.  $\Lambda_{\mathbf{M}_1} = \{\{\{p_1 \neq u_1\}\}\} \Rightarrow I_2, \{\{p_1 \neq u_2\}\} \Rightarrow I_1\}$
- 2.  $\Lambda_{\mathbf{M}_2} = \{\{\{p_1 \doteq u_1\}\}\} \Rightarrow I_2\}$
- 3.  $\Lambda_{\mathbf{M}_3} = \{\{\{p_1 \doteq u_1\}\} \Rightarrow I_1, \{\{p_1 \doteq u_1\}\} \Rightarrow I_2, \{\{p_1 \doteq u_2\}\} \Rightarrow I_2\}$

where:  $I_1 = \{ \star p_1 \doteq u_1 \}$  and  $I_2 = \{ \star p_1 \doteq u_2 \}$ . Then for every  $1 \leq i \leq 3$ ,  $\star_{\mathbf{M}_i} = \Xi_i$  given in Example 4.2.5.

**Proposition 4.2.9.** Let  $\mathbf{M}$  be a (pre-) PNmatrix. A partial  $\mathcal{L}$ -valuation v whose domain is closed under subformulas is  $\mathbf{M}$ -legal iff  $Im_v \subseteq \mathcal{U}_{\mathbf{M}}$  and v respects  $\diamond_{\mathbf{M}}$  for every  $\diamond \in \diamondsuit_{\mathcal{L}}$ .

*Proof.* The proof is completely straightforward using the definitions above. Let v be a partial  $\mathcal{L}$ -valuation whose domain is closed under subformulas.

- 1. Suppose that v is  $\mathbf{M}$ -legal. By definition,  $Im_v \subseteq \mathcal{U}_{\mathbf{M}}$ . Let  $\diamond \in \diamondsuit_{\mathcal{L}}$  and let  $\varphi = \diamond(p_1, \dots, p_{ar(\diamond)})$ . We show that v respects  $\diamond_{\mathbf{M}}$ . Let  $\sigma$  be an  $\mathcal{L}$ -substitution with  $\sigma(\{p_1, \dots, p_{ar(\diamond)}, \varphi\}) \subseteq Dom_v$ . We show that  $v(\sigma(\varphi)) \in \diamond_{\mathbf{M}}(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)}))$ . By definition, it suffices to show that for every canonical  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I \in \Lambda_{\mathbf{M}}$  for  $\diamond, \diamond(p_1, \dots, p_{ar(\diamond)}) \doteqdot v(\sigma(\varphi)) \in I$  whenever for every  $I' \in \mathcal{I}$  we have  $p_i \doteqdot v(\sigma(p_i)) \in I'$  for some  $1 \leq i \leq ar(\diamond)$ . Suppose that for every  $I' \in \mathcal{I}$ , v satisfies  $\sigma(I')$ . Since v is  $\mathbf{M}$ -legal it satisfies  $\mathcal{I} \Rightarrow I$ . Hence, v satisfies  $\sigma(I)$  (note that  $\sigma(\psi) \in Dom_v$  for each formula  $\psi$  that occurs in  $\mathcal{I} \Rightarrow I$ ). Since  $\mathcal{I} \Rightarrow I$  is a canonical  $\mathcal{L}$ -semantic condition for  $\diamond$ , it follows that  $\sigma(\varphi) \doteqdot v(\sigma(\varphi)) \in \sigma(I)$ . Thus  $\varphi \doteqdot v(\sigma(\varphi)) \in I$ .
- 2. Suppose that  $Im_v \subseteq \mathcal{U}_{\mathbf{M}}$  and v respects  $\diamond_{\mathbf{M}}$  for every  $\diamond \in \diamondsuit_{\mathcal{L}}$ . We show that v is  $\mathbf{M}$ -legal. Thus we have to show that v satisfies every  $\mathcal{L}$ -semantic condition  $\mathcal{I} \Rightarrow I$  in  $\Lambda_{\mathbf{M}}$ . Let  $\mathcal{I} \Rightarrow I \in \Lambda_{\mathbf{M}}$ . Since  $\mathbf{M}$  is a (pre-) PNmatrix,  $\mathcal{I} \Rightarrow I$  is a canonical  $\mathcal{L}$ -semantic condition for some connective  $\diamond$ . Let  $\sigma$  be an  $\mathcal{L}$ -substitution such that  $\sigma(\varphi) \in Dom_v$  for every  $\varphi$  that occurs in  $\mathcal{I} \Rightarrow I$ . Let  $\varphi = \diamond(p_1, \dots, p_{ar(\diamond)})$ . In particular,  $\varphi$  occurs in I, and thus  $\sigma(\varphi) \in Dom_v$ . Since  $Dom_v$  is closed under subformulas we also have  $\sigma(\{p_1, \dots, p_{ar(\diamond)}\}) \subseteq Dom_v$ . Suppose that v satisfies  $\sigma(I')$  for every  $I' \in \mathcal{I}$ . Then for every  $I' \in \mathcal{I}$ , there is some  $1 \leq i \leq ar(\diamond)$  such that  $p_i \doteqdot v(\sigma(p_i)) \in I$ . The definition of  $\diamond_{\mathbf{M}}$  ensures that  $\varphi \doteqdot u \in I$  for every  $u \in \diamond_{\mathbf{M}}(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)})))$ . Now, since v respects  $\diamond_{\mathbf{M}}$ , we have that  $v(\sigma(\varphi)) \in \diamond_{\mathbf{M}}(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)})))$ . Therefore,  $\varphi \doteqdot v(\sigma(\varphi)) \in I$ . It follows that v satisfies  $\sigma(I)$ .

#### 4.2.1 Effectiveness

The major attractive property of PNmatrices is their *effectiveness*: the logics induced by finite PNmatrices are all decidable. This can be easily shown for the subclass of *proper* PNmatrices, that corresponds to the Nmatrices of [17] (where no empty sets of truth values are allowed to appear in the truth tables of the logical connectives):

**Definition 4.2.10.** A (pre-) PNmatrix **M** for  $\mathcal{L}$ , is called *proper* if  $\mathcal{U}_{\mathbf{M}}$  is non-empty, and for every connective  $\diamond \in \diamond_{\mathcal{L}}$  and  $u_1, \ldots, u_{ar(\diamond)} \in \mathcal{U}_{\mathbf{M}}, \diamond_{\mathbf{M}}(u_1, \ldots, u_{ar(\diamond)}) \cap \mathcal{U}_{\mathbf{M}} \neq \emptyset$ .

**Example 4.2.11.** Let  $\mathcal{L} = \{\star^1\}$ . Consider the three pre-PNmatrices  $\mathbf{M}_1, \mathbf{M}_2$  and  $\mathbf{M}_3$  for  $\mathcal{L}$  given in Example 4.2.8. Then  $\mathbf{M}_1$  and  $\mathbf{M}_2$  are proper, but  $\mathbf{M}_3$  is not proper since  $\star_{\mathbf{M}_3}(u_1) \cap \{u_1, u_2\} = \emptyset$ .

Proper PNmatrices are exactly these in which every partial valuation (whose domain is closed under subformulas) can be extended to a full one, and thus they are *sub*-analytic many-valued systems (see Definition 3.1.16). Hence the effectiveness of proper PNmatrices directly follows from Theorem 3.1.18. In addition, the characterization of strong *sub*-analyticity in canonical sequent calculi (Corollary 4.4.1 below) will immediately follow from this observation as well.

**Proposition 4.2.12.** A (pre-) PNmatrix M (for  $\mathcal{L}$ ) is proper iff it is *sub*-analytic.

Proof (Outline). Suppose that M is proper. The extension of an M-legal partial  $\mathcal{L}$ -valuation whose domain is closed under subformulas is recursively defined by induction on the complexity of formulas. For atomic formulas that are not in the original domain, we arbitrarily choose a value in  $\mathcal{U}_M$ . For non-atomic formulas that are not in the original domain, we arbitrarily choose a value from  $\mathcal{U}_M$  that occurs in the (non-empty) set of options allowed by the corresponding PNtable.

For the converse, note first that if  $\mathcal{U}_{\mathbf{M}}$  is empty, then the empty valuation (whose domain is the empty set) cannot be extended. Otherwise, there is some  $\diamond \in \diamondsuit_{\mathcal{L}}$  and  $u_1, \ldots, u_{ar(\diamond)} \in \mathcal{U}_{\mathbf{M}}$ , such that  $\diamond_{\mathbf{M}}(u_1, \ldots, u_{ar(\diamond)}) \cap \mathcal{U}_{\mathbf{M}} = \emptyset$ . Define  $v : \{p_1, \ldots, p_{ar(\diamond)}\} \to \mathcal{U}_{\mathbf{M}}$  by  $v(p_i) = u_i$ . Obviously, v is an  $\mathbf{M}$ -legal partial  $\mathcal{L}$ -valuation, whose domain is finite and closed under subformulas. However, v cannot be extended to an  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation, since there is no truth value to assign to  $\diamond(p_1, \ldots, p_{ar(\diamond)})$ .

Corollary 4.2.13. Let M be a finite proper PNmatrix for  $\mathcal{L}$ . Given a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  or not.

*Proof.* Directly follows from Theorem 3.1.18 and Proposition 4.2.12.  $\Box$ 

Now, we show that also *non-proper* finite PNmatrices are effective. This is an immediate corollary of Theorem 3.1.19 using the following theorem:

**Theorem 4.2.14.** Let M be a finite PNmatrix for  $\mathcal{L}$ . Given an M-legal partial  $\mathcal{L}$ -valuation v, whose domain is finite and closed under subformulas, it is decidable whether v can be extended to an M-legal (full)  $\mathcal{L}$ -valuation or not.

In other words, the framework of finite PNmatrices allows to effectively check whether a certain partial valuation (whose domain is closed under subformulas) is a restriction of a full one. To prove this theorem we use the following notation and proposition:

**Notation 4.2.15.** For a (pre-) PNmatrix  $\mathbf{M}$  and  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$ , we denote by  $\mathbf{M} \cap \mathcal{U}$  the (pre-) PNmatrix which is identical to  $\mathbf{M}$  except for  $\mathcal{U}_{\mathbf{M} \cap \mathcal{U}} = \mathcal{U}$ .

Obviously, if an  $\mathcal{L}$ -valuation is  $\mathbf{M} \cap \mathcal{U}$ -legal for some  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$ , then it is  $\mathbf{M}$ -legal. It follows that for every PNmatrix  $\mathbf{M}$  and  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$ ,  $\vdash_{\mathbf{M}} \subseteq \vdash_{\mathbf{M} \cap \mathcal{U}}$ .

**Proposition 4.2.16.** Let  $\mathbf{M}$  be a (pre-) PNmatrix for  $\mathcal{L}$ , and let  $\mathcal{F} \subseteq \mathcal{L}$  be closed under subformulas. An  $\mathbf{M}$ -legal partial  $\mathcal{L}$ -valuation v, whose domain is closed under subformulas, can be extended to a (full)  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation iff v is  $\mathbf{M} \cap \mathcal{U}$ -legal for some  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$  such that  $\mathbf{M} \cap \mathcal{U}$  is proper.

*Proof.* Suppose that there is some  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$  such that  $\mathbf{M} \cap \mathcal{U}$  is proper and v is  $\mathbf{M} \cap \mathcal{U}$ -legal. By Proposition 4.2.12, there exists an  $\mathbf{M} \cap \mathcal{U}$ -legal  $\mathcal{L}$ -valuation v' that extends v. Clearly, v' is also  $\mathbf{M}$ -legal.

For the converse, let v' be an M-legal  $\mathcal{L}$ -valuation that extends v. Choose  $\mathcal{U} = Im_{v'}$ . Obviously,  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$  and v is  $\mathbf{M} \cap \mathcal{U}$ -legal. We show that  $\mathbf{M} \cap \mathcal{U}$  is proper. Obviously,  $\mathcal{U}$  is non-empty. Let  $\diamond \in \diamondsuit_{\mathcal{L}}$  and  $u_1, \dots, u_{ar(\diamond)} \in \mathcal{U}$ . Since  $\mathcal{U} = Im_{v'}$ , there are some  $\varphi_1, \dots, \varphi_{ar(\diamond)} \in \mathcal{L}$ , such that  $v'(\varphi_i) = u_i$  for every  $1 \leq i \leq ar(\diamond)$ . Since v' is M-legal,  $v'(\diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})) \in \diamond_{\mathbf{M}}(u_1, \dots, u_{ar(\diamond)})$ . By definition  $v'(\diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})) \in \mathcal{U}$ . Hence  $\diamond_{\mathbf{M} \cap \mathcal{U}}(u_1, \dots, u_{ar(\diamond)}) \cap \mathcal{U} = \diamond_{\mathbf{M}}(u_1, \dots, u_{ar(\diamond)}) \cap \mathcal{U} \neq \emptyset$ .

Proof of Theorem 4.2.14. To verify that v can be extended to an M-legal  $\mathcal{L}$ -valuation, go over all finite sets  $\mathcal{U} \subseteq \mathcal{U}_M$  such that  $M \cap \mathcal{U}$  is proper, and check whether v is  $M \cap \mathcal{U}$ -legal for each of them. Return a positive answer iff such  $\mathcal{U}$  have been found. The correctness is guaranteed by Proposition 4.2.16.

As a corollary we have the following:

Corollary 4.2.17. Let M be a finite PNmatrix for  $\mathcal{L}$ . Given a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{M}} \varphi$  or not.

*Proof.* Directly follows from Theorem 3.1.19 and Theorem 4.2.14.

Remark 4.2.18. As done for ordinary matrices (see, e.g., [93]) it is also possible to define  $\Vdash_{\wp}$ , the consequence relation induced by a family  $\wp$  of PNmatrices to be  $\bigcap_{\mathbf{M} \in \wp} \Vdash_{\mathbf{M}}$ . A PNmatrix can then be thought of as a succinct presentation of a family of proper PNmatrices: following the proof of Proposition 4.2.16, the consequence relation induced by a PNmatrix  $\mathbf{M}$  can be shown to be equivalent to the relation induced by the family of all the proper PNmatrices  $\mathbf{M} \cap \mathcal{U}$  for  $\mathcal{U} \subseteq \mathcal{U}_{\mathbf{M}}$ . Conversely, for every family of proper PNmatrices it is possible to construct an equivalent PNmatrix.

### 4.3 PNmatrices for Canonical Calculi

To show that the semantics of canonical calculi can be characterized by PNmatrices, it suffices to note that the semantic conditions induced by *canonical* rules are all canonical semantic conditions.

**Proposition 4.3.1.** Let r be a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule for a connective  $\diamond$ . Then  $\mathbf{Sem}(r)$  is a canonical  $\mathcal{L}$ -semantic condition for  $\diamond$ .

*Proof.* Directly follows from the definitions.

It follows that for every canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus G, the many-valued pre-system  $M_G$  is actually a pre-PNmatrix:

Corollary 4.3.2. Let G be a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. Then, the many-valued presystem  $\mathbf{M}_{G}$  is a pre-PNmatrix for which  $\vdash_{\mathbf{G}} = \vdash_{\mathbf{M}_{G}}$ . Furthermore, for every set  $X \subseteq \mathcal{L}$ , the logic induced by G and X is identical to the logic induced by the PNmatrix  $\mathbf{M}_{G}^{X}$ .

*Proof.* Directly follow from Proposition 4.3.1, Theorem 3.2.15, and Corollary 3.2.18

**Example 4.3.3.** Let  $\mathcal{L} = \{ \supset^2, \wedge^2, T^2 \}$ , and **G** be a canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus that consists of the following rules:

```
\begin{array}{lll} (\texttt{f}:weak) & \{\emptyset\}/\{\texttt{f}:p_1\} & (\texttt{t}:weak) & \{\emptyset\}/\{\texttt{t}:p_1\} \\ (cut) & \{\texttt{f}:p_1\}, \{\texttt{t}:p_1\}/\emptyset & (id) & \emptyset/\{\texttt{f}:p_1, \texttt{t}:p_1\} \\ (\texttt{f}:\supset) & \{\texttt{t}:p_1\}, \{\texttt{f}:p_2\}/\{\texttt{f}:p_1\supset p_2\} & (\texttt{t}:\supset) & \{\texttt{t}:p_2\}/\{\texttt{t}:p_1\supset p_2\} \\ (\texttt{f}:\land) & \{\texttt{f}:p_1, \texttt{f}:p_2\}/\{\texttt{f}:p_1\land p_2\} & (\texttt{t}:\land) & \{\texttt{t}:p_1\}, \{\texttt{t}:p_2\}/\{\texttt{t}:p_1\land p_2\} \\ (\texttt{f}:T) & \{\texttt{f}:p_2\}/\{\texttt{f}:p_1Tp_2\} & (\texttt{t}:T) & \{\texttt{t}:p_1\}/\{\texttt{t}:p_1Tp_2\} \end{array}
```

The rules for  $\wedge$  are the usual ones (those of **LK**), while the rules for  $\supset$  are those employed in primal logic [62] (see Example 2.2.6). T is standing for the "Tonk" connective, and the two rules above are equivalent to its original introduction rule from [83]. We sketch the modular construction of the pre-PNmatrix  $\mathbf{M}_{\mathbf{G}}$ .

- 1. Begin with  $G_0$ , a canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus, that does not have any rules besides the two weakening rules above. Then,  $\mathcal{V}_{\mathbf{M}_{\mathbf{G}_0}} = 2^{\mathcal{L}_2} = \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}, \{\mathbf{f}, \mathbf{t}\}\}$ . Now since  $\mathsf{P}_{\mathbf{G}_0}$  is empty,  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}_0}} = \mathcal{V}_{\mathbf{M}_{\mathbf{G}_0}}$ . Since  $\mathsf{R}_{\mathbf{G}_0}$  is empty as well, there are no semantic  $\mathcal{L}$ -conditions in  $\Lambda_{\mathbf{M}_{\mathbf{G}_0}}$ , and thus it is represented by "completely non-deterministic" PNtables, that is  $\diamond_{\mathbf{M}_{\mathbf{G}_0}}(\mathsf{X}_1, \mathsf{X}_2) = \mathcal{V}_{\mathbf{M}_{\mathbf{G}_0}}$  for every  $\diamond \in \{\supset, \land, T\}$  and  $\mathsf{X}_1, \mathsf{X}_2 \in \mathcal{V}_{\mathbf{M}_{\mathbf{G}_0}}$ .
- 2. Add ( $\mathbf{f}$ :  $\supset$ ) to  $\mathbf{G}_0$  to obtain  $\mathbf{G}_1$ . This introduces the following canonical  $\mathcal{L}$ -semantic condition for  $\supset$  in  $\Lambda_{\mathbf{M}_{\mathbf{G}_1}}$ :

$$\mathbf{Sem}((\mathbf{f}: \supset)) = \{ \{p_1 \doteqdot \{\mathbf{t}\}, p_1 \doteqdot \{\mathbf{f}, \mathbf{t}\}\}, \{p_2 \doteqdot \{\mathbf{f}\}, p_2 \doteqdot \{\mathbf{f}, \mathbf{t}\}\} \} \Rightarrow \{p_1 \supset p_2 \doteqdot \{\mathbf{f}\}, p_1 \supset p_2 \doteqdot \{\mathbf{f}, \mathbf{t}\} \}.$$

Consequently the PN table of  $\supset$  in  $\mathbf{M}_{\mathbf{G}_1}$  is:

$\supset_{\mathbf{M}_{\mathbf{G}_1}}$	Ø	$\{\mathtt{f}\}$	{t}	{f,t}
Ø	$2^{\pounds_{2}}$	$2^{\pounds_{2}}$	$2^{\pounds_{2}}$	2 <sup>£</sup> 2
{f}	$2^{\pounds_{2}}$	2 <sup>£</sup> 2	$2^{\pounds_{2}}$	2£2
{t}	$2^{\pounds_2}$	$\{\{f\}, \{f, t\}\}$	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$
ff,t	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$	$2^{\pounds_{2}}$	$\{\{f\}, \{f,t\}\}$

3. Add all other canonical rules of G to  $G_1$  to obtain  $G_2$ . This introduces one more canonical  $\mathcal{L}$ -semantic condition for  $\supset$ , two canonical  $\mathcal{L}$ -semantic conditions for  $\land$ , and two for T. The PNtables in  $M_{G_2}$  are given by:

$\supset_{\mathbf{M}_{\mathbf{G}_2}}$	Ø	{f}	{t}	$\{f,t\}$
Ø	2 <sup>£</sup> 2	2£2	$\{\{t\}, \{f, t\}\}$	$\{\{t\}, \{f, t\}\}$
$\{\mathtt{f}\}$	$2^{\pounds_{2}}$	$2^{\pounds_{2}}$	$\{\{t\}, \{f, t\}\}$	$\{\{t\},\{f,t\}\}$
{t}	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$	$\{\{t\}, \{f, t\}\}$	$\{\{f,t\}\}$
$\{f,t\}$	$\{\{t\},\{f,t\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$	$\{\{t\}, \{f, t\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$
$\wedge_{\mathbf{M}_{\mathbf{G}_2}}$	Ø	{f}	{t}	$\{f,t\}$
Ø	2 <sup>£</sup> 2	$\{\{f\}, \{f,t\}\}$	2£2	$\{\{f\}, \{f, t\}\}$
$\{\mathtt{f}\}$	$\{\{f\}, \{f, t\}\}$	$\{\{f\}, \{f, t\}\}$	$\{\{f\}, \{f,t\}\}$	$\{\{\mathtt{f}\},\{\mathtt{f},\mathtt{t}\}\}$
{t}	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$	$\{\{t\},\{f,t\}\}$	$\{\{f,t\}\}$
$\{\mathtt{f},\mathtt{t}\}$	$\{\{f\}, \{f, t\}\}$	$\{\{f\},\{f,t\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$
ı				
$T_{\mathbf{M}_{\mathbf{G}_2}}$	Ø	{f}	{t}	{f,t}
Ø	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$	$2^{\pounds_{2}}$	$\{\{f\}, \{f,t\}\}$
{f}	$2^{\pounds_{2}}$	$\{\{\mathtt{f}\},\{\mathtt{f},\mathtt{t}\}\}$	$2^{\pounds_{2}}$	$\{\{f\}, \{f, t\}\}$
{t}	$\{\{t\}, \{f, t\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$	$\{\{t\},\{f,t\}\}$	$\{\{f,t\}\}$
$\{f,t\}$	$\{\{t\}, \{f, t\}\}$	$\{\{\mathtt{f},\mathtt{t}\}\}$	$\{\{t\},\{f,t\}\}$	$\{\{f,t\}\}$

- 4. Add (id) to  $G_2$  to obtain  $G_3$ . This does not effect  $\Lambda_{M_{G_2}}$ , but it does change the legal truth values, and we now have  $\mathcal{U}_{M_{G_3}} = \{\{f\}, \{t\}, \{f, t\}\}\}$  (see Example 3.2.9).
- 5. Finally add (cut) to  $G_3$  to obtain G. This overrules another truth value, and we finally have  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \{\{\mathbf{f}\}, \{\mathbf{t}\}\}$ . Note that the PNtables of  $\mathbf{M}_G$  are the same as in  $\mathbf{M}_{\mathbf{G}_2}$ . Nevertheless, since  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \{\{\mathbf{f}\}, \{\mathbf{t}\}\}$ , we can reduce them to the following tables:

Note that  $\mathbf{M}_{\mathbf{G}}$  is not proper since  $T_{\mathbf{M}_{\mathbf{G}}}(\{\mathbf{t}\}, \{\mathbf{f}\}) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \emptyset$ .

**Example 4.3.4.** Suppose that  $\mathcal{L}$  contains a unary connective denoted by  $\neg$ , and let  $\mathbf{G}$  be a canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus. Let  $(\mathbf{f}:\neg)$  and  $(\mathbf{t}:\neg)$  be the two usual rules for  $\neg$ , that is the rules  $\{\mathbf{t}:p_1\}/\{\mathbf{f}:\neg p_1\}$  and  $\{\mathbf{f}:p_1\}/\{\mathbf{t}:\neg p_1\}$  (respectively). The following tables present  $\neg_{\mathbf{M}_{\mathbf{G}}}$  for four different options: (i) there are no canonical rules for  $\neg$  in  $\mathbf{G}$ ; (ii)  $(\mathbf{f}:\neg)$  is the only canonical rules for  $\neg$  in  $\mathbf{G}$ ; and (iv)  $(\mathbf{f}:\neg)$  and  $(\mathbf{t}:\neg)$  are the only canonical rules for  $\neg$  in  $\mathbf{G}$ .

Remark 4.3.5. The semantics of PNmatrices obtained for canonical calculi according to the definitions above coincide with the Nmatrices semantics suggested in [17, 19] (see also [21]) for the (narrower) families of canonical systems studied there. The transition from Nmatrices to PNmatrices makes it possible to provide semantics for *every* canonical calculi, while [17, 19] handle only a subset of them, called *coherent* canonical calculi. Indeed, the calculus  $\mathbf{G}$  from Example 4.3.3 is not coherent (because of the rules for T), and it is captured by the (non-proper) PNmatrix given above.

It can be easily verified that for **LK** (see Example 2.2.20) we obtain the usual twovalued truth tables semantics of classical logic. In addition, the PNmatrix obtained for **LK**'s (*cut*)-free fragment is practically equivalent to the semantics of three-valued Schütte valuations for this calculus (see [58]). Obviously, since **LK** has cut-elimination, this semantics coincides with the usual classical semantics when it comes to the provability of sequents from the *empty set of assumptions*. Similarly, the PNmatrix for **LK**'s (*id*)-free fragment coincides with the three-valued semantics introduced in [64] for this fragment.<sup>2</sup> Next, we demonstrate how this semantics can be used to prove that certain applications of cuts or identity axioms are unavoidable.

**Example 4.3.6.** Using the PNmatrix semantics of canonical calculi, one can easily verify that using (cut) is unavoidable in a given derivation. For example, we obviously have  $\{t:p_1 \supset p_2\} \vdash_{\mathbf{LK}} \{t:p_1 \supset (p_3 \supset p_2)\}$  (see Example 2.2.20 for a precise definition of  $\mathbf{LK}$ ). We show that the  $\mathcal{L}_{cl}$ -sequent  $\{t:p_1 \supset (p_3 \supset p_2)\}$  has no cut-free derivation in  $\mathbf{LK}$  from  $\{t:p_1 \supset p_2\}$ . Consider a partial  $\mathcal{L}_{cl}$ -valuation v with:

- 1.  $Dom_v = \{p_1, p_2, p_3, p_1 \supset p_2, p_3 \supset p_2, p_1 \supset (p_3 \supset p_2)\}.$
- 2.  $v(p_1) = v(p_3) = \{t\}, \ v(p_2) = v(p_3 \supset p_2) = \{f\}, \ v(p_1 \supset (p_3 \supset p_2)) = \{f\}, \ v(p_1 \supset p_2) = \{f, t\}.$

Then v is  $\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}$ -legal. To see this note that  $\mathcal{U}_{\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}} = \{\{\mathbf{f}\}, \{\mathbf{f}, \mathbf{t}\}\}, \text{ and the PNtable } \supset_{\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}} \text{ is the following one (omitting the occurrences of the non-legal truth-value } \emptyset):$ 

$\supset_{\mathbf{M}_{\mathbf{LK}_{cf}}}$	{f}	{t}	$\{f,t\}$
{f}	$\{\{t\}, \{f, t\}\}$	$\{\{t\},\{f,t\}\}$	$\{\{t\}, \{f, t\}\}$
{t}	$\{\{f\}, \{f, t\}\}$	$\{\{t\}, \{f, t\}\}$	$\{\{f,t\}\}$
$\{f,t\}$	$\{\{f,t\}\}$	$\{\{t\}, \{f, t\}\}$	$\{\{f,t\}\}$

Since  $\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}$  is proper, the partial  $\mathcal{L}_{cl}$ -valuation v can be extended to an  $\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}$ -legal  $\mathcal{L}_{cl}$ -valuation v'. v' is a model of  $\{\mathbf{t}:p_1\supset p_2\}$  but not of  $\{\mathbf{t}:p_1\supset (p_3\supset p_2)\}$ , and so we have  $\{\mathbf{t}:p_1\supset p_2\} \not\vdash_{\mathbf{M}_{\mathbf{L}\mathbf{K}_{cf}}} \{\mathbf{t}:p_1\supset (p_3\supset p_2)\}$ . Corollary 4.3.2 entails that  $\{\mathbf{t}:p_1\supset p_2\}\not\vdash_{\mathbf{L}\mathbf{K}_{cf}} \{\mathbf{t}:p_1\supset (p_3\supset p_2)\}$ .

**Example 4.3.7.** Using the PNmatrix semantics, one can easily verify that using (id) is unavoidable in a given derivation. For example, clearly  $\{\mathbf{t}:\neg p_1\} \vdash_{\mathbf{LK}} \{\mathbf{t}:\neg (p_1 \land p_2)\}$ . Consider a partial  $\mathcal{L}_{cl}$ -valuation v with  $Dom_v = \{p_1, p_2, \neg p_1, p_1 \land p_2, \neg (p_1 \land p_2)\}$ , defined by  $v(p_1) = \emptyset$ ,  $v(\neg p_1) = v(p_2) = v(p_1 \land p_2) = \{\mathbf{t}\}$ , and  $v(\neg (p_1 \land p_2)) = \{\mathbf{f}\}$ . It is easy to verify that v is  $\mathbf{M_{LK_{if}}}$ -legal:  $\mathcal{U}_{\mathbf{M_{LK_{if}}}} = \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}\}$ , the PNtable  $\wedge_{\mathbf{M_{LK_{if}}}}$  is the same as  $\wedge_{\mathbf{M_{G_2}}}$  from Example 4.3.3, and the PNtable  $\neg_{\mathbf{M_{LK_{if}}}}$  is the one given in case (iv) in Example 4.3.4. Since  $\mathbf{M_{LK_{if}}}$  is proper, v can be extended to an  $\mathbf{M_{LK_{if}}}$ -legal  $\mathcal{L}_{cl}$ -valuation v'.

<sup>&</sup>lt;sup>2</sup>Note that [58] and [64] concern also the usual quantifiers of **LK**, while here we only investigate its *propositional* fragment.

v' is a model of  $\{t:\neg p_1\}$  but not of  $\{t:\neg (p_1 \wedge p_2)\}$ . Hence,  $\{t:\neg p_1\} \nvdash_{\mathbf{M_{LK}}_{if}} \{t:\neg (p_1 \wedge p_2)\}$ . Corollary 4.3.2 entails that  $\{t:\neg p_1\} \nvdash_{\mathbf{LK}_{if}} \{t:\neg (p_1 \wedge p_2)\}$ . In other words,  $\{t:\neg (p_1 \wedge p_2)\}$  has no identity-axiom-free derivation in  $\mathbf{LK}$  from  $\{t:\neg p_1\}$ .

Now, the decidability of canonical calculi and logics induced by canonical calculi are immediate corollaries.

Corollary 4.3.8. Given a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , a finite set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, it is decidable whether  $\mathcal{S} \vdash_{\mathbf{G}} s$  or not. In addition, given a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , a set of labels  $\mathbf{X} \subseteq \mathcal{L}$ , a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{G}}^{\mathbf{X}} \varphi$  or not.

*Proof.* Construct  $\mathbf{M}_{\mathbf{G}}$  according to the definitions above and use Corollary 3.2.16, Theorem 4.2.14, and Corollary 4.3.2.

We note that the use of non-deterministic truth tables is essential for charactering the logics induced by arbitrary canonical calculi. Indeed, the use of non-deterministic semantics is unavoidable in cases of "syntactic under-specification" in the canonical rules for the connectives (see [21]). But, even when the calculus employs the usual connectives with their ordinary introduction rules, non-deterministic truth tables are required to characterize the cut-free and the identity-axiom-free fragments of the calculus. The next proposition demonstrates this claim.

**Definition 4.3.9.** Let **M** be a (pre-) PNmatrix for  $\mathcal{L}$ . A connective  $\diamond \in \diamondsuit_{\mathcal{L}}$  is called deterministic in **M** if  $\diamond_{\mathbf{M}}(u_1, \dots, u_{ar(\diamond)}) \cap \mathcal{U}_{\mathbf{M}}$  is a singleton for every  $u_1, \dots, u_{ar(\diamond)} \in \mathcal{U}_{\mathbf{M}}$ .

**Proposition 4.3.10.** Suppose that  $\mathcal{L}$  contains a unary connective denoted by  $\neg$ , and let  $\mathbf{G}$  be a canonical  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}} \rangle$ -calculus, whose rules for  $\neg$  are the usual rules:  $\{\mathbf{t}:p_1\}/\{\mathbf{f}:\neg p_1\}$  and  $\{\mathbf{f}:p_1\}/\{\mathbf{t}:\neg p_1\}$ . Let  $\mathbf{M}$  be any finite PNmatrix for  $\mathcal{L}$ . If  $\neg$  is deterministic in  $\mathbf{M}$  then:  $\Vdash_{\mathbf{M}} \neq \Vdash_{\mathbf{G}_{cf}}^{\{\mathbf{t}\}}$  and  $\Vdash_{\mathbf{M}} \neq \Vdash_{\mathbf{G}_{if}}^{\{\mathbf{t}\}}$ .

*Proof.* Assume that  $\neg$  is deterministic in  $\mathbf{M}$ .

1. Suppose that  $\Vdash_{\mathbf{M}} = \Vdash_{\mathbf{G}_{cf}}^{\{t\}}$ . Note that for every  $n \geq 0$ ,  $\{\neg^i p_1 \mid i > n\} \not\Vdash_{\mathbf{G}_{cf}}^{\{t\}} \neg^n p_1$  (it is easy to verify this using Corollary 4.3.2). Consequently,  $\{\neg^i p_1 \mid i > n\} \not\Vdash_{\mathbf{M}} \neg^n p_1$  for every  $n \geq 0$ . For every  $n \geq 0$ , let  $v_n$  be an  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation, which is a model (with respect to  $\mathbf{M}$ ) of  $\{\neg^i p_1 \mid i > n\}$ , and not of  $\neg^n p_1$ . We show that  $v_m(p_1) \neq v_n(p_1)$  for every  $n > m \geq 0$  (and so,  $\mathbf{M}$  is infinite). Let  $n > m \geq 0$ . Since  $v_m$  is a model of  $\neg^n p_1$ , and  $v_n$  is not a model of  $\neg^n p_1$ , we have  $v_m(\neg^n p_1) \neq v_n(\neg^n p_1)$ . This implies (using the fact that  $\neg_{\mathbf{M}}$  is deterministic) that  $v_m(p_1) \neq v_n(p_1)$ .

2. Suppose that  $\Vdash_{\mathbf{M}} = \Vdash_{\mathbf{G}_{if}}^{\{\mathbf{t}\}}$ . Note that  $(i) \neg^n p_1 \Vdash_{\mathbf{G}_{if}}^{\{\mathbf{t}\}} \neg^m p_1$  for every  $n, m \in \mathbb{N}_{even}$  provided that  $n \leq m$ ; and  $(ii) \neg^{n+2} p_1 \nvDash_{\mathbf{G}_{if}}^{\{\mathbf{t}\}} \neg^n p_1$  for every  $n \in \mathbb{N}$  (it is easy to verify this using Corollary 4.3.2). For every  $n \geq 0$ , let  $v_n$  be an  $\mathbf{M}$ -legal  $\mathcal{L}$ -valuation, which is a model (with respect to  $\mathbf{M}$ ) of  $\neg^{n+2} p_1$ , but not of  $\neg^n p_1$ . We show that  $v_n(p_1) \neq v_m(p_1)$  for every  $n, m \in \mathbb{N}_{even}$  such that n < m (and so,  $\mathbf{M}$  is infinite). Let  $n, m \in \mathbb{N}_{even}$  such that n < m. Then, since  $v_n$  is a model of  $\neg^{n+2} p_1$ , (i) implies that  $v_n$  is a model of  $\neg^m p_1$ . On the other hand,  $v_m$  is not a model of  $\neg^m p_1$ . This implies (using the fact that  $\neg_{\mathbf{M}}$  is deterministic) that  $v_n(p_1) \neq v_m(p_1)$ .

## 4.4 Characterization of Proof-Theoretic Properties

In addition to decision procedures, the semantics of PNmatrices is useful for checking proof-theoretic properties of canonical calculi. In this section we use the semantic characterizations of strong *sub*-analyticity, strong cut-admissibility, and axiom-expansion from the previous chapter to obtain simple *decidable* criteria for these properties in canonical calculi.

Using Theorem 3.3.3 and Proposition 4.2.12, the criterion for strong sub-analyticity is immediate:

Corollary 4.4.1. A canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is strongly *sub*-analytic iff  $\mathbf{M}_{\mathbf{G}}$  is proper.

*Proof.* By Theorem 3.3.3, **G** is strongly sub-analytic iff  $\mathbf{M}_{\mathbf{G}}$  is sub-analytic. By Proposition 4.2.12, this holds iff  $\mathbf{M}_{\mathbf{G}}$  is proper.

To characterize strong cut-admissibility, we use the following lemma:

**Lemma 4.4.2.** Let **G** be a canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. For every  $\diamond \in \diamond_{\mathcal{L}}$ , and every  $X_1, \ldots, X_{ar(\diamond)}, X'_1, \ldots, X'_{ar(\diamond)} \subseteq \mathcal{L}$  such that  $X_i \subseteq X'_i$  for every  $1 \leq i \leq ar(\diamond)$ , we have  $\diamond_{\mathbf{M}_{\mathbf{G}}}(X'_1, \ldots, X'_{ar(\diamond)}) \subseteq \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \ldots, X_{ar(\diamond)})$ .

*Proof.* Directly follows from the definitions.

**Theorem 4.4.3.** A canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** enjoys strong cut-admissibility iff  $\mathbf{M}_{\mathbf{G}}$  is proper.

Proof. First, if  $\mathbf{M}_{\mathbf{G}}$  is not proper, then (by Corollary 4.4.1)  $\mathbf{G}$  is not sub-analytic. By Proposition 4.1.5,  $\mathbf{G}$  does not enjoy strong cut-admissibility. Now suppose that  $\mathbf{M}_{\mathbf{G}}$  is proper. If  $\mathbf{G}$  has no cut rules then we are obviously done. Assume otherwise. We use Theorem 3.3.9 to show that  $\mathbf{G}$  enjoys strong cut-admissibility. Thus, we have to show that for every  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v, there exists an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation v' such that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ . Note that we have:

- $\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}} = \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  since the additional primitive  $\mathcal{L}$ -rules in  $s(\mathbf{G})$  do not have any effect on the legal truth values.
- $\mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}} = \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})}} \cup \{\pounds\}$  by Lemma 3.3.11.
- $\mathcal{L} \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  since **G** has at least one cut rule.
- $\diamond_{\mathbf{M}_{s(\mathbf{G})_{cf}}} = \diamond_{\mathbf{M}_{\mathbf{G}}}$  for every  $\diamond \in \diamondsuit_{\mathcal{L}}$  since  $s(\mathbf{G})_{cf}$  and  $\mathbf{G}$  differ only in their primitive rules.

Now, let v be an  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal  $\mathcal{L}$ -valuation. The construction of v' is done by recursion on the complexity of formulas. First, for atomic formulas, if  $v(p) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ , we choose v'(p) = v(p). Otherwise, we (arbitrarily) choose v'(p) to be an element of  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  (which is non-empty since  $\mathbf{M}_{\mathbf{G}}$  is proper). Now, let  $\diamond \in \Diamond_{\mathcal{L}}$ ,  $\varphi = \diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})$ , and suppose  $v'(\varphi_i)$  was defined for every  $1 \leq i \leq ar(\diamond)$ . We choose  $v'(\varphi)$  to be equal to  $v(\varphi)$  if the latter is in  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . Otherwise,  $v(\varphi) = \mathcal{L}$ , and we choose  $v'(\varphi)$  to be some element of  $\diamond_{\mathbf{M}_{\mathbf{G}}}(v'(\varphi_1), \dots, v'(\varphi_{ar(\diamond)})) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  (such an element exists since  $\mathbf{M}_{\mathbf{G}}$  is proper). We show that for every  $\varphi \in \mathcal{L}$ :  $v'(\varphi) = v(\varphi)$  iff  $v(\varphi) \neq \mathcal{L}$ . First, if  $v(\varphi) \neq \mathcal{L}$  then  $v(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . In this case we chose  $v'(\varphi) = v(\varphi)$ . Now, if  $v(\varphi) = \mathcal{L}$  then  $v(\varphi) \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ , and  $v'(\varphi) \neq v(\varphi)$  since we chose  $v'(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . It remains to show that v' is  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ legal. By definition,  $v'(\varphi) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}} \subseteq \mathcal{U}_{\mathbf{M}_{s(\mathbf{G})_{cf}}}$  for every  $\varphi \in \mathcal{L}$ . Suppose (for contradiction) that  $v'(\varphi) \not\in \diamond_{\mathbf{M}_{s(\mathbf{G})_{cf}}}(v'(\varphi_1), \dots, v'(\varphi_{ar(\diamond)}))$  for some formula  $\varphi = \diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})$ . Thus  $v'(\varphi) \notin \diamond_{\mathbf{M}_{\mathbf{G}}}(v'(\varphi_1), \dots, v'(\varphi_{ar(\diamond)})).$  If  $v(\varphi) = \pounds$ , then our construction ensures that  $v'(\varphi) \in \diamond_{\mathbf{M}_{\mathbf{G}}}(v'(\varphi_1), \dots, v'(\varphi_{\mathit{ar}(\diamond)})) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}. \quad \text{Hence, } v(\varphi) \neq \pounds, \text{ and so } v'(\varphi) = v(\varphi).$ Now, for every  $1 \leq i \leq ar(\diamond)$ ,  $v'(\varphi_i) = v(\varphi_i)$  iff  $v(\varphi_i) \neq \mathcal{L}$ , and thus  $v'(\varphi_i) \subseteq v(\varphi_i)$ . Lemma 4.4.2 entails that  $v(\varphi) \notin \diamond_{\mathbf{M}_{\mathbf{G}}}(v(\varphi_1), \dots, v(\varphi_{ar(\diamond)}))$ . This contradicts the fact that v is  $\mathbf{M}_{s(\mathbf{G})_{cf}}$ -legal.

As a corollary we obtain the following correspondence in canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculi:

Corollary 4.4.4. The following are equivalent for every canonical  $(\mathcal{L}, \mathcal{L})$ -calculus **G**:

- $\mathbf{M}_{\mathbf{G}}$  is proper.
- $\bullet$   ${\bf G}$  is strongly sub-analytic.
- $\bullet$   $\, {\bf G}$  enjoys strong cut-admissibility.

*Proof.* The equivalence follows by Corollary 4.4.1 and Theorem 4.4.3.  $\Box$ 

Corollary 4.4.5. The question whether a given canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus enjoys strong sub-analyticity and strong cut-admissibility is decidable.

*Proof.* Construct  $\mathbf{M}_{\mathbf{G}}$  and check whether it is proper or not.

Remark 4.4.6. A similar equivalence was proved in [17] for the (narrower) family of canonical calculi studied there. Note that [17] included also a necessary and sufficient condition for strong cut-admissibility using a simple syntactic condition of *coherence*. It can be shown that the coherence of a calculus G from the family studied in [17] is equivalent to the fact that  $M_G$  is proper.

Since canonical calculi are decidable, the property of *axiom-expansion* in them is decidable as well (see Definition 2.3.21). Indeed, one has to verify that

$$\{(\mathbf{X}:p_i) \mid \emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}, 1 \leq i \leq ar(\diamond)\} \vdash_{\mathbf{G}_{if}} (\mathbf{Y}: \diamond (p_1, \dots, p_{ar(\diamond)}))$$

for every  $Y \subseteq \mathcal{L}$  such that  $\emptyset/(Y:p_1) \in P_{\mathbf{G}}$  and connective  $\diamond \in \diamondsuit_{\mathcal{L}}$  (recall that  $\mathbf{G}_{if}$  denotes the calculus obtained from  $\mathbf{G}$  by discarding all identity axioms). Alternatively, the following semantic criterion can be used:

Corollary 4.4.7. Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus. A connective  $\diamond \in \diamondsuit_{\mathcal{L}}$  of arity n admits axiom-expansion in **G** if  $\diamondsuit_{\mathbf{M_G}}(X_1, \dots, X_n) \cap \mathcal{U}_{\mathbf{M_{G_{if}}}} \subseteq \mathcal{U}_{\mathbf{M_G}}$  for every  $X_1, \dots, X_n \in \mathcal{U}_{\mathbf{M_G}}$ . If  $\mathbf{M_{G_{if}}}$  is proper, then the converse holds as well.

Proof. Suppose that  $\diamond$  does not admit axiom-expansion in  $\mathbf{G}$ . By Corollary 3.3.12, there exists an  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal  $\mathcal{L}$ -valuation v, such that  $\mathbf{X} \cap v(p_i) \neq \emptyset$  for every  $\mathbf{X} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $1 \leq i \leq n$ , but  $\mathbf{X}_0 \cap v(\diamond(p_1,\ldots,p_n)) = \emptyset$  for some  $\mathbf{X}_0 \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{X}_0:p_1) \in \mathsf{P}_{\mathbf{G}}$ . Since v is  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal, we have  $v(\diamond(p_1,\ldots,p_n)) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}}$  and  $v(\diamond(p_1,\ldots,p_n)) \in \diamond_{\mathbf{M}_{\mathbf{G}_{if}}}(v(p_1),\ldots,v(p_n))$ . Since  $\mathbf{X}_0 \cap v(\diamond(p_1,\ldots,p_n)) = \emptyset$ , we have that  $v(\diamond(p_1,\ldots,p_n)) \not\in \mathcal{L}(\emptyset/(\mathbf{X}_0:p_1))$ , and so  $v(\diamond(p_1,\ldots,p_n)) \not\in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . By definition, we have  $\diamond_{\mathbf{M}_{\mathbf{G}_{if}}} = \diamond_{\mathbf{M}_{\mathbf{G}}}$ , and hence,  $\diamond_{\mathbf{M}_{\mathbf{G}}}(v(p_1),\ldots,v(p_n)) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}} \not\subseteq \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . Finally, note that  $v(p_1),\ldots,v(p_n) \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ , since v is  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal and  $\mathbf{X} \cap v(p_i) \neq \emptyset$  for every  $\mathbf{X} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{X}:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $1 \leq i \leq n$ .

For the converse, suppose that  $\mathbf{M}_{\mathbf{G}_{if}}$  is proper. Assume that there are  $\mathbf{X}_1, \dots, \mathbf{X}_n \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  and  $\mathbf{X} \in \diamond_{\mathbf{M}_{\mathbf{G}}}(\mathbf{X}_1, \dots, \mathbf{X}_n) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}}$  such that  $\mathbf{X} \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . Consider the partial  $\mathcal{L}$ -valuation v, defined by:  $Dom_v(v) = \{p_1, \dots, p_n, \diamond (p_1, \dots, p_n)\}$ ,  $v(p_i) = \mathbf{X}_i$  for  $1 \leq i \leq n$ , and  $v(\diamond(p_1, \dots, p_n)) = \mathbf{X}$ . v is  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal, and since  $\mathbf{M}_{\mathbf{G}_{if}}$  is proper, there is an  $\mathbf{M}_{\mathbf{G}_{if}}$ -legal (full)  $\mathcal{L}$ -valuation v' that extends v (Proposition 4.2.12). Now,  $\mathbf{Y} \cap v'(p_i) \neq \emptyset$  for every  $\mathbf{Y} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{Y}:p_1) \in \mathsf{P}_{\mathbf{G}}$  and  $1 \leq i \leq n$  (since  $\mathbf{X}_1, \dots, \mathbf{X}_n \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ ), but  $\mathbf{Y} \cap v'(\diamond(p_1, \dots, p_n)) \neq \emptyset$  for some  $\mathbf{Y} \subseteq \mathcal{L}$  such that  $\emptyset/(\mathbf{Y}:p_1) \in \mathsf{P}_{\mathbf{G}}$  (since  $\mathbf{X} \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ ). By Corollary 3.3.12,  $\diamond$  does not admit axiom-expansion in  $\mathbf{G}$ .

In the case of ordinary canonical  $\langle \mathcal{L}, \pounds_2 \rangle$ -calculi (as those studied in [17]) it is possible to obtain a simpler characterization of axiom-expansion, showing that the connectives that admit axiom-expansion are exactly the *deterministic* ones. This correspondence was

originally proved in [11]. Here we provide a new semantic proof of it, using the more general characterization given in Corollary 4.4.7.

Corollary 4.4.8. Let **G** be a pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus with  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ , in which the conclusion of each non-primitive rule takes the form  $\{\mathbf{x}: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  for some  $\mathbf{x} \in \mathcal{L}_2$  and  $\diamond \in \diamondsuit_{\mathcal{L}}$ . Suppose that  $\mathbf{M}_{\mathbf{G}}$  is proper. A connective  $\diamond \in \diamondsuit_{\mathcal{L}}^n$  admits axiom-expansion in **G** iff  $\diamond_{\mathbf{M}_{\mathbf{G}}}$  is deterministic (see Definition 4.3.9).

*Proof.* Following Example 3.2.9,  $\mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}} = \{\emptyset, \{\mathbf{f}\}, \{\mathbf{t}\}\}, \text{ and } \mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \{\{\mathbf{f}\}, \{\mathbf{t}\}\}\}.$  In addition, for the given  $\mathbf{G}$ , the following properties of  $\mathbf{M}_{\mathbf{G}}$  are easily obtained from the definitions:

- $\{\{f\}, \{t\}\} \subseteq \Diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_n) \text{ iff } \emptyset \in \Diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_n) \text{ for every } X_1, \dots, X_n \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}.$
- $\mathbf{M}_{\mathbf{G}_{if}}$  is proper. Otherwise,  $\diamond_{\mathbf{M}_{\mathbf{G}_{if}}}(\mathbf{X}_1, \dots, \mathbf{X}_{ar(\diamond)}) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}} = \emptyset$  for some  $\diamond \in \diamondsuit_{\mathcal{L}}$  and  $\mathbf{X}_1, \dots, \mathbf{X}_{ar(\diamond)} \in \mathcal{U}_{M_{\mathbf{G}_{if}}}$ . Let  $\mathbf{X}_i' = \mathbf{X}_i \cup \{\mathbf{f}\}$  for every  $1 \leq i \leq ar(\diamond)$ . By Lemma 4.4.2,  $\diamond_{\mathbf{M}_{\mathbf{G}_{if}}}(\mathbf{X}_1', \dots, \mathbf{X}_{ar(\diamond)}') \subseteq \diamond_{\mathbf{M}_{\mathbf{G}_{if}}}(\mathbf{X}_1, \dots, \mathbf{X}_{ar(\diamond)})$ , and so  $\diamond_{\mathbf{M}_{\mathbf{G}}}(\mathbf{X}_1', \dots, \mathbf{X}_{ar(\diamond)}') \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}}} = \emptyset$ . This contradicts the fact that  $\mathbf{M}_{\mathbf{G}}$  is proper.

Suppose that  $\diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_{ar(\diamond)}) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  is not a singleton for some  $X_1, \dots, X_n \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . Thus  $\{\{\mathbf{f}\}, \{\mathbf{t}\}\} \subseteq \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_n)$ . Hence,  $\emptyset \in \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_n)$ . Therefore, we have  $\emptyset \in \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \dots, X_n) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}}$  but  $\emptyset \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . By Corollary 4.4.7, since  $\mathbf{M}_{\mathbf{G}_{if}}$  is proper,  $\diamond$  does not admit axiom-expansion in  $\mathbf{G}$ .

For the converse, suppose that  $\diamond$  does not admit axiom-expansion in **G**. By Corollary 4.4.7, there are  $X_1, \ldots, X_n \in \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$  and  $X \in \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \ldots, X_n) \cap \mathcal{U}_{\mathbf{M}_{\mathbf{G}_{if}}}$  such that  $X \notin \mathcal{U}_{\mathbf{M}_{\mathbf{G}}}$ . We must have  $X = \emptyset$ , and hence  $\{\{\mathbf{f}\}, \{\mathbf{t}\}\} \subseteq \diamond_{\mathbf{M}_{\mathbf{G}}}(X_1, \ldots, X_n)$ . Hence  $\diamond_{\mathbf{M}_{\mathbf{G}}}$  is non-deterministic.

# Chapter 5

# Quasi-canonical Calculi

In the previous chapter we studied the family of *canonical calculi*, and showed how to semantically characterize each logic induced by a canonical calculus using a PNmatrix, and use this PNmatrix to give a decision procedure for the logic. However, a variety of important logics have pure sequent calculi that are *not* canonical, but still have somewhat similar nature. In particular, this is true for the family of *quasi-canonical systems*, introduced in [13]. These are propositional fully-structural two-sided systems, which in addition to the usual weakening rules, cut and identity axiom, include also logical rules with the following properties (the language is assumed to have a unary connective  $\neg$ ):

- 1. Exactly one formula is introduced in the conclusion of the rule, on exactly one of its two sides.
- 2. The formula being introduced is either of the form  $\diamond(p_1, \dots, p_{ar(\diamond)})$  or of the form  $\neg \diamond (p_1, \dots, p_{ar(\diamond)})$ .
- 3. All formulas in the premises of a rule introducing a connective  $\diamond$  belong to the set  $\{p_1, \dots, p_{ar(\diamond)}, \neg p_1, \dots, \neg p_{ar(\diamond)}\}.$
- 4. There are no restrictions on the side formulas in the rule application (i.e. the rules are *pure*).

Of course, rules of this kind are not canonical in the sense of Definition 4.1.1, due to the following two "violations": (i) the introduced formula can be not only  $\diamond(p_1, \ldots, p_{ar(\diamond)})$ , but also  $\neg \diamond (p_1, \ldots, p_{ar(\diamond)})$ , and (ii) the premises may contain not only atomic formulas  $p_i$ , but also  $\neg p_i$ . Hence the results obtained in Chapter 4 for canonical calculi do not directly apply.

In this chapter we show that the theory of canonical calculi can still be exploited for quasi-canonical systems by translating them into equivalent (in the sense defined below) canonical calculi. In fact, this is possible for a substantially larger family of many-sided calculi of which the quasi-canonical systems of [13] are particular examples. In particular,

in the calculi studied in this chapter  $\neg$  does not have any special status, and all unary connectives of a given language may occur in the premises or precede the main connective in the conclusions of the logical rules. Note that various calculi for important many-valued logics (e.g. for the propositional part of the relevance logic of first degree entailment [1]), as well as the (cut-free) calculi proposed in [13] and [8] for many paraconsistent logics (particularly, for C-systems [40]), fall in the family of quasi-canonical calculi studied in this chapter.

#### Publications Related to this Chapter

The material in this chapter was included in [29]. However, [29] concerned only two-sided quasi-canonical calculi with the usual cut and identity axiom, while here we naturally consider quasi-canonical calculi employing any finite set of labels and primitive rules.

## 5.1 Quasi-canonical Calculi

As noted above, the language of quasi-canonical systems is assumed in [13] to include a unary connective  $\neg$ . This restriction can be lifted by allowing any unary connective (possibly in addition to  $\neg$ ). Similarly, we shall not restrict ourselves to two-sided sequents, and continue working in the full framework of pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculi. The notion of a quasi-canonical rule can then be formalized in our terms as follows:

**Definition 5.1.1.** A pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule r is called quasi-canonical if its conclusion has the form  $\{X: \diamond (p_1, \ldots, p_{ar(\diamond)})\}$  or  $\{X: \star \diamond (p_1, \ldots, p_{ar(\diamond)})\}$  for some non-empty  $X \subseteq \mathcal{L}, \diamond \in \diamondsuit_{\mathcal{L}}$ , and  $\star \in \diamondsuit_{\mathcal{L}}^1$ , and its premises are all  $\langle \{p_1, \ldots, p_{ar(\diamond)}\} \cup \{\star p_i \mid \star \in \diamondsuit_{\mathcal{L}}^1, 1 \leq i \leq ar(\diamond)\}, \mathcal{L} \rangle$ -sequents. Quasi-canonical pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules will be also called  $quasi-canonical \langle \mathcal{L}, \mathcal{L} \rangle$ -rules.

Note that this definition is more liberal then Definition 4.1.1, and every canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule is also a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule.

**Example 5.1.2.** Various sequent calculi for paraconsistent logics and relevance logics employ (some of) the following quasi-canonical  $\langle \mathcal{L}_{cl}, \mathcal{L}_{\mathbf{2}} \rangle$ -rules (see [8]):

```
\begin{array}{lll} (\mathbf{f}:\neg\neg) & \{\mathbf{f}:p_1\}/\{\mathbf{f}:\neg\neg p_1\} & (\mathbf{t}:\neg\neg) & \{\mathbf{t}:p_1\}/\{\mathbf{t}:\neg\neg p_1\} \\ (\mathbf{f}:\neg\wedge) & \{\mathbf{f}:\neg p_1\}, \{\mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\wedge p_2)\} & (\mathbf{t}:\neg\wedge) & \{\mathbf{f}:\neg p_1, \mathbf{f}:\neg p_2\}/\{\mathbf{t}:\neg(p_1\wedge p_2)\} \\ (\mathbf{f}:\neg\vee) & \{\mathbf{f}:\neg p_1, \mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\vee p_2)\} & (\mathbf{t}:\neg\vee) & \{\mathbf{t}:\neg p_1\}, \{\mathbf{t}:\neg p_2\}/\{\mathbf{t}:\neg(p_1\vee p_2)\} \\ (\mathbf{f}:\neg\supset) & \{\mathbf{f}:p_1, \mathbf{f}:\neg p_2\}/\{\mathbf{f}:\neg(p_1\supset p_2)\} & (\mathbf{t}:\neg\supset) & \{\mathbf{t}:p_1\}, \{\mathbf{t}:\neg p_2\}/\{\mathbf{t}:\neg(p_1\supset p_2)\} \\ \end{array} Note that non of these rules is a canonical \langle \mathcal{L}_{cl}, \mathcal{L}_{\mathbf{2}} \rangle-rule.
```

**Example 5.1.3.** All (non-primitive) rules of  $G_{C_1}$  (see Example 2.2.22) except for the rule  $\{t:p_1\}, \{t:\neg p_1\}/\{f:\neg(p_1 \wedge \neg p_1)\}$  are quasi-canonical.

In turn, quasi-canonical calculi are defined as follows:

**Definition 5.1.4.** A quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus **G** is a pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, such that  $R_{\mathbf{G}}$  (the set of non-primitive rules of **G**) consists only of quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rules.

Various well-known logics are induced by quasi-canonical calculi. This includes many important three and four-valued logics (e.g. for the relevance logic of first degree entailment [1]), a large family of paraconsistent logics known as C-systems ([40]), for which cut-free quasi-canonical systems were proposed in [13], and various other paraconsistent extensions of positive classical logic studied in [8]. Note that the quasi-canonical systems of [13] correspond to quasi-canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculi with  $\neg \in \diamondsuit^1_{\mathcal{L}}$ ,  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}, \neg$  is the only connective that may appear in the premises of non-primitive rules, and the conclusions of non-primitive rules have the form  $\{\mathbf{x}: \diamondsuit(p_1, \dots, p_{ar(\diamondsuit)})\}$  or  $\{\mathbf{x}: \neg \diamondsuit(p_1, \dots, p_{ar(\diamondsuit)})\}$  for  $\mathbf{x} \in \mathcal{L}_2$  and  $\diamondsuit \in \diamondsuit_{\mathcal{L}}$ .

# 5.2 From Quasi-canonical to Canonical Calculi

We provide a translation of a given quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$  into a canonical one  $T(\mathbf{G})$ , which is equivalent to  $\mathbf{G}$  in a sense defined below. The idea is to "encode" the information related to the connectives from  $\diamondsuit^1_{\mathcal{L}}$  in the labels employed in  $\mathbf{G}$ , so that connectives from  $\diamondsuit^1_{\mathcal{L}}$  "violating" canonicity are removed. To this end, we use  $\mathcal{L} \cup \{\mathbf{x}^* \mid \mathbf{x} \in \mathcal{L}, \star \in \diamondsuit^1_{\mathcal{L}}\}$  as the set of labels for  $T(\mathbf{G})$ . We denote this set by  $\mathcal{L}^{\diamondsuit^1_{\mathcal{L}}}$ . Note that this may be seen as a generalization of the original idea behind two-sided sequents in classical logic. Indeed, often one-sided sequents are translated to two-sided ones by differentiating the negated formulas from the non-negated ones using two different labels (sides).

**Definition 5.2.1.** For an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\alpha$ ,  $T(\alpha)$  is the  $\mathcal{L}^{\diamondsuit^1_{\mathcal{L}}}$ -labelled  $\mathcal{L}$ -formula, defined as follows:  $T(\alpha) = \mathbf{x}^* : \varphi$  if  $\alpha = \mathbf{x} : \star \varphi$  for some  $\mathbf{x} \in \mathcal{L}$ ,  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\varphi \in \mathcal{L}$ , and otherwise  $T(\alpha) = \alpha$ . T is extended to  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and sets of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents in the obvious way (e.g.  $T(s) = \{T(\alpha) \mid \alpha \in s\}$ ).

**Example 5.2.2.** Suppose that  $\neg \in \diamondsuit^1_{\mathcal{L}}$  and  $\pounds = \pounds_2$ . Then:

$$T(\{\mathtt{f}: \neg\neg p_1, \mathtt{t}: \neg p_1, \mathtt{t}: p_2\}) = \{\mathtt{f}^\neg: \neg p_1, \mathtt{t}^\neg: p_1, \mathtt{t}: p_2\}.$$

The translation of a quasi-canonical calculus into a canonical one is given by:

**Notation 5.2.3.** Given  $X \subseteq \mathcal{L}$  and  $\star \in \diamondsuit^1_{\mathcal{L}}$ ,  $X^*$  denotes the subset  $\{x^* \mid x \in X\}$  of  $\mathcal{L}^{\diamondsuit^1_{\mathcal{L}}}$ .

**Definition 5.2.4.** Let **G** be a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus.  $T(\mathbf{G})$  is the canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus that consists of the following rules (except for the weakening rules):

- $T(\mathbf{G})$  includes all primitive  $\mathcal{L}$ -rule in  $\mathsf{P}_{\mathbf{G}}$ , and, in addition, for every primitive  $\mathcal{L}$ rule in  $\mathsf{P}_{\mathbf{G}}$ ,  $r = (\mathsf{X}_1:p_1), \ldots, (\mathsf{X}_n:p_1)/(\mathsf{X}:p_1)$ , and  $\star \in \diamondsuit^1_{\mathcal{L}}$ ,  $T(\mathbf{G})$  has the primitive  $\mathcal{L}^{\diamondsuit^1_{\mathcal{L}}}$ -rule  $(\mathsf{X}_1^{\star}:p_1), \ldots, (\mathsf{X}_n^{\star}:p_1)/(\mathsf{X}^{\star}:p_1)$ , denoted below by  $r^{\star}$ .
- For every quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule  $r = \mathcal{S}/s$  of  $\mathbf{G}$ ,  $T(\mathbf{G})$  includes the  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -rule  $T(\mathcal{S})/T(s)$ , denoted by T(r).
- For every  $\star \in \diamondsuit^1_{\mathcal{L}}$ ,  $T(\mathbf{G})$  includes the canonical  $\langle \mathcal{L}, \pounds^{\diamondsuit^1_{\mathcal{L}}} \rangle$ -rule  $\{\mathbf{x}^{\star}: p_1\}/\{\mathbf{x}: \star p_1\}$ , denoted below by  $(\mathbf{x}^{\star} \to \mathbf{x})$ .

**Example 5.2.5.** Let  $\mathcal{L} = \{\neg^1, \wedge^2\}$ . The system  $PLK[\{(\neg \wedge \Rightarrow)\}]$  from [8] is practically a quasi-canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus. Besides the weakening rules, (cut), and (id) this calculus includes the following quasi-canonical rules:

$$\begin{array}{lll} (\mathtt{f} : \land) & \{\mathtt{f} : p_1, \mathtt{f} : p_2\} / \{\mathtt{f} : p_1 \land p_2\} & (\mathtt{t} : \land) & \{\mathtt{t} : p_1\}, \{\mathtt{t} : p_2\} / \{\mathtt{t} : p_1 \land p_2\} \\ (\mathtt{t} : \neg) & \{\mathtt{f} : p_1\} / \{\mathtt{t} : \neg p_1\} & (\mathtt{f} : \neg \land) & \{\mathtt{f} : \neg p_1\}, \{\mathtt{f} : \neg p_2\} / \{\mathtt{f} : \neg (p_1 \land p_2)\} \end{array}$$

We denote this system by  $\mathbf{G}_0$ . Now,  $\mathcal{L}_{\mathbf{2}}^{\Diamond_{\mathcal{L}}^1} = \{\mathbf{f}, \mathbf{t}, \mathbf{f}^{\neg}, \mathbf{t}^{\neg}\}$ .  $T(\mathbf{G}_0)$  is the canonical  $\langle \mathcal{L}, \mathcal{L}_{\mathbf{2}}^{\Diamond_{\mathcal{L}}^1} \rangle$ -calculus, with the following rules:

```
\begin{array}{lll} (\texttt{f}:weak) & \{\emptyset\}/\{\texttt{f}:p_1\} & (\texttt{t}:weak) & \{\emptyset\}/\{\texttt{t}:p_1\} \\ (\texttt{f}^{\neg}:weak) & \{\emptyset\}/\{\texttt{f}^{\neg}:p_1\} & (\texttt{t}^{\neg}:weak) & \{\emptyset\}/\{\texttt{t}^{\neg}:p_1\} \\ (cut) & \{\texttt{f}:p_1\}, \{\texttt{t}:p_1\}/\emptyset & (id) & \emptyset/\{\texttt{f}:p_1, \texttt{t}:p_1\} \\ (cut)^{\neg} & \{\texttt{f}^{\neg}:p_1\}, \{\texttt{t}^{\neg}:p_1\}/\emptyset & (id)^{\neg} & \emptyset/\{\texttt{f}^{\neg}:p_1, \texttt{t}^{\neg}:p_1\} \\ T((\texttt{f}:\wedge)) & \{\texttt{f}:p_1, \texttt{f}:p_2\}/\{\texttt{f}:p_1 \wedge p_2\} & T((\texttt{t}:\wedge)) & \{\texttt{t}:p_1\}, \{\texttt{t}:p_2\}/\{\texttt{t}:p_1 \wedge p_2\} \\ T((\texttt{t}:\neg)) & \{\texttt{f}:p_1\}/\{\texttt{f}^{\neg}:p_1\} & T((\texttt{f}:\neg\wedge)) & \{\texttt{f}^{\neg}:p_1\}, \{\texttt{f}^{\neg}:p_2\}/\{\texttt{f}^{\neg}:p_1 \wedge p_2\} \\ (\texttt{f}^{\neg}\to\texttt{f}) & \{\texttt{f}^{\neg}:p_1\}/\{\texttt{f}:\neg p_1\} & (\texttt{t}^{\neg}\to\texttt{t}) & \{\texttt{t}^{\neg}:p_1\}/\{\texttt{t}:\neg p_1\} \end{array}
```

It is easy to see that all rules in  $T(\mathbf{G})$  are primitive  $\mathcal{L}^{\diamondsuit_{\mathcal{L}}^1}$ -rules or canonical  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -rules, and thus  $T(\mathbf{G})$  is a canonical  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -calculus. In particular, note that for a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -rule r, T(r) as defined above, can either be a canonical  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -rule (as  $T((\mathbf{f}:\neg \wedge))$  above) or a primitive  $\mathcal{L}^{\diamondsuit_{\mathcal{L}}^1}$ -rule (as  $T((\mathbf{t}:\neg))$  above).

To prove the equivalence between G and T(G) we use the following lemmas:

**Lemma 5.2.6.** Let  $\mathbf{x}:\varphi$  be an  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula, and  $\sigma$  an  $\mathcal{L}$ -substitution, such that  $\varphi \notin at_{\mathcal{L}}$  or  $\sigma(\varphi)$  is not of the form  $\star \psi$  for  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ . Then,  $T(\sigma(\mathbf{x}:\varphi)) = \sigma(T(\mathbf{x}:\varphi))$ .

*Proof.* If 
$$\varphi = \star \psi$$
 for some  $\star \in \Diamond_{\mathcal{L}}^1$  and  $\psi \in \mathcal{L}$ , then:

$$T(\sigma(\mathbf{x}:\varphi)) = T(\mathbf{x}: \star \sigma(\psi)) = \mathbf{x}^{\star}: \sigma(\psi) = \sigma(\mathbf{x}^{\star}: \psi) = \sigma(T(\mathbf{x}:\varphi)).$$

Otherwise,  $\varphi$  is either atomic or  $\varphi = \diamond(p_1, \dots, p_{ar(\diamond)})$  for  $ar(\diamond) \neq 1$ , and in both cases  $\sigma(\varphi)$  is not of the form  $\star \psi$  for  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ . Then:

$$T(\sigma(\mathbf{x}:\varphi)) = T(\mathbf{x}:\sigma(\varphi)) = \mathbf{x}:\sigma(\varphi) = \sigma(\mathbf{x}:\varphi) = \sigma(T(\mathbf{x}:\varphi)).$$

**Lemma 5.2.7.** Let **G** be a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus, and  $\sigma$  an  $\mathcal{L}$ -substitution. For every  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula  $\alpha$ :  $\{T(\sigma(\alpha))\} \vdash_{T(\mathbf{G})} \{\sigma(T(\alpha))\}$ .

Proof. Let  $\alpha = \mathbf{x}: \varphi$  (for  $\mathbf{x} \in \mathcal{L}$  and  $\varphi \in \mathcal{L}$ ). If  $\varphi \notin at_{\mathcal{L}}$  or  $\sigma(\varphi)$  is not of the form  $\star \psi$  for  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ , then by Lemma 5.2.6,  $T(\sigma(\alpha)) = \sigma(T(\alpha))$ , and obviously,  $\{T(\sigma(\alpha))\} \vdash_{T(\mathbf{G})} \{\sigma(T(\alpha))\}$ . Otherwise,  $\varphi \in at_{\mathcal{L}}$  and  $\sigma(\varphi) = \star \psi$  for some  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ . In this case,  $T(\sigma(\alpha)) = T(\mathbf{x}: \star \psi) = \mathbf{x}^*: \psi$ , and  $\sigma(T(\alpha)) = \sigma(\alpha) = \mathbf{x}: \star \psi$ . By applying the rule  $(\mathbf{x}^* \to \mathbf{x})$ , we obtain that  $\{T(\sigma(\alpha))\} \vdash_{T(\mathbf{G})} \{\sigma(T(\alpha))\}$ .

**Proposition 5.2.8.** For every quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s: if  $\mathcal{S} \vdash_{\mathbf{G}} s$  then  $T(\mathcal{S}) \vdash_{T(\mathbf{G})} T(s)$ .

*Proof.* It suffices to show that for every application of a rule of **G** deriving the  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s from the  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents  $s_1, \ldots, s_n$ , we can derive T(s) from  $T(s_1), \ldots, T(s_n)$  in  $T(\mathbf{G})$ . Consider the possible cases:

- $s = \{\mathbf{x}: \varphi\} \cup c$  is derived from c by applying a weakening rule. In this case, we can derive  $T(s) = T(\{\mathbf{x}: \varphi\}) \cup T(c)$  from T(c) by applying weakening as well.
- $s = (X:\varphi) \cup c_1 \cup ... \cup c_n$  is derived from  $s_1 = (X_1:\varphi) \cup c_1, ..., s_n = (X_n:\varphi) \cup c_n$  by applying of a primitive  $\mathcal{L}$ -rule  $r = (X_1:p_1), ..., (X_n:p_1)/(X:p_1)$  of  $\mathbf{G}$ . If  $\varphi$  does not have the form  $\star \psi$  (for  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ ), then  $T(s_i) = \{X_i:\varphi\} \cup T(c_i)$  for every  $1 \leq i \leq n$ , and  $T(s) = \{X:\varphi\} \cup T(c_1) \cup ... \cup T(c_n)$ . By applying the same primitive  $\mathcal{L}$ -rule r, we can derive T(s) from  $T(s_1), ..., T(s_n)$  in  $T(\mathbf{G})$ . Otherwise,  $\varphi = \star \psi$  (for  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ ). Here,  $T(s_i) = (X_i^*:\psi) \cup T(c_i)$  for every  $1 \leq i \leq n$ , and  $T(s) = (X^*:\psi) \cup T(c_1) \cup ... \cup T(c_n)$ . By applying  $r^*$ , we can derive T(s) from  $T(s_1), ..., T(s_n)$  in  $T(\mathbf{G})$ .
- $s = \sigma(s') \cup c_1 \cup ... \cup c_n$  is derived from  $s_1 = \sigma(s'_1) \cup c_1, ..., s_n = \sigma(s'_n) \cup c_n$  by applying a quasi-canonical rule  $r = s'_1, ..., s'_n/s'$  of  $\mathbf{G}$ . For every  $1 \leq i \leq n$ ,  $T(s_i) = T(\sigma(s'_i)) \cup T(c_i)$ , and thus by Lemma 5.2.7 (using also Proposition 2.2.17),  $T(s_i) \vdash_{T(\mathbf{G})} \sigma(T(s'_i)) \cup T(c_i)$ . By applying the rule T(r) of  $T(\mathbf{G})$  we can derive  $\sigma(T(s')) \cup T(c_1) \cup ... \cup T(c_n)$ . Since r is quasi-canonical, s' consists solely of non-atomic formulas, and by Lemma 5.2.6,  $\sigma(T(s')) = T(\sigma(s'))$ . Thus, we derived T(s) in  $T(\mathbf{G})$ .

For the other direction, we define a translation  $T^{-1}$ , mapping  $\langle \mathcal{L}, \mathcal{L}^{\Diamond_{\mathcal{L}}^{1}} \rangle$ -sequents to  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents:

**Definition 5.2.9.** For an  $\mathcal{L}^{\diamondsuit_{\mathcal{L}}^1}$ -labelled  $\mathcal{L}$ -formula  $\alpha$ ,  $T^{-1}(\alpha)$  is the  $\mathcal{L}$ -labelled  $\mathcal{L}$ -formula, defined as follows:  $T^{-1}(\alpha) = \mathbf{x} : \star \varphi$  if  $\alpha = \mathbf{x}^* : \varphi$  for some  $\mathbf{x} \in \mathcal{L}$ ,  $\star \in \diamondsuit_{\mathcal{L}}^1$  and  $\varphi \in \mathcal{L}$ , and otherwise  $T^{-1}(\alpha) = \alpha$ .  $T^{-1}$  is extended to  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequents and sets of  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequents in the obvious way (e.g.  $T^{-1}(s) = \{T^{-1}(\alpha) \mid \alpha \in s\}$ ).

**Lemma 5.2.10.**  $T^{-1}(\sigma(T(s))) = \sigma(s)$  for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, and  $\mathcal{L}$ -substitution  $\sigma$ .

**Proposition 5.2.11.** For every quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequents and  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequent s: if  $\mathcal{S} \vdash_{T(\mathbf{G})} s$  then  $T^{-1}(\mathcal{S}) \vdash_{\mathbf{G}} T^{-1}(s)$ .

*Proof.* It suffices to show that for every application of a rule of  $T(\mathbf{G})$  deriving the  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequent s from the  $\langle \mathcal{L}, \mathcal{L}^{\diamondsuit_{\mathcal{L}}^1} \rangle$ -sequents  $s_1, \dots, s_n$ , we can derive  $T^{-1}(s)$  from  $T^{-1}(s_1), \dots, T^{-1}(s_n)$  in  $\mathbf{G}$ . Consider the possible cases:

- $s = \{\mathbf{x}:\varphi\} \cup c$  is derived from c by applying a weakening rule. In this case, we can derive  $T^{-1}(s) = T^{-1}(\{\mathbf{x}:\varphi\}) \cup T^{-1}(c)$  from  $T^{-1}(c)$  by applying weakening as well.
- $s = (X:\varphi) \cup c_1 \cup ... \cup c_n$  is derived from  $s_1 = (X_1:\varphi) \cup c_1, ..., s_n = (X_n:\varphi) \cup c_n$  by applying the primitive  $\mathcal{L}$ -rule  $(X_1:p_1), ..., (X_n:p_1)/(X:p_1)$  of  $T(\mathbf{G})$ , that occurs also in  $\mathbf{G}$  itself. For every  $1 \leq i \leq n$ ,  $T^{-1}(s_i) = (X_i:\varphi) \cup T^{-1}(c_i)$ . By applying the same rule we can derive  $T^{-1}(s) = (X:\varphi) \cup T^{-1}(c_1) \cup ... \cup T^{-1}(c_n)$  in  $\mathbf{G}$ .
- $s = (X^*:\varphi) \cup c_1 \cup ... \cup c_n$  is derived from  $s_1 = (X_1^*:\varphi) \cup c_1, ..., s_n = (X_n^*:\varphi) \cup c_n$  by applying the primitive  $\mathcal{L}^{\diamondsuit_{\mathcal{L}}^1}$ -rule  $r^* = (X_1^*:p_1), ..., (X_n^*:p_1)/(X^*:p_1)$  of  $T(\mathbf{G})$ . In this case  $r = (X_1:p_1), ..., (X_n:p_1)/(X:p_1)$  is a primitive rule of  $\mathbf{G}$ . For every  $1 \leq i \leq n$ ,  $T^{-1}(s_i) = (X_i: \star \varphi) \cup T^{-1}(c_i)$ . By applying the rule r we can derive the  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent  $T^{-1}(s) = (X: \star \varphi) \cup T^{-1}(c_1) \cup ... \cup T^{-1}(c_n)$  in  $\mathbf{G}$ .
- $s = \sigma(T(s')) \cup c_1 \cup ... \cup c_n$  is derived from  $s_1 = \sigma(T(s'_1)) \cup c_1, ..., s_n = \sigma(T(s'_n)) \cup c_n$ by applying a rule T(r) of  $T(\mathbf{G})$ , where  $r = s'_1, ..., s'_n/s'$  is a rule of  $\mathbf{G}$ . By Lemma 5.2.10, for every  $1 \leq i \leq n$ ,  $T^{-1}(s_i) = \sigma(s'_1) \cup T^{-1}(c_i)$ . By applying rwe can derive  $T^{-1}(s) = \sigma(s') \cup T^{-1}(c_1) \cup ... \cup T^{-1}(c_n)$  in  $\mathbf{G}$ .
- $s = \{\mathbf{x}: \star \varphi\} \cup c$  is derived from  $s_1 = \{\mathbf{x}^*: \varphi\} \cup c$  using the rule  $(\mathbf{x}^* \to \mathbf{x})$ . In this case we have  $T^{-1}(s) = T^{-1}(s_1)$ , and obviously we are done.

It follows that  $T(\mathbf{G})$  is equivalent to the original quasi-canonical calculus  $\mathbf{G}$  in the following sense:

**Theorem 5.2.12.** For every quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents and  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s:  $\mathcal{S} \vdash_{\mathbf{G}} s$  iff  $T(\mathcal{S}) \vdash_{T(\mathbf{G})} T(s)$ .

*Proof.* By Lemma 5.2.10  $T^{-1}(T(s)) = s$  for every  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s (take the "identity substitution"). Thus the claim directly follows from Propositions 5.2.8 and 5.2.11.

A general decidability result for all quasi-canonical calculi immediately follows:

- Corollary 5.2.13. 1. Given a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , a finite set  $\mathcal{S}$  of  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequents, and an  $\langle \mathcal{L}, \mathcal{L} \rangle$ -sequent s, it is decidable whether  $\mathcal{S} \vdash_{\mathbf{G}} s$  or not.
  - 2. Given a quasi-canonical  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus  $\mathbf{G}$ , a set  $\mathbf{X} \subseteq \mathcal{L}$ , a finite set  $\Gamma$  of  $\mathcal{L}$ -formulas, and an  $\mathcal{L}$ -formula  $\varphi$ , it is decidable whether  $\Gamma \Vdash_{\mathbf{G}}^{\mathbf{X}} \varphi$  or not.

*Proof.* Directly follows from Corollary 4.3.8 and Theorem 5.2.12. Note that the construction of  $T(\mathbf{G})$  from  $\mathbf{G}$  is obviously computable.

In addition to decidability, the results of the previous chapters provide a method to obtain a (pre-) PNmatrix semantics for  $T(\mathbf{G})$  and the logics it induces. Next we show that in the most common and interesting case, dealing with a quasi-canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus  $\mathbf{G}$  with  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ , we are also able to use the semantic framework of PNmatrices to characterize  $\mathbf{G}$  itself and the logics it induces.

**Proposition 5.2.14.** Let **G** be a quasi-canonical  $\langle \mathcal{L}, \pounds_{\mathbf{2}} \rangle$ -calculus, and v an  $\mathbf{M}_{T(\mathbf{G})}$ -legal  $\mathcal{L}$ -valuation. Suppose that  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ . Then, for every  $\langle \mathcal{L}, \pounds_{\mathbf{2}} \rangle$ -sequent  $s: v \models s$  iff  $v \models T(s)$ .

Proof. Suppose that  $v \models T(s)$ . By definition, there exists some  $\pounds_{\mathbf{2}}^{\diamondsuit_{\mathcal{L}}^1}$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x} : \varphi \in T(s)$  such that  $\mathbf{x} \in v(\varphi)$ . If  $\mathbf{x} \in \pounds_{\mathbf{2}}$  then  $\mathbf{x} : \varphi \in s$  as well, and clearly  $v \models s$ . Otherwise,  $\mathbf{x} = \mathbf{y}^*$  for some  $\mathbf{y} \in \pounds_{\mathbf{2}}$  and  $\mathbf{x} \in \diamondsuit_{\mathcal{L}}^1$ , and we have  $\mathbf{y} : \mathbf{x} \varphi \in s$ . Now, since v is  $\mathbf{M}_{T(\mathbf{G})}$ -legal and  $T(\mathbf{G})$  includes the rule  $(\mathbf{y}^* \to \mathbf{y})$ , we should have  $\mathbf{y} \in v(\mathbf{x}\varphi)$ , and consequently,  $v \models s$ . To see this, note that since  $T(\mathbf{G})$  includes the rule  $(\mathbf{y}^* \to \mathbf{y})$ , we have that  $\mathbf{Sem}((\mathbf{y}^* \to \mathbf{y})) \in \Lambda_{\mathbf{M}_{T(\mathbf{G})}}$ . Since v is  $\mathbf{M}_{T(\mathbf{G})}$ -legal, it satisfies  $\mathbf{Sem}((\mathbf{y}^* \to \mathbf{y}))$ . Now,  $\mathbf{Sem}((\mathbf{y}^* \to \mathbf{y})) = \{\{p_1 \doteqdot \mathbf{X} \mid \{\mathbf{y}^*\} \subseteq \mathbf{X} \subseteq \pounds\}\} \Rightarrow \{\mathbf{x}p_1 \doteqdot \mathbf{X} \mid \{\mathbf{y}\} \subseteq \mathbf{X} \subseteq \pounds\}$ . Consequently, since  $\mathbf{y}^* \in v(\varphi)$ , we have  $\mathbf{y} \in v(\mathbf{x}\varphi)$ .

For the converse, suppose that  $v \models s$ . By definition, there exists some  $\mathcal{L}_2$ -labelled  $\mathcal{L}$ -formula  $\mathbf{x}:\varphi \in s$  such that  $\mathbf{x} \in v(\varphi)$ . If  $\varphi$  does not have a form  $\star \psi$  for  $\star \in \diamondsuit^1_{\mathcal{L}}$ , then  $\mathbf{x}:\varphi \in T(s)$ , and clearly  $v \models T(s)$ . Otherwise,  $\varphi = \star \psi$  for some  $\star \in \diamondsuit^1_{\mathcal{L}}$  and  $\psi \in \mathcal{L}$ , and  $\mathbf{x}^*:\psi \in T(s)$ . We show that we have  $\mathbf{x}^* \in v(\psi)$  in this case (and so,  $v \models T(s)$ ). First, since v is  $\mathbf{M}_{T(\mathbf{G})}$ -legal and  $(cut) \in \mathsf{P}_{T(\mathbf{G})}$ , we have  $\overline{\mathbf{x}} \notin v(\varphi)$  (where  $\overline{\mathbf{f}} = \mathbf{t}$  and  $\overline{\mathbf{t}} = \mathbf{f}$ , see Example 3.2.9). Since  $T(\mathbf{G})$  includes the rule  $(\overline{\mathbf{x}}^* \to \overline{\mathbf{x}})$ , this entails that  $\overline{\mathbf{x}}^* \notin v(\psi)$  (this is proved as in the first direction above). Finally, since  $(id)^* \in \mathsf{P}_{T(\mathbf{G})}$ , it follows that  $\mathbf{x}^* \in v(\psi)$ .

Corollary 5.2.15. Let **G** be a quasi-canonical  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculus with  $\mathsf{P}_{\mathbf{G}} = \{(cut), (id)\}$ . Then,  $\vdash_{\mathbf{G}} = \vdash_{\mathbf{M}_{T(\mathbf{G})}} \text{and } \Vdash_{\mathbf{G}}^{\mathtt{X}} = \Vdash_{\mathbf{M}_{T(\mathbf{G})}^{\mathtt{X}}} \text{ for every } \mathtt{X} \subseteq \mathcal{L}_2$ .

*Proof.* Easily follows from Proposition 5.2.14, Theorem 5.2.12, and Theorem 3.2.15.  $\Box$ 

**Example 5.2.16.** Consider the logic induced by the quasi-canonical calculus  $\mathbf{G}_0$  from Example 5.2.5 and the set of labels  $\{\mathbf{t}\}$ . The corresponding PNmatrix  $\mathbf{M} = \mathbf{M}_{T(\mathbf{G}_0)}^{\{\mathbf{t}\}}$ , which is obtained from the canonical calculus  $T(\mathbf{G}_0)$  constructed in that example, consists of:<sup>1</sup>

- 1.  $\mathcal{U}_{\mathbf{M}} = \{\{\mathsf{t},\mathsf{t}^{\neg}\},\{\mathsf{t},\mathsf{f}^{\neg}\},\{\mathsf{f},\mathsf{t}^{\neg}\}\}.$
- 2.  $\mathcal{D}_{\mathbf{M}} = \{ \{ \mathbf{t}, \mathbf{t}^{\neg} \}, \{ \mathbf{t}, \mathbf{f}^{\neg} \} \}.$
- 3.  $\wedge$  and  $\neg$  have the following PNtables:

Corollary 5.2.15 entails that  $\mathbf{M}$  characterizes the ter  $\Vdash^{\{\mathbf{t}\}}_{\mathbf{G}_0}$  (that is:  $\mathcal{T} \Vdash_{\mathbf{M}} \varphi$  iff  $\mathcal{T} \Vdash^{\{\mathbf{t}\}}_{\mathbf{G}_0} \varphi$ ). Note that  $\mathbf{M}$  is isomorphic to the three-valued Nmatrix for this logic given in [8]. Furthermore, it is easy to check that this is the case for *all* the PNmatrices obtained by this general procedure for the quasi-canonical calculi considered in [8, 13].

 $<sup>^{1}</sup>$ For simplicity, we use a reduced presentation of **M** as explained in Remark 3.1.12.

# Chapter 6

# Non-pure Sequent Calculi

For various important non-classical logics, such as modal logics and intuitionistic logic, there is no known (cut-free) pure calculus. Indeed, a major restriction in pure calculi is that unlimited context sequents may be freely used in all inference steps. Well-known sequent calculi for modal logics and intuitionistic logic do not meet this requirement, and thus they do not belong to the family of pure calculi studied in the previous chapters. For example, consider the following schemes of applications written in the usual notation of two-sided sequents:

$$(1) \frac{\Gamma, \varphi_1 \Rightarrow \varphi_2}{\Gamma \Rightarrow \varphi_1 \supset \varphi_2} \qquad (2) \frac{\Box \Gamma \Rightarrow \varphi}{\Box \Gamma \Rightarrow \Box \varphi} \qquad (3) \frac{\Gamma \Rightarrow \varphi}{\Box \Gamma \Rightarrow \Box \varphi}$$

These schemes demonstrate different possibilities regarding context sequents, and non of them can be presented as a pure rule:

- 1. Scheme (1) allows only left context formulas, that is: all context sequents should be subsets of  $\{f:\varphi \mid \varphi \in \mathcal{L}\}$ . This scheme is employed in the multiple-conclusion sequent calculus for intuitionistic logic [90].
- 2. Scheme (2) again allows only left context formulas, but all of them should begin with  $\Box$  ( $\Box\Gamma$  is an abbreviation for { $\Box\varphi \mid \varphi \in \Gamma$ }). In other words, all context sequents should be subsets of { $\mathbf{f}$ : $\Box\varphi \mid \varphi \in \mathcal{L}$ }. This scheme is employed in the usual sequent calculus for the modal logic S4 [96].
- 3. Scheme (3) exhibits more complicated treatment of the context formulas: each labelled formula  $\mathbf{f}:\varphi$  in the premise "becomes"  $\mathbf{f}:\Box\varphi$  in the conclusion of the application. In other words, any sequent  $c \subseteq \{\mathbf{f}:\varphi \mid \varphi \in \mathcal{L}\}$  can serve as a context sequent in the premise of the application, provided that  $\{\mathbf{f}:\Box\varphi \mid \mathbf{f}:\varphi \in c\}$  is the context sequent of the conclusion. This scheme is employed in the usual sequent

6.1. Basic Calculi 73

calculus for the modal logic K.<sup>1</sup>

In this chapter we introduce a general framework of sequent calculi, called basic calculi, that allow context restrictions of certain kinds (including those demonstrated above). Unlike in the previous chapters, we restrict our attention only to two-sided sequent calculi. Various sequent calculi that seem to have completely different natures can be directly presented as basic calculi. This includes standard sequent calculi for modal logics, as well as the usual multiple-conclusion systems for intuitionistic logic, its dual, and bi-intuitionistic logic. Our goal is to carry out a general and uniform semantic study of these systems, that will provide useful semantics for them, as well as semantic criteria for their proof-theoretic properties.

#### Publications Related to this Chapter

The material in this chapter was included in [15, 73].

### 6.1 Basic Calculi

In this section we precisely define the general structure of derivation rules that are allowed to appear in basic calculi. Rules of this structure will be called basic rules. As in Section 2.2, we explicitly differentiate between a rule and its applications. Derivations in a certain basic calculus consist of applications of rules, and the rules themselves are just succinct formulations of their sets of applications. In addition, for the formulation of the rules, we differentiate between two parts of their applications, namely the context part and the non-context part (see [92]). The non-context part is obtained by instantiating a rigid structure that is given in the rule. In turn, the structure of the context part is determined using context-relations. This structure is less restrictive, as the number of context formulas is completely free. Next we turn to the formal definitions.

Notation 6.1.1. Throughout this chapter, dealing only with two-sided sequents, we will not mention the set of labels in the aforementioned notions. For example, we refer to  $\mathcal{L}_2$ -labelled  $\mathcal{L}$ -formulas and  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -sequents simply as labelled  $\mathcal{L}$ -formulas and  $\mathcal{L}$ -sequents. We may use the usual sequent notation  $\Gamma \Rightarrow \Delta$ , where  $\Gamma$  and  $\Delta$  are (possibly empty) finite sets of formulas, interpreted as  $\{\mathbf{f}:\varphi \mid \varphi \in \Gamma\} \cup \{\mathbf{t}:\varphi \mid \varphi \in \Delta\}$ , and employ the standard abbreviations, e.g.,  $\Gamma, \varphi \Rightarrow \psi$  instead of  $\Gamma \cup \{\varphi\} \Rightarrow \{\psi\}$ , and  $\Gamma \Rightarrow$  instead of  $\Gamma \Rightarrow \emptyset$ .

<sup>&</sup>lt;sup>1</sup>Our intuitive distinction between context formulas and non-context formulas in rule schemes is based on the following principle: the exact number of non-context formulas is explicitly specified in the scheme, while any (finite) number of context formulas may be employed. Taking this into account, note that context formulas may "change" (as in (3)).

Name	Context-Relation	Instances (all pair of the form:)
$\pi_0$	$\{\langle \mathtt{f} : p_1, \mathtt{f} : p_1 \rangle, \langle \mathtt{t} : p_1, \mathtt{t} : p_1 \rangle \}$	$\langle \Gamma \Rightarrow \Delta , \ \Gamma \Rightarrow \Delta \rangle$
$\pi_{int}$	$\{\langle \mathtt{f} {:} p_1, \mathtt{f} {:} p_1 \rangle \}$	$\langle \Gamma \Rightarrow , \Gamma \Rightarrow \rangle$
$\pi_{d}$	$\{\langle t{:}p_1, t{:}p_1\rangle\}$	$\langle \Rightarrow \Delta , \Rightarrow \Delta \rangle$
Ø	Ø	$\langle \Rightarrow , \Rightarrow \rangle$
$\pi_{K}$	$\{\langle \mathtt{f} : p_1, \mathtt{f} : \Box p_1 \rangle \}$	$\langle \Gamma \Rightarrow , \Box \Gamma \Rightarrow \rangle$
$\pi_{K4}$	$\{\langle \mathbf{f} : p_1, \mathbf{f} : \Box p_1 \rangle, \langle \mathbf{f} : \Box p_1, \mathbf{f} : \Box p_1 \rangle\}$	$\langle \Gamma_1, \Box \Gamma_2 \Rightarrow , \ \Box \Gamma_1, \Box \Gamma_2 \Rightarrow \rangle$
$\pi_{S4}$	$\{\langle \mathtt{f} : \Box p_1, \mathtt{f} : \Box p_1 \rangle \}$	$\langle \Box \Gamma \Rightarrow \; , \; \Box \Gamma \Rightarrow \rangle$
$\pi_{B}$	$\{\langle \mathtt{f} : p_1, \mathtt{f} : \Box p_1 \rangle, \langle \mathtt{t} : \Box p_1, \mathtt{t} : p_1 \rangle\}$	$\langle \Gamma \Rightarrow \Box \Delta \; , \; \Box \Gamma \Rightarrow \Delta \rangle$
$\pi_{S5}$	$\{\langle \mathtt{f} : \Box p_1, \mathtt{f} : \Box p_1 \rangle, \langle \mathtt{t} : \Box p_1, \mathtt{t} : \Box p_1 \rangle\}$	$\langle \Box \Gamma \Rightarrow \Box \Delta \; , \; \Box \Gamma \Rightarrow \Box \Delta \rangle$

Table 6.1: Important Context-Relations

**Definition 6.1.2.** An  $\mathcal{L}$ -context-relation is a finite binary relation on the set of labelled  $\mathcal{L}$ -formulas. Given an  $\mathcal{L}$ -context-relation  $\pi$ ,  $\bar{\pi}$  is the binary relation between labelled  $\mathcal{L}$ -formulas consisting of all substitution instances of  $\pi$ , that is:

$$\bar{\pi} = \{ \langle \sigma(\alpha), \sigma(\beta) \rangle \mid \sigma \text{ is an } \mathcal{L}\text{-substitution and } \langle \alpha, \beta \rangle \in \pi \}.$$

A  $\pi$ -instance is an ordered pair of  $\mathcal{L}$ -sequents  $\langle s_1, s_2 \rangle$  for which there exist (not necessarily distinct) labelled  $\mathcal{L}$ -formulas  $\alpha_1, \ldots, \alpha_n$ , and  $\beta_1, \ldots, \beta_n$  such that  $s_1 = \{\alpha_1, \ldots, \alpha_n\}$ ,  $s_2 = \{\beta_1, \ldots, \beta_n\}$ , and  $\alpha_i \bar{\pi} \beta_i$  for every  $1 \leq i \leq n$ .

Several context-relations, that are used in the definitions and examples below, are given in Table 6.1 (in some of them the language is assumed to have a unary connective denoted by  $\square$ ).

**Definition 6.1.3.** A basic  $\mathcal{L}$ -premise is an ordered pair of the form  $\langle s, \pi \rangle$ , where s is an  $\mathcal{L}$ -sequent and  $\pi$  is an  $\mathcal{L}$ -context-relation. A basic  $\mathcal{L}$ -rule is a pair of the form  $\mathcal{P}/s$ , where  $\mathcal{P}$  is a finite set of basic  $\mathcal{L}$ -premise, and s is an  $\mathcal{L}$ -sequent. The elements of  $\mathcal{P}$  are called the *premises* of the rule, and s is called the *conclusion* of the rule. To improve readability, we usually drop the set braces of the set  $\mathcal{P}$  of premises. An application of a basic  $\mathcal{L}$ -rule  $\langle s_1, \pi_1 \rangle, \dots, \langle s_n, \pi_n \rangle/s$  is any inference step of the following form:

$$\frac{\sigma(s_1) \cup c_1 \dots \sigma(s_n) \cup c_n}{\sigma(s) \cup c'_1 \cup \dots \cup c'_n}$$

where  $\sigma$  is an  $\mathcal{L}$ -substitution, and for every  $1 \leq i \leq n$ ,  $c_i$  and  $c_i'$  are  $\mathcal{L}$ -sequents such that  $\langle c_i, c_i' \rangle$  is a  $\pi_i$ -instance. The sequents  $\sigma(s_1) \cup c_1, \dots, \sigma(s_n) \cup c_n$  are called the *premises* of the application, while  $\sigma(s) \cup c_1' \cup \dots \cup c_n'$  is called the *conclusion* of the application.

Table 6.2 provides some examples of basic rules and the forms of their applications. Note that pure  $\langle \mathcal{L}, \pounds_2 \rangle$ -rules as defined in Chapter 2 are basic  $\mathcal{L}$ -rules in which all premises have the form  $\langle s, \pi_0 \rangle$ .

6.1. Basic Calculi 75

Name	Basic Rule	Application
T	$\langle \{\mathtt{f}{:}p_1\},\pi_0 \rangle/\{\mathtt{f}{:}\Box p_1\}$	$\frac{\Gamma, \varphi \Rightarrow \Delta}{\Gamma, \Box \varphi \Rightarrow \Delta}$
(S4)	$\langle \{\mathtt{t}{:}p_1\},\pi_{S4} \rangle/\{\mathtt{t}{:}p_1\}$	$\frac{\Box\Gamma\Rightarrow\varphi}{\Box\Gamma\Rightarrow\Box\varphi}$
(K4)	$\langle \{t{:}p_1\}, \pi_{K4} \rangle / \{t{:}\Box p_1\}$	$\frac{\Gamma_1, \Box \Gamma_2 \Rightarrow \varphi}{\Box \Gamma_1, \Box \Gamma_2 \Rightarrow \Box \varphi}$
$(D_0)$	$\langle \{\mathtt{f} \mathpunct{:} p_1\}, \emptyset \rangle / \{\mathtt{f} \mathpunct{:} \Box p_1\}$	$\frac{\varphi\Rightarrow}{\Box\varphi\Rightarrow}$
(D)	$\langle\emptyset,\pi_{K} angle/\emptyset$	$\begin{array}{c} \Gamma \Rightarrow \\ \hline \Box \Gamma \Rightarrow \end{array}$
(f: ⊃)	$\langle \{\mathtt{t}:p_1\}, \pi_0 \rangle, \langle \{\mathtt{f}:p_2\}, \pi_0 \rangle / \{\mathtt{f}:p_1 \supset p_2 \}$	$\frac{\Gamma_1 \Rightarrow \varphi_1, \Delta_1  \Gamma_2, \varphi_2 \Rightarrow \Delta_2}{\Gamma_1, \Gamma_2, \varphi_1 \supset \varphi_2 \Rightarrow \Delta_1, \Delta_2}$
(f: ⊃′)	$\langle \{\mathtt{t}:p_1\}, \pi_0 \rangle, \langle \{\mathtt{f}:p_2\}, \pi_{int} \rangle / \{\mathtt{f}:p_1 \supset p_2 \}$	$\frac{\Gamma_1 \Rightarrow \varphi_1, \Delta  \Gamma_2, \varphi_2 \Rightarrow}{\Gamma_1, \Gamma_2, \varphi_1 \supset \varphi_2 \Rightarrow \Delta}$
(t: ⊃*)	$\langle \{\mathtt{f} : p_1, \mathtt{t} : p_2\}, \pi \rangle / \{\mathtt{t} : p_1 \supset p_2\}$	$\psi_1 \supset \psi_1', \dots, \psi_n \supset \psi_n', \varphi_1 \Rightarrow \varphi_2$
	where $\pi = \{\langle \mathbf{f}: p_1 \supset p_2, \mathbf{f}: p_1 \supset p_2 \rangle\}$	$\psi_1 \supset \psi_1', \dots, \psi_n \supset \psi_n' \Rightarrow \varphi_1 \supset \varphi_2$

Table 6.2: Basic Rules Examples

Convention 6.1.4. Henceforth, we identify pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules (see Definition 2.2.5) with basic  $\mathcal{L}$ -rules that employ  $\pi_0$  as the context-relation in all of their premises. Thus we refer to all pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -rules as basic rules.

**Definition 6.1.5.** A basic  $\mathcal{L}$ -calculus consists of a finite set of basic  $\mathcal{L}$ -rules, that includes the primitive  $\mathcal{L}_2$ -rules:  $(\mathfrak{f}:weak)$ ,  $(\mathfrak{t}:weak)$ , (cut) and (id) (see Page 14). A proof in a basic  $\mathcal{L}$ -calculus is defined exactly as in pure  $\langle \mathcal{L}, \mathcal{L} \rangle$ -calculus (see Definition 2.2.14), and we write  $\mathcal{S} \vdash_{\mathbf{G}} s$  to denote the existence of a proof of an  $\mathcal{L}$ -sequent s from a set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents in a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$ .

**Notation 6.1.6.** We denote by  $R_{\mathbf{G}}$  all basic  $\mathcal{L}$ -rules of a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$ , except for  $(\mathbf{f}:weak)$ ,  $(\mathbf{t}:weak)$ , (cut) and (id).  $\Pi_{\mathbf{G}}$  denotes the set of  $\mathcal{L}$ -context-relations appearing in the basic rules of  $\mathbf{G}$  (in particular, since (cut) is always included,  $\pi_0 \in \Pi_{\mathbf{G}}$  for every basic calculus  $\mathbf{G}$ ).

Note that pure  $\langle \mathcal{L}, \mathcal{L}_2 \rangle$ -calculi whose primitive rules include (cut) and (id) are obtained as a particular instance in which  $\Pi_{\mathbf{G}} = \{\pi_0\}$ . In addition, the above notion of a basic rule is sufficiently general, so that many known sequent systems for various propositional logics can be easily presented in this framework. Next, we list some of these

sequent systems, and present their formulation as basic calculi. In the sequel, we will return to some of these calculi, provide a semantics for them, and use it to study their proof-theoretic properties.

Remark 6.1.7. When we say that a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  is a calculus for a logic  $\mathbf{L} = \langle \mathcal{L}, \Vdash \rangle$ , we mean that  $\{\mathbf{t}: \psi \mid \psi \in \mathcal{T}\} \vdash_{\mathbf{G}} \{\mathbf{t}: \varphi\}$  iff  $\mathcal{T} \Vdash \varphi$ . In cases where there is difference between the local version of the logic and the global one, as happens in modal logics, we refer to the global version (see [35]). In addition, this chapter deals only with propositional logics. Throughout, when we mention a known Gentzen-type system, we refer only to its propositional fragment.

Example 6.1.8 (LJ). The most famous sequent system for intuitionistic logic is ofcourse Gentzen's LJ [56]. This system manipulates single-conclusion sequents, and thus it does not fall in our framework.<sup>2</sup> However, there is an equivalent multiple-conclusion system, called LJ' in [90], that can be naturally presented as a basic calculus, that we call LJ. Its propositional language  $\mathcal{L}_{LJ}$  is  $\{\bot^0, \wedge^2, \vee^2, \supset^2\}$  ( $\neg \varphi$  can be defined by  $\varphi \supset \bot$ ). The rules of LJ are the same rules of LK (see Example 2.2.20), except for (t:  $\supset$ ), in which  $\pi_{int}$  is used instead of  $\pi_0$  (see Table 6.1). Thus this rule has now the form  $\langle \{f:p_1, t:p_2\}, \pi_{int}\rangle/\{t:p_1 \supset p_2\}$ , and its applications allow to infer sequents of the form  $\Gamma \Rightarrow \varphi_1 \supset \varphi_2$  from  $\Gamma, \varphi_1 \Rightarrow \varphi_2$  (note that right context-formulas are forbidden). In addition, since we do not include  $\neg$  in  $\mathcal{L}_{LJ}$ , we discard its rules, and add the following basic  $\mathcal{L}_{LJ}$ -rule for  $\bot$ :  $\emptyset/\{f: \bot\}$ .

**Example 6.1.9** (**BLJ**). Bi-intuitionistic logic (see, e.g., [59]) is the extension of intuitionistic logic with a binary connective dual to implication (denoted here by  $\prec$ ). Thus its language is  $\{\bot^0, \wedge^2, \vee^2, \supset^2, \prec^2\}$ , and we denote it by  $\mathcal{L}_{\mathbf{BLJ}}$ . A sequent system for this logic (see [80]) can be presented as a basic  $\mathcal{L}_{\mathbf{BLJ}}$ -calculus, which we call **BLJ**, obtained by augmenting **LJ** with the following rules:

(f:  $\prec$ )  $\langle \{f:p_1, t:p_2\}, \pi_d \rangle / \{f:p_1 \prec p_2\}$  (t:  $\prec$ )  $\langle \{t:p_1\}, \pi_0 \rangle, \langle \{f:p_2\}, \pi_0 \rangle / \{t:p_1 \prec p_2\}$  Applications of these rules have the forms:

$$(\mathbf{f}:\prec) \quad \frac{\varphi_1 \Rightarrow \varphi_2, \Delta}{\varphi_1 \prec \varphi_2 \Rightarrow \Delta} \qquad (\mathbf{t}:\prec) \quad \frac{\Gamma_1 \Rightarrow \varphi_1, \Delta_1 \quad \Gamma_2, \varphi_2 \Rightarrow \Delta_2}{\Gamma_1, \Gamma_2 \Rightarrow \varphi_1 \prec \varphi_2, \Delta_1, \Delta_2}$$

**Example 6.1.10** (**PLJ**). Sequent systems for many paraconsistent logics that extend the positive fragment of intuitionistic logic are defined and studied in [8]. All of them belong to the family of basic calculi. For example, we present the system  $PLJ(\{(\Rightarrow \neg \supset)\})$  from [8] as a basic calculus, that we call **PLJ**. Let  $\mathcal{L}_{PLJ} = \{\neg^1, \wedge^2, \vee^2, \supset^2\}$ . The basic

<sup>&</sup>lt;sup>2</sup>Note that canonical *single-conclusion* (two-sided) *sequent* calculi, of which LJ is the prototype example, were introduced and studied in the author's M.Sc. thesis (see also [14]).

6.1. Basic Calculi 77

 $\mathcal{L}_{PLJ}$ -calculus PLJ is obtained from LJ by adding the following rules (and discarding the rule for  $\bot$ ):

$$(t:\neg)$$
  $\langle \{f:p_1\}, \pi_0 \rangle / \{t:\neg p_1\}$   $(t:\neg \supset)$   $\langle \{t:p_1\}, \pi_0 \rangle, \langle \{t:\neg p_2\}, \pi_0 \rangle / \{t:\neg (p_1 \supset p_2)\}$  Applications of these rules have the forms:

$$(t:\neg) \quad \frac{\Gamma, \varphi \Rightarrow \Delta}{\Gamma \Rightarrow \neg \varphi, \Delta} \qquad (t:\neg \supset) \quad \frac{\Gamma_1 \Rightarrow \varphi_1, \Delta_1 \quad \Gamma_2 \Rightarrow \neg \varphi_2, \Delta_2}{\Gamma_1, \Gamma_2 \Rightarrow \neg (\varphi_1 \supset \varphi_2), \Delta_1, \Delta_2}$$

**Example 6.1.11** (Systems for Modal Logics). Ordinary sequent systems for modal logics are surveyed in [96] and [81]. All of them belong to the family of basic calculi. As examples we present as basic calculi six of them (used later to demonstrate certain semantic phenomena). Let  $\mathcal{L}_{\square} = \{\square^1, \neg^1, \wedge^2, \vee^2, \supset^2\}$ . We use the basic rules (K), (B) and (S5) (in addition to some of the rules presented in Table 6.2). (K), (B) and (S5) all have the form  $\langle \{\mathbf{t}:p_1\}, \pi \rangle / \{\mathbf{t}: \square p_1\}$ , where  $\pi$  is  $\pi_K$ ,  $\pi_B$ , and  $\pi_{S5}$  respectively (see Table 6.1). Applications of these rules have the form:

$$(K) \quad \frac{\Gamma \Rightarrow \varphi}{\Box \Gamma \Rightarrow \Box \varphi} \qquad (B) \quad \frac{\Gamma \Rightarrow \varphi, \Box \Delta}{\Box \Gamma \Rightarrow \Box \varphi, \Delta} \qquad (S5) \quad \frac{\Box \Gamma \Rightarrow \varphi, \Box \Delta}{\Box \Gamma \Rightarrow \Box \varphi, \Box \Delta}$$

Based on LK, six basic  $\mathcal{L}_{\square}$ -calculi are defined as follows:

$$K = LK + (K)$$
  $K4 = LK + (K4)$   $KD = K + (D)$   $KB = LK + (B)$   $S4 = LK + (S4) + (T)$   $S5 = LK + (S5) + (T)$ 

Note that  $\square$  is the only primitive modality in the language  $\mathcal{L}_{\square}$ , and  $\Diamond \varphi$  can be defined as  $\neg \square \neg \varphi$ . For an extended language with two dual primitive modal operators, one should modify some of the context-relations in the rules for  $\square$ , and add dual rules for  $\Diamond$ . For example, for the logic S4 the following four schemes are used:

$$\begin{array}{c|c} \Box\Gamma\Rightarrow\varphi,\Diamond\Delta & \Gamma,\varphi\Rightarrow\Delta & \Box\Gamma,\varphi\Rightarrow\Diamond\Delta & \Gamma\Rightarrow\varphi,\Delta \\ \hline \Box\Gamma\Rightarrow\Box\varphi,\Diamond\Delta & \Gamma,\Box\varphi\Rightarrow\Delta & \Box\Gamma,\Diamond\varphi\Rightarrow\Diamond\Delta & \Gamma\Rightarrow\Diamond\varphi,\Delta \\ \hline \end{array}$$

**Example 6.1.12** (**GL**). The logic GL (the modal logic of provability, see [94]) is obtained by adding the axiom  $\Box(\Box\varphi\supset\varphi)\supset\Box\varphi$  to the usual Hilbert system for the modal logic K. In addition, GL has a well-known sequent system (see, e.g., [75, 84, 3]), that can be presented as a basic  $\mathcal{L}_{\Box}$ -calculus, that we call **GL**. **GL** is obtained from **LK** by adding (GL) – the basic  $\mathcal{L}_{\Box}$ -rule  $\langle \{\mathbf{f}:\Box p_1,\mathbf{t}:p_1\},\pi_{\mathsf{K4}}\rangle/\{\mathbf{t}:\Box p_1\}$ . Applications of (GL) allow to infer sequents of the form  $\Box\Gamma_1,\Box\Gamma_2\Rightarrow\Box\varphi$  from  $\Gamma_1,\Box\Gamma_2,\Box\varphi\Rightarrow\varphi$ .

**Example 6.1.13** (**IS5**). Sequent systems for intuitionistic modal logics provide an interesting source of examples to be studied in the framework of basic calculi, as they naturally employ more than one (non-trivial) context-relation. For example, the system  $G_3$  from [78] can be presented as the basic  $\mathcal{L}_{\mathbf{LJ}}^{\square}$ -calculus obtained from  $\mathbf{LJ}$  by adding the rules (S5) and (T) ( $\mathcal{L}_{\mathbf{LJ}}^{\square}$  denotes the language obtained by augmenting  $\mathcal{L}_{\mathbf{LJ}}$  with a unary connective  $\square$ ). In the sequel, we refer to this basic calculus as **IS5**.

**Example 6.1.14.** In [74] several sequent calculi for weak modal logics are introduced. All of them belong to the family of basic calculi. For example, the first system from [74] (called Mseq there), is the basic  $\mathcal{L}_{\square}$ -calculus obtained from  $\mathbf{LK}$  by adding (M) – the basic rule  $\langle \{\mathbf{f}:p_1,\mathbf{t}:p_2\},\emptyset\rangle/\{\mathbf{f}:\square p_1,\mathbf{t}:\square p_2\}$ . Its applications allow to infer sequents of the form  $\square\varphi_1\Rightarrow\square\varphi_2$  from  $\varphi_1\Rightarrow\varphi_2$ .

**Example 6.1.15.** Several sequent systems for logics of strict implication are provided in [65], and can be presented as basic calculi. For example,  $GS4^I$  (from [65]) is equivalent to the basic  $\mathcal{L}_{cl}$ -calculus, obtained from LK by replacing the rule ( $t: \supset$ ) with the rule ( $t: \supset^*$ ) (see Table 6.2). Note that  $GS4^I$  includes also a less standard rule (denoted by ( $\to K^I$ ) in [65]) that cannot be presented as one basic rule. However, one can show that it is redundant in  $GS4^I$ .

**Example 6.1.16** (**GP**). Primal logic was defined and studied in [31]. As explained there, this logic is used in the context of the access control language DKAL. We consider here the sequent system GP from [31] for primal logic with disjunction and quotations. Given a finite set Q of constants denoting "principals", let

$$\mathcal{L}_{\mathbf{GP}}^Q = \{\bot^0, \top^0, \wedge^2, \vee^2, \supset^2\} \cup \{q \text{ said}^1, q \text{ implied}^1 \mid q \in Q\}.$$

The sequent system GP (over Q) can be presented as a basic  $\mathcal{L}_{\mathbf{GP}}^Q$ -calculus, that we call  $\mathbf{GP}^Q$ . The rules of  $\mathbf{GP}^Q$  are the rules of  $\mathbf{LK}$  for  $\wedge, \vee$ , and the following rules for the other connectives (for every  $q \in Q$ ):

$$\begin{split} (\mathbf{f}:\bot) & \;\; \emptyset/\{\mathbf{f}:\bot\} \\ (\mathbf{f}:\supset) & \;\; \langle \{\mathbf{t}:p_1\},\pi_0\rangle, \langle \{\mathbf{f}:p_2\},\pi_0\rangle/\{\mathbf{f}:p_1\supset p_2\} \\ & \;\;\; (\mathbf{t}:\supset) & \;\; \langle \{\mathbf{t}:p_2\},\pi_0\rangle/\{\mathbf{t}:p_1\supset p_2\} \\ & \;\;\; (\mathbf{Said}_q) & \;\;\; \langle \{\mathbf{t}:p_1\},\pi_s^q\rangle/\{\mathbf{t}:q \; \mathbf{said} \; p_1\} \\ & \;\;\; (\mathbf{Implied}_q) & \;\; \langle \{\mathbf{t}:p_1\},\pi_i^q\rangle/\{\mathbf{t}:q \; \mathbf{implied} \; p_1\} \end{split}$$

where  $\pi_s^q = \{\langle \mathbf{f}: p_1, \mathbf{f}: q \text{ said } p_1 \rangle \}$ , and  $\pi_i^q = \pi_s^q \cup \{\langle \mathbf{f}: p_1, \mathbf{f}: q \text{ implied } p_1 \rangle \}$  for every  $q \in Q$ . Applications of these rules have the form:

### 6.1.1 Proof-Theoretic Properties

Generally speaking, the definition of the proof-theoretic properties from Chapter 2 are adapted in the obvious way to basic calculi (see Section 2.3). Formally, since we explicitly

6.1. Basic Calculi 79

required that each basic calculus includes (cut) and (id), we can not consider the calculi obtained from a basic calculus by discarding (cut) or (id) as basic calculi. Thus we find it convenient to restrict the proofs in basic calculi via proof specifications:

**Definition 6.1.17.** An  $\mathcal{L}$ -proof-specification is a triple of sets of  $\mathcal{L}$ -formulas  $\langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ . Given an  $\mathcal{L}$ -proof-specification  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ , a proof P in a basic  $\mathcal{L}$ -calculus G is called a  $\rho$ -proof if the following conditions hold:

- 1. P contains only  $\mathcal{F}$ -sequents (that is  $\mathcal{L}$ -sequents consisting only of formulas from  $\mathcal{F}$ , see Definition 2.3.3).
- 2. The cut-formula of every application of (cut) in P is in C.
- 3. The id-formula of every application of (id) in P is in A.

We write  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  if there exists a  $\rho$ -proof in a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  of an  $\mathcal{L}$ -sequent s from a set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents.

Note that  $\vdash_{\mathbf{G}}$  is a special case of  $\vdash_{\mathbf{G}\upharpoonright\rho}$ , obtained by choosing  $\rho = \langle \mathcal{L}, \mathcal{L}, \mathcal{L} \rangle$ . In addition, the following are equivalent:

- 1.  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} s \text{ for } \rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle.$
- 2.  $\{s' \in \mathcal{S} \mid frm[s'] \subseteq \mathcal{F}\} \vdash_{\mathbf{G} \upharpoonright \rho'} s \text{ for } \rho' = \langle \mathcal{F}, \mathcal{C} \cap \mathcal{F}, \mathcal{A} \cap \mathcal{F} \rangle$ .
- 3.  $S \cup \{\{f:\varphi, t:\varphi\} \mid \varphi \in A\} \vdash_{\mathbf{G} \upharpoonright \rho'} s \text{ for } \rho' = \langle \mathcal{F}, \mathcal{C}, \emptyset \rangle.$

Now, we can uniformly define  $(strong) \leq -analyticity$ , (strong) cut-admissibility, and axiom-expansion using proof-specifications.

#### **Definition 6.1.18.** Let G be a basic $\mathcal{L}$ -calculus.

- 1. Let  $\leq$  be a safe partial order on  $\mathcal{L}$ .  $\mathbf{G}$  is  $\leq$ -analytic if for every  $\mathcal{L}$ -sequent s,  $\vdash_{\mathbf{G}} s$  implies  $\vdash_{\mathbf{G}\upharpoonright\rho} s$  for  $\rho = \langle \downarrow^{\leq} [s], \mathcal{L}, \mathcal{L} \rangle$ .  $\mathbf{G}$  is  $strongly \leq$ -analytic if for every set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents and  $\mathcal{L}$ -sequent s,  $\mathcal{S} \vdash_{\mathbf{G}} s$  implies  $\mathcal{S} \vdash_{\mathbf{G}\upharpoonright\rho} s$  for  $\rho = \langle \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \mathcal{L}, \mathcal{L} \rangle$ .
- 2. **G** enjoys *cut-admissibility* if for every  $\mathcal{L}$ -sequent s,  $\vdash_{\mathbf{G}\upharpoonright\rho} s$  for  $\rho = \langle \mathcal{L}, \emptyset, \mathcal{L} \rangle$  whenever  $\vdash_{\mathbf{G}} s$ . **G** enjoys *strong cut-admissibility* if for every set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents and  $\mathcal{L}$ -sequent s,  $\mathcal{S} \vdash_{\mathbf{G}\upharpoonright\rho} s$  for  $\rho = \langle \mathcal{L}, frm[\mathcal{S}], \mathcal{L} \rangle$  whenever  $\mathcal{S} \vdash_{\mathbf{G}} s$ .
- 3. A connective  $\diamond \in \diamond_{\mathcal{L}}$  admits axiom-expansion in  $\mathbf{G}$  if  $\vdash_{\mathbf{G} \upharpoonright \rho} \{ \mathbf{f} : \varphi, \mathbf{t} : \varphi \}$  for the formula  $\varphi = \diamond(p_1, \dots, p_{ar(\diamond)})$  and  $\rho = \langle \mathcal{L}, \mathcal{L}, \{p_1, \dots, p_{ar(\diamond)}\} \rangle$ .

Note also that the consequences of (strong)  $\leq$ -analyticity discussed in Section 2.3.1 (consistency, conservativity, and decidability) hold for (strong)  $\leq$ -analytic basic calculi as well, with minor modification in their proofs.

# 6.2 Kripke-style Semantics for Basic Calculi

In this section we introduce a method for providing semantics for any given basic calculus and proof-specification. Thus, given a basic calculus  $\mathbf{G}$  and a proof-specification  $\rho$ , we show how to uniformly recognize a class of semantic structures  $\mathcal{K}_{\mathbf{G}^{\dagger}\rho}$ , that naturally induce a semantic consequence relation  $\vdash_{\mathcal{K}_{\mathbf{G}^{\dagger}\rho}}$  between sequents, for which we have soundness and completeness, i.e.  $\vdash_{\mathbf{G}^{\dagger}\rho} = \vdash_{\mathcal{K}_{\mathbf{G}^{\dagger}\rho}}$ . The semantic framework employed for this purpose is a generalization of Kripke-style semantics for modal and intuitionistic logic, where instead of usual Kripke frames and models, we will have (partial) *Kripke valuations*. These are defined as follows:

**Definition 6.2.1.** A partial Kripke  $\mathcal{L}$ -valuation (partial  $\mathcal{L}$ -Kvaluation, for short) is a function v from the Cartesian product of some set  $W_v$  (whose elements are called worlds) and some set  $Dom_v \subseteq \mathcal{L}$  to  $2^{\{f,t\}}$ . A partial  $\mathcal{L}$ -Kvaluation v with  $Dom_v = \mathcal{L}$  is also called an  $\mathcal{L}$ -Kvaluation.

Note that as in Chapter 3, the truth values are subsets of labels. Next, we introduce the semantic consequence relation associated with a given set of such valuations.

#### **Definition 6.2.2.** Let v be a partial $\mathcal{L}$ -Kvaluation.

- 1. A labelled  $\mathcal{L}$ -formula  $\mathbf{x}:\varphi$  is true in some  $w \in W_v$  with respect to v (denoted by:  $v, w \models \mathbf{x}:\varphi$ ) if  $\varphi \in Dom_v$  and  $\mathbf{x} \in v(w, \varphi)$ .
- 2. An  $\mathcal{L}$ -sequent s is true with respect to v:
  - (a) in some  $w \in W_v$  (denoted by:  $v, w \models s$ ) if  $v, w \models \alpha$  for some  $\alpha \in s$ .
  - (b) in some set  $W \subseteq W_v$  (denoted by:  $v, W \models s$ ) if  $v, w \models s$  for every  $w \in W$ .
- 3. v is a model of:
  - (a) an  $\mathcal{L}$ -sequent s (denoted by:  $v \models s$ ) if s is a  $Dom_v$ -sequent and  $v, W_v \models s$ .
  - (b) a set S of L-sequents (denoted by:  $v \models S$ ) if  $v \models s$  for every  $Dom_v$ -sequent  $s \in S$ .

Now, the given proof-specification  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  enforces some simple conditions on Kvaluations: (1) As in Chapter 3,  $Dom_v$  should consist exactly of the formulas in  $\mathcal{F}$ , those that are allowed to appear in  $\rho$ -proofs; (2) If some formula  $\varphi$  may serve as a cut-formula (i.e., if  $\varphi \in \mathcal{C} \cap \mathcal{F}$ ), the value  $\{\mathbf{f}, \mathbf{t}\}$  should be never assigned to  $\varphi$ ; and (3) Similarly,  $\emptyset$  should not be assigned to  $\varphi$ , if  $\varphi$  may serve as an id-formula (i.e., if  $\varphi \in \mathcal{A} \cap \mathcal{F}$ ). These restrictions are formulated in the next definition.

**Definition 6.2.3.** Let  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  be an  $\mathcal{L}$ -proof-specification. A partial  $\mathcal{L}$ -Kvaluation v is called  $\rho$ -legal if the following hold:

- 1.  $Dom_v = \mathcal{F}$ .
- 2.  $v(w,\varphi) \neq \{f,t\}$  for every  $w \in W_v$  and  $\varphi \in \mathcal{C} \cap \mathcal{F}$ .
- 3.  $v(w,\varphi) \neq \emptyset$  for every  $w \in W_v$  and  $\varphi \in \mathcal{A} \cap \mathcal{F}$ .

A special kind of Kvaluations will be particularly needed below:

**Definition 6.2.4.** A partial  $\mathcal{L}$ -Kvaluation is called *normal* if it is  $\langle \mathcal{F}, \mathcal{L}, \mathcal{L} \rangle$ -legal for some  $\mathcal{F} \subseteq \mathcal{L}$ .

Note that a normal partial  $\mathcal{L}$ -Kvaluation v assigns either  $\{f\}$  or  $\{t\}$  to any pair  $\langle w, \varphi \rangle \in W_v \times Dom_v$ .

Next we turn to restrictions on Kvaluations imposed by the basic system itself. As in Chapter 3, the intuitive idea is that each syntactic ingredient of  $\mathbf{G}$  imposes a certain constraint on Kvaluations. Taking all of these constraints together, we get a set of Kvaluations for which  $\mathbf{G}$  is sound and complete. The exact constraints are formulated below.

First, we associate with each context-relation  $\pi$  of  $\mathbf{G}$  a binary ("accessibility") relation on  $W_v$ , and enforce certain conditions on the associated accessibility relations.

**Definition 6.2.5.** Given a set W, a  $\langle \mathbf{G}, W \rangle$ -coupling is a function assigning a binary relation on W to every  $\pi \in \Pi_{\mathbf{G}}$ .

**Definition 6.2.6.** Let **G** be a basic  $\mathcal{L}$ -calculus, v a partial  $\mathcal{L}$ -Kvaluation, and  $\Re$  a  $\langle \mathbf{G}, W_v \rangle$ -coupling.

- 1. Given an  $\mathcal{L}$ -context-relation  $\pi$ ,  $R_{\pi}^{v}$  denotes the binary relation on  $W_{v}$  defined as follows:  $w_{1}R_{\pi}^{v}w_{2}$  iff for every two labelled  $Dom_{v}$ -formulas  $\alpha_{1}, \alpha_{2}$ , if  $\alpha_{2}\bar{\pi}\alpha_{1}$  and  $v, w_{2} \models \alpha_{2}$  then  $v, w_{1} \models \alpha_{1}$ .
- 2.  $\langle v, \Re \rangle$  is called:
  - (a)  $\pi$ -legal for some  $\pi \in \Pi_{\mathbf{G}}$  if  $\Re(\pi) \subseteq R_{\pi}^{v}$ .
  - (b)  $\Pi$ -legal for some  $\Pi \subseteq \Pi_{\mathbf{G}}$  if it is  $\pi$ -legal for every  $\pi \in \Pi$ .

**Example 6.2.7.** Let **G** be a basic  $\mathcal{L}$ -calculus, v a partial  $\mathcal{L}$ -Kvaluation, and  $\Re$  a  $\langle \mathbf{G}, W_v \rangle$ -coupling.

1. Consider the  $\mathcal{L}$ -context-relation  $\pi_0$ . By definition,  $\alpha_2 \bar{\pi_0} \alpha_1$  iff  $\alpha_2 = \alpha_1$ . Thus  $w_1 R_{\pi_0}^v w_2$  iff for every labelled  $Dom_v$ -formula  $\alpha$  such that  $v, w_2 \models \alpha$ , we have

- $v, w_1 \models \alpha$ . Equivalently,  $w_1 R_{\pi_0}^v w_2$  iff  $v(w_2, \varphi) \subseteq v(w_1, \varphi)$  for every  $\varphi \in Dom_v$ . Therefore,  $\langle v, \Re \rangle$  is  $\pi_0$ -legal iff for every  $w_1, w_2 \in W_v$  such that  $w_1 \Re(\pi_0) w_2$ , we have that  $v(w_2, \varphi) \subseteq v(w_1, \varphi)$  for every  $\varphi \in Dom_v$ . In particular, if  $\Re(\pi_0)$  is identity (on  $W_v$ ) then  $\langle v, \Re \rangle$  is  $\pi_0$ -legal. Note that if v is normal, then  $w_1 R_{\pi_0}^v w_2$  iff  $v(w_1, \varphi) = v(w_2, \varphi)$  for every  $\varphi \in Dom_v$ . In this case,  $\langle v, \Re \rangle$  is  $\pi_0$ -legal iff for every  $w_1, w_2 \in W_v$  such that  $w_1 \Re(\pi_0) w_2$ , we have that  $v(w_2, \varphi) = v(w_1, \varphi)$  for every  $\varphi \in Dom_v$ .
- 2. Suppose that  $\pi_{\text{int}}$  appears in  $\Pi_{\mathbf{G}}$ . Here,  $\alpha_2\pi_{\text{int}}^-\alpha_1$  iff  $\alpha_2=\alpha_1=\mathbf{f}:\varphi$  for some  $\varphi\in\mathcal{L}$ . Thus  $w_1R_{\pi_{\text{int}}}^vw_2$  iff for every labelled  $Dom_v$ -formula  $\alpha$  of the form  $\mathbf{f}:\varphi$ , if  $v,w_2\models\alpha$ , then  $v,w_1\models\alpha$ . Equivalently,  $w_1R_{\pi_{\text{int}}}^vw_2$  iff  $\mathbf{f}\in v(w_2,\varphi)$  implies that  $\mathbf{f}\in v(w_1,\varphi)$  for every  $\varphi\in Dom_v$ . Therefore,  $\langle v,\Re\rangle$  is  $\pi_{\text{int}}$ -legal iff for every  $w_1,w_2\in W_v$  such that  $w_1\Re(\pi_{\text{int}})w_2$ , we have that  $\mathbf{f}\in v(w_2,\varphi)$  implies  $\mathbf{f}\in v(w_1,\varphi)$  for every  $\varphi\in Dom_v$ . Note that if v is normal, then  $\langle v,\Re\rangle$  is  $\pi_{\text{int}}$ -legal iff for every  $w_1,w_2\in W_v$  such that  $w_1\Re(\pi_{\text{int}})w_2$ , we have that  $v(w_1,\varphi)=\{\mathbf{t}\}$  implies  $v(w_2,\varphi)=\{\mathbf{t}\}$  for every  $\varphi\in Dom_v$ . This restriction corresponds to the persistence (or "monotonicity") condition that is employed in intuitionistic Kripke semantics, where  $\Re(\pi_{\text{int}})$  serves as the accessibility relation.
- 3. Suppose that  $\pi_{\mathsf{K}}$  appears in  $\Pi_{\mathsf{G}}$ . Here,  $w_1 R_{\pi_{\mathsf{K}}}^v w_2$  iff  $v, w_2 \models \mathsf{f} : \varphi$  implies  $v, w_1 \models \mathsf{f} : \Box \varphi$  whenever  $\varphi \in Dom_v$  and  $\Box \varphi \in Dom_v$ . Equivalently,  $w_1 R_{\pi_{\mathsf{K}}}^v w_2$  iff  $\mathsf{f} \in v(w_2, \varphi)$  implies  $\mathsf{f} \in v(w_1, \Box \varphi)$  whenever  $\varphi \in Dom_v$  and  $\Box \varphi \in Dom_v$ . Therefore,  $\langle v, \Re \rangle$  is  $\pi_{\mathsf{K}}$ -legal iff for every  $w_1, w_2 \in W_v$  such that  $w_1 \Re(\pi_{\mathsf{K}}) w_2$ , we have that  $\mathsf{f} \in v(w_2, \varphi)$  implies  $\mathsf{f} \in v(w_1, \Box \varphi)$  whenever  $\varphi \in Dom_v$  and  $\Box \varphi \in Dom_v$ . Roughly speaking, this provides "one half" of the usual semantics of  $\Box$ .
- 4. Suppose that  $\emptyset$  appears in  $\Pi_{\mathbf{G}}$  (this context-relation is used in the rule (M), see Example 6.1.14). Since there do not exist labelled  $\mathcal{L}$ -formulas  $\alpha_1, \alpha_2$  such that  $\alpha_2 \bar{\emptyset} \alpha_1, \ w_1 R_{\emptyset}^v w_2$  trivially holds for every  $w_1, w_2 \in W_v$ . Thus  $R_{\emptyset}^v = W_v \times W_v$ , and every pair  $\langle v, \Re \rangle$  is trivially  $\emptyset$ -legal.

Next we formulate the effect of the basic rules appearing in a basic calculus.

**Notation 6.2.8.** Given a set W, a binary relation  $R \subseteq W \times W$ , and an element  $w \in W$ , we denote the set  $\{w' \in W \mid wRw'\}$  by R[w].

**Definition 6.2.9.** Let **G** be a basic  $\mathcal{L}$ -calculus, v a partial  $\mathcal{L}$ -Kvaluation, and  $\Re$  a  $\langle \mathbf{G}, W_v \rangle$ -coupling.  $\langle v, \Re \rangle$  is called:

1. r-legal for some  $r = \langle s_1, \pi_1 \rangle, \dots, \langle s_n, \pi_n \rangle / s$  in  $R_{\mathbf{G}}$  if the following condition holds for every  $w \in W_v$  and  $\mathcal{L}$ -substitution  $\sigma$  such that  $frm[\sigma(\{s_1, \dots, s_n, s\})] \subseteq Dom_v$ : if  $v, \Re(\pi_i)[w] \models \sigma(s_i)$  for every  $1 \leq i \leq n$ , then  $v, w \models \sigma(s)$ .

2. R-legal for some  $R \subseteq R_G$  if it is r-legal for every  $r \in R$ .

**Example 6.2.10.** Let **G** be a basic  $\mathcal{L}$ -calculus, v a partial  $\mathcal{L}$ -Kvaluation, and  $\Re$  a  $\langle \mathbf{G}, W_v \rangle$ -coupling.

- 1. Suppose that  $R_{\mathbf{G}}$  includes a rule r of the form  $\langle \{\mathsf{t}:p_1\}, \pi \rangle / \{\mathsf{t}:\Box p_1\}$  (such a rule appears in various basic calculi for modal logics presented above).  $\langle v, \Re \rangle$  is r-legal iff for every  $w \in W_v$  and  $\mathcal{L}$ -substitution  $\sigma$ : if  $frm[\sigma(\{\mathsf{t}:p_1, \mathsf{t}:\Box p_1\})] \subseteq Dom_v$ , and  $v, \Re(\pi)[w] \models \sigma(\{\mathsf{t}:p_1\})$ , then  $v, w \models \sigma(\{\mathsf{t}:\Box p_1\})$ . Equivalently,  $\langle v, \Re \rangle$  is r-legal iff for every  $w \in W_v$  and formula  $\varphi$ : if  $\{\varphi, \Box \varphi\} \subseteq Dom_v$ , and  $\mathsf{t} \in v(w', \varphi)$  for every  $w' \in \Re(\pi)[w]$ , then  $\mathsf{t} \in v(w, \Box \varphi)$ . Roughly speaking, this provides the "other half" of the usual semantics of  $\Box$  (see Example 6.2.7, Item 3).
- 2. Suppose that  $R_{\mathbf{G}}$  includes a rule r of the form  $\langle \{\mathbf{t}:p_1\}, \pi \rangle, \langle \{\mathbf{f}:p_2\}, \pi \rangle / \{\mathbf{f}:p_1 \supset p_2\}$  (a rule of this form appears in  $\mathbf{L}\mathbf{K}$  and  $\mathbf{L}\mathbf{J}$  with  $\pi = \pi_0$ ).  $\langle v, \Re \rangle$  is r-legal iff for every  $w \in W_v$  and  $\mathcal{L}$ -substitution  $\sigma$ : if  $frm[\sigma(\{\mathbf{t}:p_1,\mathbf{f}:p_2,\mathbf{f}:p_1 \supset p_2\})] \subseteq Dom_v$ ,  $v, \Re(\pi)[w] \models \sigma(\{\mathbf{t}:p_1\})$  and  $v, \Re(\pi)[w] \models \sigma(\{\mathbf{f}:p_2\})$ , then  $v, w \models \sigma(\{\mathbf{f}:p_1 \supset p_2\})$ . Equivalently,  $\langle v, \Re \rangle$  is r-legal iff for every  $w \in W_v$  and two formulas  $\varphi_1, \varphi_2$ : if  $\{\varphi_1, \varphi_2, \varphi_1 \supset \varphi_2\} \subseteq Dom_v$ , and for every  $w' \in \Re(\pi)[w]$  it holds that  $\mathbf{t} \in v(w', \varphi_1)$  and  $\mathbf{f} \in v(w', \varphi_2)$ , then  $\mathbf{f} \in v(w, \varphi_1 \supset \varphi_2)$ .
- 3. Suppose that  $R_{\mathbf{G}}$  includes a rule r of the form  $\langle \{\mathbf{f}:p_1,\mathbf{t}:p_2\},\pi\rangle/\{\mathbf{t}:p_1\supset p_2\}$  (a rule of this form appears in  $\mathbf{L}\mathbf{K}$  with  $\pi=\pi_0$ , and in  $\mathbf{L}\mathbf{J}$  with  $\pi=\pi_{\mathrm{int}}$ ).  $\langle v,\Re\rangle$  is r-legal iff for every  $w\in W_v$  and  $\mathcal{L}$ -substitution  $\sigma$ : if  $frm[\sigma(\{\mathbf{f}:p_1,\mathbf{t}:p_2,\mathbf{t}:p_1\supset p_2\})]\subseteq Dom_v$ , and  $v,\Re(\pi)[w]\models\sigma(\{\mathbf{f}:p_1,\mathbf{t}:p_2\})$ , then  $v,w\models\sigma(\{\mathbf{t}:p_1\supset p_2\})$ . Equivalently,  $\langle v,\Re\rangle$  is r-legal iff for every  $w\in W_v$  and two formulas  $\varphi_1,\varphi_2$ : if  $\{\varphi_1,\varphi_2,\varphi_1\supset\varphi_2\}\subseteq Dom_v$ , and  $\mathbf{f}\in v(w',\varphi_1)$  or  $\mathbf{t}\in v(w',\varphi_2)$  for every  $w'\in\Re(\pi)[w]$ , then  $\mathbf{t}\in v(w,\varphi_1\supset\varphi_2)$ .
- 4. Suppose that  $R_{\mathbf{G}}$  includes a rule r of the form  $\langle \emptyset, \pi \rangle / \emptyset$  (for example, this is the form of the rule (D), see Table 6.2). Applications of this rule allow to infer an  $\mathcal{L}$ -sequent s' from an  $\mathcal{L}$ -sequent s whenever  $\langle s, s' \rangle$  is a  $\pi$ -instance.  $\langle v, \Re \rangle$  is r-legal iff for every  $w \in W_v$ : if  $v, \Re(\pi)[w] \models \emptyset$ , then  $v, w \models \emptyset$  (note that  $\sigma(\emptyset) = \emptyset$  for every  $\mathcal{L}$ -substitution  $\sigma$ ). Since the empty sequent is not true in any world, this condition would hold iff for every world w there exists some  $w' \in \Re(\pi)[w]$ . In other words,  $\langle v, \Re \rangle$  is r-legal iff  $\Re(\pi)$  is a serial relation.

Now, by collecting the semantic restrictions introduced by the context-relations and the basic rules, as well as those of the proof-specification, we obtain the set  $\mathcal{K}_{\mathbf{G} \upharpoonright \rho}$  of partial  $\mathcal{L}$ -Kvaluations for which a given basic calculus  $\mathbf{G}$  and a proof-specification  $\rho$  are sound and complete.

**Definition 6.2.11.** Let G be a basic  $\mathcal{L}$ -calculus, and v a partial  $\mathcal{L}$ -Kvaluation.

- 1. Given a  $\langle \mathbf{G}, W_v \rangle$ -coupling  $\Re$ , the pair  $\langle v, \Re \rangle$  is called  $\mathbf{G}$ -legal if it is both  $\Pi_{\mathbf{G}}$ -legal, and  $\mathsf{R}_{\mathbf{G}}$ -legal.
- 2. v is called  $\mathbf{G}$ -legal if  $\langle v, \Re \rangle$  is  $\mathbf{G}$ -legal for some  $\langle \mathbf{G}, W_v \rangle$ -coupling  $\Re$ .
- 3. Given an  $\mathcal{L}$ -proof-specification  $\rho$ , v is called  $\mathbf{G} \upharpoonright_{\rho}$ -legal if it is both  $\rho$ -legal and  $\mathbf{G}$ -legal.  $\mathcal{K}_{\mathbf{G} \upharpoonright_{\rho}}$  denotes the set of all  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations.

**Theorem 6.2.12.**  $\vdash_{\mathbf{G} \upharpoonright \rho} = \vdash_{\mathcal{K}_{\mathbf{G} \upharpoonright \rho}}$  for every basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  and  $\mathcal{L}$ -proof-specification  $\rho$ .

The proof is given in Section 6.4. In order to obtain a very general soundness result, we chose above the set  $\mathcal{K}_{\mathbf{G} \upharpoonright \rho}$  to be as large as possible. On the other hand, a stronger completeness result can be obtained by considering a smaller set of Kvaluations:

**Notation 6.2.13.** Given a partial  $\mathcal{L}$ -Kvaluation v, we denote by  $Id_v$  the identity relation on  $W_v$ .

**Definition 6.2.14.** Let G be a basic  $\mathcal{L}$ -calculus, and v a partial  $\mathcal{L}$ -Kvaluation.

- 1.  $\Re_{\mathbf{G}}^{v}$  denotes the  $\langle \mathbf{G}, W_{v} \rangle$ -coupling defined by  $\Re_{\mathbf{G}}^{v}(\pi_{0}) = Id_{v}$ , and  $\Re_{\mathbf{G}}^{v}(\pi) = R_{\pi}^{v}$  for every other  $\pi \in \Pi_{\mathbf{G}}$  (see Definition 6.2.6).
- 2. v is called strongly  $\mathbf{G}$ -legal if  $\langle v, \Re_{\mathbf{G}}^v \rangle$  is  $\mathsf{R}_{\mathbf{G}}$ -legal.
- 3. v is called differentiated if  $R_{\pi_0}^v = Id_v$ .
- 4. Given an  $\mathcal{L}$ -proof-specification  $\rho$ , v is called  $strongly\ \mathbf{G}\!\upharpoonright_{\rho}$ -legal if it is both  $\rho$ -legal and strongly  $\mathbf{G}$ -legal.  $\mathcal{K}^*_{\mathbf{G}}\!\upharpoonright_{\rho}$  denotes the set of all strongly  $\mathbf{G}\!\upharpoonright_{\rho}$ -legal and differentiated partial  $\mathcal{L}$ -Kvaluations.

By definition (and following Example 6.2.7, Item 1), for every partial  $\mathcal{L}$ -Kvaluation v, the pair  $\langle v, \Re_{\mathbf{G}}^v \rangle$  is  $\pi$ -legal for every  $\pi \in \Pi_{\mathbf{G}}$ . Thus, a strongly  $\mathbf{G}$ -legal partial  $\mathcal{L}$ -Kvaluation is  $\mathbf{G}$ -legal.

**Remark 6.2.15.** Following Example 6.2.7 (Item 1), a normal partial  $\mathcal{L}$ -Kvaluation v is differentiated iff  $w_1 = w_2$  whenever  $v(w_1, \varphi) = v(w_2, \varphi)$  for every  $\varphi \in Dom_v$ . The name of this property is taken from [41].

**Theorem 6.2.16.**  $\vdash_{\mathbf{G} \upharpoonright \rho} = \vdash_{\mathcal{K}_{\mathbf{G} \upharpoonright \rho}^*}$  for every basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  and  $\mathcal{L}$ -proof-specification  $\rho$ .

The proof is given in Section 6.4. The two last theorems are combined in the following theorem, that provides an "interval" of possible semantics for a given basic calculus.

**Theorem 6.2.17.** Let **G** be a basic  $\mathcal{L}$ -calculus, and  $\rho$  an  $\mathcal{L}$ -proof-specification. Then,  $\vdash_{\mathbf{G}\upharpoonright\rho}=\vdash_{\mathcal{K}}$  for every set  $\mathcal{K}$  of partial  $\mathcal{L}$ -Kvaluations satisfying  $\mathcal{K}^*_{\mathbf{G}\upharpoonright\rho}\subseteq\mathcal{K}\subseteq\mathcal{K}_{\mathbf{G}\upharpoonright\rho}$ .

The proof is given in Section 6.4. The following is a useful instance that does not consider proof-specifications at all:

Corollary 6.2.18. Let G be a basic  $\mathcal{L}$ -calculus. Then,  $\vdash_{\mathbf{G}} = \vdash_{\mathcal{K}}$  for every set  $\mathcal{K}$  of normal G-legal  $\mathcal{L}$ -Kvaluations that contains all normal strongly G-legal differentiated  $\mathcal{L}$ -Kvaluations.

*Proof.* Since  $\langle \mathcal{L}, \mathcal{L}, \mathcal{L} \rangle$ -legal partial  $\mathcal{L}$ -Kvaluations are exactly normal  $\mathcal{L}$ -Kvaluations, the claim directly follows from Theorem 6.2.17.

Theorem 6.2.17 provides a general soundness and completeness result applicable to every basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  and  $\mathcal{L}$ -proof-specification  $\rho$ . Its exact content depends on the choice of set  $\mathcal{K}$  of partial  $\mathcal{L}$ -Kvaluations.  $\mathcal{K}$  should meet two conditions: first, it should contain only  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations; and second, it should contain all strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal differentiated partial  $\mathcal{L}$ -Kvaluations. In many cases, using the structure of the context-relations in  $\Pi_{\mathbf{G}}$ , it is possible to recognize some properties common to all strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal differentiated partial  $\mathcal{L}$ -Kvaluations, and derive specific soundness and completeness results with respect to the set of all  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations satisfying these properties. The following proposition is particularly useful for this purpose.

Notation 6.2.19. Given a labelled formula of the form  $\mathbf{f}:\varphi$ , we denote by  $\overline{\mathbf{f}:\varphi}$  the labelled formula  $\mathbf{t}:\varphi$ . Similarly,  $\overline{\mathbf{t}:\varphi}$  denotes the labelled formula  $\mathbf{f}:\varphi$ .

**Proposition 6.2.20.** Let v be a partial  $\mathcal{L}$ -Kvaluation, and  $\pi_1, \pi_2, \pi_3$  context-relations.

- 1. Suppose that  $\bar{\pi}_3 = \bar{\pi}_1 \cup \bar{\pi}_2$ . Then  $R^v_{\pi_3} = R^v_{\pi_2} \cap R^v_{\pi_1}$ . In particular, if  $\bar{\pi}_1 \subseteq \bar{\pi}_2$  then  $R^v_{\pi_2} \subseteq R^v_{\pi_1}$ .
- 2. Suppose that for every labelled  $Dom_v$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi}_3 \alpha_1$  then there exists  $\alpha' \in Dom_v$  such that  $\alpha_2 \bar{\pi}_1 \alpha'$  and  $\alpha' \bar{\pi}_2 \alpha_1$ . Then  $R^v_{\pi_2} \circ R^v_{\pi_1} \subseteq R^v_{\pi_3}$ . In particular, if for every labelled  $Dom_v$ -formulas  $\alpha_1$  and  $\alpha_2$ ,  $\alpha_2 \bar{\pi}_1 \alpha_1$  implies that there exists a labelled  $Dom_v$ -formula  $\alpha'$  such that  $\alpha_2 \bar{\pi}_1 \alpha'$  and  $\alpha' \bar{\pi}_1 \alpha_1$ , then  $R^v_{\pi_1}$  is a transitive relation.

<sup>&</sup>lt;sup>3</sup>Given two relations  $R_1, R_2 \subseteq A^2, aR_1 \circ R_2b$  if there exists some  $c \in A$  such that  $aR_1c$  and  $cR_2b$ .

- 3. Assume that v is normal. If  $\overline{\alpha_1}\overline{\pi_1}\overline{\alpha_2}$  whenever  $\alpha_2\overline{\pi_2}\alpha_1$ , then  $R_{\pi_1}^v \subseteq (R_{\pi_2}^v)^{-1}$ . In particular, (i) if  $\alpha_1\overline{\pi_1}\alpha_2$  implies  $\overline{\alpha_2}\overline{\pi_2}\overline{\alpha_1}$  and vice-versa, then  $R_{\pi_2}^v = (R_{\pi_1}^v)^{-1}$ , and (ii) if  $\alpha_1\overline{\pi_1}\alpha_2$  implies  $\overline{\alpha_2}\overline{\pi_1}\overline{\alpha_1}$  and vice-versa, then  $R_{\pi_1}^v$  is a symmetric relation.
- Proof. 1. Suppose first that  $w_1 R_{\pi_3}^v w_2$ . By definition, this means that for every labelled  $Dom_v$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi}_3 \alpha_1$  and  $v, w_2 \models \alpha_2$  then  $v, w_1 \models \alpha_1$ . Since  $\bar{\pi}_1 \subseteq \bar{\pi}_3$ , this implies that for every labelled  $Dom_v$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi}_1 \alpha_1$  and  $v, w_2 \models \alpha_2$  then  $v, w_1 \models \alpha_1$ . Hence,  $w_1 R_{\pi_1}^v w_2$ . Similarly,  $w_1 R_{\pi_2}^v w_2$ . For the converse, suppose that  $w_1 R_{\pi_1}^v w_2$  and  $w_1 R_{\pi_2}^v w_2$ . By definition, this means that for every labelled  $Dom_v$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi}_1 \alpha_1$  or  $\alpha_2 \bar{\pi}_2 \alpha_1$ , and  $v, w_2 \models \alpha_2$  then  $v, w_1 \models \alpha_1$ . Since  $\bar{\pi}_3 \subseteq \bar{\pi}_1 \cup \bar{\pi}_2$ , this implies that for every labelled  $Dom_v$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi}_3 \alpha_1$  and  $v, w_2 \models \alpha_2$  then  $v, w_1 \models \alpha_1$ . Hence,  $w_1 R_{\pi_3}^v w_2$ .
  - 2. Let  $w_1, w_2 \in W_v$  such that  $w_1 R_{\pi_2}^v \circ R_{\pi_1}^v w_2$ . Then there exists  $w' \in W_v$ , such that  $w_1 R_{\pi_2}^v w'$  and  $w' R_{\pi_1}^v w_2$ . We show that  $w_1 R_{\pi_3}^v w_2$ . Let  $\alpha_1, \alpha_2$  be labelled  $Dom_v$ -formulas, such that  $\alpha_2 \bar{\pi}_3 \alpha_1$ , and  $v, w_2 \models \alpha_2$ . Therefore, there exists a labelled  $Dom_v$ -formula  $\alpha'$  such that  $\alpha_2 \bar{\pi}_1 \alpha'$  and  $\alpha' \bar{\pi}_2 \alpha_1$ . Since  $w' R_{\pi_1}^v w_2$ , we have  $v, w' \models \alpha'$ . Since  $w_1 R_{\pi_2}^v w'$ , we have  $v, w_1 \models \alpha_1$ .
  - 3. Let  $wR_{\pi_1}^vw'$ . We show that  $w'R_{\pi_2}^vw$ . Let  $\alpha_1, \alpha_2$  be labelled  $Dom_v$ -formulas, such that  $\alpha_2\overline{\pi}_2\alpha_1$  and  $v, w \models \alpha_2$ . This implies that  $\overline{\alpha}_1\overline{\pi}_1\overline{\alpha}_2$ . Now, since v is normal and  $v, w \models \alpha_2$ , we have that  $v, w \not\models \overline{\alpha}_2$ . Since  $wR_{\pi_1}^vw'$ , we have  $v, w' \not\models \overline{\alpha}_1$ . Since v is normal, this entails that  $v, w' \models \alpha_1$ .

The following soundness and completeness results are easily obtained using Proposition 6.2.20:

Corollary 6.2.21. Let **G** be a basic  $\mathcal{L}$ -calculus, and  $\rho$  an  $\mathcal{L}$ -proof-specification. Suppose that  $\bar{\pi} = \bar{\pi} \circ \bar{\pi}$  for some  $\pi \in \Pi_{\mathbf{G}}$ . Let  $\mathcal{K}$  be the set of all  $\rho$ -legal partial  $\mathcal{L}$ -Kvaluations v for which there exists a  $\langle \mathbf{G}, W_v \rangle$ -coupling  $\Re$ , such that  $\Re(\pi)$  is a transitive relation, and  $\langle v, \Re \rangle$  is **G**-legal. Then,  $\vdash_{\mathbf{G}} = \vdash_{\mathcal{K}}$ .

Proof. Clearly,  $\mathcal{K}$  is a set of  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations. By Theorem 6.2.17 it suffices to show that  $\mathcal{K}$  contains all strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations. Let v be a strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluation. Then  $\langle v, \Re_{\mathbf{G}}^v \rangle$  is  $\mathbf{G}$ -legal. By Proposition 6.2.20 (Item 2),  $\Re_{\mathbf{G}}^v(\pi) = R_{\pi}^v$  is transitive. It follows that  $v \in \mathcal{K}$ .

Corollary 6.2.22. Let **G** be a basic  $\mathcal{L}$ -calculus, and  $\rho$  an  $\mathcal{L}$ -proof-specification. Suppose that for some  $\pi \in \Pi_{\mathbf{G}}$ ,  $\bar{\pi}$  includes only pairs of the form  $\langle \alpha, \alpha \rangle$ . Let  $\mathcal{K}$  be the set of all  $\rho$ -legal partial  $\mathcal{L}$ -Kvaluations v for which there exists a  $\langle \mathbf{G}, W_v \rangle$ -coupling  $\Re$ , such that  $\Re(\pi)$  is a reflexive relation, and  $\langle v, \Re \rangle$  is **G**-legal. Then,  $\vdash_{\mathbf{G}} = \vdash_{\mathcal{K}}$ .

Proof. As in the proof of Corollary 6.2.21, it suffices to show that  $\mathcal{K}$  contains all strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluations. Let v be a strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluation. Then  $\langle v, \mathfrak{R}^v_{\mathbf{G}} \rangle$  is  $\mathbf{G}$ -legal. Proposition 6.2.20 (Item 1) entails that  $R^v_{\pi_0} \subseteq R^v_{\pi}$ . Since  $Id_v \subseteq R^v_{\pi_0}, \mathfrak{R}^v_{\mathbf{G}}(\pi) = R^v_{\pi}$  is reflexive. It follows that  $v \in \mathcal{K}$ .

Semantic characterizations of  $\leq$ -analyticity, cut-admissibility, and axiom-expansion in basic calculi follow from Theorem 6.2.17. This is the topic of Section 6.3. To end this section, we prove a useful property of differentiated  $\rho$ -legal partial  $\mathcal{L}$ -Kvaluations:

**Proposition 6.2.23.** Let v be a differentiated  $\langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ -legal partial  $\mathcal{L}$ -Kvaluation.

- 1. If  $v(w,\varphi) = v(w',\varphi)$  for every  $\varphi \in \mathcal{F}$  then w = w'.
- $2. |W_v| \le 2^{|\mathcal{F} \cap \mathcal{C} \cap \mathcal{A}|} \cdot 3^{|\mathcal{F} \cap \overline{\mathcal{C}} \cap \mathcal{A}| + |\mathcal{F} \cap \mathcal{C} \cap \overline{\mathcal{A}}|} \cdot 4^{|\mathcal{F} \cap \overline{\mathcal{C}} \cap \overline{\mathcal{A}}|}.$

*Proof.* 2 directly follows from 1 by counting the number of possible functions from  $\mathcal{F}$  to  $2^{\{f,t\}}$ , that can be used in an  $\langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ -legal partial  $\mathcal{L}$ -Kvaluation. For 1, suppose that  $v(w,\varphi) = v(w',\varphi)$  for every  $\varphi \in \mathcal{F}$ . It follows that  $wR_{\pi_0}^v w'$ . Since v is differentiated,  $R_{\pi_0}^v = Id_v$ , and so w = w'.

Together with Theorem 6.2.17, the last proposition makes it possible to have a semantic decision procedure for deciding whether  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  given a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$ , an  $\mathcal{L}$ -proof-specification  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  with finite  $\mathcal{F}, \mathcal{C}$  and  $\mathcal{A}$ , finite set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents, and a single  $\mathcal{L}$ -sequent s. Indeed, it is possible to check all functions of the form  $v: W \times \mathcal{F} \to 2^{\{\mathbf{f}, \mathbf{t}\}}$ , where |W| is bounded according to the last proposition. Theorem 6.2.17 and the last proposition entail that  $\mathcal{S} \not\vdash_{\mathbf{G} \upharpoonright \rho} s$  iff one of these functions is a strongly  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Kvaluation, which is a model of  $\mathcal{S}$  but not of s. In this case the semantics is effective, leading to a counter-model search procedure. (Note that a syntactic decision procedure for this problem is trivial, as one can simply construct and check one-by-one all possible proof candidates.) It follows that we have a semantic decision procedure to decide whether  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  for strongly  $\leq$ -analytic basic calculus  $\mathbf{G}$  (where  $\leq$  is safe, see Definition 2.3.1). Indeed, in this case  $\mathcal{S} \vdash_{\mathbf{G}} s$  iff  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  for  $\rho = \langle \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \downarrow^{\leq} [\mathcal{S} \cup \{s\}] \rangle$ .

### 6.2.1 Examples

In this section we provide various examples of applications of Corollary 6.2.18, by applying it to some of the basic calculi presented above. In particular, many fundamental soundness and completeness theorems for known logics and calculi are easily obtained as special cases.

**Example 6.2.24** (LJ). Using Corollary 6.2.18, we are able to obtain a sound and complete semantics for LJ, which is practically identical to the usual Kripke semantics for intuitionistic logic. For this purpose, let  $\mathcal{K}_{LJ}$  be the set of normal  $\mathcal{L}_{LJ}$ -Kvaluations v that respect the usual truth tables of  $\wedge, \vee, \perp$  in each world (where obviously,  $\{f\}$  and  $\{t\}$  are identified with false and true), and in addition there exists a partial order  $\leq$  on  $W_v$  satisfying the following conditions:

(persistence) If  $v(w,\varphi) = \{t\}$  then  $v(w',\varphi) = \{t\}$  for every  $w' \ge w$ .

(implication) 
$$v(w, \varphi_1 \supset \varphi_2) = \{t\}$$
 iff  $v(w', \varphi_1) = \{f\}$  or  $v(w', \varphi_2) = \{t\}$  for every  $w' \geq w$ .

We show that (1)  $\mathcal{K}_{LJ}$  is a set of normal LJ-legal  $\mathcal{L}_{LJ}$ -Kvaluations, and (2)  $\mathcal{K}_{LJ}$  contains all normal strongly LJ-legal differentiated  $\mathcal{L}_{LJ}$ -Kvaluations. Corollary 6.2.18 implies then that  $\vdash_{LJ} = \vdash_{\mathcal{K}_{LJ}}$ .

- 1. Let  $v \in \mathcal{K}_{\mathbf{LJ}}$ , and let  $\leq$  be a partial order on  $W_v$  satisfying (persistence) and (implication). Recall that  $\Pi_{\mathbf{LJ}} = \{\pi_0, \pi_{\mathsf{int}}\}$ . Choose  $\Re$  to be the  $\langle \mathbf{LJ}, W_v \rangle$ -coupling assigning  $Id_v$  to  $\pi_0$ , and  $\leq$  to  $\pi_{\mathsf{int}}$ . Clearly,  $\langle v, \Re \rangle$  is  $\pi_0$ -legal. By Example 6.2.7 (Item 2), condition (persistence) ensures that  $\langle v, \Re \rangle$  is  $\pi_{\mathsf{int}}$ -legal. It is straightforward to show that  $\langle v, \Re \rangle$  is  $\mathsf{R}_{\mathsf{LJ}}$ -legal. For example, following Example 6.2.10 (Item 3), (implication) above immediately implies that  $\langle v, \Re \rangle$  is  $(\mathsf{t:} \supset)$ -legal.

Now, what happens if we simply apply Theorem 6.2.12 for **LJ** (perhaps without knowing about the usual Kripke semantics for intuitionistic logic)? In this case, we obtain that

LJ is sound and complete for the set of normal LJ-legal  $\mathcal{L}_{LJ}$ -Kvaluations. This set can be defined exactly like  $\mathcal{K}_{LJ}$  without restricting  $\leq$  to be a partial order. Thus, we obtain a semantics which is less restrictive than the usual one. On the other hand, we can apply Theorem 6.2.16, and obtain that LJ is sound and complete for the set of normal strongly LJ-legal differentiated  $\mathcal{L}_{LJ}$ -Kvaluations. This set is a subset of  $\mathcal{K}_{LJ}$  obtained by imposing also the converse of (persistence) (if  $v(w, \varphi) = \{t\}$  implies  $v(w', \varphi) = \{t\}$  for every  $\varphi$ , then  $w \leq w'$ ). Here we obtain a more restrictive semantics than the usual one.

**Example 6.2.25** (**BLJ**). Using Corollary 6.2.18, we obtain a sound and complete semantics for **BLJ**, which is practically the same as the usual Kripke semantics for bi-intuitionistic logic (see, e.g., [59]). For this purpose, let  $\mathcal{K}_{\mathbf{BLJ}}$  be the set of all normal  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluations v satisfying the conditions from Example 6.2.24, and the following additional condition:

(exclusion) 
$$v(w, \varphi_1 \prec \varphi_2) = \{f\}$$
 iff  $v(w', \varphi_1) = \{f\}$  or  $v(w', \varphi_2) = \{t\}$  for every  $w' \leq w$ .

Now,  $\mathcal{K}_{\mathbf{BLJ}}$  is a set of  $\mathbf{BLJ}$ -legal  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluations, that contains all normal strongly  $\mathbf{BLJ}$ -legal differentiated  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluations, and so  $\vdash_{\mathbf{BLJ}}=\vdash_{\mathcal{K}_{\mathbf{BLJ}}}$  by Corollary 6.2.18. This is shown similarly as for  $\mathbf{LJ}$ . In particular, the rules of  $\prec$  correspond to (exclusion), and Proposition 6.2.20 (Item 3) entails that in strongly  $\mathbf{BLJ}$ -legal  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluations  $R_{\pi_d}^v = (R_{\pi_{\mathrm{int}}}^v)^{-1}$  (since  $\alpha_1 \bar{\pi}_{\mathrm{d}} \alpha_2$  iff  $\overline{\alpha_2} \pi_{\mathrm{int}} \overline{\alpha_1}$ ).

**Example 6.2.26** (PLJ). Using Corollary 6.2.18, PLJ is sound and complete with respect to the set  $\mathcal{K}$  of normal  $\mathcal{L}_{PLJ}$ -Kvaluations v satisfying the conditions from Example 6.2.24 (ignoring the condition involving  $\bot$ ), and the following two conditions:

- If  $v(w, \varphi) = \{f\}$  then  $v(w, \neg \varphi) = \{t\}$ .
- If  $v(w, \varphi_1) = \{t\}$  and  $v(w, \neg \varphi_2) = \{t\}$  then  $v(w, \neg(\varphi_1 \supset \varphi_2)) = \{t\}$ .

To see this it suffices to show that K is a set of **PLJ**-legal  $\mathcal{L}_{\mathbf{PLJ}}$ -Kvaluations containing all normal strongly **PLJ**-legal differentiated  $\mathcal{L}_{\mathbf{PLJ}}$ -Kvaluations. This is done straightforwardly. Clearly, this semantics is non-deterministic, as the truth values of  $\varphi$  in every world may not determine the truth values of  $\neg \varphi$ . For example, in an  $\mathcal{L}_{\mathbf{PLJ}}$ -Kvaluation with a single world w, if  $v(w, p_1) = \{t\}$ , then  $v(w, \neg p_1)$  can be either  $\{t\}$  or  $\{f\}$ . Note that this semantics is different from the (three-valued) semantics given in [8] for this system.

**Example 6.2.27** (**K**). The usual Kripke semantics of the modal logic K can be described using the set  $\mathcal{K}_{\mathbf{K}}$  of normal  $\mathcal{L}_{\square}$ -Kvaluations defined as follows:  $v \in \mathcal{K}_{\mathbf{K}}$  iff v respects the usual truth tables of the classical connectives in each world, and there exists a binary relation R on  $W_v$  such that the following condition holds:

(necessity)  $v(w, \Box \varphi) = \{t\}$  iff  $v(w', \varphi) = \{t\}$  for every  $w' \in R[w]$ .

Now Corollary 6.2.18 implies that  $\vdash_{\mathbf{K}} = \vdash_{\mathcal{K}_{\mathbf{K}}}$ . To see this, we prove that  $\mathcal{K}_{\mathbf{K}}$  is a set of normal  $\mathbf{K}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations that contains all normal strongly  $\mathbf{K}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations:

- 1. Let  $v \in \mathcal{K}_{\mathbf{K}}$ , and let R be a relation on  $W_v$  satisfying (necessity). Choose  $\Re$  to be the  $\langle \mathbf{K}, W_v \rangle$ -coupling assigning  $Id_v$  to  $\pi_0$ , and R to  $\pi_{\mathbf{K}}$ . Following Example 6.2.7 (Items 1 and 3),  $\langle v, \Re \rangle$  is  $\Pi_{\mathbf{K}}$ -legal. It remains to show that  $\langle v, \Re \rangle$  is  $R_{\mathbf{K}}$ -legal. We show it here only for the rule (K). Following Example 6.2.10 (Item 1), it suffices to see that for every  $w \in W_v$  and formula  $\varphi$ : if  $v(w', \varphi) = \{t\}$  for every  $w' \in R[w]$ , then  $v(w, \Box \varphi) = \{t\}$ . This follows from the definition of  $\mathcal{K}_{\mathbf{K}}$ .
- 2. Let v be a normal strongly **K**-legal Kvaluation. It is easy to show that v respects the usual truth tables of the classical connectives in each world. We claim that  $R^v_{\pi_{K4}}$  is a relation satisfying (necessity). To see this, note that since  $\langle v, \Re^v_{\mathbf{K}} \rangle$  is (K)-legal, we have that if  $v(w', \varphi) = \{\mathbf{t}\}$  for every  $w' \in R^v_{\pi_{K4}}[w]$ , then  $v(w, \Box \varphi) = \{\mathbf{t}\}$ . The converse is obtained from the fact that (by definition)  $w_1 R^v_{\pi_{K4}} w_2$  iff  $v(w_2, \varphi) = \{\mathbf{f}\}$  implies  $v(w_1, \Box \varphi) = \{\mathbf{f}\}$  for every  $\varphi \in \mathcal{L}_{\Box}$ .

**Example 6.2.28** (Systems for modal logics). The usual Kripke semantics of the modal logics K4, KD, KB, S4 and S5 can be described as variations on the set  $\mathcal{K}_{\mathbf{K}}$  (from Example 6.2.27), obtained by imposing an additional requirement on R:

- $\mathcal{K}_{\mathbf{K4}} R$  is transitive.
- $\mathcal{K}_{KD} R$  is serial.
- $\mathcal{K}_{KB} R$  is symmetric.

- $\mathcal{K}_{\mathbf{S4}} R$  is reflexive and transitive.
- $\mathcal{K}_{S5}$  R is an equivalence relation.

For every  $G \in \{K4, KD, KB, S4, S5\}$ , Corollary 6.2.18 implies that  $\vdash_{\mathbf{G}} = \vdash_{\mathcal{K}_{\mathbf{G}}}$ . Indeed, we prove that in each of these cases  $\mathcal{K}_{\mathbf{G}}$  is a set of  $\mathbf{G}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations that contains all strongly  $\mathbf{G}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations. Let  $\mathbf{G} \in \{\mathbf{K4}, \mathbf{KD}, \mathbf{KB}\}$  (the proofs for  $\mathbf{S4}$  and  $\mathbf{S5}$  are similar and left for the reader).

- 1. Let  $v \in \mathcal{K}_{\mathbf{G}}$ , and R be a relation on  $W_v$  satisfying (necessity) and the additional condition of  $\mathcal{K}_{\mathbf{G}}$  (transitivity, seriality, or symmetry). Choose  $\Re$  to be the  $\langle \mathbf{G}, W_v \rangle$ -coupling assigning  $Id_v$  to  $\pi_0$ , and R to the other context relation in  $\Pi_{\mathbf{G}}$  ( $\pi_{\mathsf{K4}}, \pi_{\mathsf{K}}$ , or  $\pi_{\mathsf{B}}$ ). We show that v is  $\mathbf{G}$ -legal:
  - **K4** As in Example 6.2.27,  $\langle v, \Re \rangle$  is  $\pi_0$ -legal and  $\mathsf{R}_{\mathbf{K4}}$ -legal. It remains to show that  $R \subseteq R^v_{\pi_{\mathsf{K4}}}$  (and so  $\langle v, \Re \rangle$  is  $\pi_{\mathsf{K4}}$ -legal). Suppose that  $w_1 R w_2$ . Let  $\alpha_1$  and  $\alpha_2$  be labelled  $\mathcal{L}_{\square}$ -formulas such that  $\alpha_2 \pi_{\mathsf{K4}} \alpha_1$  and  $v, w_2 \models \alpha_2$ . Then,  $\alpha_2 = \mathbf{f} : \varphi$  and  $\alpha_1 = \mathbf{f} : \square \varphi$ , or  $\alpha_2 = \alpha_1 = \mathbf{f} : \square \varphi$  (for some formula  $\varphi$ ). In

the first case, (necessity) directly implies that  $v, w_1 \models \alpha_1$ . Suppose now that  $\alpha_2 = \alpha_1 = \mathbf{f} : \Box \varphi$  for some formula  $\varphi$ . Since  $v, w_2 \models \alpha_2$  (i.e.,  $v(w_2, \Box \varphi) = \{\mathbf{f}\}$ ), (necessity) entails that  $v, w \models \mathbf{f} : \varphi$  (i.e.,  $v(w, \varphi) = \{\mathbf{f}\}$ ) for some  $w \in R[w_2]$ . The transitivity of R then ensures that  $w_1 R w$ . Again (necessity) implies that  $v, w_1 \models \alpha_1$ . It follows that  $w_1 R_{\pi_{\mathsf{K}4}}^v w_2$ .

- **KD** As in Example 6.2.27,  $\langle v, \Re \rangle$  is  $\Pi_{\mathbf{KD}}$ -legal, and  $\mathsf{R}_{\mathbf{KD}} \setminus \{(D)\}$ -legal. In addition, following Example 6.2.10 (Item 4), the seriality of R ensures that  $\langle v, \Re \rangle$  is (D)-legal. Therefore v is **KD**-legal.
- KB As in Example 6.2.27,  $\langle v, \Re \rangle$  is  $\pi_0$ -legal and  $R_{KB}$ -legal. It remains to show that  $R \subseteq R_{\pi_B}^v$  (and so  $\langle v, \Re \rangle$  is also  $\pi_B$ -legal). Suppose that  $w_1 R w_2$ . Let  $\alpha_1$  and  $\alpha_2$  be labelled  $\mathcal{L}_{\square}$ -formulas such that  $\alpha_2 \overline{\pi} \alpha_1$  and  $v, w_2 \models \alpha_2$ . Then,  $\alpha_2 = \mathbf{f} : \varphi$  and  $\alpha_1 = \mathbf{f} : \square \varphi$ , or  $\alpha_2 = \mathbf{t} : \square \varphi$  and  $\alpha_1 = \mathbf{t} : \varphi$  (for some formula  $\varphi$ ). In the first case, (necessity) directly implies that  $v, w_1 \models \alpha_1$ . Suppose now that  $\alpha_2 = \mathbf{t} : \square \varphi$  and  $\alpha_1 = \mathbf{t} : \varphi$ . Since  $v, w_2 \models \alpha_2$  (i.e.,  $v(w_2, \square \varphi) = \{\mathbf{t}\}$ ), (necessity) entails  $v, w \models \alpha_1$  for every  $w \in R[w_2]$ . The symmetry of R ensures that  $w_2 R w_1$ , and so  $v, w_1 \models \alpha_1$ . It follows that  $w_1 R_{\pi_B}^v w_2$ .
- 2. Let v be a normal strongly G-legal  $\mathcal{L}_{\square}$ -Kvaluation. Similarly to Example 6.2.27, one shows that  $v \in \mathcal{K}_{\mathbf{K}}$ . In addition:
  - **K4** Since  $\pi_{K4} \subseteq \pi_{K4} \circ \pi_{K4}$ , Proposition 6.2.20 (Item 2) entails that  $R_{\pi_{K4}}^v$  is transitive.
  - **KD** Since  $\langle v, \Re_{\mathbf{KD}}^v \rangle$  is (D)-legal,  $R_{\pi_{\mathsf{D}}}^v$  is serial (see Example 6.2.10, Item 4).
  - **KB** Since  $\alpha_1 \overline{\pi}_B \alpha_2$  iff  $\overline{\alpha}_2 \overline{\pi}_B \overline{\alpha}_1$ , Proposition 6.2.20 (Item 3) entails that  $R_{\pi_B}^v$  is symmetric.

Example 6.2.29 (GL). Semantically, the modal logic GL is characterized by the set of Kripke frames whose accessibility relation is transitive and conversely well-founded. However, GL is not strongly complete with respect to models built on this set of frames (i.e. we have  $\Vdash_{GL} \varphi$  iff every such frame is a model of  $\varphi$ , but we do not have  $\mathcal{T} \Vdash_{GL} \varphi$  iff every such frame which is a model of  $\mathcal{T}$  is a model of  $\varphi$ , see [94]). Using our method, starting from the basic calculus  $\mathbf{GL}$ , we obtain a (different) strongly sound and complete semantics for GL. Indeed, by Corollary 6.2.18,  $\mathbf{GL}$  is (strongly) sound and complete with respect to the set  $\mathcal{K}_{\mathbf{GL}}$  of normal  $\mathcal{L}_{\square}$ -Kvaluations, defined similarly to  $\mathcal{K}_{\mathbf{K}}$  (see Example 6.2.27), with two additional requirements: (1) R is transitive; and (2) If  $v(w', \varphi) = \{\mathbf{f}\}$  for some  $w' \in R[w]$ , then there is some  $w'' \in R[w]$  such that  $v(w'', \varphi) = \{\mathbf{f}\}$  and  $v(w'', \square \varphi) = \{\mathbf{t}\}$ . To see this we prove that  $\mathcal{K}_{\mathbf{GL}}$  is a set of  $\mathbf{GL}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations that contains all normal strongly  $\mathbf{GL}$ -legal  $\mathcal{L}_{\square}$ -Kvaluations:

- 1. Let  $v \in \mathcal{K}_{GL}$ , and let R be a transitive relation on  $W_v$ , satisfying (necessity) and condition (2) above. Choose  $\Re$  to be the  $\langle \mathbf{GL}, W_v \rangle$ -coupling assigning  $Id_v$  to  $\pi_0$ , and R to  $\pi_{\mathsf{K4}}$ . Similarly to Example 6.2.28, one proves that  $R \subseteq R^v_{\pi_{\mathsf{K4}}}$  (using the transitivity of R), and so  $\langle v, \Re \rangle$  is  $\pi_{\mathsf{K4}}$ -legal. It remains to show that  $\langle v, \Re \rangle$  is  $\mathsf{R}_{\mathsf{GL}}$ -legal. We show it here for (GL). Let  $w \in W_v$ , and let  $\sigma$  be an  $\mathcal{L}_{\square}$ -substitution. Suppose that  $v, \Re(\pi_{\mathsf{K4}})[w] \models \sigma(\{\mathbf{f}: \square p_1, \mathbf{t}: p_1\})$ . We show that  $v, w \models \sigma(\{\mathbf{t}: \square p_1\})$ . Assume otherwise. Then  $v(w, \square \sigma(p_1)) = \{\mathbf{f}\}$ . Thus (necessity) implies that there exists some  $w' \in R[w]$ , such that  $v(w', \sigma(p_1)) = \{\mathbf{f}\}$ . By condition (2), there is some  $w'' \in R[w]$  such that  $v(w'', \sigma(p_1)) = \{\mathbf{f}\}$  and  $v(w'', \square \sigma(p_1)) = \{\mathbf{t}\}$ . Clearly,  $v, w'' \not\models \sigma(\{\mathbf{f}: \square p_1, \mathbf{t}: p_1\})$ . But, since  $\Re(\pi_{\mathsf{K4}}) = R$ , this contradicts the fact that  $v, \Re(\pi_{\mathsf{K4}})[w] \models \sigma(\{\mathbf{f}: \square p_1, \mathbf{t}: p_1\})$ .
- 2. Let v be a normal strongly  $\operatorname{GL}$ -legal  $\mathcal{L}_{\square}$ -Kvaluation. It is straightforward to show that v respects the usual truth tables of the classical connectives in each world. We show that there exists a transitive relation R on  $W_v$  satisfying (necessity) and condition (2) above. We show that  $R_{\pi_{\mathsf{K}4}}^v$  has this property (its transitivity is proved exactly as in Example 6.2.28):
  - (a) Since  $\langle v, \Re_{\mathbf{GL}}^v \rangle$  is (GL)-legal, if  $v(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_{\mathsf{K4}}}^v[w]$ , then  $v(w, \Box \varphi) = \{t\}$ . The converse holds since  $\langle v, \Re_{\mathbf{GL}}^v \rangle$  is  $\pi_{\mathsf{K4}}$ -legal.
  - (b) We prove that  $R_{\pi_{\mathsf{K4}}}^v$  satisfies condition (2) above. Suppose (for contradiction) that there exist some  $\varphi \in \mathcal{L}_{\square}$  and  $w \in W_v$ , such that  $v(w', \varphi) = \{\mathbf{f}\}$  for some  $w' \in R_{\pi_{\mathsf{K4}}}^v[w]$ , and there does not exist  $w'' \in R_{\pi_{\mathsf{K4}}}^v[w]$ , such that  $v(w'', \varphi) = \{\mathbf{f}\}$  and  $v(w'', \square \varphi) = \{\mathbf{t}\}$ . It follows that  $v, \Re(\pi_{\mathsf{K4}})[w] \models \{\mathbf{f} : \square \varphi, \mathbf{t} : \varphi\}$ . Since  $\langle v, \Re \rangle$  is (GL)-legal,  $v(w, \square \varphi) = \{\mathbf{t}\}$ . But, this contradicts (necessity).

We note it is not clear whether this semantics for GL is useful (in particular, whether it leads to a decision procedure). This question is left open for a future work.

**Example 6.2.30** ( $\mathbf{GP}^Q$ ). Using Corollary 6.2.18, we obtain a sound and complete semantics for  $\mathbf{GP}^Q$ , which is practically identical to the semantics presented in [31]. For this purpose, let  $\mathcal{K}$  be the set of  $\mathcal{L}_{\mathbf{GP}}^Q$ -Kvaluations v that respect the usual truth tables of  $\wedge, \vee, \perp, \top$  in each world, and satisfy the following conditions:

- 1. If  $v(w, \varphi_1) = \{t\}$  and  $v(w, \varphi_2) = \{f\}$  then  $v(w, \varphi_1 \supset \varphi_2) = \{f\}$ .
- 2. If  $v(w, \varphi_2) = \{t\}$  then  $v(w, \varphi_1 \supset \varphi_2) = \{t\}$ .
- 3. For every  $q \in Q$ , there exist binary relations,  $S^q$  and  $I^q$ , on  $W_v$ , satisfying:
  - (a)  $I^q \subseteq S^q$ .
  - (b)  $v(w, q \text{ said } \varphi) = \{t\} \text{ iff } v(w', \varphi) = \{t\} \text{ for every } w' \in S^q[w].$

(c) 
$$v(w, q \text{ implied } \varphi) = \{t\} \text{ iff } v(w', \varphi) = \{t\} \text{ for every } w' \in I^q[w].$$

Clearly, this semantics is non-deterministic, as the truth values of  $\varphi_1$  and  $\varphi_2$  in every world may not determine the value of  $\varphi_1 \supset \varphi_2$ . As in previous examples, it is straightforward to show that  $\mathcal{K}$  is a set of normal  $\mathbf{GP}^Q$ -legal  $\mathcal{L}_{\mathbf{GP}}^Q$ -Kvaluations, that contains all normal strongly  $\mathbf{GP}^Q$ -legal  $\mathcal{L}_{\mathbf{GP}}^Q$ -Kvaluations (the fact that in all strongly  $\mathbf{GP}^Q$ -legal  $\mathcal{L}_{\mathbf{GP}}^Q$ -Kvaluations  $\Re(\pi_i^q) \subseteq \Re(\pi_s^q)$  for every  $q \in Q$  follows from Proposition 6.2.20, Item 1).

**Example 6.2.31** (**IS5**). Using Corollary 6.2.18, we obtain a sound and complete Kripke semantics for **IS5**. For this purpose, let  $\mathcal{K}$  be the set of normal  $\mathcal{L}_{\mathbf{LJ}}^{\square}$ -Kvaluations v satisfying the conditions from Example 6.2.24, and in addition, there exists an equivalence relation  $\sim$ , such that  $v(w, \Box \varphi) = \{\mathbf{t}\}$  iff  $v(w', \varphi) = \{\mathbf{t}\}$  for every  $w' \in \sim [w]$ . (Note that if  $v \in \mathcal{K}$ , then for every  $w, w' \in W_v$ , we have that if  $w \leq w'$  and  $v(w'', \varphi) = \{\mathbf{t}\}$  for every  $w'' \in \sim [w]$ , then  $v(w'', \varphi) = \{\mathbf{t}\}$  for every  $w'' \in \sim [w']$ .) As in previous examples, it is straightforward to show that  $\mathcal{K}$  is a set of normal **IS5**-legal  $\mathcal{L}_{\mathbf{LJ}}^{\square}$ -Kvaluations, that contains all normal strongly **IS5**-legal  $\mathcal{L}_{\mathbf{LJ}}^{\square}$ -Kvaluations. Interestingly, the Kripke semantics presented in [78] is not identical to this one. In particular, in our semantics  $\sim$  should be an equivalence relation, and no direct conditions bind  $\leq$  and  $\sim$ .

# 6.3 Characterization of Proof-Theoretic Properties

In this section we use Theorem 6.2.17 to derive characterizations of strong  $\leq$ -analyticity, strong cut-admissibility, and axiom-expansion in basic calculi. First, note that the soundness part of Theorem 6.2.17 can be utilized for providing relatively simple semantic arguments for the failure of certain proof-theoretic properties. For example, by exhibiting an  $\langle \mathcal{L}, \emptyset, \mathcal{L} \rangle$ -legal **G**-legal  $\mathcal{L}$ -Kvaluation which is not a model of some sequent s, we show that every proof of s requires to use (cut). If we also have  $\vdash_{\mathbf{G}} s$ , then it follows that **G** does not enjoy cut-admissibility. Similarly, by using the other two components of the proof-specification, we can show that a certain calculus **G** is not  $\leq$ -analytic (for some  $\leq$ ) or that some connective does not admit axiom-expansion in **G**. Note that proving facts of this kind using proof-theoretic methods is sometimes very challenging! Next we provide some concrete examples of such applications.

**Example 6.3.1.** Let s be the  $\mathcal{L}_{\mathbf{BLJ}}$ -sequent  $\{\mathbf{f}:p_1, \mathbf{t}:p_2, \mathbf{t}:p_1 \supset (p_1 \prec p_2)\}$ . We show that  $\not\vdash_{\mathbf{BLJ}} \rho$  s for  $\rho = \langle \mathcal{L}_{\mathbf{BLJ}}, \emptyset, \mathcal{L}_{\mathbf{BLJ}} \rangle$  (i.e., there does not exist a proof of s in  $\mathbf{BLJ}$  without cuts). By Theorem 6.2.17, it suffices to find a  $\mathbf{BLJ} \upharpoonright_{\rho}$ -legal  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluation which is not a model of s. Let v be an  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluation defined by  $W_v = \{w_1, w_2\}, v(w, \varphi) = \{\mathbf{f}, \mathbf{t}\}$  for every  $w \in W_v$  and  $\mathcal{L}_{\mathbf{BLJ}}$ -formula  $\varphi$  except for:  $v(w_1, p_1) = v(w_2, p_1) = v(w_2, p_2) = \{\mathbf{t}\}$ ,

and  $v(w_1, p_2) = v(w_1, p_1 \supset (p_1 \prec p_2)) = v(w_2, p_1 \prec p_2) = \{f\}$ . Let  $\Re$  be the  $\langle \mathbf{BLJ}, W_v \rangle$ coupling defined by  $\Re(\pi_0) = \{\langle w_2, w_2 \rangle\}$ ,  $\Re(\pi_{\mathsf{int}}) = \{\langle w_1, w_2 \rangle\}$ , and  $\Re(\pi_{\mathsf{d}}) = \emptyset$ . One can
straightforwardly verify that  $\langle v, \Re \rangle$  is  $\mathbf{BLJ}$ -legal, and clearly,  $v \not\models s$ . However, it is easy
to find a proof for s in  $\mathbf{BLJ}$ , and thus  $\vdash_{\mathbf{BLJ}} s$ . This provides a semantic demonstration of
the fact that  $\mathbf{BLJ}$  does not enjoy cut-admissibility (the sequent s is a simplified version
of the one used in [80] to syntactically prove this fact).

**Example 6.3.2.** It is well-known that **S5** does not enjoy cut-admissibility. We provide a semantic demonstration of this fact. Let s be the  $\mathcal{L}_{\square}$ -sequent  $\{t:p_1, t: \square \neg \square p_1\}$ . It is easy to see that s is provable in **S5** (using a cut on  $\square p_1$ ). Now, let  $\rho = \langle \mathcal{L}_{\square}, \{p_1, \square \neg \square p_1\}, \mathcal{L}_{\square} \rangle$ . We show that  $\not\vdash_{\mathbf{S5} \upharpoonright \rho} s$  (and so, in particular, there does exist a cut-free proof of s). Let v be an  $\mathcal{L}_{\square}$ -Kvaluation defined by  $W_v = \{w_1, w_2\}, v(w, \varphi) = \{f, t\}$  for every  $w \in W_v$  and  $\mathcal{L}_{\square}$ -formula  $\varphi$  except for:

$$v(w_2, p_1) = v(w_2, \Box p_1) = \{t\},\$$
  
 $v(w_1, p_1) = v(w_1, \Box \neg \Box p_1) = v(w_2, \neg \Box p_1) = v(w_2, \Box \neg \Box p_1) = \{f\}.$ 

Clearly, v is  $\rho$ -legal and  $v \not\models s$ . Let  $\Re$  be the  $\langle \mathbf{S5}, W_v \rangle$ -coupling with  $\Re(\pi_0) = \{\langle w_2, w_2 \rangle\}$  and  $\Re(\pi_{S5}) = \{\langle w_1, w_2 \rangle, \langle w_2, w_2 \rangle\}$ . One can straightforwardly verify that  $\langle v, \Re \rangle$  is S5-legal, and thus v is S5-legal. For example:

- $\langle v, \Re \rangle$  is  $\pi_{S5}$ -legal since the following conditions are met: (1) if  $w\Re(\pi_{S5})w'$  and  $\mathbf{t} \in v(w', \Box \varphi)$  then  $\mathbf{t} \in v(w, \Box \varphi)$ ; (2) if  $w\Re(\pi_{S5})w'$  and  $\mathbf{f} \in v(w', \Box \varphi)$  then  $\mathbf{f} \in v(w, \Box \varphi)$ .
- $\langle v, \Re \rangle$  is (S5)-legal since the following condition is met: if  $\mathsf{t} \in v(w', \varphi)$  for every  $w' \in \Re(\pi_{\mathsf{S5}})[w]$ , then  $\mathsf{t} \in v(w, \Box \varphi)$ .
- $\langle v, \Re \rangle$  is (T)-legal since the following condition is met: if  $\mathbf{f} \in v(w', \varphi)$  for every  $u \in \Re(\pi_0)[w]$ , then  $\mathbf{f} \in v(w, \Box \varphi)$ .

**Example 6.3.3. PLJ** is not sub-analytic. This is shown in [8], by proving that the  $\mathcal{L}_{PLJ}$ -sequent  $s = \{t: p_1, t: p_2 \supset \neg(p_2 \supset p_1)\}$  is provable, but every proof of it must include a formula that does not occur in sub[s]. Using Theorem 6.2.17, we can provide a semantic demonstration of this fact. Let  $\rho = \langle sub[s], \mathcal{L}_{PLJ}, \mathcal{L}_{PLJ} \rangle$ . Consider the  $\rho$ -legal partial  $\mathcal{L}_{PLJ}$ -Kvaluation v, defined by  $W_v = \{w_1, w_2\}$ , and:

```
\begin{split} v(w_1,p_1) &= v(w_1,p_2) = v(w_1,\neg(p_2\supset p_1)) = v(w_1,p_2\supset \neg(p_2\supset p_1)) = \{\mathtt{f}\},\\ v(w_1,p_2\supset p_1) &= \{\mathtt{t}\},\\ v(w_2,\neg(p_2\supset p_1)) &= v(w_2,p_2\supset \neg(p_2\supset p_1)) = \{\mathtt{f}\},\\ v(w_2,p_1) &= v(w_2,p_2) = v(w_2,p_2\supset p_1) = \{\mathtt{t}\}. \end{split}
```

Let  $\Re$  be the  $\langle \mathbf{PLJ}, W_v \rangle$ -coupling defined by  $\Re(\pi_{\mathsf{int}}) = \{\langle w_1, w_1 \rangle, \langle w_2, w_2 \rangle, \langle w_1, w_2 \rangle\}$  and  $\Re(\pi_0) = Id_v$ . It is straightforward to show that  $\langle v, \Re \rangle$  is  $\mathbf{PLJ}$ -legal, and so v is  $\mathbf{PLJ} \upharpoonright_{\rho}$ -legal. Clearly,  $v \not\models s$ . By Theorem 6.2.17,  $\not\vdash_{\mathbf{PLJ} \upharpoonright_{\rho}} s$ . In other words, there does not exist a proof of s in  $\mathbf{PLJ}$  consisting solely of sub[s]-sequents.

Next, we present characterizations of strong  $\leq$ -analyticity, strong cut-admissibility, and axiom-expansion in basic calculi, that may be used to prove these properties. For simplicity of the presentation, we shall not discuss the weak versions of  $\leq$ -analyticity and cut-admissibility, but note that they are always implied by the strong property (by takings  $S = \emptyset$ ). We use the following additional notion:

**Definition 6.3.4.** An instance of a partial  $\mathcal{L}$ -Kvaluation v is a normal (full)  $\mathcal{L}$ -Kvaluation v' such that  $W_{v'} = W_v$ , and  $v'(w, \varphi) \subseteq v(w, \varphi)$  for every  $w \in W_v$  and  $\varphi \in Dom_v$ .

The following proposition immediately follows from the definitions.

**Proposition 6.3.5.** Let v be a partial  $\mathcal{L}$ -Kvaluation, and let v' be an instance of v. Then, for every  $Dom_v$ -sequent s: if  $v' \models s$  then  $v \models s$ . If v is  $\langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ -legal and  $frm[s] \subseteq \mathcal{C}$ , the converse holds as well.

The following characterization of strong  $\leq$ -analyticity follows from the previous results.

Corollary 6.3.6. A basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  is strongly  $\leq$ -analytic iff for every set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents and  $\mathcal{L}$ -sequent s,  $\mathcal{S} \vdash_{\mathcal{K}_1} s$  implies  $\mathcal{S} \vdash_{\mathcal{K}_2} s$ , where  $\mathcal{K}_1$  is the set of all normal  $\mathbf{G}$ -legal  $\mathcal{L}$ -Kvaluations, and  $\mathcal{K}_2$  is the set of all normal strongly  $\mathbf{G}$ -legal partial  $\mathcal{L}$ -Kvaluations whose domain is  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ .

*Proof.* Suppose that **G** is strongly  $\leq$ -analytic. Assume that  $\mathcal{S} \vdash_{\mathcal{K}_1} s$  for some set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents and  $\mathcal{L}$ -sequent s. By Corollary 6.2.18,  $\mathcal{S} \vdash_{\mathbf{G}} s$ , and so  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  for  $\rho = \langle \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \mathcal{L}, \mathcal{L} \rangle$ . Note that  $\rho = \langle \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \mathcal{L}, \mathcal{L} \rangle$ -legal partial  $\mathcal{L}$ -Kvaluations are exactly normal partial  $\mathcal{L}$ -Kvaluations whose domain is  $\downarrow^{\leq} [\mathcal{S} \cup \{s\}]$ . Therefore, by Theorem 6.2.17,  $\mathcal{S} \vdash_{\mathcal{K}_2} s$ .

For the converse, suppose that  $\mathcal{S} \vdash_{\mathbf{G}} s$ . By Corollary 6.2.18,  $\mathcal{S} \vdash_{\mathcal{K}_1} s$ , and so our assumption entails that  $\mathcal{S} \vdash_{\mathcal{K}_2} s$ . Theorem 6.2.17 again entails that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  for  $\rho = \langle \downarrow^{\leq} [\mathcal{S} \cup \{s\}], \mathcal{L}, \mathcal{L} \rangle$ .

The above characterization might be quite complicated to be used in practice. Therefore, we now present a simpler semantic criterion, that turns out to be useful for many basic calculi.

Corollary 6.3.7. Let G be a basic  $\mathcal{L}$ -calculus. Suppose that every normal strongly G-legal partial  $\mathcal{L}$ -Kvaluation whose domain is closed under  $\leq$  has a G-legal instance. Then G is strongly  $\leq$ -analytic.

Proof. We use Corollary 6.3.6. Let S be a set of L-sequents, and s a single L-sequent. Let  $K_1$  and  $K_2$  be defined as in Corollary 6.3.6. Assume that  $S \vdash_{K_1} s$ . We prove that  $S \vdash_{K_2} s$ . Let  $v \in K_2$ , and suppose that  $v \models S$ . Since  $Dom_v = \downarrow^{\leq} [S \cup \{s\}]$  is closed under  $\leq$ , our assumption entails that there exists a G-legal instance v' of v. Thus  $v' \in K_1$ . By Proposition 6.3.5, we have  $v' \models S$ . Since  $S \vdash_{K_1} s$ , we have  $v' \models s$ . Proposition 6.3.5 entails that  $v \models s$ .

Before turning to some examples of applying the criterion above, we present a characterization of strong cut-admissibility. Its proof is similar to the proof of Corollary 6.3.6.

**Definition 6.3.8.** Given a set  $C \subseteq \mathcal{L}$ , an  $\mathcal{L}$ -Kvaluation is called C-cut-restricted if it is  $\langle \mathcal{L}, \mathcal{C}, \mathcal{L} \rangle$ -legal. An  $\mathcal{L}$ -Kvaluation is called cut-restricted if it is  $\emptyset$ -cut-restricted.

Corollary 6.3.9. A basic  $\mathcal{L}$ -calculus G enjoys strong cut-admissibility iff for every set  $\mathcal{S}$  of  $\mathcal{L}$ -sequents and  $\mathcal{L}$ -sequent s,  $\mathcal{S} \vdash_{\mathcal{K}_1} s$  implies  $\mathcal{S} \vdash_{\mathcal{K}_2} s$ , where  $\mathcal{K}_1$  is the set of all normal G-legal  $\mathcal{L}$ -Kvaluations, and  $\mathcal{K}_2$  is the set of all  $frm[\mathcal{S}]$ -cut-restricted strongly G-legal  $\mathcal{L}$ -Kvaluations.

Again, the following provides a simpler *sufficient* criterion:

Corollary 6.3.10. Let G be a basic  $\mathcal{L}$ -calculus. Suppose that every cut-restricted strongly G-legal  $\mathcal{L}$ -Kvaluation has a G-legal instance. Then G enjoys strong cut-admissibility.

*Proof.* We use Corollary 6.3.9. Let  $\mathcal{S}$  be a set of  $\mathcal{L}$ -sequents, and s a single  $\mathcal{L}$ -sequent. Let  $\mathcal{K}_1$  and  $\mathcal{K}_2$  be defined as in Corollary 6.3.9. Assume that  $\mathcal{S} \vdash_{\mathcal{K}_1} s$ . We prove that  $\mathcal{S} \vdash_{\mathcal{K}_2} s$ . Let  $v \in \mathcal{K}_2$ , and suppose that  $v \models \mathcal{S}$ . Since v is cut-restricted, our assumption entails that there exists a  $\mathbf{G}$ -legal instance v' of v. Thus  $v' \in \mathcal{K}_1$ . By Proposition 6.3.5, we have  $v' \models \mathcal{S}$  (since v' is  $\langle \mathcal{L}, frm[\mathcal{S}], \mathcal{L} \rangle$ -legal). Since  $\mathcal{S} \vdash_{\mathcal{K}_1} s$ , we have  $v' \models s$ . Proposition 6.3.5 entails that  $v \models s$ .

Next we apply the previous criteria to prove strong  $\leq$ -analyticity and/or strong cutadmissibility for some of the basic calculi presented in the examples above.

**Example 6.3.11.** We use Corollary 6.3.10 to show that **LJ** enjoys strong cut-admissibility. Let v be a cut-restricted strongly **LJ**-legal  $\mathcal{L}_{\mathbf{LJ}}$ -Kvaluation. We recursively construct an instance v' of v. For every  $w \in W_v$  and for every atomic formula p,  $v'(w,p) = \{x\}$  if  $v(w,p) = \{x\}$ , and otherwise  $v'(w,p) = \{t\}$  (say). Now suppose that  $v'(w,\varphi_1)$  and  $v'(w,\varphi_2)$  were defined for every  $w \in W_v$ :

- $v'(w, \varphi_1 \supset \varphi_2)$  is defined by: if  $v(w, \varphi_1 \supset \varphi_2) = \{x\}$  then  $v'(w, \varphi_1 \supset \varphi_2) = \{x\}$ . Otherwise  $v'(w, \varphi_1 \supset \varphi_2) = \{t\}$  iff for every  $w' \in R^v_{\pi_{\text{int}}}[w]$ , either  $v'(w', \varphi_1) = \{t\}$  or  $v'(w', \varphi_2) = \{t\}$ .
- $v'(w, \varphi_1 \wedge \varphi_2)$  is defined by: if  $v(w, \varphi_1 \wedge \varphi_2) = \{x\}$  then  $v'(w, \varphi_1 \wedge \varphi_2) = \{x\}$ . Otherwise  $v'(w, \varphi_1 \wedge \varphi_2) = \{t\}$  iff  $v'(w, \varphi_1) = \{t\}$  and  $v'(w, \varphi_2) = \{t\}$ . Similar definitions are used for the other connectives of LJ.

Clearly, v' is an instance of v. Based on the fact that v is a cut-restricted strongly **LJ**-legal  $\mathcal{L}_{LJ}$ -Kvaluation, it is easy to prove that  $\langle v', \Re_{LJ}^v \rangle$  is **LJ**-legal (and so v' is **LJ**-legal).

**Example 6.3.12.** While **BLJ** does not enjoy cut-admissibility (Example 6.3.1), we use Corollary 6.3.7 to show that it is still strongly sub-analytic. This answers a question raised in [80].<sup>4</sup> Let v be a normal strongly **BLJ**-legal partial  $\mathcal{L}_{BLJ}$ -Kvaluation, whose domain is closed under subformulas. A construction of an instance v' of v, is done as in Example 6.3.11 with the following addition:

• If  $v(w, \varphi_1 \prec \varphi_2) = \{x\}$  then  $v'(w, \varphi_1 \prec \varphi_2) = \{x\}$ . Otherwise  $v'(w, \varphi_1 \prec \varphi_2) = \{f\}$  iff  $v'(w', \varphi_1) = \{f\}$  or  $v'(w', \varphi_2) = \{t\}$  for every  $w' \in R^v_{\pi_d}[w]$ .

Clearly, v' is an instance of v. Based on the facts that v is a normal strongly **BLJ**-legal partial  $\mathcal{L}_{\mathbf{BLJ}}$ -Kvaluation, and that  $Dom_v$  is closed under subformulas, it is straightforward to prove that  $\langle v', \Re_{\mathbf{BLJ}}^v \rangle$  is **BLJ**-legal.

**Example 6.3.13.** Following Example 6.3.3, **PLJ** is not sub-analytic. As a substitute, a weaker property is proved for this system in [8] (called the n-subformula property). Roughly speaking, this property means that whenever a sequent s is provable, there also exists a proof of s that includes only formulas from sub[s] and some of their negations. To be more precise, it is equivalent to strong nsub-analyticity, where nsub is the transitive closure of the union of the relation sub and

$$\{\langle \neg \varphi_i, \neg (\varphi_1 \diamond \varphi_2) \rangle \mid \varphi_1, \varphi_2 \in \mathcal{L}_{\mathbf{PLJ}}, \diamond \in \{\land, \lor, \supset\}, i = 1, 2\}.$$

Note that nsub is safe, and so strong nsub-analyticity suffices to establish decidability. Next, we prove strong nsub-analyticity for **PLJ** using Corollary 6.3.7. Let v be a normal strongly **PLJ**-legal partial  $\mathcal{L}_{\mathbf{PLJ}}$ -Kvaluation, whose domain is closed under subformulas. A construction of an instance v' of v is done as in Example 6.3.11 with the following addition:  $v'(w, \neg \varphi) = \{x\}$  if  $v(w, \neg \varphi) = \{x\}$ , and  $v'(w, \neg \varphi) = \{t\}$  otherwise. Clearly, v' is an instance of v. Based on the fact that v is a normal strongly **PLJ**-legal partial

<sup>&</sup>lt;sup>4</sup> Note that other systems for this logic, that enjoy cut-admissibility, were devised in [59] and [80]. However, these systems do not employ the standard notion of a sequent used in Gentzen-type systems, but more complicated data-structures.

 $\mathcal{L}_{\mathbf{PLJ}}$ -Kvaluation, we show that v' is  $\mathbf{PLJ}$ -legal, since  $\langle v', \Re^v_{\mathbf{PLJ}} \rangle$  is  $\mathbf{PLJ}$ -legal. To see that  $\langle v', \Re^v_{\mathbf{PLJ}} \rangle$  is  $\pi_{\mathsf{int}}$ -legal, it suffices to note that for every  $\varphi \in \mathcal{L}_{\mathbf{PLJ}}$ , if  $v'(w, \varphi) = \{t\}$  then  $v'(w', \varphi) = \{t\}$  for every  $w' \in \Re(\pi_{\mathsf{int}})[w]$ . We claim that  $\langle v', \Re^v_{\mathbf{PLJ}} \rangle$  is  $\mathsf{R}_{\mathbf{PLJ}}$ -legal. We demonstrate it here only for the rule  $(t:\neg \supset)$ . Thus we show that if  $v'(w, \varphi_1) = \{t\}$  and  $v'(w, \neg \varphi_2) = \{t\}$  then  $v'(w, \neg(\varphi_1 \supset \varphi_2)) = \{t\}$ . Assume that  $v'(w, \neg(\varphi_1 \supset \varphi_2)) = \{f\}$ . Our construction then ensures that  $\neg(\varphi_1 \supset \varphi_2) \in Dom_v$ , and  $v(w, \neg(\varphi_1 \supset \varphi_2)) = \{f\}$  as well. Since  $\{\varphi_1, \neg \varphi_2\} \subseteq \downarrow^{nsub}[\neg(\varphi_1 \supset \varphi_2)]$  and  $Dom_v$  is closed under nsub, we have that  $\{\varphi_1, \neg \varphi_2\} \subseteq Dom_v$ . Since  $\langle v', \Re^v_{\mathbf{PLJ}} \rangle$  is  $(t:\neg \supset)$ -legal, either  $v(w, \varphi_1) = \{f\}$  or  $v(w, \neg \varphi_2) = \{f\}$ . By our construction,  $v'(w, \varphi_1) = \{f\}$  or  $v'(w, \neg \varphi_2) = \{f\}$ .

Example 6.3.14. Each of the four basic calculi K, K4, KD, and S4 admits the semantic criterion given in Corollary 6.3.10 (and so they all enjoy strong cut-admissibility). To see this, let G be any one of these calculi, and v a cut-restricted strongly G-legal  $\mathcal{L}_{\square}$ -Kvaluation. We recursively construct an instance v' of v. For every  $w \in W_v$  and for every atomic formula p,  $v'(w,p) = \{x\}$  if  $v(w,p) = \{x\}$ , and otherwise  $v'(w,p) = \{t\}$  (say). Now suppose that  $v'(w,\varphi_1)$  and  $v'(w,\varphi_2)$  were defined, define  $v'(w,\varphi_1 \supset \varphi_2)$  as follows (similar definitions for the other classical connectives): if  $v(w,\varphi_1 \supset \varphi_2) = \{x\}$  then  $v'(w,\varphi_1 \supset \varphi_2) = \{x\}$ , and otherwise  $v'(w,\varphi_1 \supset \varphi_2) = \{t\}$  iff either  $v'(w,\varphi_1) = \{t\}$  or  $v'(w,\varphi_2) = \{t\}$ . In addition,  $v'(w,\square\varphi) = \{x\}$  if  $v(w,\square\varphi) = \{x\}$ , and otherwise we set  $v'(w,\square\varphi) = \{t\}$  iff  $v'(w',\varphi) = \{t\}$  for every  $v' \in R_{\pi}^v[w]$  (where  $\pi$  is the context-relation in  $\Pi_G$ , that is not  $\pi_0$ ). Clearly, v' is an instance of v. Using the fact that v is a cut-restricted strongly G-legal  $\mathcal{L}_{\square}$ -Kvaluation, it is easy to show that  $\langle v', \Re_{G}^v \rangle$  is G-legal.

Example 6.3.15. While KB and S5 do not enjoy cut-admissibility (for S5, see Example 6.3.2), Corollary 6.3.7 can be used to show that they are still strongly sub-analytic. We demonstrate it here for KB. Let v be a normal strongly KB-legal partial  $\mathcal{L}_{\square}$ -Kvaluation, whose domain is closed under subformulas. A construction of an instance v' of v is done exactly as in Example 6.3.14. We show that v' is indeed a KB-legal Kvaluation, as  $\langle v', \Re_{\mathbf{KB}}^v \rangle$  is  $\mathbf{KB}$ -legal. To see that  $\langle v', \Re_{\mathbf{KB}}^v \rangle$  is  $\pi_{\mathsf{B}}$ -legal, we show that  $R_{\pi_{\mathsf{B}}}^v \subseteq R_{\pi_{\mathsf{B}}}^{v'}$ . Suppose that  $w_1 R_{\pi_{\mathsf{B}}}^v w_2$ . Note that by Proposition 6.2.20 (Item 3), we have that  $w_2 R_{\pi_B}^v w_1$  (because of the structure of  $\pi_B$ ). We prove that  $w_1 R_{\pi_B}^{v'} w_2$ . Let  $\alpha_1$  and  $\alpha_2$ be labelled  $\mathcal{L}_{\square}$ -formulas such that  $\alpha_2 \bar{\pi}_B \alpha_1$  and  $v, w_2 \models \alpha_2$ . The structure of  $\pi_B$  ensures that there exists some  $\varphi \in \mathcal{L}_{\square}$  such that either  $\alpha_2 = \mathbf{f}:\varphi$  and  $\alpha_1 = \mathbf{f}:\square\varphi$ , or  $\alpha_2 = \mathbf{t}:\square\varphi$ and  $\alpha_1 = \mathbf{t}: \varphi$ . If  $\Box \varphi \in Dom_v$  then  $\alpha_1$  and  $\alpha_2$  are labelled  $Dom_v$ -formulas (since  $Dom_v$ is closed under subformulas). In this case, since  $w_1 R_{\pi_B}^v w_2$ , we have that  $v, w_1 \models \alpha_1$ , and we are done. Otherwise, for every  $w \in W_v$ ,  $v'(w, \Box \varphi) = \{t\}$  iff  $v'(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_R}^v[w]$ . Now, if  $\alpha_2 = \mathbf{f}:\varphi$  and  $\alpha_1 = \mathbf{f}:\Box\varphi$ , then  $v, w_2 \models \alpha_2$  directly entails that  $v, w_1 \models \alpha_1$ . Otherwise,  $\alpha_2 = t: \Box \varphi$  and  $\alpha_1 = t: \varphi$ . It follows that  $v'(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_{\mathsf{R}}}^{v}[w_2]$ . Since  $w_2 R_{\pi_{\mathsf{R}}}^{v} w_1$ ,  $v, w_1 \models \alpha_1$  in this case as well. Finally, we claim that  $\langle v', \Re_{\mathbf{KB}}^v \rangle$  is  $\mathsf{R}_{\mathbf{KB}}$ -legal. We show it here only for the rule (B). Following Example 6.2.10 (Item 1), we should prove that for every  $w \in W_v$  and  $\mathcal{L}_{\square}$ -formula  $\varphi$ : if  $v'(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_{\mathsf{B}}}^v[w]$ , then  $v'(w, \square \varphi) = \{t\}$ . Let  $w \in W_v$ , and  $\varphi \in \mathcal{L}_{\square}$ . Suppose that  $v'(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_{\mathsf{B}}}^v[w]$ . If  $\square \varphi \not\in Dom_v$ , then the construction of v' directly entails that  $v'(w, \square \varphi) = \{t\}$ . Otherwise,  $\varphi \in Dom_v$  as well, and the construction of v' entails that  $v(w', \varphi) = \{t\}$  for every  $w' \in R_{\pi_{\mathsf{B}}}^v[w]$ . Since v is strongly  $\mathsf{KB}$ -legal,  $\langle v, \Re_{\mathsf{KB}}^v \rangle$  is (B)-legal. Thus we have that  $\mathsf{t} \in v(w, \square \varphi)$  (see Definition 6.2.9, Item 1). It then follows that  $v'(w, \square \varphi) = \{\mathsf{t}\}$ .

**Example 6.3.16.** Using the semantic criterion of Corollary 6.3.10, it is easy to see that  $\mathbf{GP}^Q$  enjoys strong cut-admissibility. The construction of a  $\mathbf{GP}^Q$ -legal instance for every cut-restricted strongly  $\mathbf{GP}^Q$ -legal  $\mathcal{L}_{\mathbf{GP}}^Q$ -Kvaluation is done as for  $\mathbf{K}$  (see Example 6.3.14), with straightforward modifications for q said and q implied. In addition we replace the  $\{\mathbf{f},\mathbf{t}\}$  values assigned to formulas of the form  $\varphi_1 \supset \varphi_2$  by the value assigned to  $\varphi_2$  in each world.

**Example 6.3.17. IS5** does not enjoy cut-admissibility, since the  $\mathcal{L}_{LJ}^{\square}$ -sequent

$$s = \{ \mathbf{f} : \Box(\Box p_1 \lor p_2), \mathbf{t} : \Box p_1, \mathbf{t} : (\Box p_2 \supset \bot) \supset \bot \}$$

is provable, but not cut-free provable (see [78]). Using Theorem 6.2.12, one can semantically verify that there is no cut-free proof of s, by constructing an **IS5**-legal  $\langle \mathcal{L}_{\mathbf{LJ}}^{\square}, \emptyset, \mathcal{L}_{\mathbf{LJ}}^{\square} \rangle$ -legal  $\mathcal{L}_{\mathbf{LJ}}^{\square}$ -Kvaluation which is not a model of it. In addition, the condition for strong sub-analyticity given in Corollary 6.3.7 does not hold for **IS5**. Since this condition is only proven to be sufficient, it does not mean that **IS5** is not strongly sub-analytic, and this question remains open.

Finally, Theorem 6.2.17 also naturally leads to the following semantic characterization of axiom-expansion.

Corollary 6.3.18. Let  $\diamond \in \Diamond_{\mathcal{L}}$ , and let  $\varphi = \diamond(p_1, \dots, p_{ar(\diamond)})$ .  $\diamond$  admits axiom-expansion in a basic  $\mathcal{L}$ -calculus  $\mathbf{G}$  iff every  $\langle \mathcal{L}, \mathcal{L}, \{p_1, \dots, p_{ar(\diamond)}\} \rangle$ -legal strongly  $\mathbf{G}$ -legal  $\mathcal{L}$ -Kvaluation is also  $\langle \mathcal{L}, \mathcal{L}, \{\varphi\} \rangle$ -legal.

Proof. We prove one direction. The converse is similar. Assume that  $\diamond$  admits axiom-expansion in  $\mathbf{G}$ . By definition,  $\vdash_{\mathbf{G}\upharpoonright\rho} \{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}$  for  $\rho = \langle \mathcal{L},\mathcal{L},\{p_1,\ldots,p_{ar(\diamond)}\}\rangle$ . Theorem 6.2.17 entails that every strongly  $\mathbf{G}\upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Kvaluation is a model of  $\{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}$ . It follows that in every strongly  $\mathbf{G}\upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Kvaluation  $v,v(w,\varphi)\neq\emptyset$  for every  $w\in W_v$ . Thus, every strongly  $\mathbf{G}\upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Kvaluation is  $\langle \mathcal{L},\mathcal{L},\{\varphi\}\rangle$ -legal.

**Example 6.3.19** (LJ). Using the criterion given in Corollary 6.3.18, it is straightforward to prove that every connective of  $\mathcal{L}_{LJ}$  admits axiom-expansion in LJ. We do it here for

 $\supset$ . Let v be an  $\langle \mathcal{L}, \mathcal{L}, \{p_1, p_2\} \rangle$ -legal strongly  $\mathbf{LJ}$ -legal  $\mathcal{L}_{\mathbf{LJ}}$ -Kvaluation. We show that  $v(w, p_1 \supset p_2) \neq \emptyset$  for every  $w \in W_v$ , and so v is  $\langle \mathcal{L}, \mathcal{L}, \{p_1 \supset p_2\} \rangle$ -legal. Suppose that  $\mathsf{t} \not\in v(w, p_1 \supset p_2)$  for some  $w \in W_v$ . Since  $\langle v, \Re^v_{\mathbf{LJ}} \rangle$  is  $(\mathsf{f} : \supset)$ -legal,  $\mathsf{f} \not\in v(w', p_1)$  and  $\mathsf{t} \not\in v(w', p_2)$  for some  $w' \in \Re^v_{\mathbf{LJ}}(\pi_{\mathsf{int}})[w]$ . Since v is  $\langle \mathcal{L}, \mathcal{L}, \{p_1, p_2\} \rangle$ -legal,  $v(w', p_1) \neq \emptyset$  and  $v(w', p_2) \neq \emptyset$ . This entails that  $v(w', p_1) = \{\mathsf{t}\}$  and  $v(w', p_2) = \{\mathsf{f}\}$ . Since  $\langle v, \Re^v_{\mathbf{LJ}} \rangle$  is  $(\mathsf{f} : \supset)$ -legal and  $\Re^v_{\mathbf{LJ}}(\pi_0) = Id_v$ , we have that  $\mathsf{f} \in v(w', p_1 \supset p_2)$ . Since  $\langle v, \Re^v_{\mathbf{LJ}} \rangle$  is  $\pi_{\mathsf{int}}$ -legal,  $\mathsf{f} \in v(w, p_1 \supset p_2)$  as well.

# 6.4 Soundness and Completeness Proofs

This section is devoted to prove Theorem 6.2.17. Theorems 6.2.12 and 6.2.16 are immediately obtained as special cases. Let G be a basic  $\mathcal{L}$ -calculus, and  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  an  $\mathcal{L}$ -proof-specification. Clearly, to show that  $\vdash_{\mathbf{G} \upharpoonright \rho} = \vdash_{\mathcal{K}}$  for every set  $\mathcal{K}$  of partial  $\mathcal{L}$ -Kvaluations satisfying  $\mathcal{K}^*_{\mathbf{G} \upharpoonright \rho} \subseteq \mathcal{K} \subseteq \mathcal{K}_{\mathbf{G} \upharpoonright \rho}$ , it suffices to prove the following:

Soundness 
$$\vdash_{\mathbf{G} \upharpoonright \rho} \subseteq \vdash_{\mathcal{K}_{\mathbf{G} \upharpoonright \rho}}$$
.

Completeness 
$$\vdash_{\mathcal{K}^*_{\mathbf{G} \upharpoonright \rho}} \subseteq \vdash_{\mathbf{G} \upharpoonright \rho}$$
.

#### Soundness

For the soundness proof we use the following simple lemmas.

**Lemma 6.4.1.** Let v be a partial  $\mathcal{L}$ -Kvaluation, let  $w \in W_v$ , and let  $s_1$  and  $s_2$  be two  $\mathcal{L}$ -sequents. Then,  $v, w \models s_1 \cup s_2$  iff either  $v, w \models s_1$  or  $v, w \models s_2$ .

**Lemma 6.4.2.** Let  $\langle s, s' \rangle$  be a  $\pi$ -instance for some  $\mathcal{L}$ -context-relation  $\pi$ . Let v be partial  $\mathcal{L}$ -Kvaluation, and let  $w \in W_v$ . Suppose that  $v, w' \models s$  for some  $w' \in R_{\pi}^v[w]$ . Then either  $frm[s'] \not\subseteq Dom_v$  or  $v, w \models s'$ .

Proof. Suppose that  $frm[s'] \subseteq Dom_v$ , we show that  $v, w \models s'$ . Since  $v, w' \models s$ , we have  $v, w' \models \alpha_2$  for some  $\alpha_2 \in s$ . Since  $\langle s, s' \rangle$  is a  $\pi$ -instance, there exists  $\alpha_1 \in s'$  such that  $\alpha_2 \overline{\pi} \alpha_1$ . Note that  $frm[\alpha_2] \in Dom_v$  (because  $v, w' \models \alpha_2$ ) and  $frm[\alpha_1] \in Dom_v$  (because  $frm[s'] \subseteq Dom_v$ ). Then since  $wR_{\pi}^v w', v, w \models \alpha_1$ . It follows that  $v, w \models s'$ .

Now, assume that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s_0$ . Thus there exists a  $\rho$ -proof P in  $\mathbf{G}$  of  $s_0$  from  $\mathcal{S}$ . Let  $\mathcal{K} = \mathcal{K}_{\mathbf{G} \upharpoonright \rho}$ . We prove that  $\mathcal{S} \vdash_{\mathcal{K}} s_0$ . Let  $v \in \mathcal{K}$ . Then,  $\langle v, \Re \rangle$  is  $\mathbf{G}$ -legal for some  $\langle \mathbf{G}, W_v \rangle$ -coupling  $\Re$ . Suppose that  $v \models \mathcal{S}$ . Using induction on the length of P, we show that  $v \models s$  for every sequent s appearing in P. It then follows that  $v \models s_0$ . Note first that since v is  $\rho$ -legal,  $Dom_v = \mathcal{F}$ , and so every sequent in P is a  $Dom_v$ -sequent. Thus it suffices to prove that for every sequent s' appearing in P, we have  $v, W_v \models s'$ . This

trivially holds for the sequents of S that appear in P. We show that the property of being true in  $W_v$  is preserved by applications of the rules of G. Consider such an application in P, and assume that  $v, W_v \models s$  for every premise s of this application. We show that its conclusion is also true in  $W_v$ . Let  $w \in W_v$ .

- 1. Suppose that  $c \cup \{\mathbf{x}:\varphi\}$  is derived using from c, using  $(\mathbf{x}:weak)$   $(x \in \mathcal{L}_2)$ . Since  $v, w \models c$ , Lemma 6.4.1 entails that  $v, w \models c \cup \{\mathbf{x}:\varphi\}$ .
- 2. Suppose that  $\{f:\varphi, t:\varphi\}$  is derived using (id). In this case,  $\varphi \in \mathcal{A}$ . Since v is  $\rho$ -legal,  $v(w,\varphi) \neq \emptyset$ . This easily implies that  $v, w \models \{f:\varphi, t:\varphi\}$ .
- 3. Suppose that  $c_1 \cup c_2$  is derived from  $c_1 \cup \{\mathbf{f}:\varphi\}$  and  $c_2 \cup \{\mathbf{t}:\varphi\}$  using (cut). In this case,  $\varphi \in \mathcal{C}$ . Since v is  $\rho$ -legal,  $v(w,\varphi) \neq \{\mathbf{f},\mathbf{t}\}$ . This easily implies that either  $v,w \not\models \{\mathbf{f}:\varphi\}$  or  $v,w \not\models \{\mathbf{t}:\varphi\}$ . Since  $v,w \models c_1 \cup \{\mathbf{f}:\varphi\}$ , Lemma 6.4.1 entails that either  $v,w \models c_1$  or  $v,w \models \{\mathbf{f}:\varphi\}$ . Similarly, either  $v,w \models c_2$  or  $v,w \models \{\mathbf{t}:\varphi\}$ . This entails that either  $v,w \models c_1$  or  $v,w \models c_2$ . Therefore Lemma 6.4.1 entails that  $v,w \models c_1 \cup c_2$ .
- 4. Suppose that  $\sigma(s) \cup c'_1 \cup ... \cup c'_n$  is derived from  $\sigma(s_1) \cup c_1, ..., \sigma(s_n) \cup c_n$  using a basic  $\mathcal{L}$ -rule  $r = \langle s_1, \pi_1 \rangle, ..., \langle s_n, \pi_n \rangle / s'$  in  $R_{\mathbf{G}}$ . Thus  $\langle c_i, c'_i \rangle$  is a  $\pi_i$ -instance for every  $1 \leq i \leq n$ . Now, if  $v, w \models c'_i$  for some  $1 \leq i \leq n$ , then by Lemma 6.4.1,  $v, w \models \sigma(s) \cup c'_1 \cup ... \cup c'_n$ , and we are done. Assume otherwise. We show that  $v, \Re(\pi_i)[w] \models \sigma(s_i)$  for every  $1 \leq i \leq n$ . Let  $1 \leq i \leq n$  and  $w' \in \Re(\pi_i)[w]$ . Since  $\langle v, \Re \rangle$  is  $\pi_i$ -legal, we have that  $wR^v_{\pi_i}w'$ . Now, since  $\langle c_i, c'_i \rangle$  is a  $\pi_i$ -instance, Lemma 6.4.2 entails that  $v, w' \not\models c_i$ . By Lemma 6.4.1 (since we assumed that  $v, w' \models \sigma(s_i) \cup c_i$ ,  $v, w' \models \sigma(s_i)$ . Finally, we have  $v, w \models \sigma(s)$  since  $\langle v, \Re \rangle$  is r-legal and  $frm[\sigma(\{s_1, ..., s_n, s\})] \subseteq Dom_v$ . By Lemma 6.4.1,  $v, w \models \sigma(s) \cup c'_1 \cup ... \cup c'_n$ .

### Completeness

Recall that by extended  $\mathcal{L}$ -sequent we mean a (possibly infinite) set of labelled  $\mathcal{L}$ -formulas (see Definition 3.3.6). For this completeness proof, we call an extended  $\mathcal{L}$ -sequent  $\mu$  provable if  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s$  for some  $\mathcal{L}$ -sequent  $s \subseteq \mu$ . Otherwise, we say that  $\mu$  is unprovable. In addition, we call an extended  $\mathcal{L}$ -sequent  $\mu^*$  maximal if it satisfies the following conditions: (1)  $frm[\mu^*] \subseteq \mathcal{F}$ ; (2)  $\mu^*$  is unprovable; and (3) for every labelled  $\mathcal{F}$ -formula  $\alpha \notin \mu^*$ ,  $\{\alpha\} \cup \mu^*$  is provable. As in Section 3.4, it is straightforward to show that:

(a) For every unprovable extended  $\mathcal{F}$ -sequent  $\mu$ , there is a maximal extended  $\mathcal{L}$ -sequent  $\mu^*$  such that  $\mu \subseteq \mu^*$ .

Now, suppose that  $\mathcal{S} \not\vdash_{\mathbf{G} \upharpoonright \rho} s_0$ . We show that  $\mathcal{S} \not\vdash_{\mathcal{K}^*_{\mathbf{G} \upharpoonright \rho}} s_0$ . Let v be the partial  $\mathcal{L}$ Kvaluation defined by:  $W_v$  is the set of all maximal extended  $\mathcal{L}$ -sequents;  $Dom_v = \mathcal{F}$ ;
and  $v(\mu, \varphi) = \{ \mathbf{x} \in \{ \mathbf{f}, \mathbf{t} \} \mid \mathbf{x} : \varphi \not\in \mu \}$  for every  $\mu \in W_v$  and  $\varphi \in \mathcal{F}$ . We show that  $v \in \mathcal{K}^*_{\mathbf{G} \upharpoonright \rho}$ , and that v is a model of  $\mathcal{S}$  but not of  $s_0$ .

Note first that v is  $\rho$ -legal. By definition  $Dom_v = \mathcal{F}$ . To see that  $v(\mu, \varphi) \neq \{\mathbf{f}, \mathbf{t}\}$  for every  $\varphi \in \mathcal{C}$ , it suffices to prove that if  $\varphi \in \mathcal{C} \cap \mathcal{F}$  then  $\mathbf{f}: \varphi \in \mu$  or  $\mathbf{t}: \varphi \in \mu$ . Assume by way of contradiction that  $\mathbf{f}: \varphi \notin \mu$  and  $\mathbf{t}: \varphi \notin \mu$  for some  $\varphi \in \mathcal{C} \cap \mathcal{F}$ . It follows that there exist  $\mathcal{F}$ -sequents  $s_1, s_2 \subseteq \mu$  such that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s_1 \cup \{\mathbf{f}: \varphi\}$  and  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s_2 \cup \{\mathbf{t}: \varphi\}$ . Since  $\varphi \in \mathcal{C}$ , a (legal) application of (cut) ensured that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s_1 \cup s_2$ . But this contradicts the properties of  $\mu$ . To see that  $v(\mu, \varphi) \neq \emptyset$  for every  $\varphi \in \mathcal{A}$ , it suffices to prove that  $\varphi \in \mathcal{A} \cap \mathcal{F}$  implies that  $\mathbf{f}: \varphi \notin \mu$  or  $\mathbf{t}: \varphi \notin \mu$ . Note that if  $\varphi \in \mathcal{A} \cap \mathcal{F}$  then  $\{\mathbf{f}: \varphi, \mathbf{t}: \varphi\}$  is a (legal) application of (id), and so  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} \{\mathbf{f}: \varphi, \mathbf{t}: \varphi\}$ . Since  $\mu$  is maximal, either  $\mathbf{f}: \varphi \notin \mu$  or  $\mathbf{t}: \varphi \notin \mu$ .

Now, for every  $\pi \in \Pi_{\mathbf{G}}$ , let  $R_{\pi}^{\mathcal{F}}$  denote the binary relation on  $W_v$  defined by:  $\mu_1 R_{\pi}^{\mathcal{F}} \mu_2$  iff for every labelled  $\mathcal{F}$ -formulas  $\alpha_1, \alpha_2$ , if  $\alpha_2 \bar{\pi} \alpha_1$  and  $\alpha_2 \in \mu_2$  then  $\alpha_1 \in \mu_1$ . We claim that the following hold:

(b) For every labelled  $\mathcal{F}$ -formula  $\alpha$  and  $\mu \in W_v$ :  $v, \mu \models \alpha$  iff  $\alpha \notin \mu$ .

*Proof.* Suppose that  $\alpha = \mathbf{x}:\varphi$  where  $\mathbf{x} \in \{\mathbf{f}, \mathbf{t}\}$ . Then,  $\mathbf{x} \in v(\mu, \varphi)$  iff  $\alpha \notin \mu$ . Equivalently,  $v, \mu \models \alpha$  iff  $\alpha \notin \mu$ .

(c)  $R_{\pi}^{\mathcal{F}} = R_{\pi}^{v}$  for every  $\pi \in \Pi_{\mathbf{G}}$ , and  $R_{\pi_{0}}^{\mathcal{F}} = Id_{v}$ .

Proof. For every context-relation  $\pi$ ,  $R_{\pi}^{\mathcal{F}} = R_{\pi}^{v}$  follows from (b). To see that  $R_{\pi_0}^{\mathcal{F}} = Id_v$ , note that  $\alpha_2 \bar{\pi}_0 \alpha_1$  iff  $\alpha_2 = \alpha_1$ . Thus,  $\mu_1 R_{\pi_0}^{\mathcal{F}} \mu_2$  iff for every labelled  $\mathcal{F}$ -formula  $\alpha$ ,  $v, \mu_2 \models \alpha$  implies that  $v, \mu_1 \models \alpha$ . By (b), we obtain that  $\mu_1 R_{\pi_0}^{\mathcal{F}} \mu_2$  iff  $\mu_1 \subseteq \mu_2$ . Therefore, obviously,  $\mu R_{\pi_0}^{\mathcal{F}} \mu$  for every  $\mu \in W_v$ . For the converse, we show that if  $\mu_1, \mu_2 \in W_v$  and  $\mu_1 \subseteq \mu_2$ , then  $\mu_1 = \mu_2$ . Assume (by way of contradiction) that  $\mu_1 \subseteq \mu_2$  and there exists  $\alpha \in \mu_2 \setminus \mu_1$ . Since  $\mu_1$  is maximal, there exists an  $\mathcal{F}$ -sequent  $s \subseteq \mu_1$  such that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s \cup \{\alpha\}$ . But,  $s \cup \{\alpha\} \subseteq \mu_2$ , and this contradicts the fact that  $\mu_2$  is unprovable.

(d) For every  $\mathcal{F}$ -sequent s and  $\mu \in W_v$ :  $s \not\subseteq \mu$  iff  $v, \mu \models s$ .

*Proof.* Easily follows from  $(\mathbf{b})$ .

(e) For every  $\mathcal{F}$ -sequent s and  $\mu \in W_v$ : if there exists an  $\mathcal{F}$ -sequent  $s' \subseteq \mu$  such that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} s \cup s'$ , then  $v, \mu \models s$ .

*Proof.* Assume that there exists a sequent  $s' \subseteq \mu$  such that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s \cup s'$ . Since  $\mu$  is unprovable,  $s \not\subseteq \mu$ . Therefore, (d) entails that  $v, \mu \models s$ .

(f) For every  $\mathcal{F}$ -sequent  $s, \mu \in W_v$ , and  $\pi \in \Pi_{\mathbf{G}}$ : if  $v, R_{\pi}^{\mathcal{F}}[\mu] \models s$ , then there exists a  $\pi$ -instance  $\langle c, c' \rangle$  such that  $c' \subseteq \mu$  and  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s \cup c$ .

Proof. Assume that there is no  $\pi$ -instance  $\langle c, c' \rangle$  such that  $c' \subseteq \mu$  and  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s \cup c$ . We show that  $v, R_{\pi}^{\mathcal{F}}[\mu] \not\models s$ . Let  $\mu^* = \{\alpha \mid frm[\alpha] \in \mathcal{F} \ and \ \exists \beta \in \mu.\alpha \overline{\pi}\beta\}$ . Since  $(\mathbf{f}:weak)$  and  $(\mathbf{t}:weak)$  are available, our assumption implies that  $s \cup \mu^*$  is unprovable. Since  $frm[s \cup \mu^*] \subseteq \mathcal{F}$ , (a) entails that there exists maximal extended  $\mathcal{L}$ -sequent  $\mu'$ , such that  $s \cup \mu^* \subseteq \mu'$ . (d) entails that  $v, \mu' \not\models s$ . By definition,  $\mu R_{\pi}^{\mathcal{F}} \mu'$ . Hence,  $v, R_{\pi}^{\mathcal{F}}[\mu] \not\models s$ .

(g) For every  $\mathcal{L}$ -sequent  $s: v \models s \text{ iff } \mathcal{S} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} s$ .

Proof. Note first that if  $frm[s] \not\subseteq \mathcal{F}$ , then by definition,  $v \not\models s$  and  $\mathcal{S} \not\vdash_{\mathbf{G} \upharpoonright \rho} s$ . Assume now that  $frm[s] \subseteq \mathcal{F}$ . One direction easily follows from (e). For the converse, assume that  $\mathcal{S} \not\vdash_{\mathbf{G} \upharpoonright \rho} s$ . We show that  $v \not\models s$ . Because of the presence of  $(\mathbf{f}:weak)$  and  $(\mathbf{t}:weak)$ , there does not exist  $s' \subseteq s$  such that  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s'$ . By (a), there exists  $\mu \in W_v$ , such that  $s \subseteq \mu$ . (d) entails that  $v, \mu \not\models s$ . Hence,  $v \not\models s$ .  $\square$ 

Next, we show that v is strongly  $\mathbf{G}$ -legal. Thus we prove that  $\langle v, \mathfrak{R}_{\mathbf{G}}^{v} \rangle$  is  $\mathsf{R}_{\mathbf{G}}$ -legal. Let  $r = \langle s_{1}, \pi_{1} \rangle, \ldots, \langle s_{n}, \pi_{n} \rangle / s$  be a rule in  $\mathsf{R}_{\mathbf{G}}$ . Let  $\mu \in W_{v}$ , and let  $\sigma$  be an  $\mathcal{L}$ -substitution. Suppose that  $frm[\sigma(\{s_{1}, \ldots, s_{n}, s\})] \subseteq \mathcal{F}$ , and that  $v, \mathfrak{R}_{\mathbf{G}}^{v}(\pi_{i})[\mu] \models \sigma(s_{i})$  for every  $1 \leq i \leq n$ . We prove that  $v, \mu \models \sigma(s)$ . By  $(\mathbf{c}), \mathfrak{R}_{\mathbf{G}}^{v}(\pi_{i}) = R_{\pi_{i}}^{\mathcal{F}}$  for every  $1 \leq i \leq n$ . Thus  $(\mathbf{f})$  entails that for every  $1 \leq i \leq n$ , there exists a  $\pi$ -instance  $\langle c_{i}, c_{i}' \rangle$  such that  $c'_{i} \subseteq \mu$  and  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} \sigma(s_{i}) \cup c_{i}$ . Now we can use these proofs, and the rule r to obtain  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} \sigma(s) \cup c'_{1} \cup \ldots \cup c'_{n}$ , where  $c'_{1} \cup \ldots \cup c'_{n} \subseteq \mu$ .  $(\mathbf{e})$  entails that  $v, \mu \models \sigma(s)$ .

It follows that  $v \in \mathcal{K}_{\mathbf{G} \upharpoonright \rho}^*$  (note that by  $(\mathbf{c})$ , v is differentiated). Finally, we show that  $v \models \mathcal{S}$  but  $v \not\models s_0$ . Since obviously  $\mathcal{S} \vdash_{\mathbf{G} \upharpoonright \rho} s'$  for every  $\mathcal{F}$ -sequent  $s' \in \mathcal{S}$ ,  $(\mathbf{g})$  implies that  $v \models s'$  for every such s'. Since  $\mathcal{S} \not\vdash_{\mathbf{G} \upharpoonright \rho} s_0$ ,  $(\mathbf{g})$  also implies that  $v \not\models s_0$ .

# Chapter 7

# Canonical Gödel Hypersequent Calculi

Gödel logic, known also as Gödel-Dummett logic, is perhaps the most prominent intermediate logic, and one of the three fundamental fuzzy logics [63]. It was introduced in [48] both semantically, by an infinite-valued matrix, and syntactically, with a simple axiomatization, namely the extension of (an axiomatization of) intuitionistic logic with the axiom scheme  $(\varphi_1 \supset \varphi_2) \lor (\varphi_2 \supset \varphi_1)$  of linearity. The quest for a (cut-free) Gentzen-type formulation for (propositional) Gödel logic began later, and several calculi were proposed (see, e.g., [86, 46, 2, 50, 25, 51]). One of the most important cut-free calculi for Gödel logic is the calculus **HG**, introduced in [4] (see also [22] and [76]). **HG** is relatively simple, especially due to the fact that its logical rules are practically the same rules as in LJ, the well-known single-conclusion sequent calculus for intuitionistic logic. This is obtained by working in the slightly richer framework of (single-conclusion) hypersequents, that provides a natural generalization of Gentzen's original sequents framework.<sup>1</sup> The structural part of **HG** consists of all the usual structural rules, both on the sequent level (internal) and on the hypersequent level (external). In addition, it includes the communication rule that allows "exchange of information" between two hypersequents [7], and, needless to say, the identity axiom and the (admissible) cut rule.

In this chapter we introduce and study the family of canonical Gödel hypersequent calculi of which **HG** is the prototype example. The idea, just like in canonical sequent calculi (following [17], see Chapter 4), is to allow any "ideal" logical rules for introducing the logical connectives. Thus we define canonical single-conclusion hypersequent rules, and in turn, canonical hypersequent Gödel calculi are (two-sided, single-conclusion) hypersequent calculi that include all standard structural rules, the cut rule, the identity

Note that hypersequents are currently the main proof-theoretic framework for fuzzy logics [76].

7.1. Preliminaries 105

axiom, the communication rule, and an arbitrary finite set of canonical single-conclusion hypersequent rules.<sup>2</sup> Then, as in the previous chapters, we study canonical Gödel calculi from a *semantic* point of view. Note that we could similarly study *multiple-conclusion* hypersequent calculi in which some of the rules do not allow right context formulas (as the right rules for implication in the multiple-conclusion calculus for intuitionistic logic). We choose to work in the single-conclusion framework both in order to demonstrate the applicability of our methods in this framework, and because the single-conclusion framework is more common when it comes to Gödel logic.

First and foremost, our study includes a general method to obtain a sound and complete semantics for *every* canonical Gödel calculus. The semantics is based on totally ordered algebraic structures with (possibly) non-deterministic interpretations of the different connectives. Here we also consider the semantic effect of the cut rule and the identity axiom, and obtain semantics for canonical Gödel calculi in which these rules are restricted to apply only on some given set of formulas. This semantics is then used to characterize proof-theoretic properties of canonical Gödel calculi, and particularly to identify the "good" ones, namely those that enjoy (strong) cut-admissibility. In fact, we show that the simple *coherence* criterion of [17, 14] characterizes strong cut-admissibility in canonical Gödel calculi as well.

#### Publications Related to this Chapter

The material in this chapter was included in [71, 69]. Note that canonical *single-conclusion* (two-sided) *sequent* calculi were introduced and studied in the author's M.Sc. thesis (see also [14]).

# 7.1 Preliminaries

As in the previous chapter, in this chapter we only consider two-sided sequents, refer to them as  $\mathcal{L}$ -sequents, and employ the standard notation and abbreviations (see Notation 6.1.1). In turn, hypersequents are defined as follows:

**Definition 7.1.1.** An  $\mathcal{L}$ -hypersequent is a finite set of  $\mathcal{L}$ -sequents. Given a set  $\mathcal{F} \subseteq \mathcal{L}$ , an  $\mathcal{F}$ -hypersequent is a hypersequent consisting solely of  $\mathcal{F}$ -sequents.

<sup>&</sup>lt;sup>2</sup>In fully-structural single-conclusion sequent calculi weakening on the right side can only be applied on a sequent whose right side is empty. Similarly, in fully-structural single-conclusion hypersequent calculi right internal weakening can only be applied on a hypersequent including a component with an empty right-hand side.

**Notation 7.1.2.** We usually use H as a metavariable for  $\mathcal{L}$ -hypersequents. We usually denote an  $\mathcal{L}$ -hypersequent  $\{s_1, \ldots, s_n\}$  by  $s_1 \mid \ldots \mid s_n$ , and employ the standard abbreviations, e.g.  $H_1 \mid H_2$  instead of  $H_1 \cup H_2$ , and  $H \mid s$  instead of  $H \cup \{s\}$ .

**Remark 7.1.3.** As we did for sequents, we defined hypersequents using *sets*. This immediately entails that the external exchange rule, the external contraction rule and the external expansion rule (the converse of contraction) are built-in in all hypersequent calculi that we study.

Unlike previous chapters, we study here *single-conclusion* calculi:

**Definition 7.1.4.** An  $\mathcal{L}$ -sequent s satisfying  $|\{\varphi \in \mathcal{L} \mid \mathbf{t}: \varphi \in s\}| \leq 1$  is called a *single-conclusion*  $\mathcal{L}$ -sequent. An  $\mathcal{L}$ -sequent s is called *negative* if  $\{\varphi \in \mathcal{L} \mid \mathbf{t}: \varphi \in s\} = \emptyset$ . A *single-conclusion*  $\mathcal{L}$ -hypersequent is an  $\mathcal{L}$ -hypersequent that consists solely of single-conclusion  $\mathcal{L}$ -sequents.

We usually use the metavariable E and F for singleton or empty sets of formulas (to represent the "right-side" of a single-conclusion sequent). Since in this chapter we discuss only the single-conclusion framework, we shall omit the prefix "single-conclusion" and refer to single-conclusion  $\mathcal{L}$ -(hyper)sequents simply as  $\mathcal{L}$ -(hyper)sequents.

## 7.2 Canonical Gödel Calculi

As defined below, all canonical Gödel calculi include the external and internal weakening rules ( $(\mathbf{f}:weak)$ ) and  $(\mathbf{t}:weak)$ ), and the rules (com), (cut) and (id):<sup>3</sup>

**External Weakening** This rule allows to infer  $H \mid s$  from H for every  $\mathcal{L}$ -hypersequent H and  $\mathcal{L}$ -sequent s.

- (f:weak) This rule allows to infer  $H \mid s \cup \{f:\varphi\}$  from  $H \mid s$  for every  $\mathcal{L}$ -hypersequent H,  $\mathcal{L}$ -sequent s, and  $\mathcal{L}$ -formula  $\varphi$ .
- (t:weak) This rule allows to infer  $H \mid s \cup \{t:\varphi\}$  from  $H \mid s$  for every  $\mathcal{L}$ -hypersequent H, negative  $\mathcal{L}$ -sequent s, and  $\mathcal{L}$ -formula  $\varphi$ .
- (com) This rule allows to infer  $H \mid s_1 \cup c_2 \mid s_2 \cup c_1$  from  $H \mid s_1 \cup c_1$  and  $H \mid s_2 \cup c_2$  for every  $\mathcal{L}$ -hypersequent H,  $\mathcal{L}$ -sequents  $s_1$  and  $s_2$ , and negative  $\mathcal{L}$ -sequents  $c_1$  and  $c_2$ . In the more usual notation, applications of (com) have the form:

$$\frac{H \mid \Gamma_1, \Gamma_2' \Rightarrow E_1 \quad H \mid \Gamma_2, \Gamma_1' \Rightarrow E_2}{H \mid \Gamma_1, \Gamma_1' \Rightarrow E_1 \mid \Gamma_2, \Gamma_2' \Rightarrow E_2}$$

<sup>&</sup>lt;sup>3</sup>We use the names (f:weak), (t:weak), (cut) and (id) in this context as well, but strictly speaking these are not the same rules that were defined in Chapter 2, but their hypersequential versions.

- (cut) This rule allows to infer  $H \mid s_1 \cup s_2$  from  $H \mid s_1 \cup \{f:\varphi\}$  and  $H \mid s_2 \cup \{t:\varphi\}$  for every  $\mathcal{L}$ -hypersequent H,  $\mathcal{L}$ -sequent  $s_1$ , and negative  $\mathcal{L}$ -sequent  $s_2$ .
- (id) This rule provides all axioms of the form  $\{\{f:\varphi,t:\varphi\}\}\$  for every  $\mathcal{L}$ -formula  $\varphi$ .

Unlike the structural rules, the logical rules of canonical Gödel calculi are not predefined, and they vary according to the concrete language of the calculus. Next we define the general form of the allowed logical rules. We first define right-introduction rules and their applications, and then deal with left-introduction rules.

**Definition 7.2.1.** A canonical right single-conclusion hypersequent  $\mathcal{L}$ -rule is a pair of the form  $\mathcal{S}/\{\mathsf{t}: \diamond (p_1, \dots, p_{ar(\diamond)})\}$ , where  $\diamond \in \diamond_{\mathcal{L}}$  and  $\mathcal{S}$  is a finite set of  $\{p_1, \dots, p_{ar(\diamond)}\}$ -sequents. The elements of  $\mathcal{S}$  are called the *premises* of the rule, and  $\{\mathsf{t}: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  is called the *conclusion* of the rule. An *application* of  $\{s_1, \dots, s_n\}/\{\mathsf{t}: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  is any inference step of the following form:

$$\frac{H \mid \sigma(s_1) \cup c \quad \dots \quad H \mid \sigma(s_n) \cup c}{H \mid \{\mathsf{t} : \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))\} \cup c}$$

where  $\sigma$  is an  $\mathcal{L}$ -substitution, c is a negative  $\mathcal{L}$ -sequent (called *context sequent*), and H is an  $\mathcal{L}$ -hypersequent (called *context hypersequent*).  $H \mid \sigma(s_1) \cup c, ..., H \mid \sigma(s_n) \cup c$  are called the *premises* of the application, while  $H \mid \{\mathbf{t}: \sigma(\diamond(p_1, ..., p_{ar(\diamond)}))\} \cup c$  is called the *conclusion* of the application.

**Example 7.2.2.** Suppose that  $\wedge \in \diamondsuit^2_{\mathcal{L}}$ , and consider the following canonical right single-conclusion hypersequent  $\mathcal{L}$ -rule:

$$\{\{\mathtt{t}{:}p_1\},\{\mathtt{t}{:}p_2\}\}/\{\mathtt{t}{:}p_1\wedge p_2\}$$

Applications of this rule have the form:

$$\frac{H \mid \Gamma \Rightarrow \varphi_1 \quad H \mid \Gamma \Rightarrow \varphi_2}{H \mid \Gamma \Rightarrow \varphi_1 \land \varphi_2}$$

Obviously, in applications of canonical right single-conclusion hypersequent rules, we can not allow non-negative context sequents (otherwise there will not be "enough space" for the conclusion). The following definition of left rules is slightly more complicated, since a non-negative context sequent may be added to the negative premises. To have a general notion here, we allow also cases in which some negative premises disallow non-negative context (just like the negative premises of the right rules). Hence, the set of premises of canonical left rules is divided into two different sets: premises that disallow non-negative context (including all non-negative premises), and premises that allow non-negative context (of-course all of them are negative).

<sup>&</sup>lt;sup>4</sup>A similar division of the premises was used in the definition of canonical left single-conclusion *sequent* rules in [14].

**Definition 7.2.3.** A canonical left single-conclusion hypersequent  $\mathcal{L}$ -rule is a triple of the form  $S_1, S_2/\{f: \diamond (p_1, \dots, p_{ar(\diamond)})\}$ , where  $\diamond \in \diamond_{\mathcal{L}}$ ,  $S_1$  is a finite set of  $\{p_1, \dots, p_{ar(\diamond)}\}$ -sequents, and  $S_2$  is a finite set of negative  $\{p_1, \dots, p_{ar(\diamond)}\}$ -sequents. The elements of  $S_1 \cup S_2$  are called the *premises* of the rule, and  $\{f: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  is called the *conclusion* of the rule. An application of  $\{s_1, \dots, s_n\}, \{s'_1, \dots, s'_m\}/\{f: \diamond (p_1, \dots, p_{ar(\diamond)})\}$  is any inference step of the following form:

where  $\sigma$  is an  $\mathcal{L}$ -substitution, c is a negative  $\mathcal{L}$ -sequent, and c' is an  $\mathcal{L}$ -sequent. The premises, conclusion, context sequents, and context hypersequent are defined exactly as in Definition 7.2.1.

**Example 7.2.4.** Suppose that  $\star \in \diamondsuit^1_{\mathcal{L}}$ , and consider the following canonical left single-conclusion hypersequent  $\mathcal{L}$ -rules:

$$\emptyset, \{\{f:p_1\}\}/\{f:\star p_1\}$$
  $\{\{f:p_1\}\}, \emptyset/\{f:\star p_1\}$ 

Applications of these rules have respectively the forms:

$$\begin{array}{c|c} H \mid \Gamma, \varphi \Rightarrow & H \mid \Gamma, \varphi \Rightarrow E \\ \hline H \mid \Gamma, \star \varphi \Rightarrow & H \mid \Gamma, \star \varphi \Rightarrow E \end{array}$$

In this chapter we shall refer to "canonical right (left) single-conclusion hypersequent  $\mathcal{L}$ -rules" simply as "canonical right (left)  $\mathcal{L}$ -rules". By "canonical  $\mathcal{L}$ -rules" we mean either canonical right  $\mathcal{L}$ -rules or canonical left  $\mathcal{L}$ -rules. In addition, we say that a canonical  $\mathcal{L}$ -rule r is a rule  $for \diamond if \diamond is$  the connective that occurs in the conclusion of r.

Remark 7.2.5. Since both internal and external weakening rules are present in every hypersequent calculus we study, it is always possible to incorporate weakenings in the applications of the rules. Thus for example, we could have defined an application of a canonical right  $\mathcal{L}$ -rule as an inference step deriving  $H \mid \sigma(s) \cup c$  from the  $\mathcal{L}$ -hypersequents  $H_i \mid \sigma(s_i) \cup c_i$  for every  $1 \leq i \leq n$ , where  $H, H_1, \ldots, H_n$  are  $\mathcal{L}$ -hypersequents such that  $H_1 \cup \ldots \cup H_n \subseteq H$ ,  $C, C_1, \ldots, C_n$  are negative  $\mathcal{L}$ -sequents such that  $C_1 \cup \ldots \cup C_n \subseteq C$ , and  $C_n \subseteq C$  is an  $C_n \subseteq C_n$  substitution. A similar definition is possible for the canonical left rules. Henceforth, we may use freely this kind of applications (which formally might involve additional applications of the weakening rules).

In Table 7.1, we present all logical rules of the hypersequent system **HG** for the standard propositional Gödel logic (see [22]) as canonical rules.<sup>5</sup> It is also possible to introduce new connectives using canonical rules:

<sup>&</sup>lt;sup>5</sup>By saying that **HG** is a system for propositional Gödel logic, we mean that  $\{\{\{\mathbf{t}:\psi\}\}\mid\psi\in\mathcal{T}\}\vdash_{\mathbf{HG}}\{\{\mathbf{t}:\varphi\}\}\}$  iff  $\mathcal{T}\Vdash\varphi$  in propositional Gödel logic.

Canonical Rule	Application scheme
$\emptyset,\emptyset/\{ extsf{f}\colon ot\}$	$H \mid \Gamma, \bot \Rightarrow E$
	$\frac{H \mid \Gamma \Rightarrow \varphi_1  H \mid \Gamma \Rightarrow \varphi_2}{H \mid \Gamma \Rightarrow \varphi_1 \land \varphi_2}$
$\emptyset, \{\{\mathtt{f} : p_1, \mathtt{f} : p_2\}\} / \{\mathtt{f} : p_1 \wedge p_2\}$	$\frac{H \mid \Gamma, \varphi_1, \varphi_2 \Rightarrow E}{H \mid \Gamma, \varphi_1 \land \varphi_2 \Rightarrow E}$
$\{\{\mathtt{t}{:}p_1\}\}/\{\mathtt{t}{:}p_1\vee p_2\}$	$\frac{H \mid \Gamma \Rightarrow \varphi_1}{H \mid \Gamma \Rightarrow \varphi_1 \vee \varphi_2}$
$\{\{\mathtt{t} : p_2\}\}/\{\mathtt{t} : p_1 \vee p_2\}$	$\frac{H \mid \Gamma \Rightarrow \varphi_2}{H \mid \Gamma \Rightarrow \varphi_1 \vee \varphi_2}$
$\emptyset, \{\{\mathtt{f}:p_1\}, \{\mathtt{f}:p_2\}\}/\{\mathtt{f}:p_1 \lor p_2\}$	$ \frac{H \mid \Gamma, \varphi_1 \Rightarrow E  H \mid \Gamma, \varphi_2 \Rightarrow E}{H \mid \Gamma, \varphi_1 \lor \varphi_2 \Rightarrow E} $
$\big\{ \{ \mathtt{f} {:} p_1, \mathtt{t} {:} p_2 \} \} / \{ \mathtt{t} {:} p_1 \supset p_2 \}$	$\frac{H \mid \Gamma, \varphi_1 \Rightarrow \varphi_2}{H \mid \Gamma \Rightarrow \varphi_1 \supset \varphi_2}$
$\{\{\mathtt{t}:p_1\}\}, \{\{\mathtt{f}:p_2\}\}/\{\mathtt{f}:p_1\supset p_2\}$	$ \frac{H \mid \Gamma_1 \Rightarrow \varphi_1  H \mid \Gamma_2, \varphi_2 \Rightarrow E}{H \mid \Gamma_1, \Gamma_2, \varphi_1 \supset \varphi_2 \Rightarrow E} $

Table 7.1: The Logical Rules of  $\mathbf{H}\mathbf{G}$ 

**Example 7.2.6.** A primal-implication connective (see [62]) can be introduced with the following two rules:

$$\{\{\mathtt{t} : p_2\}\}/\{\mathtt{t} : p_1 \leadsto p_2\} \qquad \{\{\mathtt{t} : p_1\}\}, \{\{\mathtt{f} : p_2\}\}/\{\mathtt{f} : p_1 \leadsto p_2\}$$

Applications of the left rule are like those of the left rule of implication in  $\mathbf{HG}$ , while applications of the right rule allow us to infer a hypersequent of the form  $H \mid \Gamma \Rightarrow \varphi_1 \rightsquigarrow \varphi_2$  from a hypersequent of the form  $H \mid \Gamma \Rightarrow \varphi_2$ .

**Example 7.2.7.** It is possible to combine the usual right rule for conjunction with the usual left rule for disjunction, and introduce a new binary connective X with the following rules:

$$\{\{\mathbf{t}:p_1\}, \{\mathbf{t}:p_2\}\}/\{\mathbf{t}:p_1 \otimes p_2\}$$
  $\emptyset, \{\{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2\}\}/\{\mathbf{f}:p_1 \otimes p_2\}$ 

Applications of the right rule are like those of the right rule of conjunction in **HG**, while applications of the left rule are like those of the left rule of disjunction in **HG** (see Table 7.1).

We can now define canonical Gödel calculi:

**Definition 7.2.8.** A canonical Gödel  $\mathcal{L}$ -calculus consists of all structural rules listed above (see Page 106), and any finite set of canonical  $\mathcal{L}$ -rules. The notion of a proof in a canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$  of an  $\mathcal{L}$ -hypersequent H from a set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents is defined as usual (see Definition 2.2.14). We write  $\mathcal{H} \vdash_{\mathbf{G}} H$  to denote the existence of such a proof.

To speak about restricted proofs, as needed in order to define and characterize prooftheoretic, we will consider *proof-specifications*. These are defined exactly as in Chapter 6 (Definition 6.1.17), with the obvious modifications in the definition of a  $\rho$ -proof (reflecting the transition from sequents to hypersequents). We shall also employ the same notation and write  $\mathcal{H} \vdash_{\mathbf{G} \mid \rho} H$  if there exists a  $\rho$ -proof in  $\mathbf{G}$  of H from  $\mathcal{H}$ , for a given  $\mathcal{L}$ -proofspecification  $\rho$ . For canonical Gödel calculi, we have the following:

**Proposition 7.2.9.** Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus, and let  $\rho = \langle \mathcal{L}, \mathcal{C}, \mathcal{A} \rangle$  be an  $\mathcal{L}$ -proof-specification. If  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$ , then  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho'} H$  for  $\rho' = \langle sub[\mathcal{C}] \cup sub[H], \mathcal{C}, \mathcal{A} \rangle$ .

*Proof.* The claim is proved by usual induction on the length of the proof in  $\mathbf{G}$ . Note that all rules in canonical Gödel  $\mathcal{L}$ -calculi except for cut have the "local subformula property" (i.e., for each rule, the premises of its applications consist only of formulas occurring as subformulas of the corresponding conclusions).

The proof-theoretic properties of (strong) sub-analyticity, (strong) cut-admissibility and axiom-expansion in canonical Gödel calculi are also defined as for basic calculi (Definition 6.1.18) with obvious modifications. From the last proposition, it easily follows that

if a canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$  enjoys strong cut-admissibility (i.e.  $\mathcal{H} \vdash_{\mathbf{G}} H$  implies that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$  for  $\rho = \langle \mathcal{L}, frm[\mathcal{H}], \mathcal{L} \rangle$ ), then it is strongly sub-analytic (i.e.  $\mathcal{H} \vdash_{\mathbf{G}} H$  implies that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$  for  $\rho = \langle sub[\mathcal{H} \cup \{H\}], \mathcal{L}, \mathcal{L} \rangle$ ). In the sequel, we show that these two properties are actually equivalent for canonical Gödel calculi.

We end this section with two propositions that will turn out to be useful in connection with the proof of Theorem 7.3.17 below.

**Proposition 7.2.10** (Generalized Communication). Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus, and let  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  be an  $\mathcal{L}$ -proof-specification. For all  $\mathcal{F}$ -hypersequents  $H_1, H_2$ , integers  $n, m \geq 0$ , n + m  $\mathcal{F}$ -sequents  $s_1, \ldots, s_n, s'_1, \ldots, s'_m$ , and two negative  $\mathcal{F}$ -sequents c, c', the  $\mathcal{L}$ -hypersequent  $H_1 \mid H_2 \mid s_1 \cup c' \mid \ldots \mid s_n \cup c' \mid s'_1 \cup c \mid \ldots \mid s'_m \cup c'$  has a  $\rho$ -proof in **G** from the  $\mathcal{L}$ -hypersequents  $H_1 \mid s_1 \cup c \mid \ldots \mid s_n \cup c$  and  $H_2 \mid s'_1 \cup c' \mid \ldots \mid s'_m \cup c'$ .

*Proof.* We prove this by induction on n+m. First, when n=0 or m=0, the claim follows by applying external weakening. Assume that n, m>0, n+m=l and that the claim holds for every n, m such that n+m < l. By the induction hypothesis, the following two hypersequents have a  $\rho$ -proof in **HIF** from  $H_1 \mid s_1 \cup c \mid ... \mid s_n \cup c$  and  $H_2 \mid s'_1 \cup c' \mid ... \mid s'_m \cup c'$ :

$$H_1 \mid s_n \cup c \mid H_2 \mid s_1 \cup c' \mid \dots \mid s_{n-1} \cup c' \mid s'_1 \cup c \mid \dots \mid s'_m \cup c \mid H_1 \mid H_2 \mid s'_m \cup c' \mid s_1 \cup c' \mid \dots \mid s_{n-1} \cup c' \mid s'_1 \cup c \mid \dots \mid s'_{m-1} \cup c \mid S'_{m-1$$

An application of (com) on these two hypersequents provides the desired result.

**Proposition 7.2.11.** Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus,  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  an  $\mathcal{L}$ -proof-specification, and  $r = \mathcal{S}_1, \mathcal{S}_2/\{\mathbf{f}:\varphi\}$  a canonical left  $\mathcal{L}$ -rule of **G**, with  $|\mathcal{S}_2| > 0$ . Let  $\sigma$  be an  $\mathcal{L}$ -substitution such that  $frm[\sigma(\mathcal{S}_1 \cup \mathcal{S}_2)] \cup \{\sigma(\varphi)\} \subseteq \mathcal{F}$ . Let  $\mathcal{S}_2 = \{s_1, \dots, s_m\}$  and  $s_0 = \{\mathbf{f}:\sigma(\varphi)\}$ . Let c be a negative  $\mathcal{F}$ -sequent and H an  $\mathcal{F}$ -hypersequent. Denote by  $\mathcal{H}_1$  the set  $\{H \mid c \cup \sigma(s)\}_{s \in \mathcal{S}_1}$ . Then, for every  $n_1, \dots, n_m \geq 0$ ,  $\mathcal{F}$ -hypersequent H' such that  $H \subseteq H'$ , and  $n_1 + \dots + n_m$   $\mathcal{F}$ -sequents  $s_1^1, \dots, s_{n_1}^1, \dots, s_{n_m}^m$ ,

$$H' \mid c \cup s_1^1 \cup s_0 \mid ... \mid c \cup s_{n_1}^1 \cup s_0 \mid ... \mid c \cup s_1^m \cup s_0 \mid ... \mid c \cup s_{n_m}^m \cup s_0$$

has a  $\rho$ -proof in **G** from

$$\mathcal{H}_1 \cup \{H' \mid s_1^i \cup \sigma(s_i) \mid \dots \mid s_{n_i}^i \cup \sigma(s_i)\}_{1 \leq i \leq m}.$$

Proof. First, if  $n_i = 0$  for some  $1 \le i \le m$ , the claim follows by applying external weakening on H'. Next, we prove the claim for the case that  $n_1 = n_2 = \dots = n_m = 1$ , by induction on the size S of the set  $\{s_1^1, \dots, s_1^m\}$ . If S = 1, then one application of r suffices. Now let  $S \ge 2$ , and assume that the claim holds for sets of size S - 1. Let  $s_1^1, \dots, s_1^m$  be  $\mathcal{F}$ -sequents sets such that  $|\{s_1^1, \dots, s_1^m\}| = S$ , and let H' be some  $\mathcal{F}$ -hypersequent such that  $H \subseteq H'$ . Let  $G_0 = H' \mid c \cup s_1^1 \cup s_0 \mid \dots \mid c \cup s_1^m \cup s_0$ , and for every  $1 \le i \le m$ ,  $G_i = H' \mid s_1^i \cup \sigma(s_i)$ . Let  $i_1, i_2$  be two indices such that  $s_1^{i_1} \ne s_1^{i_2}$ , and let  $I_1 = \{1 \le i \le m \mid s_1^i = s_1^{i_1}\}$  and

 $I_2 = \{1 \le i \le m \mid s_1^i = s_1^{i_2}\}$ . For every  $j_1 \in I_1$  and  $j_2 \in I_2$ , we have that (using one application of (com)):

$$\{G_{j_1}, G_{j_2}\} \vdash_{\mathbf{G} \upharpoonright \rho} H' \mid s_1^{i_1} \cup \sigma(s_{j_2}) \mid s_1^{i_2} \cup \sigma(s_{j_1}).$$

For every  $j_1 \in I_1$ , the induction hypothesis and the availability of external weakening entail that  $G_0 \mid s_1^{i_2} \cup \sigma(s_{j_1})$  has a  $\rho$ -proof in **G** from

$$\mathcal{H}_1 \cup \{G_j\}_{j \notin I_2} \cup \{H' \mid s_1^{i_1} \cup \sigma(s_{j_2}) \mid s_1^{i_2} \cup \sigma(s_{j_1})\}_{j_2 \in I_2}.$$

The induction hypothesis and the availability of external weakening again imply that  $G_0$  has a  $\rho$ -proof in  $\mathbf{G}$  from  $\mathcal{H}_1 \cup \{G_j\}_{j \notin I_1} \cup \{G_0 \mid s_1^{i_2} \cup \sigma(s_{j_1})\}_{j_1 \in I_1}$ . Together, we have  $\mathcal{H}_1 \cup \{G_i\}_{1 \leq i \leq m} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} G_0$ .

Next we prove the claim for any  $n_1, \ldots, n_m \geq 1$  by induction on  $n_1 + \ldots + n_m$ . Assume that  $n_1 + \ldots + n_m = l$  and that the claim holds for every  $n_1, \ldots, n_m$  such that  $n_1 + \ldots + n_m < l$ . Let H' be an  $\mathcal{F}$ -hypersequent that extends H, and let  $s_1^1, \ldots, s_{n_1}^1, \ldots, s_1^m, \ldots, s_{n_m}^m$  be  $\mathcal{F}$ -sequents. Let  $G_0$  denote the hypersequent

$$H' \mid c \cup s_1^1 \cup s_0 \mid \dots \mid c \cup s_{n_1}^1 \cup s_0 \mid \dots \mid c \cup s_1^m \cup s_0 \mid \dots \mid c \cup s_{n_m}^m \cup s_0,$$

and  $\mathcal{H} = \{H' \mid s_1^i \cup \sigma(s_i) \mid ... \mid s_{n_i}^i \cup \sigma(s_i)\}_{1 \leq i \leq m}$ . For every  $1 \leq i \leq m$ , the induction hypothesis and the availability of external weakening entail that  $\mathcal{H}_1 \cup \mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} G_0 \mid s_1^i \cup \sigma(s_i)$ . By the proof for the case  $n_1 = n_2 = ... = n_m = 1$ , we have that

$$\mathcal{H}_1 \cup \{G_0 \mid s_1^i \cup \sigma(s_i)\}_{1 \le i \le m} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} G_0. \qquad \Box$$

**Example 7.2.12.** Suppose that **G** includes the left rule  $\emptyset$ ,  $\{\{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2\}\}/\{\mathbf{f}:p_1 \times p_2\}$  (see Example 7.2.7). By Proposition 7.2.11, the following rule (given by a scheme) is cut-free derivable in **G**:

$$H \mid \Gamma_1^1, \varphi_1 \Rightarrow E_1 \mid \dots \mid \Gamma_{n_1}^1, \varphi_1 \Rightarrow E_{n_1} \quad H \mid \Gamma_1^2, \varphi_2 \Rightarrow F_1 \mid \dots \mid \Gamma_{n_2}^2, \varphi_2 \Rightarrow F_{n_2}$$

$$H \mid \Gamma_1^1, \varphi_1 \bowtie \varphi_2 \Rightarrow E_1 \mid \dots \mid \Gamma_{n_1}^1, \varphi_1 \bowtie \varphi_2 \Rightarrow E_{n_1} \mid \Gamma_1^2, \varphi_1 \bowtie \varphi_2 \Rightarrow F_1 \mid \dots \mid \Gamma_{n_2}^2, \varphi_1 \bowtie \varphi_2 \Rightarrow F_{n_2}$$

## 7.3 Many-Valued Semantics

In this section we provide a method to obtain sound and complete many-valued semantics for any given canonical Gödel calculus and proof-specification. The semantic structures introduced for this task are called *Gödel valuations*. The truth values in these structures should form a *propositional Gödel set*, defined as follows:

**Definition 7.3.1.** A (propositional) Gödel set is a bounded linearly ordered set  $\mathcal{V} = \langle V, \leq \rangle$ . We denote by  $0_{\mathcal{V}}$  and  $1_{\mathcal{V}}$  the maximal and minimal elements (respectively) of V with respect to  $\leq$ . The operations  $\min_{\mathcal{V}}$ ,  $\max_{\mathcal{V}}$  are defined as usual (where  $\min_{\mathcal{V}} \emptyset = 1_{\mathcal{V}}$  and  $\max_{\mathcal{V}} \emptyset = 0_{\mathcal{V}}$ ). For every two elements  $u_1, u_2 \in V$ ,  $u_1 \to_{\mathcal{V}} u_2$  is defined to be  $1_{\mathcal{V}}$  if  $u_1 \leq u_2$ , and  $u_2$  otherwise. The relations  $\geq$ , <, and > are also defined in the

obvious way. We omit the subscript  $\mathcal{V}$  when it is clear from the context, and sometimes identify  $\mathcal{V}$  with the set V (e.g., when referring to the elements of V as elements of  $\mathcal{V}$ ).

Now, (partial) Gödel  $\mathcal{L}$ -valuations are defined as follows:

**Definition 7.3.2.** A partial Gödel  $\mathcal{L}$ -valuation (partial  $\mathcal{L}$ -Gvaluation, for short) is a triple  $\langle \mathcal{V}, Dom, v \rangle$ , where  $\mathcal{V}$  is a Gödel set,  $Dom \subseteq \mathcal{L}$ , and v is a function from Dom to  $\mathcal{V} \times \mathcal{V}$ . A partial  $\mathcal{L}$ -Gvaluation  $\langle \mathcal{V}, Dom, v \rangle$  with  $Dom = \mathcal{L}$  is also called an  $\mathcal{L}$ -Gvaluation.

**Notation 7.3.3.** Throughout, we identify a partial  $\mathcal{L}$ -Gvaluation  $\langle \mathcal{V}, Dom, v \rangle$  with its underlying function v, and denote the Gödel set  $\mathcal{V}$  by  $\mathcal{V}_v$ , and the set Dom by  $Dom_v$ . In addition, given a partial  $\mathcal{L}$ -Gvaluation v, we denote by  $v^{\mathfrak{f}}$  and  $v^{\mathfrak{t}}$  "the left and right projections of v", that is  $v^{\mathfrak{f}}(\varphi) = u_1$  and  $v^{\mathfrak{t}}(\varphi) = u_2$  iff  $v(\varphi) = \langle u_1, u_2 \rangle$ .

There are two main ideas behind the definition of a partial  $\mathcal{L}$ -Gvaluation above. First, as in the previous chapters, these semantic structures may not assign truth values to all formulas. Their exact domain Dom should be determined according to the formulas that are allowed to appear in  $\rho$ -proofs for a given proof-specification  $\rho$ . Second, note that a partial  $\mathcal{L}$ -Gvaluation v assigns a  $pair \langle v^{\mathfrak{t}}(\varphi), v^{\mathfrak{t}}(\varphi) \rangle$  of truth values to each formula  $\varphi \in Dom_v$ . Intuitively,  $v^{\mathfrak{t}}(\varphi)$  is the value of  $\varphi$  when it is " $\mathfrak{t}$ -labelled" (occurs on the "left side" of a sequent), and  $v^{\mathfrak{t}}(\varphi)$  is its value when it is " $\mathfrak{t}$ -labelled" (occurs on the "right side"). Roughly speaking, to have a complete semantics for proof-specifications in which (cut) and/or (id) may not be applied on some formula  $\varphi$ , we have to "disconnect"  $v^{\mathfrak{t}}(\varphi)$  and  $v^{\mathfrak{t}}(\varphi)$ . Obviously, certain restrictions on the relation between  $v^{\mathfrak{t}}(\varphi)$  and  $v^{\mathfrak{t}}(\varphi)$  should be put when (cut) and/or (id) are allowed to apply on  $\varphi$  (see Definition 7.3.6 below). Before turning to these restrictions, we define the semantic consequence relation between hypersequents induced by a set of partial  $\mathcal{L}$ -Gvaluations.

#### **Definition 7.3.4.** Let v be a partial $\mathcal{L}$ -Gvaluation.

- 1. Given a  $Dom_v$ -sequent s,  $v^{\mathbf{f}}(s)$ ,  $v^{\mathbf{t}}(s)$  and v(s) denote the element of  $\mathcal{V}_v$  defined by:
  - (a)  $v^{\mathbf{f}}(s) = \min\{v^{\mathbf{f}}(\varphi) \mid \mathbf{f}: \varphi \in s\}.$
  - $\text{(b)} \ v^{\mathtt{t}}(s) = \max\{v^{\mathtt{t}}(\varphi) \mid \mathtt{t} : \varphi \in s\}.$
  - (c)  $v(s) = v^{\mathsf{f}}(s) \to v^{\mathsf{t}}(s)$ .
- 2. v is a model of:
  - (a) an  $\mathcal{L}$ -sequent s if s is a  $Dom_v$ -sequent and v(s) = 1.
  - (b) an  $\mathcal{L}$ -hypersequent H if H is a  $Dom_v$ -hypersequent and v is a model of some component  $s \in H$ .

(c) a set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents if it is a model of every  $Dom_v$ -hypersequent  $H \in \mathcal{H}$ .

We write  $v \models X$  to denote that v is a model of X (here X is either a sequent, a hypersequent, or a set of hypersequents).

Note that in order to check whether  $v \models s$ , one only needs  $v^{\mathfrak{f}}(\varphi)$  for every formula  $\varphi$  such that  $\mathfrak{t}:\varphi \in s$ , and  $v^{\mathfrak{t}}(\varphi)$  for every formula  $\varphi$  such that  $\mathfrak{t}:\varphi \in s$ . In turn, hypersequents are interpreted as "meta-disjunctions" of sequents. The consequence relation between hypersequents induced by a set  $\mathcal{G}$  of partial  $\mathcal{L}$ -Gvaluations is defined as follows:

**Definition 7.3.5.** An  $\mathcal{L}$ -hypersequent H follows from a set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents with respect to a set  $\mathcal{G}$  of partial  $\mathcal{L}$ -Gvaluations (denoted by:  $\mathcal{H} \vdash_{\mathcal{G}} H$ ) if for every  $v \in \mathcal{G}$ :  $v \models H$  whenever  $v \models \mathcal{H}$ .

As noted above, when (cut) and/or (id) are allowed to apply on some formula  $\varphi$  (according to the proof-specification), we have to "connect"  $v^{\mathbf{f}}(\varphi)$  and  $v^{\mathbf{t}}(\varphi)$ . Intuitively, (cut) and (id) have opposite semantic roles – while (cut) forces the "f-value" to be greater than or equal to the "t-value", (id) forces the "f-value" to be lower than or equal to the t-value. This is formulated in the next definition.

**Definition 7.3.6.** Let  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  be an  $\mathcal{L}$ -proof-specification. A partial  $\mathcal{L}$ -Gvaluation v is called  $\rho$ -legal if the following hold:

- 1.  $Dom_v = \mathcal{F}$ .
- 2.  $v^{t}(\varphi) \leq v^{f}(\varphi)$  for every  $\varphi \in \mathcal{C} \cap \mathcal{F}$ .
- 3.  $v^{f}(\varphi) \leq v^{t}(\varphi)$  for every  $\varphi \in \mathcal{A} \cap \mathcal{F}$ .

Note that when (cut) and (id) can be used for all formulas (that is:  $\mathcal{C} = \mathcal{A} = \mathcal{L}$ ), we require that  $v^{\mathbf{f}}(\varphi) = v^{\mathbf{t}}(\varphi)$  for every  $\varphi \in Dom_v$ . Such Gvaluations will be called *normal*:

**Definition 7.3.7.** A partial  $\mathcal{L}$ -Gvaluation v is called *normal* if  $v^{\mathfrak{f}}(\varphi) = v^{\mathfrak{t}}(\varphi)$  for every  $\varphi \in \mathcal{L}$ . For normal partial  $\mathcal{L}$ -Gvaluation v, we may write  $v(\varphi)$  instead of  $v^{\mathfrak{f}}(\varphi)$  (equivalently,  $v^{\mathfrak{t}}(\varphi)$ ).

Next, we turn to the semantic effect of the canonical rules included in the given canonical Gödel calculus. Roughly speaking, the idea here is that each rule for a connective  $\diamond$  induces a function from  $(\mathcal{V} \times \mathcal{V})^{ar(\diamond)}$  to  $\mathcal{V}$ . By applying these functions on the truth values of  $\varphi_1, \ldots, \varphi_{ar(\diamond)}$ , one obtain bounds on the truth value of  $\diamond(\varphi_1, \ldots, \varphi_{ar(\diamond)})$ . The functions induced by the right canonical rules for  $\diamond$  provide lower bounds on  $v^{\mathbf{t}}(\diamond(\varphi_1, \ldots, \varphi_{ar(\diamond)}))$ , while those induced by the left rules for  $\diamond$  are used as upper bounds on  $v^{\mathbf{f}}(\diamond(\varphi_1, \ldots, \varphi_{ar(\diamond)}))$ . Below we precisely formulate this idea.

**Notation 7.3.8.** Let v be a partial  $\mathcal{L}$ -Gvaluation. Given a finite set  $\mathcal{S}$  of  $Dom_v$ -sequents,  $v(\mathcal{S}) = \min\{v(s) \mid s \in \mathcal{S}\}$  and  $v^{\mathfrak{f}}(\mathcal{S}) = \max\{v^{\mathfrak{f}}(s) \mid s \in \mathcal{S}\}.$ 

**Definition 7.3.9.** Let  $\diamond$  be an *n*-ary connective of  $\mathcal{L}$ ,  $\mathcal{V}$  a Gödel set, and r a canonical  $\mathcal{L}$ -rule for  $\diamond$ .  $\diamond^r_{\mathcal{V}}$  is a function from  $(\mathcal{V} \times \mathcal{V})^n$  to  $\mathcal{V}$  defined by:

$$\diamond_{\mathcal{V}}^{r}(\langle u_{1}^{\mathtt{f}}, u_{1}^{\mathtt{t}} \rangle, \dots, \langle u_{n}^{\mathtt{f}}, u_{n}^{\mathtt{t}} \rangle) = \begin{cases} v(\mathcal{S}) & r = \mathcal{S}/\{\mathtt{t}:\varphi\} \\ v(\mathcal{S}_{1}) \to v^{\mathtt{f}}(\mathcal{S}_{2}) & r = \mathcal{S}_{1}, \mathcal{S}_{2}/\{\mathtt{f}:\varphi\} \end{cases}$$

where v is the partial  $\mathcal{L}$ -Gvaluation defined by  $\mathcal{V}_v = \mathcal{V}$ ,  $Dom_v = \{p_1, \dots, p_n\}$ , and  $v(p_i) = \langle u_i^{\mathbf{f}}, u_i^{\mathbf{t}} \rangle$  for every  $1 \leq i \leq n$ .

**Example 7.3.10.** Suppose that  $\wedge \in \diamondsuit^2_{\mathcal{L}}$ , and consider the usual rules:

$$(t:\land) = \{\{t:p_1\}, \{t:p_2\}\}/\{t:p_1 \land p_2\} = (f:\land) = \emptyset, \{\{f:p_1, f:p_2\}\}/\{f:p_1 \land p_2\}$$

Let  $\mathcal{V}$  be a Gödel set, and  $\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle$ ,  $\langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V}$ . We calculate  $\wedge_{\mathcal{V}}^{(\mathbf{t}:\wedge)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle)$ , and  $\wedge_{\mathcal{V}}^{(\mathbf{f}:\wedge)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle)$ . Let v be the partial  $\mathcal{L}$ -Gvaluation defined by  $\mathcal{V}_v = \mathcal{V}$ ,  $Dom_v = \{p_1, p_2\}$ , and  $v(p_i) = \langle u_i^{\mathbf{f}}, u_i^{\mathbf{t}} \rangle$  for  $1 \leq i \leq 2$ . We have

$$v(\{\mathtt{t}:p_1\}) = v^{\mathtt{f}}(\{\mathtt{t}:p_1\}) \to v^{\mathtt{t}}(\{\mathtt{t}:p_1\}) = 1 \to v^{\mathtt{t}}(p_1) = v^{\mathtt{t}}(p_1) = u_1^{\mathtt{t}}.$$

Similarly,  $v(\{t:p_2\}) = u_2^t$ . Thus

$$\wedge_{\mathcal{V}}^{(\mathtt{t}:\wedge)}(\langle u_1^{\mathtt{f}}, u_1^{\mathtt{t}} \rangle, \langle u_2^{\mathtt{f}}, u_2^{\mathtt{t}} \rangle) = v(\{\{\mathtt{t}:p_1\}, \{\mathtt{t}:p_2\}\}) = \min\{u_1^{\mathtt{t}}, u_2^{\mathtt{t}}\}.$$

In addition,  $v(\emptyset) = 1$ , and

$$v^{\mathbf{f}}(\{\{\mathbf{f}:p_1,\mathbf{f}:p_2\}\}) = v^{\mathbf{f}}(\{\mathbf{f}:p_1,\mathbf{f}:p_2\}) = \min\{v^{\mathbf{f}}(p_1),v^{\mathbf{f}}(p_2)\} = \min\{u_1^{\mathbf{f}},u_2^{\mathbf{f}}\}.$$

Therefore,

$$\wedge_{\mathcal{V}}^{(\mathbf{f}:\wedge)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle) = v(\emptyset) \rightarrow v^{\mathbf{f}}(\{\{\mathbf{f}: p_1, \mathbf{f}: p_2\}\}) = \min\{u_1^{\mathbf{f}}, u_2^{\mathbf{f}}\}.$$

Now, if we replace  $(\mathbf{f}:\wedge)$  by  $\emptyset$ ,  $\{\{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2\}\}/\{\mathbf{f}:p_1\wedge p_2\}$  (see Example 7.2.7) we obtain  $\wedge_{\mathcal{V}}^{(\mathbf{f}:\wedge)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}}\rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}}\rangle) = v(\emptyset) \rightarrow v^{\mathbf{f}}(\{\{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2\}\}) = \max\{u_1^{\mathbf{f}}, u_2^{\mathbf{f}}\}.$ 

**Example 7.3.11.** Suppose that  $\supset \in \diamondsuit^2_{\mathcal{L}}$ , and consider the usual rules:

$$(\mathtt{t}:\supset) \quad \{\{\mathtt{f}:p_1,\mathtt{t}:p_2\}\}/\{\mathtt{t}:p_1\supset p_2\} \qquad (\mathtt{f}:\supset) \quad \{\{\mathtt{t}:p_1\}\},\{\{\mathtt{f}:p_2\}\}/\{\mathtt{f}:p_1\supset p_2\}$$

Let  $\mathcal{V}$  be a Gödel set, and  $\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle$ ,  $\langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V}$ . We calculate  $\supset_{\mathcal{V}}^{(\mathbf{t}:\supset)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle)$ , and  $\supset_{\mathcal{V}}^{(\mathbf{f}:\supset)}(\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle)$ . Let v be the partial  $\mathcal{L}$ -Gvaluation defined by  $\mathcal{V}_v = \mathcal{V}$ ,  $Dom_v = \{p_1, p_2\}$ , and  $v(p_i) = \langle u_i^{\mathbf{f}}, u_i^{\mathbf{t}} \rangle$  for  $1 \leq i \leq 2$ . Then:

$$\supset_{\mathcal{V}}^{(\mathtt{t}:\supset)}(\langle u_1^{\mathtt{f}}, u_1^{\mathtt{t}} \rangle, \langle u_2^{\mathtt{f}}, u_2^{\mathtt{t}} \rangle) = v(\{\{\mathtt{f}:p_1, \mathtt{t}:p_2\}\}) = v^{\mathtt{f}}(\{\mathtt{f}:p_1, \mathtt{t}:p_2\}) \to v^{\mathtt{t}}(\{\mathtt{f}:p_1, \mathtt{t}:p_2\}) = v^{\mathtt{f}}(p_1) \to v^{\mathtt{t}}(p_2) = u_1^{\mathtt{f}} \to u_2^{\mathtt{t}}$$

$$\supset_{\mathcal{V}}^{(\mathtt{f}:\supset)}(\langle u_{1}^{\mathtt{f}}, u_{1}^{\mathtt{t}} \rangle, \langle u_{2}^{\mathtt{f}}, u_{2}^{\mathtt{t}} \rangle) = v(\{\{\mathtt{t}:p_{1}\}\}) \to v^{\mathtt{f}}(\{\{\mathtt{f}:p_{2}\}\}) = v^{\mathtt{t}}(p_{1}) \to v^{\mathtt{f}}(p_{2}) = u_{1}^{\mathtt{t}} \to u_{2}^{\mathtt{f}}$$

Now, if we replace (t:  $\supset$ ) by  $\{\{t:p_2\}\}/\{t:p_1\supset p_2\}$  (see Example 7.2.6) we obtain

$$\supset_{\mathcal{V}}^{(\mathtt{t}:\supset)}(\langle u_1^{\mathtt{f}}, u_1^{\mathtt{t}} \rangle, \langle u_2^{\mathtt{f}}, u_2^{\mathtt{t}} \rangle) = v(\{\{\mathtt{t}:p_2\}\}) = 1 \to v^{\mathtt{t}}(p_2) = u_2^{\mathtt{t}}$$

Note that the calculation of  $\diamond_{\mathcal{V}}^r$  requires only applying "atomic Gödel functions", namely min, max, 0, 1, and Gödel implication. These functions enforce restrictions on Gvaluations as follows:

#### **Definition 7.3.12.** A partial $\mathcal{L}$ -Gvaluation v is called r-legal:

• for a canonical right rule  $r = \mathcal{S}/\{\mathsf{t}: \diamond(p_1, \dots, p_n)\}$  if for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\sigma(\{p_1, \dots, p_n, \diamond(p_1, \dots, p_n)\}) \subseteq Dom_v$ :

$$\diamond^r_{\mathcal{V}_v}(v(\sigma(p_1)),\ldots,v(\sigma(p_n))) \leq v^{\mathtt{t}}(\sigma(\diamond(p_1,\ldots,p_n))).$$

• for a canonical left rule  $r = S_1, S_2/\{f: \diamond (p_1, ..., p_n)\}$  if for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\sigma(\{p_1, ..., p_n, \diamond (p_1, ..., p_n)\}) \subseteq Dom_v$ :

$$v^{\mathbf{f}}(\sigma(\diamond(p_1,\ldots,p_n))) \leq \diamond_{\mathcal{V}_n}^r(v(\sigma(p_1)),\ldots,v(\sigma(p_n))).$$

Note that the right rules for  $\diamond$  impose restrictions on  $v^{\mathsf{t}}(\sigma(\diamond(p_1,\ldots,p_n)))$ , while the left rules for  $\diamond$  restrict  $v^{\mathsf{f}}(\sigma(\diamond(p_1,\ldots,p_n)))$ .

**Example 7.3.13.** Let v be a  $\rho$ -legal partial  $\mathcal{L}$ -Gvaluation for  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$ . Below we present the condition for v to be r-legal for the canonical rules presented in the examples above. Note that the condition on the right should hold for every  $\mathcal{L}$ -substitution  $\sigma$  such that  $\{\sigma(p_1 \diamond p_2), \sigma(p_1), \sigma(p_2)\} \subseteq \mathcal{F}$  where  $\diamond$  is the connective in the corresponding rule.

Canonical Rule Semantic Condition  $(t:\land) \{\{t:p_1\}, \{t:p_2\}\}/\{t:p_1 \land p_2\}$  $\min\{v^{\mathsf{t}}(\sigma(p_1)), v^{\mathsf{t}}(\sigma(p_2))\} < v^{\mathsf{t}}(\sigma(p_1 \land p_2))$  $v^{\mathbf{f}}(\sigma(p_1 \wedge p_2)) < \min\{v^{\mathbf{f}}(\sigma(p_1)), v^{\mathbf{f}}(\sigma(p_2))\}$  $(f:\land) \emptyset, \{\{f:p_1, f:p_2\}\}/\{f:p_1 \land p_2\}$  $v^{\mathsf{f}}(\sigma(p_1)) \to v^{\mathsf{t}}(\sigma(p_2)) < v^{\mathsf{t}}(\sigma(p_1) \supset p_2)$  $(t: \supset) \{\{f:p_1, t:p_2\}\}/\{t:p_1 \supset p_2\}$  $v^{\mathbf{f}}(\sigma(p_1 \supset p_2)) < v^{\mathbf{t}}(\sigma(p_1)) \rightarrow v^{\mathbf{f}}(\sigma(p_2))$  $(f: \supset) \{\{t:p_1\}\}, \{\{f:p_2\}\}/\{f:p_1 \supset p_2\}$  $v^{\mathsf{t}}(\sigma(p_2)) \leq v^{\mathsf{t}}(\sigma(p_1 \leadsto p_2))$  $(\mathsf{t}:\leadsto) \{\{\mathsf{t}:p_2\}\}/\{\mathsf{t}:p_1\leadsto p_2\}$  $v^{\mathbf{f}}(\sigma(p_1 \leadsto p_2)) \leq v^{\mathbf{t}}(\sigma(p_1)) \to v^{\mathbf{f}}(\sigma(p_2))$  $(f: \leadsto) \{\{t:p_1\}\}, \{\{f:p_2\}\}/\{f:p_1 \leadsto p_2\}$  $\min\{v^{\mathsf{t}}(\sigma(p_1)), v^{\mathsf{t}}(\sigma(p_2))\} \leq v^{\mathsf{t}}(\sigma(p_1 \bowtie p_2))$  $(t:X) \{\{t:p_1\}, \{t:p_2\}\}/\{t:p_1X|p_2\}$  $(f:X) \emptyset, \{\{f:p_1\}, \{f:p_2\}\}/\{f:p_1 \times p_2\}$  $v^{\mathbf{f}}(\sigma(p_1 \times p_2)) < \max\{v^{\mathbf{f}}(\sigma(p_1)), v^{\mathbf{f}}(\sigma(p_2))\}$ 

Now, if we only consider normal (full)  $\mathcal{L}$ -Gvaluations, these conditions can be simplified as follows.

# $\begin{array}{ll} Canonical \ Rule & Semantic \ Condition \\ (\texttt{t}:\land) \ \{\{\texttt{t}:p_1\}, \{\texttt{t}:p_2\}\}/\{\texttt{t}:p_1 \land p_2\} & \min\{v(\varphi_1), v(\varphi_2)\} \leq v(\varphi_1 \land \varphi_2) \\ (\texttt{f}:\land) \ \emptyset, \{\{\texttt{f}:p_1, \texttt{f}:p_2\}\}/\{\texttt{f}:p_1 \land p_2\} & v(\varphi_1 \land \varphi_2) \leq \min\{v(\varphi_1), v(\varphi_2)\} \\ (\texttt{t}:\supset) \ \{\{\texttt{f}:p_1, \texttt{t}:p_2\}\}/\{\texttt{t}:p_1 \supset p_2\} & v(\varphi_1) \rightarrow v(\varphi_2) \leq v(\varphi_1 \supset \varphi_2) \\ (\texttt{f}:\supset) \ \{\{\texttt{t}:p_1\}\}, \{\{\texttt{f}:p_2\}\}/\{\texttt{f}:p_1 \supset p_2\} & v(\varphi_1 \supset \varphi_2) \leq v(\varphi_1) \rightarrow v(\varphi_2) \\ (\texttt{t}:\leadsto) \ \{\{\texttt{t}:p_2\}\}/\{\texttt{t}:p_1 \leadsto p_2\} & v(\varphi_2) \leq v(\varphi_1 \leadsto \varphi_2) \\ \end{array}$

$$\begin{array}{ll} (\mathbf{f}:\leadsto) & \{\{\mathbf{t}:p_1\}\}, \{\{\mathbf{f}:p_2\}\}/\{\mathbf{f}:p_1\leadsto p_2\} & v(\varphi_1\leadsto\varphi_2)\leq v(\varphi_1)\to v(\varphi_2) \\ (\mathbf{t}:\aleph) & \{\{\mathbf{t}:p_1\}, \{\mathbf{t}:p_2\}\}/\{\mathbf{t}:p_1\aleph p_2\} & \min\{v(\varphi_1), v(\varphi_2)\}\leq v(\varphi_1\aleph\varphi_2) \\ (\mathbf{f}:\aleph) & \emptyset, \{\{\mathbf{f}:p_1\}, \{\mathbf{f}:p_2\}\}/\{\mathbf{f}:p_1\aleph p_2\} & v(\varphi_1\aleph\varphi_2)\leq \max\{v(\varphi_1), v(\varphi_2)\} \end{array}$$

In this case the condition in the right should hold for every  $\varphi_1, \varphi_2 \in \mathcal{L}$ . Recall that for normal Kvaluations we write  $v(\varphi)$  instead of  $v^{\mathbf{f}}(\varphi)$  (or equivalently  $v^{\mathbf{t}}(\varphi)$ ). Note that the two rules for  $\wedge$  together impose the usual Gödel logic semantics of  $\wedge$ :

$$v(\varphi_1 \wedge \varphi_2) = \min\{v(\varphi_1), v(\varphi_2)\}.$$

Similarly, the two rules for  $\supset$  impose its usual semantics:

$$v(\varphi_1 \supset \varphi_2) = v(\varphi_1) \to v(\varphi_2).$$

Now, the rules for  $\rightsquigarrow$  and X together impose the conditions:

$$\begin{split} v(\varphi_2) &\leq v(\varphi_1 \leadsto \varphi_2) \leq v(\varphi_1) \to v(\varphi_2), \\ \min\{v(\varphi_1), v(\varphi_2)\} &\leq v(\varphi_1 \bowtie \varphi_2) \leq \max\{v(\varphi_1), v(\varphi_2)\}. \end{split}$$

In these two cases we obtain non-deterministic semantics since the value assigned to  $\varphi_1 \rightsquigarrow \varphi_2$  (similarly, to  $\varphi_1 \bowtie \varphi_2$ ) is not uniquely determined by the value assigned to  $\varphi_1$  and  $\varphi_2$ . A deterministic semantics is obtained only when the lower bound determined by the right rules is equal to the upper bound determined by the left rules.

Remark 7.3.14. Note that in non-normal (partial)  $\mathcal{L}$ -Gvaluations, when we may have  $v^{\mathbf{f}}(\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)})) \neq v^{\mathbf{t}}(\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)}))$ , then the semantics of  $\diamond$  is non-deterministic by definition, since both  $v^{\mathbf{f}}(\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)}))$  and  $v^{\mathbf{t}}(\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)}))$  are restricted only from one side. Therefore, even when  $\diamond$  is a usual connective with ordinary introduction rules, non-deterministic semantics is employed to handle proof-specifications in which (cut) and/or (id) are not allowed on formulas of the form  $\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)})$ .

The following technical lemma will be useful below.

**Lemma 7.3.15.** Let v be a partial  $\mathcal{L}$ -Gvaluation, r a canonical  $\mathcal{L}$ -rule for an n-ary connective  $\diamond$ , and  $\sigma$  an  $\mathcal{L}$ -substitution such that  $\sigma(\{p_1, \dots, p_n\}) \subseteq Dom_v$ . Then  $\diamond_{\mathcal{V}_v}^r(v(\sigma(p_1)), \dots, v(\sigma(p_n)))$  is equal to:

- $v(\sigma(S))$  when  $r = S/\{t: \diamond (p_1, ..., p_n)\}$  is a right rule.
- $v(\sigma(S_1)) \to v^f(\sigma(S_2))$  when  $r = S_1, S_2/\{f: \diamond (p_1, \dots, p_n)\}$  is a left rule.

We are now ready to identify the set of Gvaluations for which a given canonical Gödel calculus and a proof-specification are sound and complete.

**Definition 7.3.16.** Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus, and  $\rho$  an  $\mathcal{L}$ -proof-specification. A partial  $\mathcal{L}$ -Gvaluation v is called **G**-legal if it is r-legal for every canonical  $\mathcal{L}$ -rule r of **G**. v is called  $\mathbf{G} \upharpoonright_{\rho}$ -legal if it is both  $\rho$ -legal and  $\mathbf{G}$ -legal.  $\mathcal{G}_{\mathbf{G} \upharpoonright_{\rho}}$  denotes the set of all  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluations.

Next, we state the general soundness and completeness theorem connecting  $\vdash_{\mathbf{G} \upharpoonright \rho}$  and  $\vdash_{\mathcal{G}_{\mathbf{G} \upharpoonright \rho}}$ . Its proof is given in Section 7.5.

**Theorem 7.3.17.** For every canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$ ,  $\mathcal{L}$ -proof-specification  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  where  $\mathcal{F}$  is closed under subformulas, set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents, and  $\mathcal{F}$ -hypersequent  $H: \mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H \text{ iff } \mathcal{H} \vdash_{\mathcal{G}_{\mathbf{G} \upharpoonright \rho}} H.$ 

For the simpler case without a proof-specification we obtain the following:

Corollary 7.3.18. For every canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$ ,  $\vdash_{\mathbf{G}} = \vdash_{\mathcal{G}}$ , where  $\mathcal{G}$  is the set of all  $\mathbf{G}$ -legal normal  $\mathcal{L}$ -Gvaluations.

*Proof.* Since  $\langle \mathcal{L}, \mathcal{L}, \mathcal{L} \rangle$ -legal partial  $\mathcal{L}$ -Gvaluations are exactly normal  $\mathcal{L}$ -Gvaluations, the claim directly follows from Theorem 7.3.17.

The following is an immediate corollary of the completeness proof.

Corollary 7.3.19. Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus, and  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  an  $\mathcal{L}$ -proof-specification. If  $\mathcal{H} \not\vdash_{\mathbf{G} \upharpoonright \rho} H$ , then there exists a  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation v, which is a model of  $\mathcal{H}$  but not of H, satisfying  $|\mathcal{V}_v| \leq 2|\mathcal{F}| + 2$ .

*Proof.* Directly follows from the fact that the Gödel set  $V_v$  constructed in the completeness proof contains at most  $2|\mathcal{F}| + 2$  elements (see Section 7.5).

It follows that the semantics of  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluations is effective, in the sense that it naturally induces a procedure to decide whether  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\rho}} H$  or not for a given canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$ ,  $\mathcal{L}$ -proof-specification  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  with finite  $\mathcal{F}, \mathcal{C}$  and  $\mathcal{A}$ , finite set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents and single  $\mathcal{L}$ -hypersequent  $\mathcal{H}$ . Note that a syntactic decision procedure in this case is trivial, since the number of  $\mathcal{F}$ -hypersequents is bounded by  $M = 2^{2^{2|\mathcal{F}|}}$ . Obviously, one can enumerate all lists of  $\mathcal{F}$ -hypersequents of size at most M, and return "true" iff one of them is a  $\rho$ -proof in  $\mathbf{G}$  of  $\mathcal{H}$  from  $\mathcal{H}$ . Of-course, the problem is more interesting when  $\rho$  is *not* given, and one has to decide whether  $\mathcal{H} \vdash_{\mathbf{G}} H$  or not. We consider this problem in the next section.

**Remark 7.3.20.** The linearly ordered set of truth values employed in the completeness proof is countable, and can be embedded into the unit interval [0,1]. Thus we can fix  $\mathcal{V} = [0,1]$  in the definition of a Gvaluation, and obtain "standard" semantics.

## 7.4 Applications of the Semantics

In this section we use Theorem 7.3.17 to derive semantic characterizations of strong subaanalyticity, strong cut-admissibility, and axiom-expansion in canonical Gödel calculi, and

use them to obtain a general decidability result for these calculi. First, note that the soundness part of Theorem 7.3.17 can be utilized to prove that certain applications of identity axioms or cuts are unavoidable, or perhaps to show that in all derivations of some hypersequent H, a certain formula  $\varphi$  appears.

**Example 7.4.1.** Let  $H = \{\{\mathbf{t}: (p_1 \supset p_2) \lor (p_2 \supset p_1)\}\}$  and  $\mathcal{F} = sub[H]$ . We show that all proofs of H in  $\mathbf{HG}$  that consist solely of formulas from  $\mathcal{F}$  include the application  $\{\{\mathbf{f}: p_1, \mathbf{t}: p_1\}\}\}$  of (id). Let  $\rho = \langle \mathcal{F}, \mathcal{F}, \mathcal{F} \setminus \{p_1\}\rangle$ . By Theorem 7.3.17, it suffices to provide a  $\mathbf{HG} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation which is not a model of H. For that we can choose  $\mathcal{V}_v = [0, 1]$ ,  $Dom_v = \mathcal{F}$  and:

- $v^{\mathbf{f}}(p_1) = 1, v^{\mathbf{t}}(p_1) = 0, v^{\mathbf{f}}(p_2) = v^{\mathbf{t}}(p_2) = 0.5,$
- $v^{\mathbf{f}}(p_1 \supset p_2) = v^{\mathbf{t}}(p_1 \supset p_2) = 0.5,$
- $v^{f}(p_2 \supset p_1) = v^{t}(p_2 \supset p_1) = 0$ ,
- $v^{\mathbf{f}}((p_1 \supset p_2) \lor (p_2 \supset p_1)) = v^{\mathbf{t}}((p_1 \supset p_2) \lor (p_2 \supset p_1)) = 0.5.$

It is straightforward to verify that v is a  $\mathbf{HG} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation which is not a model of H.

Next, we present a simple *coherence* criterion and prove that it is necessary and sufficient for strong *sub*-analyticity and strong cut-admissibility in canonical Gödel calculi.

**Definition 7.4.2.** A set R of canonical  $\mathcal{L}$ -rules for a connective  $\diamond \in \diamondsuit_{\mathcal{L}}$  is called *coherent* if  $\mathcal{S} \cup \mathcal{S}_1 \cup \mathcal{S}_2$  is classically inconsistent whenever R contains both  $\mathcal{S}/\{\mathfrak{t}: \diamond(p_1, \dots, p_{ar(\diamond)})\}$  and  $\mathcal{S}_1, \mathcal{S}_2/\{\mathfrak{f}: \diamond(p_1, \dots, p_{ar(\diamond)})\}$ . A canonical Gödel  $\mathcal{L}$ -calculus G is called *coherent* if for for each  $\diamond \in \diamondsuit_{\mathcal{L}}$ , the set of rules in G for  $\diamond$  is coherent.

**Example 7.4.3.** Clearly, the set of sequents  $\{\{t:p_2\}, \{t:p_1\}, \{f:p_2\}\}$  is classically inconsistent. Thus the two rules of  $\leadsto$  from Example 7.2.6 form a coherent set of rules.

Note that this is exactly the same criterion used for single-conclusion canonical sequent systems in [14]. It is easy to verify that  $\mathbf{HG}$  is coherent. Moreover, all sets of rules considered in previous examples are coherent. Coherence is a natural requirement for any canonical Gödel calculus. Indeed, in non-coherent calculi the existence of one provable hypersequent of the form  $\{\{\mathbf{t}:\varphi\}\}$  and another provable hypersequent of the form  $\{\{\mathbf{f}:\varphi\}\}$  implies that all (non-empty) hypersequents are provable:

**Proposition 7.4.4.** Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus. If **G** is not coherent then  $\{\{\mathbf{t}:p_1\}\}, \{\{\mathbf{f}:p_2\}\} \vdash_{\mathbf{G}} H \text{ for every non-empty } \mathcal{L}$ -hypersequent H.

*Proof.* Similar to the proof of Theorem 4.10 in [14] for single conclusion canonical systems. The fact that G manipulates hypersequents is immaterial here.

This easily entails that non-coherent calculi do not enjoy strong cut-admissibility. To see this, take  $H = \{\emptyset\}$ , i.e. the hypersequent consisting solely of the empty sequent. Obviously, in any canonical Gödel  $\mathcal{L}$ -calculus there does not exist an  $\langle \mathcal{L}, \{p_1, p_2\}, \mathcal{L} \rangle$ -proof of H from  $\{\{\mathbf{t}:p_1\}\}$  and  $\{\{\mathbf{f}:p_2\}\}$ . Similarly, it is also clear that there does not exist a  $\langle \{p_1, p_2\}, \mathcal{L}, \mathcal{L} \rangle$ -proof of H from  $\{\{\mathbf{t}:p_1\}\}$  and  $\{\{\mathbf{f}:p_2\}\}$ , and so non-coherent calculi are not strongly sub-analytic.

Next we show that coherence is sufficient for strong sub-analyticity and strong cut-admissibility.

**Lemma 7.4.5.** Let **G** be a coherent canonical Gödel  $\mathcal{L}$ -calculus. For every Gödel set  $\mathcal{V}$ , n-ary connective  $\diamond \in \diamondsuit_{\mathcal{L}}$ , canonical right and left  $\mathcal{L}$ -rules  $r_{\mathsf{t}}, r_{\mathsf{f}}$  for  $\diamond$  of **G** (respectively), and  $u_1^{\mathsf{f}}, u_1^{\mathsf{t}}, \dots, u_n^{\mathsf{f}}, u_n^{\mathsf{t}} \in \mathcal{V}$  such that  $u_i^{\mathsf{f}} \geq u_i^{\mathsf{t}}$  for every  $1 \leq i \leq n$ :

$$\diamond^{r_{\mathtt{t}}}_{\mathcal{V}}(\langle u_{1}^{\mathtt{f}}, u_{1}^{\mathtt{t}} \rangle, \ldots, \langle u_{n}^{\mathtt{f}}, u_{n}^{\mathtt{t}} \rangle) \leq \diamond^{r_{\mathtt{f}}}_{\mathcal{V}}(\langle u_{1}^{\mathtt{f}}, u_{1}^{\mathtt{t}} \rangle, \ldots, \langle u_{n}^{\mathtt{f}}, u_{n}^{\mathtt{t}} \rangle).$$

Proof. Suppose that  $\diamond_{\mathcal{V}}^{r_{\mathsf{t}}}(\langle u_{1}^{\mathsf{f}}, u_{1}^{\mathsf{t}} \rangle, \dots, \langle u_{n}^{\mathsf{f}}, u_{n}^{\mathsf{t}} \rangle) > \diamond_{\mathcal{V}}^{r_{\mathsf{f}}}(\langle u_{1}^{\mathsf{f}}, u_{1}^{\mathsf{t}} \rangle, \dots, \langle u_{n}^{\mathsf{f}}, u_{n}^{\mathsf{t}} \rangle)$  for canonical  $\mathcal{L}$ -rules  $r_{\mathsf{t}} = \mathcal{S}/\{\mathsf{t}: \diamond(p_{1}, \dots, p_{n})\}$  and  $r_{\mathsf{f}} = \mathcal{S}_{1}, \mathcal{S}_{2}/\{\mathsf{f}: \diamond(p_{1}, \dots, p_{n})\}$ . Let v be the partial  $\mathcal{L}$ -Gvaluation with  $\mathcal{V}_{v} = \mathcal{V}$ ,  $Dom_{v} = \{p_{1}, \dots, p_{n}\}$ , and  $v(p_{i}) = \langle u_{i}^{\mathsf{f}}, u_{i}^{\mathsf{t}} \rangle$  for every  $1 \leq i \leq n$ . Then, by definition,  $v(\mathcal{S}) > v(\mathcal{S}_{1}) \to v^{\mathsf{f}}(\mathcal{S}_{2})$ . Hence  $v(\mathcal{S}) > v^{\mathsf{f}}(\mathcal{S}_{2})$  and  $v(\mathcal{S}_{1}) > v^{\mathsf{f}}(\mathcal{S}_{2})$ . Consider the classical valuation c on  $p_{1}, \dots, p_{n}$  defined by  $c(p_{i}) = t$  iff  $v^{\mathsf{t}}(p_{i}) > v^{\mathsf{f}}(\mathcal{S}_{2})$ . We prove that c satisfies every  $\mathcal{L}$ -sequent in  $\mathcal{S} \cup \mathcal{S}_{1} \cup \mathcal{S}_{2}$ , and so  $\mathbf{G}$  is not coherent.

Let  $s \in \mathcal{S} \cup \mathcal{S}_1$ . Since  $v(\mathcal{S}) > v^{\mathbf{f}}(\mathcal{S}_2)$  and  $v(\mathcal{S}_1) > v^{\mathbf{f}}(\mathcal{S}_2)$ ,  $v(s) > v^{\mathbf{f}}(\mathcal{S}_2)$ , and so  $v^{\mathbf{f}}(s) \to v^{\mathbf{f}}(s) > v^{\mathbf{f}}(\mathcal{S}_2)$ . If  $v^{\mathbf{t}}(s) > v^{\mathbf{f}}(\mathcal{S}_2)$ , it follows that  $\mathbf{t}: p_i \in s$  for some  $1 \leq i \leq n$  and  $v^{\mathbf{t}}(p_i) > v^{\mathbf{f}}(\mathcal{S}_2)$ , and so  $c(p_i) = t$ . Thus c classically satisfies s. Assume now that  $v^{\mathbf{t}}(s) \leq v^{\mathbf{f}}(\mathcal{S}_2)$ . This implies that  $v^{\mathbf{f}}(s) \leq v^{\mathbf{t}}(s)$ . It follows that  $v^{\mathbf{f}}(s) \leq v^{\mathbf{f}}(\mathcal{S}_2)$ , and so there exists some  $\mathbf{f}: p_i \in s$  such that  $v^{\mathbf{f}}(p_i) \leq v^{\mathbf{f}}(\mathcal{S}_2)$ . Since  $u_i^{\mathbf{f}} \geq u_i^{\mathbf{t}}$ ,  $v^{\mathbf{t}}(p_i) \leq v^{\mathbf{f}}(p_i)$ , and so  $c(p_i) = f$ . Thus c classically satisfies s.

Now, let  $s \in \mathcal{S}_2$ . Obviously,  $v^{\mathbf{f}}(s) \leq v^{\mathbf{f}}(\mathcal{S}_2)$ . This implies that there exists some  $\mathbf{f}: p_i \in s$  such that  $v^{\mathbf{f}}(p_i) \leq v^{\mathbf{f}}(\mathcal{S}_2)$ . Since  $u_i^{\mathbf{f}} \geq u_i^{\mathbf{t}}$ ,  $v^{\mathbf{t}}(p_i) \leq v^{\mathbf{f}}(p_i)$ , and so  $c(p_i) = f$ . Again c classically satisfies s.

**Theorem 7.4.6.** All coherent canonical Gödel  $\mathcal{L}$ -calculi enjoy strong cut-admissibility.

*Proof.* Let **G** be a coherent canonical Gödel  $\mathcal{L}$ -calculus,  $\mathcal{H}$  a set of  $\mathcal{L}$ -hypersequents and H an  $\mathcal{L}$ -hypersequent. Let  $\rho = \langle \mathcal{L}, \mathcal{L}, \mathcal{A} \rangle$  and  $\rho' = \langle \mathcal{L}, frm[\mathcal{H}], \mathcal{A} \rangle$ . Suppose that  $\mathcal{H} \not\vdash_{\mathbf{G} \upharpoonright \rho'} H$ . We show that  $\mathcal{H} \not\vdash_{\mathbf{G} \upharpoonright \rho} H$ .<sup>6</sup> By Theorem 7.3.17, there is some  $\mathbf{G} \upharpoonright_{\rho'}$ -legal

<sup>&</sup>lt;sup>6</sup>In fact, this proves a stronger claim:  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$  for  $\rho = \langle \mathcal{L}, \mathcal{L}, \mathcal{A} \rangle$  implies that  $\mathcal{H} \nvDash_{\mathbf{G} \upharpoonright \rho'} H$  for  $\rho' = \langle \mathcal{L}, frm[\mathcal{H}], \mathcal{A} \rangle$ . Roughly speaking, this means that one never has to introduce new applications of the identity axioms for eliminating cuts. The usual notion of strong cut-admissibility is obtained by taking  $\mathcal{A} = \mathcal{L}$ .

 $\mathcal{L}$ -Gvaluation q, such that  $q \models \mathcal{H}$  but  $q \not\models H$ . We construct a  $\mathbf{G} \upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Gvaluation v, such that  $v \models \mathcal{H}$  but  $v \not\models H$ . By Theorem 7.3.17, it then follows that  $\mathcal{H} \not\vdash_{\mathbf{G} \upharpoonright_{\rho}} H$ . First, we define  $\mathcal{V}_v = \mathcal{V}_q$ . Next, for every  $\varphi \in \mathcal{L}$ ,  $v^{\mathbf{f}}(\varphi)$  and  $v^{\mathbf{t}}(\varphi)$  are determined as follows. If  $q^{\mathbf{t}}(\varphi) \leq q^{\mathbf{f}}(\varphi)$  then  $v^{\mathbf{f}}(\varphi) = q^{\mathbf{f}}(\varphi)$  and  $v^{\mathbf{t}}(\varphi) = q^{\mathbf{t}}(\varphi)$ . Otherwise, if  $\varphi$  is an atomic formula,  $v^{\mathbf{f}}(\varphi) = v^{\mathbf{t}}(\varphi) = q^{\mathbf{f}}(\varphi)$ . Finally, if  $\varphi$  is a compound formula of the form  $\diamond(\varphi_1, \dots, \varphi_{ar(\diamond)})$  and  $q^{\mathbf{f}}(\varphi) < q^{\mathbf{t}}(\varphi)$ , we define:

$$v^{\mathbf{f}}(\varphi) = v^{\mathbf{t}}(\varphi) = \begin{cases} q^{\mathbf{f}}(\varphi) & v^{\mathbf{G}}(\varphi) < q^{\mathbf{f}}(\varphi) \\ v^{\mathbf{G}}(\varphi) & q^{\mathbf{f}}(\varphi) \le v^{\mathbf{G}}(\varphi) \le q^{\mathbf{t}}(\varphi) \\ q^{\mathbf{t}}(\varphi) & q^{\mathbf{t}}(\varphi) < v^{\mathbf{G}}(\varphi) \end{cases}$$

where  $v^{\mathbf{G}}(\varphi) = \min\{\diamond_{\mathcal{V}_q}^r(v(\varphi_1), \dots, v(\varphi_{ar(\diamond)})) \mid r \text{ is a canonical left } \mathcal{L}\text{-rule for } \diamond \text{ in } \mathbf{G}\}$ . Note that  $v^{\mathbf{G}}(\varphi)$  depends only on the values assigned to proper subformulas of  $\varphi$ , and hence this construction is well-defined. We first show that v is  $\rho$ -legal. Obviously,  $v^{\mathbf{t}}(\varphi) \leq v^{\mathbf{f}}(\varphi)$  for every formula  $\varphi$ . It remains to prove that  $v^{\mathbf{f}}(\varphi) \leq v^{\mathbf{t}}(\varphi)$  for every  $\varphi \in \mathcal{A}$ . To see this, note that the only case in which we have  $v^{\mathbf{f}}(\varphi) \neq v^{\mathbf{t}}(\varphi)$  is when  $q^{\mathbf{t}}(\varphi) < q^{\mathbf{f}}(\varphi)$ . Since q is  $\rho'$ -legal, this can only happen for  $\varphi \notin \mathcal{A}$ . Next, we show that v is  $\mathbf{G}$ -legal. For this we use the following properties:

- 1.  $q^{\mathbf{f}}(\varphi) \leq v^{\mathbf{f}}(\varphi)$  and  $v^{\mathbf{t}}(\varphi) \leq q^{\mathbf{t}}(\varphi)$  for every formula  $\varphi$ .
- 2.  $q^{\mathtt{f}}(s) \leq v^{\mathtt{f}}(s)$  and  $v^{\mathtt{t}}(s) \leq q^{\mathtt{t}}(s)$  for every sequent s.
- 3.  $v(s) \le q(s)$  for every sequent s.
- 4.  $v(S) \leq q(S)$  and  $q^{f}(S) \leq v^{f}(S)$  for every finite set S of sequents.
- 5.  $q(S_1) \to q^{\mathfrak{f}}(S_2) \leq v(S_1) \to v^{\mathfrak{f}}(S_2)$  for every finite sets  $S_1, S_2$  of sequents.

The proofs of these properties easily follow from the definitions (note that if  $u_1 \leq u_2$  and  $u_3 \leq u_4$  then  $u_2 \to u_3 \leq u_1 \to u_4$ ). Now, we show that v is **G**-legal.

- Let  $r = \mathcal{S}/\{\mathsf{t}: \diamond(p_1, \dots, p_n)\}$  be a canonical right  $\mathcal{L}$ -rule of  $\mathbf{G}$ , and  $\sigma$  be an  $\mathcal{L}$ -substitution. We show that  $\diamond_{\mathcal{V}_q}^r(v(\sigma(p_1)), \dots, v(\sigma(p_n))) \leq v^{\mathsf{t}}(\sigma(\diamond(p_1, \dots, p_n)))$ . Let  $\varphi = \sigma(\diamond(p_1, \dots, p_n))$  and  $u = \diamond_{\mathcal{V}_q}^r(v(\sigma(p_1)), \dots, v(\sigma(p_n)))$ . Since q is  $\mathbf{G}$ -legal, we have  $\diamond_{\mathcal{V}_q}^r(q(\sigma(p_1)), \dots, q(\sigma(p_n))) \leq q^{\mathsf{t}}(\varphi)$ . By Lemma 7.3.15,  $u = v(\sigma(\mathcal{S}))$ , and  $\diamond_{\mathcal{V}_q}^r(q(\sigma(p_1)), \dots, q(\sigma(p_n))) = q(\sigma(\mathcal{S}))$ . Thus if  $v^{\mathsf{t}}(\varphi) = q^{\mathsf{t}}(\varphi)$ , the claim follows by Item 4. Otherwise, the construction of v ensures that  $v^{\mathsf{t}}(\varphi) = \max\{v^{\mathsf{G}}(\varphi), q^{\mathsf{f}}(\varphi)\}$ . Thus  $v^{\mathsf{G}}(\varphi) \leq v^{\mathsf{t}}(\varphi)$ . Now, by Lemma 7.4.5, the coherence of  $\mathbf{G}$  entails that  $u \leq v^{\mathsf{G}}(\varphi)$ .
- Let  $r = S_1, S_2/\{f: \diamond(p_1, ..., p_n)\}$  be a canonical left  $\mathcal{L}$ -rule of  $\mathbf{G}$ , and  $\sigma$  an  $\mathcal{L}$ substitution. We show that  $v^{\mathbf{f}}(\sigma(\diamond(p_1, ..., p_n))) \leq \diamond_{\mathcal{V}_q}^r(v(\sigma(p_1)), ..., v(\sigma(p_n)))$ . Let  $\varphi = \sigma(\diamond(p_1, ..., p_n))$  and  $u = \diamond_{\mathcal{V}_q}^r(v(\sigma(p_1)), ..., v(\sigma(p_n)))$ . Since q is  $\mathbf{G}$ -legal, we have

 $q^{\mathbf{f}}(\varphi) \leq \diamond_{\mathcal{V}_q}^r(q(\sigma(p_1)), \dots, q(\sigma(p_n)))$ . By Lemma 7.3.15,  $u = v(\sigma(\mathcal{S}_1)) \to v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$  and  $\diamond_{\mathcal{V}_q}^r(q(\sigma(p_1)), \dots, q(\sigma(p_n))) = q(\sigma(\mathcal{S}_1)) \to q^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ . Thus if  $v^{\mathbf{t}}(\varphi) = q^{\mathbf{t}}(\varphi)$ , then the claim follows by Item 5. Otherwise, the construction of v ensures that we have  $v^{\mathbf{f}}(\varphi) = \min\{v^{\mathbf{G}}(\varphi), q^{\mathbf{t}}(\varphi)\}$ , and so  $v^{\mathbf{f}}(\varphi) \leq u$ .

It remains to show that  $v \models \mathcal{H}$  but  $v \not\models H$ . Let  $H' \in \mathcal{H}$ . Since  $q \models \mathcal{H}$ , there exists some  $s \in H'$  such that  $q^{\mathbf{f}}(s) \leq q^{\mathbf{t}}(s)$ . Since q is  $\rho'$ -legal and  $frm[s] \subseteq frm[\mathcal{H}]$ ,  $q^{\mathbf{t}}(\varphi) \leq q^{\mathbf{f}}(\varphi)$  for every  $\varphi \in frm[s]$ . The construction of v ensures that  $v(\varphi) = q(\varphi)$  whenever  $\varphi \in frm[s]$ . Hence,  $v^{\mathbf{f}}(s) \leq v^{\mathbf{t}}(s)$ , and consequently  $v \models H'$ . Finally, we show that  $v \not\models H$ . Let  $s \in H$ . Since  $q \not\models H$ , we have  $q \not\models s$ . Thus q(s) < 1. Item 3 above entails that v(s) < 1 as well, and so  $v \not\models s$ .

**Example 7.4.7.** Since **HG** is coherent, it enjoys strong cut-admissibility. The extension of **HG** with the rules for  $\rightsquigarrow$  and  $\bowtie$  from Examples 7.2.6 and 7.2.7 enjoys strong cut-admissibility as well.

This leads to the following triple equivalence in canonical Gödel calculi:

Corollary 7.4.8. The following are equivalent for a canonical Gödel  $\mathcal{L}$ -calculus G:

- **G** is coherent.
- **G** is strongly *sub*-analytic.
- G enjoys strong cut-admissibility.

*Proof.* Following Proposition 7.2.9, if  $\mathbf{G}$  enjoys strong cut-admissibility then it is strongly sub-analytic. In addition, following Proposition 7.4.4, if  $\mathbf{G}$  is not coherent then it is not strongly sub-analytic. The third link follows by Theorem 7.4.6.

The decidability of coherent calculi is an easy corollary.

Corollary 7.4.9. Given a coherent canonical Gödel  $\mathcal{L}$ -calculus  $\mathbf{G}$ , a finite set  $\mathcal{H}$  of  $\mathcal{L}$ -hypersequents and an  $\mathcal{L}$ -hypersequent H, it is decidable whether  $\mathcal{H} \vdash_{\mathbf{G}} H$  or not.

*Proof.* By Corollary 7.4.8, if **G** is coherent then it is strongly sub-analytic. Thus it suffices to check whether  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$  for  $\rho = \langle sub[\mathcal{H} \cup \{H\}], sub[\mathcal{H} \cup \{H\}], sub[\mathcal{H} \cup \{H\}] \rangle$ . This can be as described in the discussion after Corollary 7.3.19.

Note that the same equivalence of Corollary 7.4.8 holds for the family of canonical (two-sided) sequent systems [17] and their single-conclusion counterparts [14]. Another similarity between canonical Gödel calculi and canonical sequent calculi arises when studying the property of axiom-expansion. Indeed, as shown in Corollary 4.4.8, canonical sequent calculi exhibit a strong connection between determinism of the semantics of a

certain connective, and the fact that this connective admits axiom-expansion (see also [11, 43]). Next, we show that a similar connection exists in canonical Gödel calculi.

Roughly speaking, a connective  $\diamond$  is deterministic in a calculus  $\mathbf{G}$  if for every formula  $\diamond(\varphi_1,\ldots,\varphi_{ar(\diamond)})$ , the truth values assigned to  $\varphi_1,\ldots,\varphi_{ar(\diamond)}$  in  $\mathbf{G}$ -legal normal  $\mathcal{L}$ -Gvaluations uniquely determine the truth value assigned to  $\diamond(\varphi_1,\ldots,\varphi_n)$ . Formally, this property is defined follows:

**Definition 7.4.10.** Let **G** be a canonical Gödel  $\mathcal{L}$ -calculus. An n-ary connective  $\diamond \in \diamond_{\mathcal{L}}$  is deterministic in **G** if for every Gödel set  $\mathcal{V}$  and elements  $u_1, \ldots, u_n \in \mathcal{V}$ , there exist canonical right and left  $\mathcal{L}$ -rule  $r_t, r_f$  for  $\diamond$  in **G** (respectively), such that

$$\diamond_{\mathcal{V}}^{r_{\mathsf{t}}}(\langle u_1, u_1 \rangle, \dots, \langle u_n, u_n \rangle) = \diamond_{\mathcal{V}}^{r_{\mathsf{f}}}(\langle u_1, u_1 \rangle, \dots, \langle u_n, u_n \rangle).$$

Indeed, if v is a **G**-legal normal  $\mathcal{L}$ -Gvaluation, then for every compound formula  $\varphi = \diamond(\varphi_1, \dots, \varphi_n), \ v(\varphi)$  is forced to be greater than or equal to  $\diamond_{\mathcal{V}}^r(v(\varphi_1), \dots, v(\varphi_n))$  for every right rule r for  $\diamond$ , and less than or equal to  $\diamond_{\mathcal{V}}^r(v(\varphi_1), \dots, v(\varphi_n))$  for every left rule r for  $\diamond$ . If the condition above holds, then this leaves exactly one option for  $v(\varphi)$  given  $v(\varphi_1), \dots, v(\varphi_n)$ .

**Theorem 7.4.11.** Let G be a coherent canonical Gödel  $\mathcal{L}$ -calculus. A connective admits axiom-expansion in G iff it is deterministic in G.

*Proof.* Let  $\diamond$  be an *n*-ary connective, and let  $\varphi = \diamond(p_1, \dots, p_n)$ . By definition,  $\diamond$  admits axiom-expansion in  $\mathbf{G}$  iff  $\vdash_{\mathbf{G} \upharpoonright \rho} H$  for  $H = \{\{\mathbf{f}: \varphi, \mathbf{t}: \varphi\}\}$  and  $\rho = \langle \mathcal{L}, \mathcal{L}, \{p_1, \dots, p_n\} \rangle$ .

- Suppose that  $\diamond$  is deterministic in  $\mathbf{G}$ . We show that  $\vdash_{\mathbf{G} \upharpoonright \rho} H$ . By Theorem 7.3.17, it suffices to show that every  $\mathbf{G} \upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Gvaluation is a model of H. Let v be a  $\mathbf{G} \upharpoonright_{\rho}$ -legal  $\mathcal{L}$ -Gvaluation. Since  $\diamond$  is deterministic in  $\mathbf{G}$ , there are canonical right and left  $\mathcal{L}$ -rules  $r_{\mathbf{t}}, r_{\mathbf{f}}$  for  $\diamond$  in  $\mathbf{G}$ , such that  $\diamond_{\mathcal{V}}^{r_{\mathbf{t}}}(v(p_1), \dots, v(p_n)) = \diamond_{\mathcal{V}}^{r_{\mathbf{f}}}(v(p_1), \dots, v(p_n))$ . Since v is  $\mathbf{G}$ -legal,  $\diamond_{\mathcal{V}}^{r_{\mathbf{f}}}(v(p_1), \dots, v(p_n)) \leq v^{\mathbf{t}}(\varphi)$  and  $v^{\mathbf{f}}(\varphi) \leq \diamond_{\mathcal{V}}^{r_{\mathbf{t}}}(v(p_1), \dots, v(p_n))$ . It follows that  $v^{\mathbf{f}}(\varphi) \leq v^{\mathbf{t}}(\varphi)$ , and so  $v \models H$ .
- Suppose that  $\diamond$  is not deterministic in  $\mathbf{G}$ . Hence, there is a Gödel set  $\mathcal{V}$  and  $u_1, \ldots, u_n \in \mathcal{V}$ , such that  $\diamond_{\mathcal{V}}^{r_{\mathbf{t}}}(\langle u_1, u_1 \rangle, \ldots, \langle u_n, u_n \rangle) \neq \diamond_{\mathcal{V}}^{r_{\mathbf{t}}}(\langle u_1, u_1 \rangle, \ldots, \langle u_n, u_n \rangle)$  for every canonical right and left  $\mathcal{L}$ -rules  $r_{\mathbf{t}}, r_{\mathbf{f}}$  for  $\diamond$  in  $\mathbf{G}$ . Let  $\mathcal{F} = \{p_1, \ldots, p_n, \varphi\}$ , and define a partial  $\mathcal{L}$ -Gvaluation v by:  $\mathcal{V}_v = \mathcal{V}$ ,  $Dom_v = \mathcal{F}$ ,  $v(p_i) = \langle u_i, u_i \rangle$  for every  $1 \leq i \leq n$ , and

 $v(\varphi) = \langle \min\{\diamond_{\mathcal{V}}^r(v(p_1), \dots, v(p_n)) \mid r \text{ is a canonical left } \mathcal{L}\text{-rule for } \diamond \text{ in } \mathbf{G} \},$  $\max\{\diamond_{\mathcal{V}}^r(v(p_1), \dots, v(p_n)) \mid r \text{ is a canonical right } \mathcal{L}\text{-rule for } \diamond \text{ in } \mathbf{G} \} \rangle.$ 

Then our assumption entails that  $v^{\mathbf{t}}(\varphi) \neq v^{\mathbf{f}}(\varphi)$ . Let  $\rho' = \langle \mathcal{F}, \mathcal{F}, \{p_1, \dots, p_n\} \rangle$ . It is easy to see that v is a  $\mathbf{G} \upharpoonright \rho'$ -legal partial  $\mathcal{L}$ -Gvaluation. In particular,  $v^{\mathbf{t}}(\varphi) \leq v^{\mathbf{f}}(\varphi)$ 

since **G** is coherent (see Lemma 7.4.5). Hence  $v^{\mathbf{t}}(\varphi) < v^{\mathbf{f}}(\varphi)$ , and so v is not a model of H. By Theorem 7.3.17,  $\not\vdash_{\mathbf{G}\upharpoonright \rho'} H$ . Now, suppose for contradiction that  $\vdash_{\mathbf{G}\upharpoonright \rho} H$ . By the proof of Theorem 7.4.6 (again using the fact that **G** is coherent), it follows that  $\vdash_{\mathbf{G}\upharpoonright \rho_1} H$  for  $\rho_1 = \langle \mathcal{L}, \emptyset, \{p_1, \dots, p_n\} \rangle$ . By Proposition 7.2.9,  $\vdash_{\mathbf{G}\upharpoonright \rho_2} H$  for  $\rho_2 = \langle sub[H], \emptyset, \{p_1, \dots, p_n\} \rangle$ . This contradicts the fact that  $\not\vdash_{\mathbf{G}\upharpoonright \rho'} H$ .

# 7.5 Soundness and Completeness Proofs

In this section we prove Theorem 7.3.17. For the rest of this section, let G be a canonical Gödel  $\mathcal{L}$ -calculus,  $\rho = \langle \mathcal{F}, \mathcal{C}, \mathcal{A} \rangle$  an  $\mathcal{L}$ -proof-specification where  $\mathcal{F}$  is closed under subformulas,  $\mathcal{H}$  a set of  $\mathcal{L}$ -hypersequents, and  $H_0$  an  $\mathcal{F}$ -hypersequent.

#### Soundness

Suppose that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_0$ , and so there exists a  $\rho$ -proof P in  $\mathbf{G}$  of  $H_0$  from  $\mathcal{H}$ . Let v be a  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation, which is a model of  $\mathcal{H}$ . Using induction on the length of P, we show that  $v \models H$  for every  $\mathcal{L}$ -hypersequent H appearing in P. It then follows that  $v \models H_0$ . Note that since all  $\mathcal{L}$ -hypersequents in P are  $\mathcal{F}$ -hypersequents, it suffices to prove that v is a model of some component  $s \in H$  for every  $\mathcal{L}$ -hypersequent H in P. This trivially holds for the  $\mathcal{L}$ -hypersequents of  $\mathcal{H}$  that appear in P. We show that this property is also preserved by applications in P of the rules of  $\mathbf{G}$ . Consider such an application, and assume that v is a model of some component of every premise of this application. We show that v is also a model of some component of the conclusion:

Weakenings For applications of the weakening rules, this is obvious.

- (cut) Suppose that  $H \mid s_1 \cup s_2$  is derived from  $H \mid s_1 \cup \{f:\varphi\}$  and  $H \mid s_2 \cup \{t:\varphi\}$  using (cut) (here  $s_2$  must be a negative sequent). Thus  $\varphi \in \mathcal{C}$ . If  $v \models s$  for some component  $s \in H$ , then we are done. Otherwise,  $v \models s_1 \cup \{f:\varphi\}$  and  $v \models s_2 \cup \{t:\varphi\}$ . We show that  $v \models s_1 \cup s_2$ . By definition, we have  $v^f(s_1 \cup \{f:\varphi\}) \leq v^t(s_1)$  and  $v^f(s_2) \leq v^t(\varphi)$ . Since  $\varphi \in \mathcal{C}$ ,  $v^t(\varphi) \leq v^f(\varphi)$ , and so  $v^f(s_2) \leq v^f(\varphi)$ . It follows that  $v^f(s_1 \cup s_2) \leq v^f(s_1 \cup \{f:\varphi\})$ . Therefore,  $v^f(s_1 \cup s_2) \leq v^t(s_1)$ , and so  $v \models s_1 \cup s_2$ .
- (id) Suppose that  $\{\{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}\}\$  is derived using (id). Thus  $\varphi \in \mathcal{A}$ , and so  $v^{\mathbf{f}}(\varphi) \leq v^{\mathbf{t}}(\varphi)$ . Consequently,  $v \models \{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}$ .
- (com) Suppose that  $H \mid s_1 \cup c_2 \mid s_2 \cup c_1$  is derived from  $H \mid s_1 \cup c_1$  and  $H \mid s_2 \cup c_2$  using (com), where  $c_1, c_2$  are negative sequents. If  $v \models s$  for some  $s \in H$ , then we are done. Otherwise,  $v \models s_1 \cup c_1$  and  $v \models s_2 \cup c_2$ . We show that  $v \models s_1 \cup c_2$  or  $v \models s_2 \cup c_1$ .

Our assumption entails that  $v^{\mathbf{f}}(s_1 \cup c_1) \leq v^{\mathbf{t}}(s_1 \cup c_1)$  and  $v^{\mathbf{f}}(s_2 \cup c_2) \leq v^{\mathbf{t}}(s_2 \cup c_2)$ . By definition,  $v^{\mathbf{t}}(s_1 \cup c_1) = v^{\mathbf{t}}(s_1)$  and  $v^{\mathbf{t}}(s_2 \cup c_2) = v^{\mathbf{t}}(s_2)$ . Hence, we have both  $v^{\mathbf{f}}(s_1 \cup c_1) \leq v^{\mathbf{t}}(s_1)$  and  $v^{\mathbf{f}}(s_2 \cup c_2) \leq v^{\mathbf{t}}(s_2)$ . If  $v^{\mathbf{f}}(s_1) \leq v^{\mathbf{t}}(s_1)$  then  $v \models s_1 \cup c_2$  and we are done. Similarly, if  $v^{\mathbf{f}}(s_2) \leq v^{\mathbf{t}}(s_2)$ , then  $v \models s_2 \cup c_1$  and we are done. Otherwise, we obtain that  $v^{\mathbf{f}}(c_1) \leq v^{\mathbf{t}}(s_1)$  and  $v^{\mathbf{f}}(c_2) \leq v^{\mathbf{t}}(s_2)$ . Now, if  $v^{\mathbf{t}}(s_1) \leq v^{\mathbf{f}}(c_2)$ , we obtain that  $v^{\mathbf{f}}(c_1) \leq v^{\mathbf{t}}(s_2)$  and so  $v \models s_2 \cup c_1$ , and otherwise (using the fact that  $\leq$  is linear) we have  $v^{\mathbf{f}}(c_2) < v^{\mathbf{t}}(s_1)$  and so  $v \models s_1 \cup c_2$ .

Right rules Suppose  $H \mid \{\mathsf{t}: \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))\} \cup c$  is derived from  $\{H \mid \sigma(s) \cup c\}_{s \in \mathcal{S}}$ , using the right rule  $r = \mathcal{S}/\{\mathsf{t}: \diamond(p_1, \dots, p_{ar(\diamond)})\}$  (here c must be a negative sequent). If  $v \models s$  for some  $s \in H$ , then we are done. Otherwise,  $v \models \sigma(s) \cup c$  for every  $s \in \mathcal{S}$ . Let  $\varphi = \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))$ , and suppose for a contradiction that  $v \not\models \{\mathsf{t}:\varphi\} \cup c$ . Since c is a negative sequent, we have  $v^{\mathsf{f}}(c) > v^{\mathsf{t}}(\varphi)$ . Now, since  $\varphi \in \mathcal{F}$  and  $\mathcal{F}$  is closed under subformulas, we have also  $\sigma(\{p_1, \dots, p_n\}) \subseteq \mathcal{F}$ . Since v is r-legal we have  $\diamond_{\mathcal{V}_v}^r(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)}))) \leq v^{\mathsf{t}}(\varphi)$ . By Lemma 7.3.15,  $\diamond_{\mathcal{V}_v}^r(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)}))) = v(\sigma(\mathcal{S}))$ , and so  $v^{\mathsf{f}}(c) > v(\sigma(\mathcal{S}))$ . By definition, we have  $v^{\mathsf{f}}(c) > v(\sigma(s)) = v^{\mathsf{f}}(\sigma(s)) \to v^{\mathsf{t}}(\sigma(s))$  for some  $s \in \mathcal{S}$ . It follows that  $v^{\mathsf{f}}(\sigma(s)) > v^{\mathsf{t}}(\sigma(s))$  and  $v^{\mathsf{f}}(c) > v^{\mathsf{t}}(\sigma(s))$ . But then  $v^{\mathsf{f}}(\sigma(s) \cup c) > v^{\mathsf{t}}(\sigma(s) \cup c)$ , in contradiction to the fact that  $v \models \sigma(s) \cup c$ .

**Left rules** Suppose  $H \mid \{f: \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))\} \cup c \cup c' \text{ is derived from } \{H \mid \sigma(s) \cup c\}_{s \in \mathcal{S}_1}$ and  $\{H \mid \sigma(s_i') \cup c'\}_{s \in \mathcal{S}_2}$ , using the left rule  $r = \mathcal{S}_1, \mathcal{S}_2/\{f: \diamond(p_1, \dots, p_{ar(\diamond)})\}$  (here c must be a negative sequent). If  $v \models s$  for some  $s \in H$ , then we are done. Otherwise,  $v \models \sigma(s) \cup c$  for every  $s \in \mathcal{S}_1$ , and  $v \models \sigma(s) \cup c'$  for every  $s \in \mathcal{S}_2$ . Hence: (1) for every  $s \in \mathcal{S}_1$ , either  $v^{\mathbf{f}}(c) \leq v^{\mathbf{t}}(\sigma(s))$  or  $v^{\mathbf{f}}(\sigma(s)) \leq v^{\mathbf{t}}(\sigma(s))$ ; and (2) either  $v^{\mathbf{f}}(c') \leq v^{\mathbf{t}}(c')$ , or  $v^{\mathbf{f}}(\sigma(s)) \leq v^{\mathbf{t}}(c')$  for every  $s \in \mathcal{S}_2$ . Let  $\varphi = \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))$ . Suppose for a contradiction that  $v \not\models \{f:\varphi\} \cup c \cup c'$ . Then by definition we have  $v^{\mathbf{f}}(\{\mathbf{f}:\varphi\} \cup c \cup c') > v^{\mathbf{t}}(c')$ . Therefore: (3)  $v^{\mathbf{t}}(c') < v^{\mathbf{f}}(c)$  and  $v^{\mathbf{t}}(c') < v^{\mathbf{f}}(c')$ ; and (4)  $v^{\mathsf{t}}(c') < v^{\mathsf{f}}(\varphi)$ . From (2) and (3) we obtain (5):  $v^{\mathsf{f}}(\sigma(s)) \leq v^{\mathsf{t}}(c')$  for every  $s \in \mathcal{S}_2$ . Now, since  $\varphi \in \mathcal{F}$  and  $\mathcal{F}$  is closed under subformulas, we have also  $\sigma(\{p_1,\ldots,p_n\})\subseteq \mathcal{F}$ . Since v is r-legal,  $v^{\mathbf{f}}(\varphi)\leq \diamond_{\mathcal{V}_v}^r(v(\sigma(p_1)),\ldots,v(\sigma(p_{ar(\diamond)})))$ . Let  $x = v(\sigma(S_1)) \to v^{\mathbf{f}}(\sigma(S_2))$ . By Lemma 7.3.15,  $\diamond_{\mathcal{V}_v}^r(v(\sigma(p_1)), \dots, v(\sigma(p_{ar(\diamond)}))) = x$ , and so  $v^{\mathbf{f}}(\varphi) \leq x$ . Together with (4), we have that  $v^{\mathbf{t}}(c') < x$ . By (5), we obtain that  $v^{\mathbf{f}}(\sigma(s)) < x$  for every  $s \in \mathcal{S}_2$ . Let  $s_0$  be a sequent in  $\mathcal{S}_2$  such that  $v^{\mathbf{f}}(s)$  obtains a maximal value (i.e.  $v^{\mathbf{f}}(s_0) = v^{\mathbf{f}}(\sigma(S_2))$ ). In particular,  $v^{\mathbf{f}}(s_0) < v(\sigma(S_1)) \to v^{\mathbf{f}}(s_0)$ . This entails that  $v(\sigma(S_1)) \leq v^{\mathbf{f}}(s_0)$ . Now, (3) and (5) imply that  $v^{\mathbf{f}}(s_0) < v^{\mathbf{f}}(c)$ . It then follows that  $v(\sigma(S_1)) < v^{\mathbf{f}}(c)$ . Hence  $v(\sigma(s)) < v^{\mathbf{f}}(c)$  for some  $s \in S_1$ . Equivalently,  $v^{\mathbf{f}}(\sigma(s)) \to v^{\mathbf{t}}(\sigma(s)) < v^{\mathbf{f}}(c)$ . This implies that  $v^{\mathbf{t}}(\sigma(s)) < v^{\mathbf{f}}(\sigma(s))$ 

and  $v^{t}(\sigma(s)) < v^{f}(c)$ . But this contradicts (1) above.

### Completeness

Suppose that  $\mathcal{H} \not\vdash_{\mathbf{G} \upharpoonright \rho} H_0$ . We construct a  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation v, which is a model of  $\mathcal{H}$ , but not of  $H_0$ .

As the previous completeness proofs, v is constructed using a "maximal" hypersequent. For this purpose, we introduce extended hypersequents:

**Definition 7.5.1.** An extended  $\mathcal{L}$ -hypersequent is a (possibly infinite) set of extended  $\mathcal{L}$ -sequents.<sup>7</sup> Given two extended hypersequents  $\Omega_1, \Omega_2$ , we write  $\Omega_1 \sqsubseteq \Omega_2$  (and say that  $\Omega_2$  extends  $\Omega_1$ ) if for every  $\mu_1 \in \Omega_1$ , there exists  $\mu_2 \in \Omega_2$  such that  $\mu_1 \subseteq \mu_2$ .

We shall use the same notations as above for extended hypersequents. For example, we write  $\Omega \mid \mu$  instead of  $\Omega \cup \{\mu\}$ . An extended  $\mathcal{F}$ -hypersequent is also defined as expected (namely, an extended hypersequent that consists only of formulas from  $\mathcal{F}$ ). In addition, for the rest of this proof call an extended  $\mathcal{L}$ -hypersequent  $\Omega$ :

- 1. *finite* if  $|\Omega|$  is finite, and so is  $|\mu|$  for every  $\mu \in \Omega$ .
- 2. provable if  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} H$  for some (ordinary)  $\mathcal{L}$ -hypersequent  $H \sqsubseteq \Omega$ .
- 3. *unprovable* if it is not provable.
- 4. maximal with respect to an  $\mathcal{L}$ -formula  $\varphi$  if for every  $\mu \in \Omega$  and  $\mathbf{x} \in \mathcal{L}_2$ , the extended  $\mathcal{L}$ -hypersequent  $\Omega \mid \mu \cup \{\mathbf{x}: \varphi\}$  is provable whenever  $\mathbf{x}: \varphi \notin \mu$ .
- 5. internally maximal if it is maximal with respect to any  $\varphi \in \mathcal{F}$ .
- 6. maximal with respect to an  $\mathcal{L}$ -sequent s if  $\Omega \mid s$  is provable whenever  $\{s\} \not\sqsubseteq \Omega$ .
- 7. externally maximal if it is maximal with respect to any  $\mathcal{F}$ -sequent.
- 8. maximal if it is an extended  $\mathcal{F}$ -hypersequent, unprovable, internally maximal, and externally maximal.

Less formally, an extended hypersequent  $\Omega$  is internally maximal if every formula added on some side of some component of  $\Omega$  would make it provable. Similarly,  $\Omega$  is externally maximal if every sequent added to  $\Omega$  would make it provable.

**Proposition 7.5.2.** Let  $\Omega$  be an extended  $\mathcal{L}$ -hypersequent.

- Assume that  $\Omega$  is maximal with respect to a formula  $\varphi \in \mathcal{F}$ . For every  $\mu \in \Omega$ :
  - If  $\mathbf{f}: \varphi \notin \mu$ , then  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H \mid s_1 \cup \{\mathbf{f}: \varphi\} \mid ... \mid s_n \cup \{\mathbf{f}: \varphi\} \text{ for some } \mathcal{L}$ hypersequent  $H \sqsubseteq \Omega$  and  $\mathcal{L}$ -sequents  $s_1, ..., s_n \subseteq \mu$ .

<sup>&</sup>lt;sup>7</sup>Recall that extended  $\mathcal{L}$ -sequents are (possibly infinite) set of labelled  $\mathcal{L}$ -formulas. In particular, extended  $\mathcal{L}$ -sequents are not necessarily single-conclusion.

- If  $t:\varphi \not\in \mu$ , then  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H \mid s \cup \{t:\varphi\}$  for some  $\mathcal{L}$ -hypersequent  $H \sqsubseteq \Omega$  and negative  $\mathcal{L}$ -sequent  $s \subseteq \mu$ .
- Assume that  $\Omega$  is maximal with respect to an  $\mathcal{F}$ -sequent s. Then, if  $\{s\} \not\sqsubseteq \Omega$ , then  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\Omega}} H \mid s$  for some  $\mathcal{L}$ -hypersequent  $H \sqsubseteq \Omega$ .

*Proof.* Directly follow from our definitions and the availability of the weakening rules.

Next, we show that  $H_0$  can be extended to a maximal extended  $\mathcal{L}$ -hypersequent  $\Omega$ .

**Lemma 7.5.3.** Suppose that  $\Omega = \mu_1 \mid ... \mid \mu_n$  is an unprovable finite extended  $\mathcal{F}$ -hypersequent. Let  $\varphi \in \mathcal{F}$ , and let s be an  $\mathcal{F}$ -sequent. Then there exists an unprovable finite extended  $\mathcal{F}$ -hypersequent  $\Omega'$ , such that:

- $\Omega' = \mu'_1 \mid \dots \mid \mu'_{n'}$ , where  $n' \in \{n, n+1\}$ ,  $\mu_i \subseteq \mu'_i$  for every  $1 \le i \le n$ .
- $\Omega'$  is maximal with respect to  $\varphi$ .
- $\Omega'$  is maximal with respect to s.

Proof. First, if  $\Omega \mid s$  is unprovable, let n' = n + 1 and define  $\mu_{n+1} = s$ . Otherwise, let n' = n. We recursively define a finite sequence of finite extended  $\mathcal{F}$ -hypersequents,  $\Omega_0 = \mu_1^0 \mid \ldots \mid \mu_{n'}^0, \ldots, \Omega_{n'} = \mu_1^{n'} \mid \ldots \mid \mu_{n'}^{n'}$ , in which  $\mu_j^i \subseteq \mu_j^{i+1}$  for every  $1 \leq j \leq n'$  and  $0 \leq i \leq n' - 1$ . First, define  $\mu_j^0 = \mu_j$  for every  $1 \leq j \leq n'$ . Let  $0 \leq i \leq n' - 1$ . Assume that  $\Omega_i = \mu_1^i \mid \ldots \mid \mu_{n'}^i$  is defined. To construct  $\Omega_{i+1} = \mu_1^{i+1} \mid \ldots \mid \mu_{n'}^{i+1}$ , we take some maximal set  $\mathbf{X} \subseteq \pounds_2$  (with respect to set inclusion) for which  $\mu_1^i \mid \ldots \mid \mu_{i+1}^i \cup (\mathbf{X} : \varphi) \mid \ldots \mid \mu_{n'}^i$  is unprovable, and define  $\mu_{i+1}^{i+1} = \mu_{i+1}^i \cup (\mathbf{X} : \varphi)$ , and  $\mu_j^{i+1} = \mu_j^i$  for every  $j \neq i+1$ . It is easy to verify that  $\Omega_{n'} = \mu_1^{n'} \mid \ldots \mid \mu_{n'}^{n'}$  has all of the required properties. For example, we show that  $\Omega_{n'}$  is maximal with respect to  $\varphi$ . Let  $\mu_i^{n'} \in \Omega_{n'}$ , and assume that  $\mathbf{x} : \varphi \notin \mu_i^{n'}$  (for  $x \in \pounds_2$ ). This implies that  $\mu_1^{i-1} \mid \ldots \mid \mu_i^i \cup \{\mathbf{x} : \varphi\} \mid \ldots \mid \mu_{n'}^{i-1}$  is provable. Using weakenings, it easily follows that  $\Omega_{n'} \mid \mu_i^{n'} \cup \{\mathbf{x} : \varphi\}$  is provable.

#### **Lemma 7.5.4.** There is some maximal extended $\mathcal{L}$ -hypersequent $\Omega$ that extends $H_0$ .

Proof. Let  $\varphi_0, \varphi_1, \ldots$  be an enumeration of all  $\mathcal{F}$ -formulas, in which every formula appears an infinite number of times. Let  $s_0, s_1, \ldots$  be an enumeration of all  $\mathcal{F}$ -sequents (with repetitions, if there is only finite number of them). We recursively define an infinite sequence of unprovable finite extended  $\mathcal{F}$ -hypersequents,  $H_0 = s_1^0 \mid \ldots \mid s_{n_0}^0, H_1 = s_1^1 \mid \ldots \mid s_{n_1}^1, \ldots$ , in which: (a)  $n_0 \leq n_1 \leq \ldots$  and (b)  $s_j^i \subseteq s_j^{i+1}$  for every  $i \geq 0$  and  $1 \leq j \leq n_i$ . First, let  $n_0 = n$  and let  $s_1^0 \mid \ldots \mid s_{n_0}^0$  be the original  $\mathcal{F}$ -hypersequent  $H_0$ . Let  $i \geq 0$ . Assume  $H_i = s_1^i \mid \ldots \mid s_{n_i}^i$  is defined. By Lemma 7.5.3, there exists an unprovable finite extended  $\mathcal{F}$ -hypersequent H' such that:

•  $H' = s'_1 \mid ... \mid s'_{n'}$  where  $n' \in \{n_i, n_i + 1\}$ , and  $s_i \subseteq s'_i$  for every  $1 \le i \le n_i$ .

- H' is maximal with respect to  $\varphi_i$ .
- H' is maximal with respect to  $s_i$ .

Let  $n_{i+1} = n'$ , and  $s_j^{i+1} = s_j'$  for every  $1 \le j \le n_{i+1}$ . Note that after every step we have an unprovable finite extended  $\mathcal{F}$ -hypersequent, so Lemma 7.5.3 can be applied. Finally, let  $N = \max\{n_0, n_1, ...\} + 1$ , if such a maximum exists, and infinity otherwise. Let  $n(j) = \min\{i \mid j \le n_i\}$  for every  $1 \le j < N$ . Define  $\mu_j = \bigcup_{i \ge n(j)} s_j^i$  for every  $1 \le j < N$ . Let  $\Omega$  be the extended  $\mathcal{F}$ -hypersequent  $\mu_1 \mid \mu_2 \mid ....$  Obviously,  $\Omega$  extends  $H_0$ . We prove that it is maximal:

Unprovability Suppose by way of contradiction that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$  for an  $\mathcal{L}$ -hypersequent  $H \sqsubseteq \Omega$ . Assume that  $H = s_1 \mid ... \mid s_n$ . The construction of  $\Omega$  ensures that for every  $1 \leq i \leq n$ , there exists  $k_i \geq 1$  such that  $s_i \subseteq \mu_{k_i}$ . This entails that for every  $1 \leq i \leq n$ , there exists  $m_i \geq 0$  such that  $s_i \subseteq s_{k_i}^{m_i}$ . By the construction of the  $s_j^i$ 's, we have that for every  $1 \leq i \leq n$  and  $l \geq m_i$ ,  $s_i \subseteq s_{k_i}^l$ . Let  $m = \max\{m_1, ..., m_n\}$ . Then, by definition  $H \sqsubseteq H_m$ . Since  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H$ , it follows that  $H_m$  is provable. But, this contradicts the fact that  $H_0$  is unprovable, and that each application of Lemma 7.5.3 yields an unprovable extended  $\mathcal{L}$ -hypersequent.

Internal Maximality Let  $\varphi \in \mathcal{F}$ ,  $\mu_j \in \Omega$ , and  $\mathbf{x} \in \mathcal{L}_2$ . Since we included  $\varphi$  infinite number of times in the enumeration of  $\mathcal{F}$ -formulas, there exists some  $i \geq n(j)$  such that  $\varphi_i = \varphi$ . Our construction ensures that  $H_{i+1}$  is maximal with respect to  $\varphi$ , and so if  $\mathbf{x}: \varphi \notin s_j^{i+1}$  then  $H_{i+1} \mid s_j^{i+1} \cup \{\mathbf{x}: \varphi\}$  is provable. Since  $H_{i+1} \sqsubseteq \Omega$ , it follows that if  $\mathbf{x}: \varphi \notin \mu_j$  then  $\Omega \mid \mu_j \cup \{\mathbf{x}: \varphi\}$  is provable.

**External Maximality** Let s be an  $\mathcal{F}$ -sequent. Assume that  $s = s_i$   $(i \geq 0)$ , our construction ensures that  $H_{i+1}$  is maximal with respect to s. Hence,  $H_{i+1} \mid s$  is provable whenever  $\{s\} \not\sqsubseteq H_{i+1}$ . Since  $H_{i+1} \sqsubseteq \Omega$ , we have that  $\Omega \mid s$  is provable whenever  $\{s\} \not\sqsubseteq \Omega$ .

Using the maximal extended  $\mathcal{L}$ -hypersequent  $\Omega$  (that extends  $H_0$ ), we are not ready to construct a  $\mathbf{G} \upharpoonright_{\rho}$ -legal partial  $\mathcal{L}$ -Gvaluation v which is a model of  $\mathcal{H}$ , but not of  $H_0$ . For every  $\varphi \in \mathcal{F}$ , define  $L(\varphi)$  and  $R(\varphi)$  as follows:

$$L(\varphi) = \{ \mu \in \Omega \mid \mathbf{f} : \varphi \in \mu \} \quad \text{and} \quad R(\varphi) = \{ \mu \in \Omega \mid \mathbf{t} : \varphi \not\in \mu \}.$$

v is defined as follows:

- 1.  $\mathcal{V}_v = \langle V, \subseteq \rangle$ , where:  $V = \{L(\varphi) \mid \varphi \in \mathcal{F}\} \cup \{R(\varphi) \mid \varphi \in \mathcal{F}\} \cup \{\Omega, \emptyset\}$  ( $\subseteq$  denotes set inclusion).
- 2.  $Dom_v = \mathcal{F}$ .
- 3. For every  $\varphi \in \mathcal{F}$ ,  $v(\varphi) = \langle L(\varphi), R(\varphi) \rangle$ .

First, we show that  $\mathcal{V}_v = \langle V, \subseteq \rangle$  is indeed a Gödel set (see Definition 7.3.1). Clearly,  $\subseteq$  is a partial order on V,  $\Omega$  is a maximal element, and  $\emptyset$  is a minimal one. To see that V is linearly ordered by  $\subseteq$ , it suffices to prove the following:

1.  $L(\varphi_1) \subseteq L(\varphi_2)$  or  $L(\varphi_2) \subseteq L(\varphi_1)$  for every  $\varphi_1, \varphi_2 \in \mathcal{F}$ . To see this, suppose for a contradiction that  $L(\varphi_1) \not\subseteq L(\varphi_2)$  and  $L(\varphi_2) \not\subseteq L(\varphi_1)$  for some  $\varphi_1, \varphi_2 \in \mathcal{F}$ . Thus there exist  $\mu_1, \mu_2 \in \Omega$ , such that  $\mu_1 \in L(\varphi_1) \setminus L(\varphi_2)$  and  $\mu_2 \in L(\varphi_2) \setminus L(\varphi_1)$ . Hence,  $\mathbf{f}: \varphi_1 \in \mu_1$ ,  $\mathbf{f}: \varphi_1 \not\in \mu_2$ ,  $\mathbf{f}: \varphi_2 \in \mu_2$  and  $\mathbf{f}: \varphi_2 \not\in \mu_1$ . By Proposition 7.5.2, there exist an  $\mathcal{L}$ -hypersequent  $H_1 \sqsubseteq \Omega$  and  $\mathcal{L}$ -sequents  $s_1, \dots, s_n \subseteq \mu_1$  such that

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid s_1 \cup \{ \mathbf{f} : \varphi_2 \} \mid \dots \mid s_n \cup \{ \mathbf{f} : \varphi_2 \}.$$

Similarly, there exist  $H_2 \sqsubseteq \Omega$  and  $s'_1, \ldots, s'_m \subseteq \mu_2$ , such that

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_2 \mid s_1' \cup \{\mathbf{f} : \varphi_1\} \mid \ldots \mid s_m' \cup \{\mathbf{f} : \varphi_1\}.$$

By Proposition 7.2.10:

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid H_2 \mid s_1 \cup \{\mathbf{f} : \varphi_1\} \mid ... \mid s_n \cup \{\mathbf{f} : \varphi_1\} \mid s'_1 \cup \{\mathbf{f} : \varphi_2\} \mid ... \mid s'_m \cup \{\mathbf{f} : \varphi_2\}.$$
  
But,  $\Omega$  extends this hypersequent, and this contradicts the fact that  $\Omega$  is unprovable.

2.  $R(\varphi_1) \subseteq R(\varphi_2)$  or  $R(\varphi_2) \subseteq R(\varphi_1)$  for every  $\varphi_1, \varphi_2 \in \mathcal{F}$ . To see this, suppose for a contradiction that  $R(\varphi_1) \not\subseteq R(\varphi_2)$  and  $R(\varphi_2) \not\subseteq R(\varphi_1)$  for some  $\varphi_1, \varphi_2 \in \mathcal{F}$ . Thus there are  $\mu_1, \mu_2 \in \Omega$ , such that  $\mu_1 \in R(\varphi_1) \setminus R(\varphi_2)$  and  $\mu_2 \in R(\varphi_2) \setminus R(\varphi_1)$ . Hence,  $\mathfrak{t}: \varphi_1 \not\in \mu_1$ ,  $\mathfrak{t}: \varphi_1 \in \mu_2$ ,  $\mathfrak{t}: \varphi_2 \in \mu_1$  and  $\mathfrak{t}: \varphi_2 \not\in \mu_2$ . By Proposition 7.5.2, there exist  $\mathcal{L}$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega$  and negative  $\mathcal{L}$ -sequents  $S_1, S_2$  such that

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid s_1 \cup \{\mathsf{t} : \varphi_1\} \quad \text{and} \quad \mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_2 \mid s_2 \cup \{\mathsf{t} : \varphi_2\}.$$

By applying (com), we obtain

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} H_1 \mid H_2 \mid s_2 \cup \{\mathsf{t} : \varphi_1\} \mid s_1 \cup \{\mathsf{t} : \varphi_2\}.$$

Again, since  $\Omega$  extends this  $\mathcal{L}$ -hypersequent, this contradicts the fact that  $\Omega$  is unprovable.

3.  $L(\varphi_1) \subseteq R(\varphi_2)$  or  $R(\varphi_2) \subseteq L(\varphi_1)$  for every  $\varphi_1 \in \mathcal{F}$  and  $\varphi_2 \in \mathcal{F}$ . To see this, suppose for a contradiction that  $L(\varphi_1) \not\subseteq R(\varphi_2)$  and  $R(\varphi_2) \not\subseteq L(\varphi_1)$  for some  $\varphi_1 \in \mathcal{F}$  and  $\varphi_2 \in \mathcal{F}$ . Let  $\mu_1, \mu_2 \in \Omega$ , such that  $\mu_1 \in L(\varphi_1) \setminus R(\varphi_2)$  and  $\mu_2 \in R(\varphi_2) \setminus L(\varphi_1)$ . Hence,  $\mathbf{f}: \varphi_1 \in \mu_1$ ,  $\mathbf{f}: \varphi_1 \not\in \mu_2$ ,  $\mathbf{t}: \varphi_2 \in \mu_1$  and  $\mathbf{t}: \varphi_2 \not\in \mu_2$ . By Proposition 7.5.2, there exist  $\mathcal{L}$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega$ ,  $\mathcal{L}$ -sequents  $s_1, \ldots, s_n \subseteq \mu_2$  and a negative  $\mathcal{L}$ -sequent  $s' \subseteq \mu_2$ , such that

 $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid s_1 \cup \{ \mathbf{f} : \varphi_1 \} \mid ... \mid s_n \cup \{ \mathbf{f} : \varphi_1 \} \quad \text{and} \quad \mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_2 \mid s' \cup \{ \mathbf{t} : \varphi_2 \}.$  By Proposition 7.2.10,  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid H_2 \mid s_1 \cup s' \mid ... \mid s_n \cup s' \mid \{ \mathbf{f} : \varphi_1, \mathbf{t} : \varphi_2 \}.$  Again, this contradicts the fact that  $\Omega$  is unprovable.

Now, it is easy to verify that v is a model of  $\mathcal{H}$  but not of  $H_0$ :

- Let H be an  $\mathcal{F}$ -hypersequent in  $\mathcal{H}$ . Since  $\Omega$  is unprovable,  $H \not\sqsubseteq \Omega$ . Thus there exists some  $\mathcal{L}$ -sequent  $s \in H$  such that  $s \not\subseteq \mu$  for every  $\mu \in \Omega$ . This implies that for every  $\mu \in \Omega$ , either  $\mu \not\in L(\varphi)$  for some  $\varphi$  such that  $\mathbf{f}: \varphi \in s$ , or  $\mu \in R(\varphi)$  for some  $\varphi$  such that  $\mathbf{t}: \varphi \in s$ . It follows that for every  $\mu \in \Omega$ , we have  $\mu \not\in v^{\mathbf{f}}(s)$  or  $\mu \in v^{\mathbf{t}}(s)$ . Thus  $v^{\mathbf{f}}(s) \subseteq v^{\mathbf{t}}(s)$ , and so  $v \models s$ . Consequently,  $v \models H$ .
- Let  $s \in H_0$ . Since  $H_0 \sqsubseteq \Omega$ , there exists  $\mu \in \Omega$  such that  $s \subseteq \mu$ . Hence  $\mu \in L(\varphi)$  for every  $\varphi$  such that  $\mathbf{f} : \varphi \in s$ , and  $\mu \not\in R(\varphi)$  for every  $\varphi$  such that  $\mathbf{t} : \varphi \in s$ . It follows that  $\mu \in v^{\mathbf{f}}(s)$  and  $\mu \not\in v^{\mathbf{t}}(s)$ . Therefore,  $v^{\mathbf{f}}(s) \not\subseteq v^{\mathbf{t}}(s)$ , and so  $v \not\models s$ . Consequently,  $v \not\models H_0$ .

It remains to show that v is  $\mathbf{G} \upharpoonright_{\rho}$ -legal. First, we show that it is  $\rho$ -legal. By definition  $Dom_v = \mathcal{F}$ . In addition:

- Let  $\varphi \in \mathcal{C} \cap \mathcal{F}$ . Assume for a contradiction that  $v^{\mathbf{t}}(\varphi) \not\subseteq v^{\mathbf{f}}(\varphi)$ , and thus there exists some  $\mu \in \Omega$  such that  $\mu \in R(\varphi)$  but  $\mu \not\in L(\varphi)$ . Thus  $\mathbf{t}: \varphi \not\in \mu$  and  $\mathbf{f}: \varphi \not\in \mu$ . By Proposition 7.5.2, there exist  $\mathcal{L}$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega$ ,  $\mathcal{L}$ -sequents  $s_1, \ldots, s_n \subseteq \mu$  and a negative  $\mathcal{L}$ -sequent  $s' \subseteq \mu$ , such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid s_1 \cup \{\mathbf{f}: \varphi\} \mid \ldots \mid s_n \cup \{\mathbf{f}: \varphi\}$  and  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_2 \mid s' \cup \{\mathbf{t}: \varphi\}$ . By n consecutive applications of (cut) (on  $\varphi$ , which is an element of  $\mathcal{C}$ ), we obtain  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid H_2 \mid s_1 \cup s' \mid \ldots \mid s_n \cup s'$ , but this contradicts the fact that  $\Omega$  is unprovable.
- Let  $\varphi \in \mathcal{A} \cap \mathcal{F}$ . Thus  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} \{ \{ \mathbf{f} : \varphi, \mathbf{t} : \varphi \} \}$  (by applying (id) with  $\varphi$ , which is an element of  $\mathcal{A}$ ). Since  $\Omega$  is unprovable, either  $\mathbf{f} : \varphi \notin \mu$  or  $\mathbf{t} : \varphi \notin \mu$  for every  $\mu \in \Omega$ . Equivalently, for every  $\mu \in \Omega$ , either  $\mu \notin L(\varphi)$  or  $\mu \in R(\varphi)$ . It follows that  $L(\varphi) \subseteq R(\varphi)$ .

To show that v is G-legal, we first prove that the following hold:

(a) For every  $\mathcal{F}$ -sequent s, if  $v^{\mathfrak{t}}(s) \subseteq v^{\mathfrak{t}}(s)$  then there exists an  $\mathcal{L}$ -hypersequent  $H' \sqsubseteq \Omega$  such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H' \mid s$ .

Proof. Suppose that there does not exist  $H' \sqsubseteq \Omega$  such that  $\mathcal{H} \vdash_{\mathbf{G} \restriction \rho} H' \mid s$ . Then Proposition 7.5.2 implies that  $\{s\} \sqsubseteq \Omega$ . Hence  $s \subseteq \mu$  for some  $\mu \in \Omega$ . By definition,  $\mu \in L(\varphi)$  for every  $\varphi$  such that  $\mathbf{f} : \varphi \in s$ , and  $\mu \notin R(\varphi)$  for  $\varphi$  such that  $\mathbf{t} : \varphi \in s$ . It follows that  $\mu \in v^{\mathbf{f}}(s)$  and  $\mu \notin v^{\mathbf{t}}(s)$ .

(b) For every  $\mathcal{F}$ -sequent s, if  $\mu \in v(s)$ , then there exist an  $\mathcal{L}$ -hypersequent  $H' \sqsubseteq \Omega$ , and a negative  $\mathcal{L}$ -sequent  $s' \subseteq \mu$ , such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H' \mid s \cup s'$ .

*Proof.* Suppose that  $\mu \in v(s) = v^{\mathfrak{t}}(s) \to v^{\mathfrak{t}}(s)$ . Now, if  $v^{\mathfrak{t}}(s) \subseteq v^{\mathfrak{t}}(s)$ , then the claim follows by (a) (take  $s' = \emptyset$ ). Otherwise,  $\mu \in v^{\mathfrak{t}}(s)$ . Hence there is some

 $\varphi \in \mathcal{L}$  such that  $\mathbf{t}: \varphi \in s$  and  $\mu \in R(\varphi)$ . Thus  $\mathbf{t}: \varphi \notin \mu$ . By (possibly) using weakening, Proposition 7.5.2 implies that there exist an  $\mathcal{L}$ -hypersequent  $H' \sqsubseteq \Omega$ , and a negative  $\mathcal{L}$ -sequent  $s' \subseteq \mu$ , such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \Omega} H' \mid s \cup s'$ .

(c) For every negative  $\mathcal{F}$ -sequent s, if  $\mu \not\in v^{\mathfrak{f}}(s)$  then there exist an  $\mathcal{L}$ -hypersequent  $H' \sqsubseteq \Omega$  and  $\mathcal{L}$ -sequents  $s_1, \ldots, s_n \subseteq \mu$  such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} H' \mid s \cup s_1 \mid \ldots \mid s \cup s_n$ .

*Proof.* Suppose that  $\mu \notin v^{\mathbf{f}}(s)$ . Thus there exists some  $\varphi \in \mathcal{L}$  such that  $\mathbf{f}:\varphi \in s$  and  $\mu \notin L(\varphi)$ . Hence,  $\mathbf{f}:\varphi \notin \mu$ . The claim follows from Proposition 7.5.2 (possibly by using weakenings).

Next, we show that v is r-legal for each canonical  $\mathcal{L}$ -rule r of  $\mathbf{G}$ :

• Let  $r = \mathcal{S}/\{\mathsf{t}: \diamond(p_1, \dots, p_{ar(\diamond)})\}$  be a right canonical  $\mathcal{L}$ -rule of  $\mathbf{G}$ , and  $\sigma$  an  $\mathcal{L}$ -substitution, such that  $\sigma(\{p_1, \dots, p_n\}) \cup \{\varphi\} \subseteq \mathcal{F}$ , where  $\varphi = \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))$ . Using Lemma 7.3.15, it suffices to prove that  $v(\sigma(\mathcal{S})) \subseteq R(\varphi)$ . Let  $\mu \in v(\sigma(\mathcal{S}))$ . Suppose that  $\mathcal{S} = \{s_1, \dots, s_n\}$ . Since  $\mu \in v(\sigma(\mathcal{S}))$ , (b) entails that for every  $1 \leq i \leq n$ , there exist an  $\mathcal{L}$ -hypersequent  $H_i \sqsubseteq \Omega$  and a negative  $\mathcal{L}$ -sequent  $s_i' \subseteq \mu$ , such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} H_i \mid s_i' \cup \sigma(s_i)$ . By applying the rule r, we obtain that

$$\mathcal{H} \vdash_{\mathbf{G} \upharpoonright_{\mathcal{O}}} H_1 \mid \ldots \mid H_n \mid s'_1 \cup \ldots \cup s'_n \cup \{\mathsf{t} : \varphi\}.$$

The fact that  $\Omega$  is unprovable then entails that  $\mathbf{t}:\varphi\notin\mu$ , and so  $\mu\in R(\varphi)$ .

- Let  $r = \mathcal{S}_1, \mathcal{S}_2/\{\mathbf{f}: \diamond(p_1, \dots, p_{ar(\diamond)})\}$  be a left canonical  $\mathcal{L}$ -rule of  $\mathbf{G}$ , and  $\sigma$  an  $\mathcal{L}$ -substitution, such that  $\sigma(\{p_1, \dots, p_n\}) \cup \{\varphi\} \subseteq \mathcal{F}$ , where  $\varphi = \sigma(\diamond(p_1, \dots, p_{ar(\diamond)}))$ . Using Lemma 7.3.15, it suffices to prove that  $L(\varphi) \subseteq v(\sigma(\mathcal{S}_1)) \to v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ . Let  $\mu \in \Omega$  and suppose that  $\mu \not\in v(\sigma(\mathcal{S}_1)) \to v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ . Hence,  $v(\sigma(\mathcal{S}_1)) \not\subseteq v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$  and  $\mu \not\in v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ . Let  $\mu' \in v(\sigma(\mathcal{S}_1))$  such that  $\mu' \not\in v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ . Suppose that  $\mathcal{S}_1 = \{s_1, \dots, s_n\}$  and  $\mathcal{S}_2 = \{s_1', \dots, s_m'\}$ . We have the following:
  - (1) Since  $\mu' \in v(\sigma(S_1))$ , we have  $\mu' \in v(\sigma(s_i))$  for every  $1 \leq i \leq n$ . (b) entails that for every  $1 \leq i \leq n$ , there exist an  $\mathcal{L}$ -hypersequent  $H_i \sqsubseteq \Omega$ , and a negative  $\mathcal{L}$ -sequent  $c_i \subseteq \mu'$ , such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_i \mid c_i \cup \sigma(s_i)$ . The availability of weakenings entail that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_1 \mid c \cup \sigma(s_i)$  for every  $1 \leq i \leq n$ , where  $H = H_1 \mid ... \mid H_n$  and  $c = c_1 \cup ... \cup c_n$ .
  - (2) Since  $\mu' \not\in v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$ ,  $\mu' \not\in v^{\mathbf{f}}(\sigma(\{\psi \mid \mathbf{f}: \psi \in s_i'\}))$  for every  $1 \leq i \leq m$ . (c) entails that for every  $1 \leq i \leq m$ , there exist an  $\mathcal{L}$ -hypersequent  $H_i' \sqsubseteq \Omega$  and  $\mathcal{L}$ -sequents  $s_1^i, \ldots, s_{n_i}^i \subseteq \mu'$  such that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_i' \mid s_1^i \cup \sigma(s_i') \mid \ldots \mid s_{n_i}^i \cup \sigma(s_i')$ . The availability of weakenings entail that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H' \mid s_1^i \cup \sigma(s_i') \mid \ldots \mid s_{n_i}^i \cup \sigma(s_i')$  for every  $1 \leq i \leq m$ , where  $H' = H_1' \mid \ldots \mid H_m'$ .

(3) Similarly, the fact that  $\mu \not\in v^{\mathbf{f}}(\sigma(\mathcal{S}_2))$  entails that for every  $1 \leq i \leq m$ , there exist an  $\mathcal{L}$ -hypersequent  $H_i'' \sqsubseteq \Omega$  and  $\mathcal{L}$ -sequents  $c_1^i, \ldots, c_{n_i'}^i \subseteq \mu$  for which we have  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_i'' \mid c_1^i \cup \sigma(s_i') \mid \ldots \mid c_{n_i'}^i \cup \sigma(s_i')$ . The availability of weakenings entail that  $\mathcal{H} \vdash_{\mathbf{G} \upharpoonright \rho} H_i'' \mid c_1^i \cup \sigma(s_i') \mid \ldots \mid c_{n_i'}^i \cup \sigma(s_i')$  for every  $1 \leq i \leq m$ , where  $H'' = H_1'' \mid \ldots \mid H_m''$ .

Let  $H^* = H \mid H' \mid H''$ . From (1) and (2), by Proposition 7.2.11 we obtain that the following  $\mathcal{L}$ -hypersequent has a  $\rho$ -proof in  $\mathbf{G}$  of  $\mathcal{H}$ :

 $H^* \mid c \cup s_1^1 \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup s_{n_1}^1 \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup s_1^m \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup s_{n_m}^m \cup \{\mathbf{f}:\varphi\}.$ Similarly, from (1) and (3), by Proposition 7.2.11 we obtain that the following  $\mathcal{L}$ -hypersequent has a  $\rho$ -proof in  $\mathbf{G}$  of  $\mathcal{H}$ :

 $H^* \mid c \cup c_1^1 \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup c_{n'_1}^1 \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup c_1^m \cup \{\mathbf{f}:\varphi\} \mid ... \mid c \cup c_{n'_m}^m \cup \{\mathbf{f}:\varphi\}.$ By Proposition 7.2.10, it follows that the following hypersequent has a  $\rho$ -proof in  $\mathbf{G}$  of  $\mathcal{H}$ :

$$\begin{split} H^* \mid c \cup s_1^1 \mid ... \mid c \cup s_{n_1}^1 \mid ... \mid c \cup s_1^m \mid ... \mid c \cup s_{n_m}^m \mid \\ c_1^1 \cup \{\mathbf{f}{:}\varphi\} \mid ... \mid c_{n_1'}^1 \cup \{\mathbf{f}{:}\varphi\} \mid ... \mid c_1^m \cup \{\mathbf{f}{:}\varphi\} \mid ... \mid c_{n_m'}^m \cup \{\mathbf{f}{:}\varphi\}. \end{split}$$

Now, if  $\mathbf{f}:\varphi\in\mu$ , then  $\Omega$  extends this  $\mathcal{L}$ -hypersequent, and this contradicts the fact that  $\Omega$  is unprovable. Therefore,  $\mathbf{f}:\varphi\not\in\mu$ , and consequently  $\mu\not\in L(\varphi)$ .

# Chapter 8

# Calculus for First-Order Gödel Logic

So far we have considered logics and calculi only at the propositional level. However, the ideas and methods described in the previous chapters are applicable for first-order calculi as well. In this chapter we demonstrate our methods for the hypersequent calculus **HIF** for standard first-order Gödel logic (standard means that the real interval [0,1] can be used as the underlying set of truth values). HIF, introduced in [30], is obtained from **HG** (the original hypersequent calculus for propositional Gödel logic, see Chapter 7) by adding standard (hypersequential versions of) rules for the quantifiers  $\forall$  and  $\exists$ . It was proved in [30] that **HIF** is sound and complete for standard first-order Gödel logic by showing its equivalence to an Hilbert system for this logic (see [63]). Furthermore, it was shown in [30] and [22] that **HIF** admits cut-elimination. As a corollary, one obtains Herbrand theorem for the prenex fragment of this logic [76].

In this chapter we obtain alternative semantic proofs for these facts, by using the ideas and techniques from Chapter 7. First, we briefly present standard first-order Gödel logic (from a many-valued semantic point of view), and the hypersequent calculus **HIF**. Then we prove the soundness of **HIF** for standard first-order Gödel logic. The completeness proof is tied together with cut-admissibility and involves two stages: (i) We present a non-deterministic semantics and show its completeness for the cut-free fragment of **HIF**; (ii) It is shown that from every non-deterministic counter-model, one can extract a usual counter-model. From these two facts together, it easily follows that **HIF** enjoys cut-admissibility, and that it is complete for standard first-order Gödel logic.

Note that unlike previous chapters, this chapters studies one specific calculus for a

<sup>&</sup>lt;sup>1</sup>Note that Gödel logic is the only fundamental fuzzy logic whose first-order version is recursively axiomatizable [76].

<sup>&</sup>lt;sup>2</sup>In fact, the first syntactic proof in [30] of cut-elimination was erroneous. A corrected syntactic proof appear in [22]. There has also been a gap in the proof given in [4] for **HG** in its handling of the case of disjunction.

particular logic, aiming to demonstrate the usefulness of semantic tools in studying first-order calculi. It should be also possible to define and study general canonical rules for quantifiers in this context as done in [18, 98]. This chapter also serves as a preparation for the next one, where we extend of **HIF** with rules for *second-order* quantifiers. Cut-admissibility for **HIF** itself can be derived as a corollary of the results of the next chapter. However, to make our presentation more accessible we first include here the full proof for **HIF**.

Remark 8.0.5. The method of proving cut-admissibility in this chapter is similar to what we did for (coherent) canonical Gödel hypersequent calculi in Chapter 7. Indeed, a general cut-admissibility theorem (Theorem 7.4.6) was proven in Chapter 7 by (uniformly) extracting a G-legal  $\mathcal{L}$ -Gvaluation from a  $\mathbf{G} \upharpoonright \langle \mathcal{L}, \emptyset, \mathcal{L} \rangle$ -legal  $\mathcal{L}$ -Gvaluation. The non-deterministic semantics of the cut-free fragment of HIF developed in this chapter is a natural first-order version of the semantics obtained in Chapter 7 for cut-free canonical Gödel hypersequent calculi (i.e., using the proof-specification  $\langle \mathcal{L}, \emptyset, \mathcal{L} \rangle$ ).

#### Publications Related to this Chapter

The results of this chapter were included in [16] and [72]. Nevertheless, our method here is completely different (and it allows generalization for second-order). In [16, 72] we extended the results of [10] concerning the propositional calculus **HG** to the first-order calculus **HIF**. As in [10] this was done by proving the completeness of the cut-free part of **HIF** for its Kripke-style semantics (thereby proving both completeness of the calculus and the admissibility of the cut rule in it). In this chapter we take a novel approach which is very close to the methods of Chapter 7. In particular, we consider many-valued semantics rather than Kripke-style one. Note also that [16] introduced a multiple-conclusion calculus for first-order Gödel logic that is not included here.

## 8.1 Preliminaries

In what follows,  $\mathcal{L}^1$  denotes an arbitrary first-order language, defined by:

**Definition 8.1.1.** A first-order language consists of the following:

- 1. Infinitely many variables  $\nu_1, \nu_2, \dots$  We use the metavariables x, y, z (with or without subscripts) for variables.
- 2. A propositional constant  $\perp$ .
- 3. Binary connectives  $\land, \lor, \supset$ . We use  $\diamond$  as a metavariable for the binary connectives.
- 4. Quantifiers  $\forall$  and  $\exists$ . We use Q as a metavariable for the quantifiers.

8.1. Preliminaries 135

5. An arbitrary set of constant symbols. The metavariable c is used to range over constant symbols.

- 6. An arbitrary set of function symbols. The metavariable f is used to range over them.
- 7. An arbitrary set of predicate symbols. The metavariable p is used to range over them.
- 8. Parentheses '(' and ')'.

**Definition 8.1.2.** The set of  $\mathcal{L}^1$ -terms consists of: (i) all variables of  $\mathcal{L}^1$ ; (ii) all constant symbols of  $\mathcal{L}^1$ ; and (iii) if f is an n-ary function symbol of  $\mathcal{L}^1$  and  $t_1, \ldots, t_n$  are  $\mathcal{L}^1$ -terms then  $f(t_1, \ldots, t_n)$  is an  $\mathcal{L}^1$ -term. We use t (with or without subscripts) as a metavariable for  $\mathcal{L}^1$ -terms. The set of variables occurring in an  $\mathcal{L}^1$ -term t is defined as usual, and denoted by Fv[t].

Following the convention of Girard in [58], we define a first-order formula as an equivalence class of what we call concrete formulas, so that two formulas that differ only by the names of their bound variables are considered the same.<sup>3</sup> This is convenient for handling the bureaucracy of free and bound variables. Moreover, it simplifies the non-deterministic semantics below (see Remark 8.5.7).

#### **Definition 8.1.3.** Concrete $\mathcal{L}^1$ -formulas are inductively defined as follows:

- 1.  $p(t_1, ..., t_n)$  is a concrete  $\mathcal{L}^1$ -formula for every predicate symbol p of arity n and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ .
- 2.  $\perp$  is a concrete  $\mathcal{L}^1$ -formula.
- 3. If  $\Phi_1$  and  $\Phi_2$  are concrete  $\mathcal{L}^1$ -formulas, so are  $(\Phi_1 \wedge \Phi_2)$ ,  $(\Phi_1 \vee \Phi_2)$ , and  $(\Phi_1 \supset \Phi_2)$ .
- 4. If  $\Phi$  is a concrete  $\mathcal{L}^1$ -formula, and x is a variable of  $\mathcal{L}^1$ , then  $(\forall x\Phi)$  and  $(\exists x\Phi)$  are concrete  $\mathcal{L}^1$ -formulas.

We use  $\Phi$  (with or without subscripts) as a metavariable for concrete  $\mathcal{L}^1$ -formulas. Free and bound variables in concrete  $\mathcal{L}^1$ -formulas are defined as usual. We denote by  $Fv[\Phi]$ , the set of variables occurring free in a concrete  $\mathcal{L}^1$ -formula  $\Phi$ . Alpha-equivalence between concrete  $\mathcal{L}^1$ -formulas is defined as usual (renaming of bound variables). We denote by  $[\Phi]_{\alpha}$  the set of all concrete  $\mathcal{L}^1$ -formulas which are alpha-equivalent to  $\Phi$  (i.e. the equivalence class of  $\Phi$  under alpha-equivalence).  $cp[\Phi]$ , the complexity of a concrete  $\mathcal{L}^1$ -formula  $\Phi$  is the sum of the numbers of occurrences of quantifiers, connectives (including  $\bot$ ), and atomic concrete formulas (formulas of the form  $p(t_1, \ldots, t_n)$ ) in  $\Phi$ .

<sup>&</sup>lt;sup>3</sup>Since [58] does not provide all the technical details for this convention, we do it here.

**Definition 8.1.4.** An  $\mathcal{L}^1$ -formula is an equivalence class of concrete  $\mathcal{L}^1$ -formulas under alpha-equivalence. We mainly use  $\varphi, \psi$  (with or without subscripts) as metavariables for  $\mathcal{L}^1$ -formulas. The set of free variables and the complexity of an  $\mathcal{L}^1$ -formula are defined using representatives, i.e. for an  $\mathcal{L}^1$ -formula  $\varphi$ ,  $Fv[\varphi] = Fv[\Phi]$  and  $cp[\varphi] = cp[\Phi]$  for some  $\Phi \in \varphi$ .

In the last definition and henceforth, it is easy to verify that all notions defined using representatives are well-defined.

**Definition 8.1.5.** We define two operations on  $\mathcal{L}^1$ -formulas:

- For  $\diamond \in \{\land, \lor, \supset\}$ , and  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$ :  $(\varphi_1 \diamond \varphi_2) = [(\Phi_1 \diamond \Phi_2)]_{\alpha} \text{ for some } \Phi_1 \in \varphi_1 \text{ and } \Phi_2 \in \varphi_2.$
- For  $Q \in \{ \forall, \exists \}$ , a variable x of  $\mathcal{L}^1$ , and an  $\mathcal{L}^1$ -formula  $\varphi$ :

$$(Qx\varphi) = [(Qx\Phi)]_{\alpha}$$
 for some  $\Phi \in \varphi$ .

The next proposition allows us to use inductive definitions and to prove claims by induction on complexity of formulas:

**Proposition 8.1.6.** Exactly one of the following holds for every  $\mathcal{L}^1$ -formula  $\varphi$ :

- $cp[\varphi] = 1$  and exactly one of the following holds:
  - $-\varphi = \{p(t_1, \dots, t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, \dots, t_n$ .
  - $\varphi = \{\bot\}.$
- $\varphi = (\varphi_1 \diamond \varphi_2)$  for some  $\diamond \in \{\land, \lor, \supset\}$ , and unique  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ .
- For every variable  $x \notin Fv[\varphi]$ ,  $\varphi = (Qx\psi)$  for some  $Q \in \{\forall, \exists\}$ , and unique  $\mathcal{L}^1$ formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ .

Substitutions are defined as follows:

**Definition 8.1.7.** Let t be an  $\mathcal{L}^1$ -term and x a variable of  $\mathcal{L}^1$ .

1. Given an  $\mathcal{L}^1$ -term t',  $t'\{t/x\}$  is inductively defined by:

$$t'\{t/x\} = \begin{cases} t & t' = x \\ t' & t' = y \text{ for } y \neq x, \text{ or } t' = c \\ f(t_1\{t/x\}, \dots, t_n\{t/x\}) & t' = f(t_1, \dots, t_n) \end{cases}$$

2. Given an  $\mathcal{L}^1$ -formula  $\varphi$ ,  $\varphi\{t/x\}$  is inductively defined by:

$$\varphi\{t/x\} = \begin{cases} \{p(t_1\{t/x\}, \dots, t_n\{t/x\})\} & \varphi = \{p(t_1, \dots, t_n)\} \\ \varphi & \varphi = \{\bot\} \\ (\varphi_1\{t/x\} \diamond \varphi_2\{t/x\}) & \varphi = (\varphi_1 \diamond \varphi_2) \\ (Qy\psi\{t/x\}) & \varphi = (Qy\psi) \text{ for } y \notin Fv[t] \cup \{x\} \end{cases}$$

Note that the above substitution operations are well-defined. In particular, the choice of the variable y is immaterial.

# 8.2 Standard First-Order Gödel Logic

In this section we briefly present standard first-order Gödel logic from a (many-valued) semantic point of view (see, e.g., [63, 76] for more detailed presentations). The first component of the semantics is the set of truth values. These should form a Gödel set:

**Definition 8.2.1.** A (standard) Gödel set  $\mathcal{V} = \langle V, \leq \rangle$  is defined just like a propositional Gödel set (see Definition 7.3.1), with the additional restriction that  $\mathcal{V}$  is a complete totally ordered set. The operations  $\inf_{\mathcal{V}}$  and  $\sup_{\mathcal{V}}$  are defined as usual.

Next, the semantic structures include a domain and an interpretation function defined as follows:

**Definition 8.2.2.** A *domain* is a non-empty set  $\mathcal{D}$ . Given a domain  $\mathcal{D}$ , an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -interpretation is a function I assigning an element in  $\mathcal{D}$  to every constant symbol of  $\mathcal{L}^1$ , and a function in  $\mathcal{D}^n \to \mathcal{D}$  to every n-ary function symbol of  $\mathcal{L}^1$ .

To interpret predicate symbols we use *fuzzy subsets*:

**Definition 8.2.3.** Given a Gödel set  $\mathcal{V}$ , and some non-empty set  $\mathcal{D}$ , a function D from  $\mathcal{D}$  to  $\mathcal{V}$  is called a *fuzzy subset* of  $\mathcal{D}$  over  $\mathcal{V}$ .

Predicate symbols are naturally interpreted as fuzzy subsets of tuples of elements of  $\mathcal{D}$ .

**Definition 8.2.4.** An  $\mathcal{L}^1$ -structure is a triple  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$ , where:

- 1.  $\mathcal{V}$  is a Gödel set.
- 2.  $\mathcal{D}$  is a domain.
- 3. I is an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -interpretation.
- 4. P is a function assigning a fuzzy subset of  $\mathcal{D}^n$  over  $\mathcal{V}$  to every n-ary predicate symbol of  $\mathcal{L}^1$ .

As usual, an additional function is used for interpreting the *free* variables.

#### **Definition 8.2.5.** Let $\mathcal{D}$ be a domain.

- 1. An  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment is a function assigning elements of  $\mathcal{D}$  to the variables of  $\mathcal{L}^1$ .
- 2. Given an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -interpretation I and an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\sigma^I$  is the function assigning elements of  $\mathcal{D}$  to all  $\mathcal{L}^1$ -terms, recursively defined by:
  - $\sigma^{I}[c] = I[c]$  for every constant symbol c of  $\mathcal{L}^{1}$ .
  - $\sigma^{I}[x] = \sigma[x]$  for every variable x of  $\mathcal{L}^{1}$ .
  - $\sigma^I[f(t_1,\ldots,t_n)] = I[f](\sigma^I[t_1],\ldots,\sigma^I[t_n])$  for every *n*-ary function symbol f of  $\mathcal{L}^1$  and n  $\mathcal{L}^1$ -terms  $t_1,\ldots,t_n$ .
- 3. Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Given a variable x of  $\mathcal{L}^1$  and  $d \in \mathcal{D}^I$ , we denote by  $\sigma_{x:=d}$  the  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment that is identical to  $\sigma$  except for  $\sigma_{x:=d}[x] = d$ . This notation is naturally extended to several distinct variables (e.g.  $\sigma_{\nu_1:=d_1,\nu_2:=d_2}$ ).

**Lemma 8.2.6.** Let  $\mathcal{D}$  be a domain,  $\sigma$  an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment, t an  $\mathcal{L}^1$ -term, and x a variable of  $\mathcal{L}^1$ . For every  $\mathcal{L}^1$ -term t':  $\sigma^I[t'\{t/x\}]] = \sigma^I_{x:=\sigma^I[t]}[t']$ .

*Proof.* By usual induction on the structure of t'.

We can now define the truth value assigned by a given structure to an arbitrary formula with respect to some assignment. This definition generalizes in a natural way the usual recursive definition used in classical higher-order logics, where instead of the usual truth tables we use their counterparts of Gödel logic:  $\land$  corresponds to min,  $\lor$  to max, and the implication  $\supset$  is interpreted as the  $\rightarrow$  operation. For the quantifiers, we take inf and sup. Since the set of truth values is complete by definition, inf and sup are always defined.

**Definition 8.2.7.** Let  $W = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^1$ -structure. For every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $W[\varphi, \sigma]$  is the element of V inductively defined as follows:

$$\mathcal{W}[\varphi, \sigma] = \begin{cases} P[p][\sigma^{I}[t_{1}], \dots, \sigma^{I}[t_{n}]] & \varphi = \{p(t_{1}, \dots, t_{n})\} \\ 0 & \varphi = \{\bot\} \\ \min\{\mathcal{W}[\varphi_{1}, \sigma], \mathcal{W}[\varphi_{2}, \sigma]\} & \varphi = (\varphi_{1} \land \varphi_{2}) \\ \max\{\mathcal{W}[\varphi_{1}, \sigma], \mathcal{W}[\varphi_{2}, \sigma]\} & \varphi = (\varphi_{1} \lor \varphi_{2}) \\ \mathcal{W}[\varphi_{1}, \sigma] \to \mathcal{W}[\varphi_{2}, \sigma] & \varphi = (\varphi_{1} \supset \varphi_{2}) \\ \inf_{d \in \mathcal{D}} \mathcal{W}[\psi, \sigma_{x:=d}] & \varphi = (\exists x \psi) \end{cases}$$

It can be verified that the choice of x in the last definition is immaterial, and  $\mathcal{W}[\varphi, \sigma]$  is well-defined. The following usual lemmas will be needed below:

**Lemma 8.2.8.** Let  $W = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^1$ -structure, x a variable of  $\mathcal{L}^1$ , and d an element of  $\mathcal{D}$ . Then  $W[\varphi, \sigma_{x:=d}] = W[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  such that  $x \notin Fv[\varphi]$ , and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .

*Proof.* After proving the claim for  $\mathcal{L}^1$ -terms (that  $\sigma^I_{x:=d}[t] = \sigma^I[t]$  for every  $\mathcal{L}^1$ -term such that  $x \notin Fv[t]$ , and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ), the claim is obtained by usual induction on the complexity of  $\varphi$ .

**Lemma 8.2.9.** Let  $W = \langle V, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^1$ -structure, t an  $\mathcal{L}^1$ -term, and x a variable of  $\mathcal{L}^1$ . For every  $\mathcal{L}^1$ -formula  $\varphi$ , and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ :

$$\mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}] = \mathcal{W}[\varphi\{t/x\}, \sigma].$$

*Proof.* We prove the claim by induction on the complexity of  $\varphi$ . First, suppose that  $cp[\varphi] = 1$ , and let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Exactly one of the following holds:

- $\varphi = \{\bot\}$ . In this case the claim obviously holds.
- $\varphi = \{p(t_1, \dots, t_n)\}$ . In this case,  $\varphi\{t/x\} = \{p(t_1\{t/x\}, \dots, t_n\{t/x\})\}$ . Thus:  $\mathcal{W}[\varphi\{t/x\}, \sigma] = P[p][\sigma^I[t_1\{t/x\}], \dots, \sigma^I[t_n\{t/x\}]]$ .

By Lemma 8.2.6,  $P[p][\sigma^I[t_1\{t/x\}], \dots, \sigma^I[t_n\{t/x\}]] = P[p][\sigma^I_{x:=\sigma^I[t]}[t_1], \dots, \sigma^I_{x:=\sigma^I[t]}[t_n]]$ . By definition, this is equal to  $\mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}]$ .

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^1$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for some  $\diamond \in \{\land, \lor, \supset\}$ , and  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . By definition,  $\varphi\{t/x\} = (\varphi_1\{t/x\} \diamond \varphi_2\{t/x\})$ . We continue with  $\diamond = \supset$  (the proof is similar for  $\land$  and  $\lor$ ). By the induction hypothesis,  $\mathcal{W}[\varphi\{t/x\}, \sigma] = \mathcal{W}[\varphi_1\{t/x\}, \sigma] \to \mathcal{W}[\varphi_2\{t/x\}, \sigma] = \mathcal{W}[\varphi_1, \sigma_{x:=\sigma^I[t]}] \to \mathcal{W}[\varphi_2, \sigma_{x:=\sigma^I[t]}]$ . By definition, this is equal to  $\mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}]$ .
- $\varphi = (Qy\psi)$  for some  $Q \in \{\forall, \exists\}$ , variable  $y \notin \{x\} \cup Fv[t]$  of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . By definition,  $\varphi\{t/x\} = (Qy\psi\{t/x\})$ . Continuing with  $Q = \forall$  ( $\exists$  is similar), we have  $\mathcal{W}[\varphi\{t/x\}, \sigma] = \inf_{d \in \mathcal{D}} \mathcal{W}[\psi\{t/x\}, \sigma_{y:=d}]$ . By the induction hypothesis,  $\mathcal{W}[\psi\{t/x\}, \sigma_{y:=d}] = \mathcal{W}[\psi, \sigma_{y:=d,x:=\sigma^I[t]}]$  for every  $d \in \mathcal{D}$  (note that  $y \neq x$ ), and so  $\mathcal{W}[\varphi\{t/x\}] = \inf_{d \in \mathcal{D}} \mathcal{W}[\psi, \sigma_{y:=d,x:=\sigma^I[t]}]$ . By definition,  $\inf_{d \in \mathcal{D}} \mathcal{W}[\psi, \sigma_{y:=d,x:=\sigma^I[t]}] = \mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}]$ .

Finally, we define standard first-order Gödel logic. For simplicity, unlike in the previous chapters, we identify a logic with its set of theorems and do not consider consequence relations.

**Definition 8.2.10.** For an  $\mathcal{L}^1$ -formula  $\varphi$ , we write  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^1}} \varphi$  if  $\mathcal{W}[\varphi, \sigma] = 1$  for every  $\mathcal{L}^1$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .  $\mathbf{G}\ddot{o}_{\mathcal{L}^1}$  is the logic consisting of all formulas  $\varphi$  such that  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^1}} \varphi$ .

### 8.3 The Hypersequent Calculus HIF

In this section we present the hypersequent calculus **HIF** for  $\mathbf{G\ddot{o}}_{\mathcal{L}^1}$  from [30] (adapted to our definitions, where formulas are equivalence classes of concrete formulas). (single-conclusion)  $\mathcal{L}^1$ -sequents and  $\mathcal{L}^1$ -hypersequents are defined exactly as in Chapter 7. **HIF** is obtained by augmenting the calculus  $\mathbf{HG}$  (see Chapter 7, and Table 7.1 in particular), with the following rules for first-order quantifiers:

$$(\mathbf{f}:\forall) \quad \frac{H \mid \Gamma, \varphi\{t/x\} \Rightarrow E}{H \mid \Gamma, (\forall x\varphi) \Rightarrow E} \qquad (\mathbf{t}:\forall) \quad \frac{H \mid \Gamma \Rightarrow \varphi}{H \mid \Gamma \Rightarrow (\forall x\varphi)}$$

(f:
$$\exists$$
)  $\frac{H \mid \Gamma, \varphi \Rightarrow E}{H \mid \Gamma, (\exists x \varphi) \Rightarrow E}$  (t: $\exists$ )  $\frac{H \mid \Gamma \Rightarrow \varphi\{t/x\}}{H \mid \Gamma \Rightarrow (\exists x \varphi)}$ 

Applications of the rules  $(t:\forall)$  and  $(f:\exists)$  must obey the eigenvariable condition: x is not a free variable in the lower hypersequent.

Below, we write  $\vdash_{\mathbf{HIF}} H$  to denote that an  $\mathcal{L}^1$ -hypersequent H is provable in  $\mathbf{HIF}$ , and  $\vdash_{\mathbf{HIF}}^{cf} H$  to denote that H is provable in  $\mathbf{HIF}$  without applying (cut). Several clarifications should be noted:

- 1. The above rules are formulated by schemes using metavariables. For example, an  $\mathcal{L}^1$ -hypersequent  $H_1$  can be derived from an  $\mathcal{L}^1$ -hypersequent  $H_2$  by applying the rule  $(\mathbf{f}:\forall)$  iff  $H_1 = H \cup \{\Gamma \Rightarrow (\forall x\varphi)\}$  and  $H_2 = H \cup \{\Gamma \cup \varphi\{t/x\} \Rightarrow E\}$  for some  $\mathcal{L}^1$ -hypersequent H, finite set  $\Gamma$  of  $\mathcal{L}^1$ -formulas, variable x of  $\mathcal{L}^1$ ,  $\mathcal{L}^1$ -formula  $\varphi$ ,  $\mathcal{L}^1$ -term t, and singleton or empty set E of  $\mathcal{L}^1$ -formulas.
- 2. Since formulas are equivalence classes, the rules (t:∀), (f:∃) could be written as well as:

$$(\mathbf{t}:\forall) \quad \frac{H \mid \Gamma \Rightarrow \varphi\{y/x\}}{H \mid \Gamma \Rightarrow (\forall x \varphi)} \qquad (\mathbf{f}:\exists) \quad \frac{H \mid \Gamma, \varphi\{y/x\} \Rightarrow E}{H \mid \Gamma, (\exists x \varphi) \Rightarrow E}$$

where y is not a free variable in the lower hypersequent.

The following standard lemma establishes the admissibility of substitution:

8.4. Soundness 141

**Lemma 8.3.1.** If  $\vdash_{\mathbf{HIF}} H$ , then  $\vdash_{\mathbf{HIF}} H\{y/x\}$  for every variables x, y of  $\mathcal{L}^1$ , such that  $y \notin Fv[H]$ . The same holds for  $\vdash_{\mathbf{HIF}}^{cf}$ .

Next, the rule  $(f:\exists)$  can be generalized as follows.

**Proposition 8.3.2.** For every  $n \geq 0$ ,  $\mathcal{L}^1$ -hypersequent H,  $\mathcal{L}^1$ -sequents  $s_1, \ldots, s_n$ ,  $\mathcal{L}^1$ -formula  $\varphi$ , and variable  $x \notin Fv[H \mid s_1 \mid \ldots \mid s_n]$ : if  $\vdash_{\mathbf{HIF}} H \mid s_1 \cup \{\mathbf{f}:\varphi\} \mid \ldots \mid s_n \cup \{\mathbf{f}:(\exists x\varphi)\} \mid$ 

Proof. We use induction on n. The claim is trivial for n=0. Now assume that the claim holds for n-1, we prove it for n. Let H be an  $\mathcal{L}^1$ -hypersequent,  $s_1, \ldots, s_n$   $\mathcal{L}^1$ -sequents,  $\varphi$  an  $\mathcal{L}^1$ -formula, and x a variable of  $\mathcal{L}^1$  such that  $x \notin Fv[H \mid s_1 \mid \ldots \mid s_n]$ . Let  $H_0 = H \mid s_1 \cup \{\mathbf{f}:\varphi\} \mid \ldots \mid s_n \cup \{\mathbf{f}:\varphi\}$ . Suppose that  $\vdash_{\mathbf{HIF}} H_0$ . Let y be a variable of  $\mathcal{L}^1$  such that  $y \notin Fv[H_0]$ . By Lemma 8.3.1,  $\vdash_{\mathbf{HIF}} H_0\{y/x\}$ . Using a generalized version of (com) (see Proposition 7.2.10), the following  $\mathcal{L}^1$ -hypersequent is cut-free derivable from  $H_0$  and  $H_0\{y/x\}$ :  $\{H \mid s_n \cup \{\mathbf{f}:\varphi\} \mid s_1 \cup \{\mathbf{f}:\varphi\{y/x\}\} \mid \ldots \mid s_{n-1} \cup \{\mathbf{f}:\varphi\{y/x\}\}\}$  (to see this, take  $H_1 = H \mid s_n \cup \{\mathbf{f}:\varphi\}$  and  $H_2 = H \mid s_1 \cup \{\mathbf{f}:\varphi\{y/x\}\} \mid \ldots \mid s_{n-1} \cup \{\mathbf{f}:\varphi\{y/x\}\}\}$ . By an application of  $(\mathbf{f}:\exists)$ , we obtain:  $H \mid s_n \cup \{\mathbf{f}:(\exists x\varphi)\} \mid s_1 \cup \{\mathbf{f}:\varphi\{y/x\}\} \mid \ldots \mid s_{n-1} \cup \{\mathbf{f}:\varphi\{y/x\}\}\}$ . The induction hypothesis entails that  $\vdash_{\mathbf{HIF}} H \mid s_1 \cup \{\mathbf{f}:(\exists x\varphi)\} \mid \ldots \mid s_n \cup \{\mathbf{f}:(\exists x\varphi)\}\}$ . Since cuts were not involved in this proof, the proof for  $\vdash_{\mathbf{HIF}}^{cf}$  is exactly the same.

### 8.4 Soundness

In this section we prove the soundness of **HIF** for  $G\ddot{o}_{\mathcal{L}^1}$ . The following definition is the first-order version of Definition 7.3.4.

**Definition 8.4.1.** Let  $W = \langle V, D, I, P \rangle$  be an  $\mathcal{L}^1$ -structure.

- 1. Given an  $\mathcal{L}^1$ -sequent s and an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\mathcal{W}^{\mathfrak{f}}[s, \sigma]$ ,  $\mathcal{W}^{\mathfrak{t}}[s, \sigma]$  and  $\mathcal{W}[s, \sigma]$  are defined as follow:
  - (a)  $\mathcal{W}^{f}[s,\sigma] = \min\{\mathcal{W}[\varphi,\sigma] \mid f:\varphi \in s\}.$
  - (b)  $\mathcal{W}^{\mathsf{t}}[s,\sigma] = \max\{\mathcal{W}[\varphi,\sigma] \mid \mathsf{t}: \varphi \in s\}.$
  - (c)  $\mathcal{W}[s,\sigma] = \mathcal{W}^{\mathfrak{t}}[s,\sigma] \to \mathcal{W}^{\mathfrak{t}}[s,\sigma].$
- 2. An  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$  is a model (with respect to  $\mathcal{W}$ ) of:
  - (a) an  $\mathcal{L}^1$ -sequent s (denoted by:  $\mathcal{W}, \sigma \models s$ ) if  $\mathcal{W}[s, \sigma] = 1$ .
  - (b) an  $\mathcal{L}^1$ -hypersequent H (denoted by:  $\mathcal{W}, \sigma \models H$ ) if  $\mathcal{W}, \sigma \models s$  for some  $s \in H$ .
- 3. W is a *model* of an  $\mathcal{L}^1$ -hypersequent H if  $W, \sigma \models H$  for every  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .

**Theorem 8.4.2.** Let H be an  $\mathcal{L}^1$ -hypersequent. If  $\vdash_{\mathbf{HIF}} H$ , then every  $\mathcal{L}^1$ -structure is a model of H.

*Proof.* Let  $W = \langle V, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^1$ -structure, where  $V = \langle V, \leq \rangle$ . It suffices to prove soundness of each possible application of a rule of **HIF**. For the weakening rules, (cut), (com), and (id) this is as in Section 7.5. We do here several other cases and leave the rest to the reader:

- (t:  $\supset$ ) Suppose that  $H \mid s \cup \{t: (\varphi_1 \supset \varphi_2)\}$  is derived from  $H \mid s \cup \{f: \varphi_1, t: \varphi_2\}$  using (t:  $\supset$ ) (here s must be a negative  $\mathcal{L}^1$ -sequent). Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. If  $\mathcal{W}, \sigma \models s'$  for some  $s' \in H$ , then we are done. Otherwise,  $\mathcal{W}, \sigma \models s \cup \{f: \varphi_1, t: \varphi_2\}$ . Thus, either  $\mathcal{W}[\varphi_1, \sigma] \leq \mathcal{W}^f[s, \sigma]$  and  $\mathcal{W}[\varphi_1, \sigma] \leq \mathcal{W}[\varphi_2, \sigma]$ , or  $\mathcal{W}^f[s, \sigma] \leq \mathcal{W}[\varphi_2, \sigma]$ . In both cases, it follows that  $\mathcal{W}^f[s, \sigma] \leq \mathcal{W}[\varphi_1, \sigma] \to \mathcal{W}[\varphi_2, \sigma]$ . By definition,  $\mathcal{W}[\varphi_1, \sigma] \to \mathcal{W}[\varphi_2, \sigma] = \mathcal{W}[\varphi_1, \sigma] \to \varphi_2, \sigma$ . It follows that  $\mathcal{W}, \sigma \models s \cup \{t: (\varphi_1 \supset \varphi_2)\}$ , and so  $\mathcal{W}, \sigma \models H \mid s \cup \{t: (\varphi_1 \supset \varphi_2)\}$ .
- (f:  $\supset$ ) Suppose that  $H \mid s' \cup s \cup \{f: (\varphi_1 \supset \varphi_2 \Rightarrow E \text{ is derived from } H \mid s' \cup \{t: \varphi_1\} \text{ and } H \mid s \cup \{f: \varphi_2\} \text{ using } (f: \supset) \text{ (here } s' \text{ must be a negative } \mathcal{L}^1\text{-sequent)}.$  Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. If  $\mathcal{W}, \sigma \models s''$  for some  $s'' \in H$ , then we are done. Otherwise,  $\mathcal{W}, \sigma \models s' \cup \{t: \varphi_1\} \text{ and } \mathcal{W}, \sigma \models s \cup \{f: \varphi_2\}.$  Let  $u_1 = \mathcal{W}^f[s' \cup s, \sigma]$  and  $u_2 = \mathcal{W}^t[s, \sigma].$  If  $u_1 \leq u_2$ , then  $\mathcal{W}, \sigma \models \Gamma, (\varphi_1 \supset \varphi_2) \Rightarrow E$ , and we are done. Otherwise, we have  $u_1 \leq \mathcal{W}[\varphi_1, \sigma], \mathcal{W}[\varphi_2, \sigma] \leq u_2$ , and so  $\mathcal{W}[\varphi_2, \sigma] < \mathcal{W}[\varphi_1, \sigma].$  It follows that

$$\mathcal{W}[(\varphi_1 \supset \varphi_2), \sigma] = \mathcal{W}[\varphi_1, \sigma] \to \mathcal{W}[\varphi_2, \sigma] = \mathcal{W}[\varphi_2, \sigma] \le u_2.$$
  
Consequently,  $\mathcal{W}, \sigma \models s' \cup s \cup \{f: (\varphi_1 \supset \varphi_2)\}$  in this case as well.

- (t: $\forall$ ) Suppose that  $H = H' \mid s \cup \{t:(\forall x\varphi)\}$  is derived from  $H' \mid s \cup \{t:\varphi\}$  using (t: $\forall$ ) (where  $x \notin Fv[H]$ , and s is a negative  $\mathcal{L}^1$ -sequent). Assume that  $\mathcal{W}$  is not a model of H. Thus there exists an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ , such that  $\mathcal{W}, \sigma \not\models H$ . Hence,  $\mathcal{W}, \sigma \not\models s'$  for every  $s' \in H'$ , and  $\mathcal{W}, \sigma \not\models s \cup \{t:(\forall x\varphi)\}$ . Thus we have  $\mathcal{W}^{\mathbf{f}}[s,\sigma] > \mathcal{W}[(\forall x\varphi),\sigma]$ . It follows that  $\mathcal{W}^{\mathbf{f}}[s,\sigma] > \mathcal{W}[\varphi,\sigma_{x:=d}]$  for some  $d \in \mathcal{D}$ . Since  $x \notin Fv[s]$ , by Lemma 8.2.8 we have that  $\mathcal{W}[\psi,\sigma_{x:=d}] = \mathcal{W}[\psi,\sigma]$  for every  $\psi$  that occurs in s. It follows that  $\mathcal{W}, \sigma_{x:=d} \not\models s \cup \{t:\varphi\}$ . Moreover, since  $x \notin Fv[H']$ , again using by Lemma 8.2.8, we obtain that  $\mathcal{W}, \sigma_{x:=d} \not\models s'$  for every  $s' \in H'$ . Hence,  $\mathcal{W}, \sigma_{x:=d} \not\models H' \mid s \cup \{t:\varphi\}$ , and consequently  $\mathcal{W}$  is not a model of  $H' \mid s \cup \{t:\varphi\}$ .
- (f: $\forall$ ) Suppose that  $H = H' \mid s \cup \{f:(\forall x\varphi)\}$  is derived from  $H' \mid s \cup \{f:\varphi\{t/x\}\}\}$  using (f: $\forall$ ). Assume that  $\mathcal{W}, \sigma \not\models H$  for some  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ . Hence,  $\mathcal{W}, \sigma \not\models s'$  for every  $s' \in H'$ , and  $\mathcal{W}, \sigma \not\models s \cup \{f:(\forall x\varphi)\}$ . Let  $u = \mathcal{W}^{\mathsf{t}}[s,\sigma]$ . The assumption that  $\mathcal{W}, \sigma \not\models s \cup \{f:(\forall x\varphi)\}$  entails that  $\mathcal{W}^{\mathsf{f}}[s,\sigma] > u$  and  $\mathcal{W}[(\forall x\varphi),\sigma] > u$ . By definition,

 $\mathcal{W}[(\forall x\varphi), \sigma] = \inf_{d \in \mathcal{D}} \mathcal{W}[\varphi, \sigma_{x:=d}].$  Thus  $\mathcal{W}[\varphi, \sigma_{x:=d}] > u$  for every  $d \in \mathcal{D}$ . In particular,  $\mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}] > u$ . Lemma 8.2.9 implies that  $\mathcal{W}[\varphi\{t/x\}, \sigma] > u$ . It follows that  $\mathcal{W}, \sigma \not\models s \cup \{\mathbf{f}: \varphi\{t/x\}\}$ . Hence  $\mathcal{W}$  is not a model of  $H' \mid s \cup \{\mathbf{f}: \varphi\{t/x\}\}$ .  $\square$ 

Soundness for  $G\ddot{o}_{\mathcal{L}^1}$  is an obvious corollary:

Corollary 8.4.3. For every  $\mathcal{L}^1$ -formula  $\varphi$ , if  $\vdash_{\mathbf{HIF}} \{\{\mathbf{t}:\varphi\}\}$  then  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^1}} \varphi$ .

*Proof.* Follows from Theorem 8.4.2, since for every  $\mathcal{L}^1$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ :  $\sigma$  is a model of  $\{\{\mathbf{t}:\varphi\}\}$  with respect to  $\mathcal{W}$  iff  $\mathcal{W}[\varphi, \sigma] = 1$ .

### 8.5 Complete Non-deterministic Semantics

In this section we present a non-deterministic semantics for which the cut-free fragment of **HIF** is complete. This semantics will be used in the next section, where we show that ordinary counter-models can be extracted out of non-deterministic ones. As corollaries, we will obtain the completeness of **HIF** for the (usual) semantics described above, and the fact that (cut) is admissible in **HIF**.

The non-deterministic semantics is based on  $quasi-\mathcal{L}^1$ -structures. The idea behind these structures is similar to what we had in Chapter 7 for accommodating systems in which (cut) is not available (i.e. for the proof-specification  $\langle \mathcal{L}, \emptyset, \mathcal{L} \rangle$ ). Thus quasi- $\mathcal{L}^1$ -structures assign two truth values to each formula – one for its "f-labelled" occurrences, and one for its "t-labelled" ones (see Page 113).

**Definition 8.5.1.** Let  $\mathcal{V} = \langle V, \leq \rangle$  be a Gödel set. Given some non-empty set  $\mathcal{D}$ , a function  $\mathcal{D}$  from  $\mathcal{D}$  to  $\{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \leq u^{\mathbf{t}} \}$  is called a *quasi fuzzy subset* of  $\mathcal{D}$  over  $\mathcal{V}$ .

**Definition 8.5.2.** A quasi- $\mathcal{L}^1$ -structure is a tuple  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$ , where:

- 1. V, D, and I are defined as in  $\mathcal{L}^1$ -structures (Definition 8.2.4).
- 2. P is a function assigning a *quasi* fuzzy subset of  $\mathcal{D}^n$  over  $\mathcal{V}$  to every n-ary predicate symbol of  $\mathcal{L}^1$ .
- 3. v is a function assigning a pair in  $\{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \leq u^{\mathbf{t}} \}$  to every ordered pair of the form  $\langle \varphi, \sigma \rangle$ , where  $\varphi$  is an  $\mathcal{L}^1$ -formula and  $\sigma$  is an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment, such that  $v[\varphi, \sigma_{x:=d}] = v[\varphi\{y/x\}, \sigma_{y:=d}]$  for every variable x, variable  $y \notin Fv[\varphi]$ , and  $d \in \mathcal{D}$ .

Note that quasi-structures include a function v that assigns truth values to every formula. This is related to the fact that the semantics is non-deterministic. Thus, in

contrast to (ordinary) structures, in quasi structures the values of the atomic formulas do not uniquely determine the values of all compound formulas. The function v is then used to "store" the values of the compound formulas. Obviously, in order to be able to extract an ordinary counter-model out of a quasi-structure, further conditions should be imposed:

**Notation 8.5.3.** For each function F whose range is  $\{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \leq u^{\mathbf{t}} \}$  (e.g. the functions P[p] for every predicate symbol p and v from Definition 8.5.2), we denote by  $F^{\mathbf{f}}$  and  $F^{\mathbf{t}}$  the functions obtained from F by taking only the left and the right components (respectively). For instance, for every  $\varphi$  and  $\sigma$ ,  $v^{\mathbf{f}}[\varphi, \sigma]$  is the left component of the pair  $v[\varphi, \sigma]$ .

**Definition 8.5.4.** Let  $Q = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  be a quasi- $\mathcal{L}^1$ -structure. For every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\underline{v}_{\mathcal{O}}[\varphi, \sigma]$  is the pair defined as follows:

$$\frac{\partial}{\partial t} \left\{ \begin{array}{ll} \mathcal{L}(t, \mathcal{D}) & \text{defined as follows.} \\
P[p][\sigma^I[t_1], \dots, \sigma^I[t_n]] & \varphi = \{p(t_1, \dots, t_n)\} \\
\langle 0, 0 \rangle & \varphi = \{\bot\} \\
\langle \min\{v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{f}}[\varphi_2, \sigma]\}, \min\{v^{\mathbf{t}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma]\} \rangle & \varphi = (\varphi_1 \wedge \varphi_2) \\
\langle \max\{v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{f}}[\varphi_2, \sigma]\}, \max\{v^{\mathbf{t}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma]\} \rangle & \varphi = (\varphi_1 \vee \varphi_2) \\
\langle v^{\mathbf{t}}[\varphi_1, \sigma] \to v^{\mathbf{f}}[\varphi_2, \sigma], v^{\mathbf{f}}[\varphi_1, \sigma] \to v^{\mathbf{t}}[\varphi_2, \sigma] \rangle & \varphi = (\varphi_1 \supset \varphi_2) \\
\langle \inf_{d \in \mathcal{D}} v^{\mathbf{f}}[\psi, \sigma_{x:=d}], \inf_{d \in \mathcal{D}} v^{\mathbf{t}}[\psi, \sigma_{x:=d}] \rangle & \varphi = (\exists x \psi) \\
\langle \sup_{d \in \mathcal{D}} v^{\mathbf{f}}[\psi, \sigma_{x:=d}], \sup_{d \in \mathcal{D}} v^{\mathbf{t}}[\psi, \sigma_{x:=d}] \rangle & \varphi = (\exists x \psi) \\
\end{array}$$

The condition on v in Definition 8.5.2 ensures that  $\mathcal{Q}$  is well-defined, namely that the choice of x is immaterial. It is straightforward to verify that  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma]^{\mathbf{f}} \leq \underline{v}_{\mathcal{Q}}[\varphi, \sigma]^{\mathbf{t}}$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$  (for  $\supset$ , note that if  $u_1 \leq u_2$  and  $u_3 \leq u_4$  then  $u_2 \to u_3 \leq u_1 \to u_4$ ).

**Notation 8.5.5.** Let  $\mathcal{V} = \langle V, \leq \rangle$  be a Gödel set. For  $u \in \mathcal{V}$  and a pair  $\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V}$  with  $u^{\mathbf{f}} \leq u^{\mathbf{t}}$ , we write  $u \in \langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle$  if  $u^{\mathbf{f}} \leq u \leq u^{\mathbf{t}}$ . For two pairs  $\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle, \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle$  in  $\{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \leq u^{\mathbf{t}} \}$ , we write  $\langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle \subseteq \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle$  if  $u_1^{\mathbf{f}} \geq u_2^{\mathbf{f}}$  and  $u_1^{\mathbf{t}} \leq u_2^{\mathbf{t}}$ .

**Definition 8.5.6.** A quasi- $\mathcal{L}^1$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  is called *legal* if we have  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] \subseteq v[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .

We can now demonstrate the non-deterministic nature of the semantics. For instance, consider an  $\mathcal{L}^1$ -formula of the form  $(\varphi_1 \wedge \varphi_2)$ . Suppose that  $v[\varphi_1, \sigma] = \langle u_1^{\mathbf{f}}, u_1^{\mathbf{t}} \rangle$  and  $v[\varphi_2, \sigma] = \langle u_2^{\mathbf{f}}, u_2^{\mathbf{t}} \rangle$ . All we require from  $v[(\varphi_1 \wedge \varphi_2), \sigma]$  is that

$$\langle \min\{u_1^{\mathtt{f}}, u_2^{\mathtt{f}}\}, \min\{u_1^{\mathtt{t}}, u_2^{\mathtt{t}}\} \rangle \subseteq v[(\varphi_1 \wedge \varphi_2), \sigma].$$

In other words, every pair  $\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle$  such that  $u^{\mathbf{f}} \leq \min\{u_1^{\mathbf{f}}, u_2^{\mathbf{f}}\}$  and  $\min\{u_1^{\mathbf{t}}, u_2^{\mathbf{t}}\} \leq u^{\mathbf{t}}$  can be chosen as a value for  $v[(\varphi_1 \wedge \varphi_2), \sigma]$ . Thus, in contrast to ordinary structures, here the values of  $\langle \varphi_1, \sigma \rangle$  and  $\langle \varphi_2, \sigma \rangle$  do not uniquely determine the value of  $\langle (\varphi_1 \wedge \varphi_2), \sigma \rangle$ .

Remark 8.5.7. Since formulas are defined to be alpha equivalence classes of concrete formulas, we do not have to explicitly enforce that two alpha-equivalent formulas obtain the same value. Previous works on non-deterministic semantics for languages with quantifiers, such as [18], studied structures in which truth values are non-deterministically assigned to concrete formulas. In this case, additional restrictions are needed.

The notion of model for quasi- $\mathcal{L}^1$ -structures is a natural first-order version of what we had for Gvaluations in Chapter 7 (see Definition 7.3.4):

**Definition 8.5.8.** Let  $Q = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  be a quasi- $\mathcal{L}^1$ -structure.

- 1. Given an  $\mathcal{L}^1$ -sequent s and an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\mathcal{Q}^{\mathbf{f}}[s, \sigma]$ ,  $\mathcal{Q}^{\mathbf{t}}[s, \sigma]$  and  $\mathcal{Q}[s, \sigma]$  are defined as follow:
  - (a)  $Q^{\mathbf{f}}[s, \sigma] = \min\{v^{\mathbf{f}}[\varphi, \sigma] \mid \mathbf{f}: \varphi \in s\}.$
  - (b)  $\mathcal{Q}^{\mathsf{t}}[s,\sigma] = \max\{v^{\mathsf{t}}[\varphi,\sigma] \mid \mathsf{t}: \varphi \in s\}.$
  - (c)  $Q[s,\sigma] = Q^{f}[s,\sigma] \to Q^{t}[s,\sigma]$ .
- 2. An  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$  is a model (with respect to  $\mathcal{Q}$ ) of:
  - (a) an  $\mathcal{L}^1$ -sequent s (denoted by:  $\mathcal{Q}, \sigma \models s$ ) if  $\mathcal{Q}[s, \sigma] = 1$ .
  - (b) an  $\mathcal{L}^1$ -hypersequent H (denoted by:  $\mathcal{Q}, \sigma \models H$ ) if  $\mathcal{Q}, \sigma \models s$  for some  $s \in H$ .
- 3.  $\mathcal{Q}$  is a model of an  $\mathcal{L}^1$ -hypersequent H if  $\mathcal{Q}, \sigma \models H$  for every  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .

Now we can state the main completeness theorem.

**Theorem 8.5.9.** Suppose that  $\not\vdash_{\mathbf{HIF}}^{cf} H_0$  for some  $\mathcal{L}^1$ -hypersequent  $H_0$ . Then there exists a legal quasi- $\mathcal{L}^1$ -structure that is not a model of  $H_0$ .

The rest of this section is devoted to prove this theorem. First, we introduce the two main ingredients of this proof: maximal extended hypersequents and Herbrand domains.

### Maximal Extended Hypersequents

As in Chapter 7, maximal extended  $\mathcal{L}^1$ -hypersequents will play a crucial role in the completeness proof below. These are defined exactly as in Section 7.5, but here we add one more requirement:

**Definition 8.5.10.** An extended  $\mathcal{L}^1$ -sequent  $\mu$  admits the witness property if the following hold for every  $\mathcal{L}^1$ -formula  $\varphi$ , and variable x of  $\mathcal{L}^1$ :

- 1. If  $\mathbf{t}:(\forall x\varphi) \in \mu$ , then  $\mathbf{t}:\varphi\{y/x\} \in \mu$  for some variable y of  $\mathcal{L}^1$ .
- 2. If  $\mathbf{f}:(\exists x\varphi) \in \mu$ , then  $\mathbf{f}:\varphi\{y/x\} \in \mu$  for some variable y of  $\mathcal{L}^1$ .

An extended  $\mathcal{L}^1$ -hypersequent  $\Omega$  admits the witness property if every  $\mu \in \Omega$  admits the witness property.  $\Omega$  is called maximal if it is unprovable, internally maximal, externally maximal, and it admits the witness property (see Section 7.5).<sup>4</sup>

As in Section 7.5, the following hold:

**Proposition 8.5.11.** Let  $\Omega$  be an extended  $\mathcal{L}^1$ -hypersequent.

- Assume that  $\Omega$  is maximal with respect to an  $\mathcal{L}^1$ -formula  $\varphi$ . For every  $\mu \in \Omega$ :
  - If  $\mathbf{f}:\varphi \notin \mu$ , then  $\vdash_{\mathbf{HIF}}^{cf} H \mid s_1 \cup \{\mathbf{f}:\varphi\} \mid \dots \mid s_n \cup \{\mathbf{f}:\varphi\} \text{ for some } \mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega$  and  $\mathcal{L}^1$ -sequents  $s_1, \dots, s_n \subseteq \mu$ .
  - If  $t:\varphi \notin \mu$ , then  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{t:\varphi\}$  for some  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega$  and negative  $\mathcal{L}^1$ -sequent  $s \subseteq \mu$ .
- Assume that  $\Omega$  is maximal with respect to an  $\mathcal{L}^1$ -sequent s. Then, if  $\{s\} \not\sqsubseteq \Omega$ , then there exists an  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega$  such that  $\vdash_{\mathbf{HIF}}^{cf} H \mid s$ .

**Lemma 8.5.12.** Every unprovable  $\mathcal{L}^1$ -hypersequent can be extended to a *maximal* extended  $\mathcal{L}^1$ -hypersequent.

*Proof.* The proof proceeds similarly to the proof of Lemma 7.5.4. To obtain an extended  $\mathcal{L}^1$ -hypersequent that admits the witness property, we add another step in the definition of  $H_{i+1}$ , based on the following claim:

Let  $\Omega = \mu_1 \mid ... \mid \mu_n$  be an unprovable finite extended  $\mathcal{L}^1$ -hypersequent. Then there exists an unprovable finite extended  $\mathcal{L}^1$ -hypersequent  $\Omega'$  of the form  $\mu'_1 \mid ... \mid \mu'_n$ , such that  $\mu_i \subseteq \mu'_i$  for every  $1 \leq i \leq n$ , and  $\Omega'$  admits the witness property.

To prove this claim describe an extension of  $\Omega$  to  $\Omega'$ . This extension is done in steps.<sup>5</sup> In every step, we take some extended  $\mathcal{L}^1$ -sequent  $\mu \in \Omega$ , and proceed as follows:

• If  $\mu$  contains a labelled formula of the form  $\mathbf{t}:(\forall x\varphi)$ , we take a variable y of  $\mathcal{L}^1$ , which is not free in the current hypersequent, and add  $\mathbf{t}:\varphi\{y/x\}$  to  $\mu$ .

<sup>&</sup>lt;sup>4</sup>Obviously, instead of  $\mathcal{H} \vdash_{\mathbf{G} \mid \rho} H$  that we had in the definition of "provable extended hypersequent" in Section 7.5, we should have  $\vdash_{\mathbf{HIF}}^{cf} H$  for the present case.

<sup>&</sup>lt;sup>5</sup>Formally, this extension should be defined inductively, but the intention should be clear.

• If  $\mu$  contains a labelled formula of the form  $\mathbf{f}:(\exists x\varphi)$ , we take a variable y of  $\mathcal{L}^1$ , which is not free in the current hypersequent, and add  $\mathbf{f}:\varphi\{y/x\}$  to  $\mu$ .

We continue this procedure until the obtained extended  $\mathcal{L}^1$ -hypersequent admits the witness property. Note that since the number of formulas in  $\Omega$  is finite, and the complexity of the formulas which are added is decreasing, this procedure would terminate after a finite number of steps.  $\Omega'$  is the finite extended  $\mathcal{L}^1$ -hypersequent obtained from  $\Omega$  by this procedure. We show that every such extension keeps the extended  $\mathcal{L}^1$ -hypersequent unprovable (and thus  $\Omega'$  is unprovable):

- Suppose that an unprovable extended  $\mathcal{L}^1$ -hypersequent  $\Omega_1$  contains an extended  $\mathcal{L}^1$ -sequent  $\mu$ , that contains a labelled formula of the form  $\mathfrak{t}:(\forall x\varphi)$ . Let  $\Omega_2$  be the extended  $\mathcal{L}^1$ -hypersequent obtained from  $\Omega_1$  by adding  $\mathfrak{t}:\varphi\{y/x\}$  to  $\mu$ , where y is a variable which does not occur in  $Fv[\Omega_1]$ . Assume for contradiction that  $\Omega_2$  is provable. Hence there exist an  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega_2$ , and a negative  $\mathcal{L}^1$ -sequent  $s' \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H \mid s' \cup \{\mathfrak{t}:\varphi\{y/x\}\}$ . By applying  $(\mathfrak{t}:\forall)$ , we obtain  $\vdash_{\mathbf{HIF}}^{cf} H \mid s' \cup \{\mathfrak{t}:(\forall x\varphi)\}$ . This contradicts the fact the  $\Omega_1$  is unprovable.
- Suppose that an unprovable extended  $\mathcal{L}^1$ -hypersequent  $\Omega_1$  contains an extended  $\mathcal{L}^1$ -sequent  $\mu$ , that contains a labelled formula of the form  $\mathfrak{f}:(\exists x\varphi)$ . Let  $\Omega_2$  be the extended  $\mathcal{L}^1$ -hypersequent obtained from  $\Omega_1$  by adding  $\mathfrak{f}:\varphi\{y/x\}$  to  $\mu$ , where y is a variable which does not occur in  $Fv[\Omega_1]$ . Assume for contradiction that  $\Omega_2$  is provable. Hence  $\vdash_{\mathbf{HIF}}^{cf} H \mid s'_1 \cup \{\mathfrak{f}:\varphi\{y/x\}\} \mid \ldots \mid s'_n \cup \{\mathfrak{f}:\varphi\{y/x\}\}\}$  for some  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega_2$ , and  $\mathcal{L}^1$ -sequents  $s'_1, \ldots, s'_n \subseteq \mu$ . Proposition 8.3.2 entails that  $\vdash_{\mathbf{HIF}}^{cf} H \mid s'_1 \cup \{\mathfrak{f}:(\exists x\varphi)\} \mid \ldots \mid s'_n \cup \{\mathfrak{f}:(\exists x\varphi)\}$ . This contradicts the fact the  $\Omega_1$  is unprovable.

#### The Herbrand Domain

**Definition 8.5.13.** The Herbrand domain for  $\mathcal{L}^1$ , denoted by  $\mathcal{D}^{\mathcal{L}^1}$ , is the domain consisting of all  $\mathcal{L}^1$ -terms. The Herbrand interpretation for  $\mathcal{L}^1$ , denoted by  $I^{\mathcal{L}^1}$ , is the  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -interpretation defined by: I[c] = c for every constant symbol c of  $\mathcal{L}^1$ , and  $I[f] = \lambda t_1, \ldots, t_n \in \mathcal{D}^{\mathcal{L}^1}$ .  $f(t_1, \ldots, t_n)$  for every n-ary function symbol f of  $\mathcal{L}^1$ .

Below, given an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment  $\sigma$  and an  $\mathcal{L}^1$ -term t, we write  $\sigma[t]$  instead of  $\sigma^{I_{\mathcal{L}^1}}[t]$  (see Definition 8.2.5). In addition,  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignments are extended to apply on  $\mathcal{L}$ -formulas. Roughly speaking, every occurrence of a free variable x in a formula  $\varphi$  is replaced in  $\sigma[\varphi]$  by  $\sigma[x]$ . Formally, this is defined as follows.

**Definition 8.5.14.** Let  $\varphi$  be an  $\mathcal{L}^1$ -formula, and  $\sigma$  an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment. The set of free variables of the pair  $\langle \varphi, \sigma \rangle$  (denoted by  $Fv[\langle \varphi, \sigma \rangle]$ ) consists of the variables of  $\sigma[x]$  for every variable  $x \in Fv[\varphi]$ .

**Definition 8.5.15.**  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignments are extended to  $\mathcal{L}^1$ -formulas, according to the following inductive definition:

$$\sigma[\varphi] = \begin{cases} \{p(\sigma[t_1], \dots, \sigma[t_n])\} & \varphi = \{p(t_1, \dots, t_n)\} \\ \{\bot\} & \varphi = \{\bot\} \\ (\sigma[\varphi_1] \diamond \sigma[\varphi_2]) & \varphi = (\varphi_1 \diamond \varphi_2) \\ (Qx\sigma_{x:=x}[\psi]) & \varphi = (Qx\psi) \text{ for } x \notin Fv[\langle \varphi, \sigma \rangle] \end{cases}$$

Note that the choice of x in the last definition is immaterial, and thus  $\sigma[\varphi]$  is well-defined. The following technical lemmas are needed in the completeness proof.

**Lemma 8.5.16.** Let t be an  $\mathcal{L}^1$ -term.

- 1.  $\sigma_{x:=t}[t'] = \sigma_{y:=t}[t'\{y/x\}]$  for every  $\mathcal{L}^1$ -term t',  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment  $\sigma$ , and variables x, y such that  $y \notin Fv[t']$ .
- 2.  $\sigma_{x:=t}[\varphi] = \sigma_{y:=t}[\varphi\{y/x\}]$  for every  $\mathcal{L}^1$ -formula  $\varphi$ ,  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment  $\sigma$ , and variables x, y such that  $y \notin Fv[\varphi]$ .

*Proof.* This first claim is proved by induction on the structure of t':

- Suppose that t' = c for some constant symbol c of  $\mathcal{L}^1$ , or t' = z for some variable  $z \notin \{x, y\}$ . Then  $t'\{y/x\} = t'$ , and  $\sigma_{x:=t}[t'] = \sigma_{y:=t}[t']$ .
- Suppose that t' = x. Then  $\sigma_{x:=t}[t'] = t$ , and  $\sigma_{y:=t}[t'\{y/x\}] = \sigma_{y:=t}[y] = t$ .
- Suppose that  $t' = f(t_1, ..., t_n)$  for some n-ary function symbol f of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ . Then,  $\sigma_{x:=t}[t'] = f(\sigma_{x:=t}[t_1], ..., \sigma_{x:=t}[t_n])$ . By the induction hypotheses this term equals  $f(\sigma_{y:=t}[t_1\{y/x\}], ..., \sigma_{y:=t}[t_n\{y/x\}])$ , which in turn equals  $\sigma_{y:=t}[f(t_1, ..., t_n)\{y/x\}]$ .

Next, we prove the second claim in the lemma by induction on the complexity of  $\varphi$ . First, suppose that  $cp[\varphi] = 1$ . Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment, and x, y variables such that  $y \notin Fv[\varphi]$ . Exactly one of the following holds:

- $\varphi = \{p(t_1, ..., t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^1$  and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ . Then, by definition  $\sigma_{x:=t}[\varphi] = \{p(\sigma_{x:=t}[t_1], ..., \sigma_{x:=t}[t_n])\}$ . Since  $y \notin Fv[t_i]$  for every  $1 \leq i \leq n$ , this formula equals  $\{p(\sigma_{y:=t}[t_1\{y/x\}], ..., \sigma_{y:=t}[t_n\{y/x\}])\}$ , which is, by definition,  $\sigma_{y:=t}[\varphi\{y/x\}]$ .
- $\varphi = \{\bot\}$ . Then,  $\sigma_{x:=t}[\varphi] = \{\bot\} = \sigma_{y:=t}[\varphi\{y/x\}]$ .

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^1$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment, and x, y variables such that  $y \notin Fv[\varphi]$ . Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for  $\diamond \in \{\land, \lor, \supset\}$  and  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . Then,  $\sigma_{x:=t}[\varphi] = (\sigma_{x:=t}[\varphi_1] \diamond \sigma_{x:=t}[\varphi_2])$ . By the induction hypothesis, this  $\mathcal{L}^1$ -formula is equal to  $(\sigma_{y:=t}[\varphi_1\{y/x\}] \diamond \sigma_{y:=t}[\varphi_2\{y/x\}])$ . And, by definition, this is equal to  $\sigma_{y:=t}[\varphi\{y/x\}]$ .
- $\varphi = (Qz\psi)$  for  $Q \in \{\forall, \exists\}$ ,  $\mathcal{L}^1$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and variable  $z \notin \{x, y\} \cup \sigma[\varphi] \cup Fv[t]$ . Then,  $\sigma_{x:=t}[\varphi] = (Qz\sigma_{x:=t,z:=z}[\psi])$ . By the induction hypothesis, this  $\mathcal{L}^1$ -formula is equal to  $(Qz\sigma_{y:=t,z:=z}[\psi\{y/x\}])$ . And this is (by definition) equal to  $\sigma_{y:=t}[(Qz\psi\{y/x\})]$ , which is equal to  $\sigma_{y:=t}[\varphi\{y/x\}]$ .

#### **Lemma 8.5.17.** Let t be an $\mathcal{L}^1$ -term.

- 1.  $\sigma_{x:=z}[t']\{t/z\} = \sigma_{x:=t}[t']$  for every  $\mathcal{L}^1$ -term t',  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment  $\sigma$ , and variables x, z such that  $z \notin Fv[\sigma[t']]$ .
- 2.  $\sigma_{x:=z}[\varphi]\{t/z\} = \sigma_{x:=t}[\varphi]$  for every  $\mathcal{L}^1$ -formula  $\varphi$ ,  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment  $\sigma$ , and variables x, z such that  $z \notin Fv[\sigma[\varphi]]$ .

*Proof.* The first claim is proved by induction on the structure of t':

• Suppose that t' = c for some constant symbol c of  $\mathcal{L}^1$ . Then:

$$\sigma_{x:=z}[t']\{t/z\} = I^{\mathcal{L}^1}[c]\{t/z\} = c\{t/z\} = c = I^{\mathcal{L}^1}[c] = \sigma_{x:=t}[t'].$$

- Suppose that t' = y for a variable  $y \neq x$ . Then  $\sigma_{x:=z}[t']\{t/z\} = \sigma[y]\{t/z\}$ . Since  $z \notin Fv[\sigma[y]]$ , we have  $\sigma[y]\{t/z\} = \sigma[y]$ . The claim follows since  $\sigma[y] = \sigma_{x:=t}[y]$ .
- Suppose that t' = x. Then,  $\sigma_{x:=z}[t']\{t/z\} = z\{t/z\} = t = \sigma_{x:=t}[x]$ .
- Suppose that  $t' = f(t_1, ..., t_n)$  for some *n*-ary function symbol f of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ . Then,

$$\sigma_{x:=z}[t']\{t/z\} = f(\sigma_{x:=z}[t_1], \dots, \sigma_{x:=z}[t_n])\{t/z\} = f(\sigma_{x:=z}[t_1]\{t/z\}, \dots, \sigma_{x:=z}[t_n]\{t/z\}).$$

By the induction hypotheses this term equals  $f(\sigma_{x:=t}[t_1], \dots, \sigma_{x:=t}[t_n])$ , which in turn equals  $\sigma_{x:=t}[t']$ .

Next, we prove the second claim in the lemma by induction on the complexity of  $\varphi$ . First, suppose that  $cp[\varphi] = 1$ . Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment, and x, z variables such that  $z \notin Fv[\sigma[\varphi]]$ . Exactly one of the following holds:

•  $\varphi = \{p(t_1, ..., t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ . Then, by definition

$$\sigma_{x:=z}[\varphi]\{t/z\} = \{p(\sigma_{x:=z}[t_1], \dots, \sigma_{x:=z}[t_n])\}\{t/z\} =$$

$$= \{p(\sigma_{x:=z}[t_1]\{t/z\}, \dots, \sigma_{x:=z}[t_n]\{t/z\})\}.$$

Since  $z \notin Fv[\sigma[t_i]]$  for every  $1 \le i \le n$ , the claim above for terms entails that this formula equals  $\{p(\sigma_{x:=t}[t_1], \dots, \sigma_{x:=t}[t_n])\}$ , which is, by definition,  $\sigma_{x:=t}[\varphi]$ .

•  $\varphi = \{\bot\}$ . Then,  $\sigma_{x:=z}[\varphi]\{t/z\} = \{\bot\} = \sigma_{x:=t}[\varphi]$ .

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^1$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D}^{\mathcal{L}^1} \rangle$ -assignment, and x, z variables such that  $z \notin Fv[\sigma[\varphi]]$ . Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for  $\diamond \in \{\land, \lor, \supset\}$  and  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . Then,  $\sigma_{x:=x}[\varphi]\{t/x\} = (\sigma_{x:=z}[\varphi_1]\{t/z\} \diamond \sigma_{x:=z}[\varphi_2]\{t/z\})$ . By the induction hypothesis, this  $\mathcal{L}^1$ -formula is equal to  $(\sigma_{x:=t}[\varphi_1] \diamond \sigma_{x:=t}[\varphi_2])$ . And, by definition, this is equal to  $\sigma_{x:=t}[\varphi]$ .
- $\varphi = (Qy\psi)$  for  $Q \in \{\forall, \exists\}$ ,  $\mathcal{L}^1$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and variable  $y \notin Fv[t] \cup \{x, z\} \cup Fv[\sigma[\varphi]]$ . Then,

$$\sigma_{x:=z}[\varphi]\{t/z\} = (Qy\sigma_{x:=z,y:=y}[\psi])\{t/z\} = (Qy\sigma_{x:=z,y:=y}[\psi]\{t/z\}).$$

By the induction hypothesis, this  $\mathcal{L}^1$ -formula is equal to  $(Qy\sigma_{x:=t,y:=y}[\psi])$ . And this is (by definition) equal to  $\sigma_{x:=t}[\varphi]$ .

#### Proof of Theorem 8.5.9

Suppose that  $\mathcal{F}_{HIF}^{ef}$   $H_0$ . The availability of external and internal weakenings ensures that  $H_0$  is unprovable. By Lemma 8.5.12, there exists a maximal extended  $\mathcal{L}^1$ -hypersequent  $\Omega^*$  such that  $H_0 \sqsubseteq \Omega^*$ . We use  $\Omega^*$  to construct a counter-model for  $H_0$  in the form of a quasi- $\mathcal{L}^1$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$ .

First, we define a bounded linearly ordered set  $\mathcal{V}_0$ , that will be used to construct (using the Dedekind-MacNeille completion) the Gödel set  $\mathcal{V}$ . For every  $\mathcal{L}^1$ -formula  $\varphi$  we define:

$$L[\varphi] = \{\mu \in \Omega^* \mid \mathbf{f} : \varphi \in \mu\}, \quad R[\varphi] = \{\mu \in \Omega^* \mid \mathbf{t} : \varphi \not\in \mu\}.$$

Let  $\mathcal{V}_0 = \langle V_0, \subseteq \rangle$ , where

$$V_0 = \{L(\varphi) \mid \varphi \text{ is an } \mathcal{L}^1\text{-formula}\} \cup \{R(\varphi) \mid \varphi \text{ is an } \mathcal{L}^1\text{-formula}\} \cup \{\Omega^*, \emptyset\}.$$

Clearly,  $\mathcal{V}$  is partially ordered set, bounded by  $0 = \emptyset$  and  $1 = \Omega^*$ . The proof that V is linearly ordered by  $\subseteq$  proceeds exactly as in Section 7.5. Now, since  $\mathcal{V}_0$  might not be complete, we consider its Dedekind-MacNeille completion  $\mathcal{V} = \langle V, \subseteq \rangle$  defined by:

$$V = \{ \Pi \subseteq V_0 \mid (\Pi^{\uparrow})^{\downarrow} = \Pi \}$$

where  $\Pi^{\uparrow} = \{\Omega \in V_0 \mid \Omega' \subseteq \Omega \text{ for all } \Omega' \in \Pi\}$  and  $\Pi^{\downarrow} = \{\Omega \in V_0 \mid \Omega \subseteq \Omega' \text{ for all } \Omega' \in \Pi\}$ .  $\mathcal{V}$  is a bounded complete linearly ordered set (see [76]), and thus it forms a Gödel set. Note that using  $\subseteq$  as the order relation, min and max are sets intersection and sets union

(respectively). In addition, the function  $\eta: V_0 \to V$  defined by  $\eta(\Omega) = \{\Omega\}^{\downarrow}$  is injective and it satisfies the following properties:<sup>6</sup>

- $\{\emptyset\} = \eta(\emptyset)$ .
- For every  $\Omega, \Omega' \in V_0$ :

$$-\Omega \subseteq \Omega' \text{ iff } \eta(\Omega) \subseteq \eta(\Omega'). \qquad -\eta(\Omega) \cup \eta(\Omega') = \eta(\Omega \cup \Omega'). \\ -\eta(\Omega) \cap \eta(\Omega') = \eta(\Omega \cap \Omega'). \qquad -\eta(\Omega) \to \eta(\Omega') = \eta(\Omega \to \Omega')$$

- For every  $\Omega \in V_0$  and  $\Pi \subseteq V_0$ :
  - If  $\Omega \subseteq \bigcap_{\Omega' \in \Pi} \Omega'$ , then  $\eta(\Omega) \subseteq \inf_{\Omega' \in \Pi} \eta(\Omega')$ .
  - If  $\bigcap_{\Omega' \in \Pi} \Omega' \subseteq \Omega$ , then  $\inf_{\Omega' \in \Pi} \eta(\Omega') \subseteq \eta(\Omega)$ .
  - If  $\Omega \subseteq \bigcup_{\Omega' \in \Pi} \Omega'$ , then  $\eta(\Omega) \subseteq \sup_{\Omega' \in \Pi} \eta(\Omega')$ .
  - If  $\bigcup_{\Omega' \in \Pi} \Omega' \subseteq \Omega$ , then  $\sup_{\Omega' \in \Pi} \eta(\Omega') \subseteq \eta(\Omega)$ .

The proofs of these properties are straightforward (note that the linearity of  $V_0$  is needed in some of them).

Henceforth, we will identify the elements of  $V_0$  of the form  $\{\Omega\}^{\downarrow}$  with the (unique) corresponding element  $\Omega$ , and freely use the properties above.

Next, for every formula  $\varphi$ , let  $\Omega^*[\varphi]$  be the pair defined by:  $\Omega^*[\varphi] = \langle L[\varphi], R[\varphi] \rangle$ . Note that  $\Omega^*[\varphi] \in \{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \subseteq u^{\mathbf{t}} \}$  for every  $\mathcal{L}^1$ -formula  $\varphi$ . Indeed, in the presence of (id), either  $\mathbf{f}:\varphi \notin \mu$  or  $\mathbf{t}:\varphi \notin \mu$  for every  $\mu \in \Omega$  and  $\mathcal{L}^1$ -formula  $\varphi$  (otherwise,  $\{\{\mathbf{f}:\varphi,\mathbf{t}:\varphi\}\} \sqsubseteq \Omega$ , contradicting the fact that  $\Omega$  is unprovable), and consequently,  $L[\varphi] \subseteq R[\varphi]$ . Let  $\mathcal{D}$  be the Herbrand domain for  $\mathcal{L}^1$ , I the Herbrand interpretation for  $\mathcal{L}^1$ , and define P and v as follows:

- For every n-ary predicate symbol p of  $\mathcal{L}^1$ ,  $P[p] = \lambda t_1, \dots, t_n \in \mathcal{D}.\Omega^*[\{p(t_1, \dots, t_n)\}].$
- For every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $v[\varphi, \sigma] = \Omega^*[\sigma[\varphi]]$ .

It is easy to verify the condition on v from Definition 8.5.2. Indeed, Lemma 8.5.16 ensures that if  $y \notin Fv[\varphi]$ , then for every  $\mathcal{L}^1$ -term we have  $\sigma_{x:=t}[\varphi] = \sigma_{y:=t}[\varphi\{y/x\}]$ . This implies that  $v[\varphi, \sigma_{x:=t}] = v[\varphi\{y/x\}, \sigma_{y:=t}]$  for every  $t \in \mathcal{D}$ .

We show that  $\mathcal{Q}$  is not a model of  $H_0$ . Consider the  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma_{id}$  defined by  $\sigma_{id}[x] = x$  for every variable x of  $\mathcal{L}^1$ . Let  $s \in H_0$ . Since  $H_0 \sqsubseteq \Omega^*$ , there exists some  $\mu \in \Omega^*$ , such that  $s \subseteq \mu$ . We claim that  $\mu \in v^{\mathfrak{r}}[\varphi, \sigma_{id}]$  whenever  $\mathfrak{f}: \varphi \in s$ , and  $\mu \notin v^{\mathfrak{r}}[\varphi, \sigma_{id}]$ whenever  $\mathfrak{t}: \varphi \in s$ . To see this, it suffices to note that  $\sigma_{id}[\varphi] = \varphi$  for every  $\mathcal{L}^1$ -formula  $\varphi$ . This fact follows from the definition of  $\sigma_{id}[\varphi]$ . Consequently,  $v^{\mathfrak{r}}[\varphi, \sigma_{id}] \not\subseteq v^{\mathfrak{r}}[\varphi, \sigma_{id}]$ , and so  $\mathcal{Q}, \sigma_{id} \not\models s$ .

<sup>&</sup>lt;sup>6</sup>All operations notations from Definition 8.2.1 are adopted to the set  $\mathcal{V}_0$  in the obvious way.

It remains to prove that  $\mathcal{Q}$  is legal, namely that  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] \subseteq v[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ . Let  $\varphi$  be an  $\mathcal{L}^1$ -formula, and  $\sigma$  an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Then, exactly one of the following holds:

•  $\varphi = \{p(t_1, ..., t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, ..., t_n$ . Then, by definition:

$$\underline{v}_{\mathcal{O}}[\varphi,\sigma] = P[p][\sigma[t_1],\ldots,\sigma[t_n]] = \Omega^*[\{p(\sigma[t_1],\ldots,\sigma[t_n])\}] = \Omega^*[\sigma[\varphi]] = v[\varphi,\sigma].$$

- $\varphi = \{\bot\}$ . Then,  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] = \langle \emptyset, \emptyset \rangle$ . To see that  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] \subseteq v[\varphi, \sigma]$ , it suffices to note that  $v^{\mathbf{f}}[\varphi, \sigma] = \emptyset$ . This follows from the fact that  $\mathbf{f}:\sigma[\varphi] = \mathbf{f}:\{\bot\} \not\in \mu$  for every  $\mu \in \Omega^*$ . (Otherwise,  $\{\{\mathbf{f}:\{\bot\}\}\}\subseteq \Omega^*$ , but  $\vdash_{\mathbf{HIF}}^{cf} \{\{\mathbf{f}:\{\bot\}\}\}$  by applying the rule  $(\mathbf{f}:\bot)$ .)
- $\varphi = (\varphi_1 \wedge \varphi_2)$  for some  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$ . Then:

$$\underline{v}_{\mathcal{Q}}[\varphi,\sigma] = \langle v^{\mathbf{f}}[\varphi_1,\sigma] \cap v^{\mathbf{f}}[\varphi_2,\sigma], v^{\mathbf{t}}[\varphi_1,\sigma] \cap v^{\mathbf{t}}[\varphi_2,\sigma] \rangle.$$

We first prove that  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq v^{\mathbf{f}}[\varphi_1, \sigma] \cap v^{\mathbf{f}}[\varphi_2, \sigma]$ . Let  $\mu \notin v^{\mathbf{f}}[\varphi_1, \sigma]$  ( $\mu \notin v^{\mathbf{f}}[\varphi_2, \sigma]$  is symmetric). Thus  $\mathbf{f}: \sigma[\varphi_1] \notin \mu$ . We show that  $\mu \notin v^{\mathbf{f}}[\varphi, \sigma]$ . By Proposition 8.5.11, since  $\mathbf{f}: \sigma[\varphi_1] \notin \mu$ , there exist an  $\mathcal{L}^1$ -hypersequent  $H_1 \sqsubseteq \Omega^*$ , and  $\mathcal{L}^1$ -sequents  $s_1, \ldots, s_n \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_1 \mid s_1 \cup \{\mathbf{f}: \sigma[\varphi_1]\} \mid \ldots \mid s_n \cup \{\mathbf{f}: \sigma[\varphi_1]\}$ . The availability of weakening and  $(\mathbf{f}: \wedge)$  entails that  $\vdash_{\mathbf{HIF}}^{cf} H$  for

$$H = H_1 \mid s_1 \cup \{ \mathtt{f} : (\sigma[\varphi_1] \land \sigma[\varphi_2]) \} \mid \dots \mid s_n \cup \{ \mathtt{f} : (\sigma[\varphi_1] \land \sigma[\varphi_2]) \}.$$

Since  $\Omega^*$  is unprovable,  $H \not\sqsubseteq \Omega^*$ , and thus  $\mathbf{f}:(\sigma[\varphi_1] \wedge \sigma[\varphi_2]) \not\in \mu$ . By definition,  $(\sigma[\varphi_1] \wedge \sigma[\varphi_2]) = \sigma[\varphi]$ . It follows that  $\mu \not\in v^{\mathbf{f}}[\varphi, \sigma]$ .

Next, we prove that  $v^{\mathbf{t}}[\varphi_1, \sigma] \cap v^{\mathbf{t}}[\varphi_2, \sigma] \subseteq v^{\mathbf{t}}[\varphi, \sigma]$ . Let  $\mu \in v^{\mathbf{t}}[\varphi_1, \sigma] \cap v^{\mathbf{t}}[\varphi_2, \sigma]$ . Then we have  $\mathbf{t}:\sigma[\varphi_1] \not\in \mu$  and  $\mathbf{t}:\sigma[\varphi_2] \not\in \mu$ . By Proposition 8.5.11, there exist  $\mathcal{L}^1$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega^*$ , and negative  $\mathcal{L}^1$ -sequents  $s_1, s_2 \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_1 \mid s_1 \cup \{\mathbf{t}:\sigma[\varphi_1]\}$  and  $\vdash_{\mathbf{HIF}}^{cf} H_2 \mid s_2 \cup \{\mathbf{t}:\sigma[\varphi_2]\}$ . The availability of  $(\mathbf{t}:\wedge)$  entails that  $\vdash_{\mathbf{HIF}}^{cf} H$  for  $H = H_1 \mid H_2 \mid s_1 \cup s_2 \cup \{\mathbf{t}:(\sigma[\varphi_1] \wedge \sigma[\varphi_2])\}$ . Since  $\Omega^*$  is unprovable,  $H \not\sqsubseteq \Omega^*$ , and thus  $\mathbf{t}:\sigma[\varphi] = \mathbf{t}:(\sigma[\varphi_1] \wedge \sigma[\varphi_2]) \not\in \mu$ . It follows that  $\mu \in v^{\mathbf{t}}[\varphi, \sigma]$ .

•  $\varphi = (\varphi_1 \vee \varphi_2)$  for some  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$ . Then:

$$\underline{v}_{\mathcal{O}}[\varphi,\sigma] = \langle v^{\mathsf{t}}[\varphi_1,\sigma] \cup v^{\mathsf{t}}[\varphi_2,\sigma], v^{\mathsf{f}}[\varphi_1,\sigma] \cup v^{\mathsf{f}}[\varphi_2,\sigma] \rangle.$$

We first prove that  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq v^{\mathbf{f}}[\varphi_1, \sigma] \cup v^{\mathbf{f}}[\varphi_2, \sigma]$ . Let  $\mu \not\in v^{\mathbf{f}}[\varphi_1, \sigma] \cup v^{\mathbf{f}}[\varphi_2, \sigma]$ . We prove that  $\mu \not\in v^{\mathbf{f}}[\varphi, \sigma]$ . Our assumption entails that  $\mathbf{f}: \sigma[\varphi_1] \not\in \mu$  and  $\mathbf{f}: \sigma[\varphi_2] \not\in \mu$ . By Proposition 8.5.11, there exist  $\mathcal{L}^1$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega^*$ , and  $\mathcal{L}^1$ -sequents  $s_1, \ldots, s_n, s'_1, \ldots, s'_m \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_1 \mid s_1 \cup \{\mathbf{f}: \sigma[\varphi_1]\} \mid \ldots \mid s_n \cup \{\mathbf{f}: \sigma[\varphi_1]\}$  and

 $\vdash_{\mathbf{HIF}}^{cf} H_2 \mid s_1' \cup \{\mathbf{f}: \sigma[\varphi_2]\} \mid ... \mid s_m' \cup \{\mathbf{f}: \sigma[\varphi_2]\}$ . As in the proof of Proposition 7.2.11, it is possible to use (com) and  $(\mathbf{f}: \vee)$  to obtain that  $\vdash_{\mathbf{HIF}}^{cf} H$  for

$$H = H_1 \mid H_2 \mid s_1 \cup \{\mathbf{f}: (\sigma[\varphi_1] \vee \sigma[\varphi_2])\} \mid \dots \mid s_n \cup \{\mathbf{f}: (\sigma[\varphi_1] \vee \sigma[\varphi_2])\} \mid s_1' \cup \{\mathbf{f}: (\sigma[\varphi_1] \vee \sigma[\varphi_2])\} \mid \dots \mid s_m' \cup \{\mathbf{f}: (\sigma[\varphi_1] \vee \sigma[\varphi_2])\}.$$

Since  $\Omega^*$  is unprovable,  $H \not\sqsubseteq \Omega^*$ , and thus  $\mathbf{f}:(\sigma[\varphi_1] \vee \sigma[\varphi_2]) \not\in \mu$ . By definition,  $(\sigma[\varphi_1] \vee \sigma[\varphi_2]) = \sigma[\varphi]$ . It follows that  $\mu \not\in v^{\mathbf{f}}[\varphi, \sigma]$ .

Next, we prove that  $v^{\mathbf{t}}[\varphi_1, \sigma] \cup v^{\mathbf{t}}[\varphi_2, \sigma] \subseteq v^{\mathbf{t}}[\varphi, \sigma]$ . Let  $\mu \in v^{\mathbf{t}}[\varphi_1, \sigma]$ , and so  $\mathbf{t}: \sigma[\varphi_1] \not\in \mu$  (the case that  $\mu \in v^{\mathbf{t}}[\varphi_2, \sigma]$  is symmetric). By Proposition 8.5.11, there exist an  $\mathcal{L}^1$ -hypersequent  $H_1 \sqsubseteq \Omega^*$ , and a negative  $\mathcal{L}^1$ -sequent  $s \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_1 \mid s \cup \{\mathbf{t}: \sigma[\varphi_1]\}$ . The availability of weakening and  $(\mathbf{t}: \vee)$  entails that  $\vdash_{\mathbf{HIF}}^{cf} H$  for  $H = H_1 \mid s \cup \{\mathbf{t}: (\sigma[\varphi_1] \vee \sigma[\varphi_2])\}$ . Since  $\Omega^*$  is unprovable,  $H \not\sqsubseteq \Omega^*$ , and thus  $\mathbf{t}: \sigma[\varphi] = \mathbf{t}: (\sigma[\varphi_1] \vee \sigma[\varphi_2]) \not\in \mu$ . It follows that  $\mu \in v^{\mathbf{t}}[\varphi, \sigma]$ .

•  $\varphi = (\varphi_1 \supset \varphi_2)$  for some  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$ . Then:

$$\underline{v}_{\mathcal{Q}}[\varphi,\sigma] = \langle v^{\mathbf{f}}[\varphi_1,\sigma] \to v^{\mathbf{t}}[\varphi_2,\sigma], v^{\mathbf{t}}[\varphi_1,\sigma] \to v^{\mathbf{f}}[\varphi_2,\sigma] \rangle.$$

We first prove that  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq v^{\mathbf{t}}[\varphi_1, \sigma] \to v^{\mathbf{f}}[\varphi_2, \sigma]$ . Let  $\mu \notin v^{\mathbf{t}}[\varphi_1, \sigma] \to v^{\mathbf{f}}[\varphi_2, \sigma]$ . Then,  $v^{\mathbf{t}}[\varphi_1, \sigma] \not\subseteq v^{\mathbf{f}}[\varphi_2, \sigma]$  and  $\mu \notin v^{\mathbf{f}}[\varphi_2, \sigma]$ . Let  $\mu' \in \Omega^*$  such that  $\mu' \in v^{\mathbf{t}}[\varphi_1, \sigma]$ , and  $\mu' \notin v^{\mathbf{f}}[\varphi_2, \sigma]$ . Hence,  $\mathbf{t}: \sigma[\varphi_1] \not\in \mu'$  and  $\mathbf{f}: \sigma[\varphi_2] \not\in \mu'$ . By Proposition 8.5.11, there exist  $\mathcal{L}^1$ -hypersequents  $H_1, H_2 \sqsubseteq \Omega^*$ , a negative  $\mathcal{L}^1$ -sequent  $s' \subseteq \mu'$ , and  $\mathcal{L}^1$ -sequents  $s'_1, \dots, s'_n \subseteq \mu'$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_1 \mid s' \cup \{\mathbf{t}: \sigma[\varphi_1]\}$ , and

$$\vdash_{\mathbf{HIF}}^{cf} H_2 \mid s_1' \cup \{\mathbf{f} \mathpunct{:}\! \sigma[\varphi_2]\} \mid \ldots \mid s_n' \cup \{\mathbf{f} \mathpunct{:}\! \sigma[\varphi_2]\}.$$

By n consecutive applications of  $(f: \supset)$  (note that  $(\sigma[\varphi_1] \supset \sigma[\varphi_2]) = \sigma[\varphi]$ ), we obtain that

$$\vdash_{\mathbf{HIF}}^{cf} H_1 \mid H_2 \mid s' \cup s'_1 \cup \{\mathbf{f} : \sigma[\varphi]\} \mid \dots \mid s' \cup s'_n \cup \{\mathbf{f} : \sigma[\varphi]\}. \tag{8.1}$$

Since  $\mu \notin v^{\mathbf{f}}[\varphi_2, \sigma]$ , we also have  $\mathbf{f}:\sigma[\varphi_2] \notin \mu$ . Proposition 8.5.11 entails that there also exist  $\mathcal{L}^1$ -hypersequent  $H_3 \sqsubseteq \Omega^*$ , and  $\mathcal{L}^1$ -sequents  $s_1, \ldots, s_m \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H_3 \mid s_1 \cup \{\mathbf{f}:\sigma[\varphi_2]\} \mid \ldots \mid s_m \cup \{\mathbf{f}:\sigma[\varphi_2]\}$ . By another m applications of  $(\mathbf{f}: \supset)$ , we obtain that

$$\vdash_{\mathbf{HIF}}^{cf} H_1 \mid H_3 \mid s' \cup s_1 \cup \{\mathbf{f}: \sigma[\varphi]\} \mid \dots \mid s' \cup s_m \cup \{\mathbf{f}: \sigma[\varphi]\}. \tag{8.2}$$

Using a generalized version of (com) (see Proposition 7.2.10) we obtain from (8.1) and (8.2) above:

$$\vdash_{\mathbf{HIF}}^{cf} H_1 \mid H_2 \mid H_3 \mid s' \cup s'_1 \mid ... \mid s' \cup s'_n \mid s_1 \cup \{ \mathbf{f} : \sigma[\varphi] \} \mid ... \mid s_m \cup \{ \mathbf{f} : \sigma[\varphi] \}.$$

Now, if  $\mathbf{f}:\sigma[\varphi] \in \mu$ , then  $\Omega^*$  extends this hypersequent, and this contradicts the fact that  $\Omega^*$  is unprovable. Therefore,  $\mathbf{f}:\sigma[\varphi] \notin \mu$ , and consequently  $\mu \notin v^{\mathbf{f}}[\sigma[\varphi]]$ .

Next, we prove that  $v^{\mathbf{f}}[\varphi_1, \sigma] \to v^{\mathbf{t}}[\varphi_2, \sigma] \subseteq v^{\mathbf{t}}[\varphi, \sigma]$ . Suppose that  $\mu \notin v^{\mathbf{t}}[\varphi, \sigma]$ , and so  $\mathbf{t}$ : $\sigma[\varphi] \in \mu$ . To show that  $\mu \notin v^{\mathbf{f}}[\varphi_1, \sigma] \to v^{\mathbf{t}}[\varphi_2, \sigma]$ , we first show that

 $\mu \not\in v^{\mathsf{t}}[\varphi_2, \sigma]$  and then we show that  $v^{\mathsf{f}}[\varphi_1, \sigma] \not\subseteq v^{\mathsf{t}}[\varphi_2, \sigma]$ :

- 1. Assume for contradiction that  $\mu \in v^{\mathsf{t}}[\varphi_2, \sigma]$ , and thus  $\mathsf{t}:\sigma[\varphi_2] \not\in \mu$ . Then by Proposition 8.5.11, there exist an  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega^*$ , and a negative  $\mathcal{L}^1$ -sequent  $s \subseteq \mu$ , such that  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{\mathsf{t}:\sigma[\varphi_2]\}$ . By applying internal weakening we obtain  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{\mathsf{f}:\sigma[\varphi_1], \mathsf{t}:\sigma[\varphi_2]\}$ . Using  $(\mathsf{t}: \supset)$  we obtain  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{\mathsf{t}:\sigma[\varphi]\}$ . This contradicts the fact that  $\Omega^*$  is unprovable (because  $H \mid s \cup \{\mathsf{t}:\sigma[\varphi]\} \sqsubseteq \Omega^*$ ).
- 2. Note that the fact that  $\Omega^*$  is unprovable and the availability of  $(\mathbf{t}: \supset)$  also entail that  $\not\vdash_{\mathbf{HIF}}^{cf} H \mid \{\mathbf{f}:\sigma[\varphi_1], \mathbf{t}:\sigma[\varphi_2]\}$ . Therefore, Proposition 8.5.11 entails that  $\{\mathbf{f}:\sigma[\varphi_1], \mathbf{t}:\sigma[\varphi_2]\} \sqsubseteq \Omega^*$ . Thus there is an extended  $\mathcal{L}^1$ -sequent  $\mu' \in \Omega^*$ , such that  $\mathbf{f}:\sigma[\varphi_1] \in \mu'$  and  $\mathbf{t}:\sigma[\varphi_2] \in \mu'$ . Consequently,  $\mu' \in v^{\mathbf{f}}[\varphi_1, \sigma]$  and  $\mu' \notin v^{\mathbf{t}}[\varphi_2, \sigma]$ . Hence  $v^{\mathbf{f}}[\varphi_1, \sigma] \not\subseteq v^{\mathbf{t}}[\varphi_2, \sigma]$ .
- $\varphi = (\exists x \psi)$  for some variable  $x \notin Fv[\sigma[\varphi]]$  and  $\mathcal{L}^1$ -formula  $\psi$ . Then:

$$\underline{v}_{\mathcal{Q}}[\varphi, \sigma] = \langle \sup_{t \in \mathcal{D}} v^{\mathsf{t}}[\psi, \sigma_{x:=t}], \sup_{t \in \mathcal{D}} v^{\mathsf{f}}[\psi, \sigma_{x:=t}] \rangle.$$

We first prove that  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq \sup_{t \in \mathcal{D}} v^{\mathbf{f}}[\psi, \sigma_{x:=t}]$ . Suppose that  $\mu \in v^{\mathbf{f}}[\varphi, \sigma]$ . Thus  $\mathbf{f}: \sigma[\varphi] \in \mu$ . By definition,  $\sigma[\varphi] = (\exists x \sigma_{x:=x}[\psi])$ . Since  $\Omega^*$  admits the witness property, there exists a variable y of  $\mathcal{L}^1$ , such that  $\mathbf{f}: \sigma_{x:=x}[\psi]\{y/x\} \in \mu$ . By Lemma 8.5.17,  $\sigma_{x:=x}[\psi]\{y/x\} = \sigma_{x:=y}[\psi]$ . It follows that  $\mu \in v^{\mathbf{f}}[\psi, \sigma_{x:=y}]$ , and therefore  $\mu \in \bigcup_{t \in \mathcal{D}} v^{\mathbf{f}}[\psi, \sigma_{x:=t}]$ .

Next, we prove that  $\sup_{t\in\mathcal{D}} v^{\mathsf{t}}[\psi, \sigma_{x:=t}] \subseteq v^{\mathsf{t}}[\varphi, \sigma]$ . Let  $\mu \in \bigcup_{t\in\mathcal{D}} v^{\mathsf{t}}[\psi, \sigma_{x:=t}]$ . Thus  $\mu \in v^{\mathsf{t}}[\psi, \sigma_{x:=t}]$  for some  $t \in \mathcal{D}$ . By definition,  $\mathsf{t}:\sigma_{x:=t}[\psi] \not\in \mu$ . By Lemma 8.5.17,  $\sigma_{x:=t}[\psi] = \sigma_{x:=x}[\psi]\{t/x\}$ . By Proposition 8.5.11,  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{\mathsf{t}:\sigma_{x:=x}[\psi]\{t/x\}\}$  for some  $\mathcal{L}^1$ -hypersequent  $H \sqsubseteq \Omega^*$ , and negative  $\mathcal{L}^1$ -sequent  $s \subseteq \mu$ . By an application of  $(\mathsf{t}:\exists)$ , we obtain  $\vdash_{\mathbf{HIF}}^{cf} H \mid s \cup \{\mathsf{t}:(\exists x \sigma_{x:=x}[\psi])\}$ . Since  $\Omega^*$  is unprovable,  $\mathsf{t}:(\exists x \sigma_{x:=x}[\psi]) \not\in \mu$ . By definition,  $(\exists x \sigma_{x:=x}[\psi]) = \sigma[\varphi]$ . It follows that  $\mu \in v^{\mathsf{t}}[\varphi, \sigma]$ .

• The case  $\varphi = (\forall x \psi)$  is handled similarly.

### 8.6 Completeness for the Ordinary Semantics

In this section we use the complete semantics of quasi-structures to prove the completeness of **HIF** for the (ordinary) structures of first-order Gödel logic. To do so, we show that from every legal quasi-structure which is a counter-model of some hypersequent H, it is possible to extract an (ordinary) structure, which is also not a model of H. **Theorem 8.6.1.** Let  $Q = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  be a legal quasi- $\mathcal{L}^1$ -structure. There exists an  $\mathcal{L}^1$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P' \rangle$ , such that  $\mathcal{W}[\varphi, \sigma] \in v[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ .

Proof. Define P' by  $P'[p] = P[p]^f$  for every predicate symbol p. We prove that the  $\mathcal{L}^1$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P' \rangle$  satisfies the requirement in the theorem. Let  $\mathcal{V} = \langle V, \leq \rangle$ . We use induction on the complexity of  $\varphi$  to show that  $\mathcal{W}[\varphi, \sigma] \in v[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ . Note that since  $\mathcal{Q}$  is legal, it suffices to show that  $\mathcal{W}[\varphi, \sigma] \in \underline{v}_{\mathcal{Q}}[\varphi, \sigma]$  for every  $\mathcal{L}^1$ -formula  $\varphi$  and  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ . First, suppose that  $cp[\varphi] = 1$ , and let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Exactly one of the following holds:

- $\varphi = \{p(t_1, \dots, t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -terms  $t_1, \dots, t_n$ . By definition,  $\mathcal{W}[\varphi, \sigma] = P'[p][\sigma^I[t_1], \dots, \sigma^I[t_n]] = P[p]^{\mathfrak{r}}[\sigma^I[t_1]], \dots, \sigma^I[t_n]] \in \underline{v}_{\mathcal{O}}[\varphi, \sigma]$ .
- $\varphi = \{\bot\}$ . Then by definition,  $\mathcal{W}[\varphi, \sigma] = 0 \in \langle 0, 0 \rangle = \underline{v}_{\mathcal{Q}}[\varphi, \sigma]$ .

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^1$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment. Exactly one of the following holds:

•  $\varphi = (\varphi_1 \wedge \varphi_2)$  for  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . By the induction hypothesis,

$$\mathcal{W}[\varphi_1, \sigma] \in \langle v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_1, \sigma] \rangle$$
 and  $\mathcal{W}[\varphi_2, \sigma] \in \langle v^{\mathbf{f}}[\varphi_2, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma] \rangle$ .  
Hence,

$$\min\{\mathcal{W}[\varphi_1,\sigma],\mathcal{W}[\varphi_2,\sigma]\} \in \langle \min\{v^{\mathbf{f}}[\varphi_1,\sigma],v^{\mathbf{f}}[\varphi_2,\sigma]\}, \min\{v^{\mathbf{t}}[\varphi_1,\sigma],v^{\mathbf{t}}[\varphi_2,\sigma]\}\rangle,$$
 and so  $\mathcal{W}[\varphi,\sigma] \in \underline{v}_{\mathcal{O}}[\varphi,\sigma].$ 

- $\varphi = (\varphi_1 \vee \varphi_2)$  for  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . This case is similar to the previous case (replace min by max).
- $\varphi = (\varphi_1 \supset \varphi_2)$  for  $\mathcal{L}^1$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . By the induction hypothesis,

$$\mathcal{W}[\varphi_1, \sigma] \in \langle v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_1, \sigma] \rangle \quad \text{and} \quad \mathcal{W}[\varphi_2, \sigma] \in \langle v^{\mathbf{f}}[\varphi_2, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma] \rangle.$$
  
Since  $u_1 \leq u' \leq u_2$  and  $u_3 \leq u'' \leq u_4$  imply that  $u_2 \to u_3 \leq u' \to u'' \leq u_1 \to u_4$ , we obtain that:  $\mathcal{W}[\varphi_1, \sigma] \to \mathcal{W}[\varphi_2, \sigma] \in \langle v^{\mathbf{t}}[\varphi_1, \sigma] \to v^{\mathbf{f}}[\varphi_2, \sigma], v^{\mathbf{f}}[\varphi_1, \sigma] \to v^{\mathbf{t}}[\varphi_2, \sigma] \rangle,$  and so  $\mathcal{W}[\varphi, \sigma] \in \underline{v}_{\mathcal{Q}}[\varphi, \sigma].$ 

•  $\varphi = (Qx\psi)$  for some  $Q \in \{\forall, \exists\}$ , variable x of  $\mathcal{L}^1$ , and  $\mathcal{L}^1$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . We continue with  $Q = \forall$  (the proof is similar for  $\exists$ ). By the induction hypothesis, for every  $d \in \mathcal{D}$ ,  $\mathcal{W}[\psi, \sigma_{x:=d}] \in v[\psi, \sigma_{x:=d}]$ . Hence,

$$\mathcal{W}[\varphi,\sigma] = \inf_{d \in \mathcal{D}} \mathcal{W}[\psi,\sigma_{x:=d}] \in \langle \inf_{d \in \mathcal{D}} v^{\mathbf{f}}[\psi,\sigma_{x:=d}], \inf_{d \in \mathcal{D}} v^{\mathbf{t}}[\psi,\sigma_{x:=d}] \rangle = \underline{v}_{\mathcal{Q}}[\varphi,\sigma]. \qquad \Box$$

Corollary 8.6.2. If  $\not\vdash_{\mathbf{HIF}}^{cf} H$ , then there exists an  $\mathcal{L}^1$ -structure which is not a model of H.

Proof. Suppose that  $\not\vdash_{\mathbf{HIF}}^{cf} H$ . Then, by Theorem 8.5.9, there exists a legal quasi- $\mathcal{L}^1$ structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$ , which is not a model of H. This implies that there exists
an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -assignment  $\sigma$ , such that  $\mathcal{Q}, \sigma \not\models s$  for every  $s \in H$ . Let  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P' \rangle$  be
an  $\mathcal{L}^1$ -structure satisfying the requirement in Theorem 8.6.1. We show that  $\mathcal{W}, \sigma \not\models H$ .
Let  $s \in H$ . Since  $\mathcal{Q}, \sigma \not\models s$ , we have  $\mathcal{Q}^{\mathbf{f}}[s, \sigma] > \mathcal{Q}^{\mathbf{t}}[s, \sigma]$ . The fact that  $\mathcal{W}[\varphi, \sigma] \in v[\varphi, \sigma]$ for every  $\varphi$  entails that  $\mathcal{W}^{\mathbf{f}}[\varphi, \sigma] \geq v^{\mathbf{f}}[\varphi, \sigma]$  and  $\mathcal{W}^{\mathbf{t}}[\varphi, \sigma] \leq v^{\mathbf{t}}[\varphi, \sigma]$  for every  $\varphi$ . It follows
that  $\mathcal{W}^{\mathbf{f}}[s, \sigma] > \mathcal{W}^{\mathbf{t}}[s, \sigma]$ , and so  $\mathcal{W}, \sigma \not\models s$ .

Corollary 8.6.3. For every  $\mathcal{L}^1$ -formula  $\varphi$ , if  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^1}} \varphi$  then  $\vdash_{\mathbf{HIF}} \{\{\mathbf{t}:\varphi\}\}$ .

Finally, we automatically obtain the admissibility of the cut rule:

Corollary 8.6.4. If an  $\mathcal{L}^1$ -hypersequent is provable in HIF then it is provable in HIF without applying (cut).

*Proof.* Note that if an  $\mathcal{L}^1$ -hypersequent H is provable in **HIF**, then every  $\mathcal{L}^1$ -structure is a model of H. By Corollary 8.6.2, this implies that  $\vdash_{\mathbf{HIF}}^{cf} H$ .

**Remark 8.6.5.** While we allowed any Gödel set to serve as the set of truth values in  $\mathcal{L}^1$ -structures, we could equivalently take the real interval [0,1]. Obviously, soundness for [0,1] is a particular instance. Completeness for [0,1] can be obtained by embedding the set  $V_0$  in the proof of Theorem 8.5.9 into the rational numbers in [0,1], and continuing the proof with  $\mathcal{V} = [0,1]$ .

## Chapter 9

# Calculus for Second-Order Gödel Logic

Fuzzy logics, and Gödel logic in particular, have a wide variety of applications, as they provide a reasonable model of certain very common vagueness phenomena. Both their propositional and first-order versions are well-studied by now (see, e.g., [63]). Clearly, for many interesting applications (see, e.g., [38] and Section 5.5.2 in Chapter I of [45]), propositional and first-order fuzzy logics do not suffice, and one has to use higher-order versions. These are much less developed (see, e.g., [95] and [45]), especially from the proof-theoretic perspective. Evidently, higher-order fuzzy logics deserve a proof-theoretic study, with the aim of providing a basis for automated deduction methods, as well as a complimentary point of view in the investigation of these logics.

In this chapter we study the extension of **HIF** with usual rules for second-order quantifiers. These consist of the single-conclusion hypersequent version of the rules for introducing the second-order quantifiers in the ordinary sequent calculus for classical logic (see, e.g., [58, 90]). We denote by **HIF**<sup>2</sup> the extension of **HIF** with these rules. To the best of our knowledge, this system is studied here for the first time. Our main results is that **HIF**<sup>2</sup> is sound and complete for second-order Gödel logic, and that (cut) is admissible in **HIF**<sup>2</sup>. It should be noted that like in the case of second-order classical logic, the obtained calculus characterizes Henkin-style second-order Gödel logic. Thus second-order quantifiers range over a domain (of fuzzy sets) that is directly specified in the second-order structure, and this domain should admit full comprehension. This is in contrast to what is called the standard semantics, where second-order quantifiers range over all subsets of the universe. Hence **HIF**<sup>2</sup> is practically a system for two-sorted first-order Gödel logic together with the comprehension axiom (see also [37]).

Our approach in proving cut-admissibility for HIF<sup>2</sup> is (of-course) semantic, and it is similar to the one taken in Chapter 8 for HIF. Note that unlike in first-order calculi,

usual syntactic arguments for cut-elimination dramatically fail for the rules of second-order quantifiers. Thus the first proof of cut-admissibility for the extension of **LK** with rules for second-order quantifiers was also semantic (see also the discussion in Chapter 1, page 8).

#### Publications Related to this Chapter

The material in this chapter was not published before.

### 9.1 Preliminaries

For the simplicity of the presentation, we follow [58] and restrict ourselves to simplified second-order languages, in which the second-order part of the signature consists only of one predicate symbol  $\varepsilon$  (with the intuitive meaning of set inclusion). This is formulated in the next definition.

**Definition 9.1.1.** A *simple second-order language* is obtained by augmenting a first-order language (see Definition 8.1.1) with the following:

- 1. Infinitely many set variables  $\chi_1, \chi_2, ...$  We use the metavariables X, Y, Z (with or without subscripts) for set variables. To avoid confusion, we shall refer the variables  $\nu_1, \nu_2, ...$  of the underlying first-order language as individual variables.
- 2. Set quantifiers  $\forall^s$  and  $\exists^s$ . We use  $Q^s$  as a metavariable for the set quantifiers. We shall refer the quantifiers  $\forall$ ,  $\exists$  of the underlying first-order language as *individual* quantifiers, and usually denote them by  $\forall^i$  and  $\exists^i$ .
- 3. An arbitrary set of set constant symbols. The metavariable C are is to range over set constant symbols. We shall refer the constant symbols of the underlying first-order language as individual constant symbols.
- 4. A predicate symbol  $\varepsilon$  with two places, the first for individuals and the second for sets.

In what follows,  $\mathcal{L}^2$  denotes an arbitrary simple second-order language.

**Definition 9.1.2.** The set of  $\mathcal{L}^2$ -terms consists of first-order  $\mathcal{L}^2$ -terms and second-order  $\mathcal{L}^2$ -terms. First-order  $\mathcal{L}^2$ -terms are defined as in Definition 8.1.2, while second-order  $\mathcal{L}^2$ -terms consists of all set variables of  $\mathcal{L}^2$  and all set constant symbols of  $\mathcal{L}^2$ . We use T (with or without subscripts) as a metavariable for second-order  $\mathcal{L}^2$ -terms. The set of (individual) variables occurring in a first-order  $\mathcal{L}^2$ -term t is defined as usual, and denoted by Fv[t]. Similarly, the set of set variables occurring in a second-order  $\mathcal{L}^2$ -term

9.1. Preliminaries 159

T is denoted by Fv[T]. Substitutions in first-order  $\mathcal{L}^2$ -terms (denoted by  $t'\{t/x\}$ ) are defined as in Definition 8.1.7.

**Definition 9.1.3.** Concrete  $\mathcal{L}^2$ -formulas are defined as for a first-order language (Definition 8.1.3), with the following additions:

- 1.  $(t\varepsilon T)$  is a concrete  $\mathcal{L}^2$ -formula for every first-order  $\mathcal{L}^2$ -term t and second-order  $\mathcal{L}^2$ -term T.
- 2. If  $\Phi$  is a concrete  $\mathcal{L}^2$ -formula, and X is a set variable of  $\mathcal{L}^2$ , then  $(\forall^s X \Phi)$  and  $(\exists^s X \Phi)$  are concrete  $\mathcal{L}^2$ -formulas.

 $Fv[\Phi]$ ,  $[\Phi]_{\alpha}$ , and  $cp[\Phi]$  for a concrete  $\mathcal{L}^2$ -formula  $\Phi$ , are defined as in first-order languages (Definition 8.1.3), where concrete  $\mathcal{L}^2$ -formulas of the form  $(t\varepsilon T)$  are also considered as atomic concrete formulas, whose complexity is 1. As for first-order language,  $\mathcal{L}^2$ -formulas are defined as equivalence classes of concrete  $\mathcal{L}^2$ -formulas (see Definition 8.1.4), and  $Fv[\varphi]$  and  $cp[\varphi]$  for an  $\mathcal{L}^2$ -formula  $\varphi$  are defined exactly as for  $\mathcal{L}^1$ -formulas (using representatives). Similarly,  $(\varphi_1 \diamond \varphi_2)$  for  $\mathcal{L}^2$ -formulas  $\varphi_1, \varphi_2$  and  $\varphi \in \{\wedge, \vee, \supset\}$ , as well as  $(Q^i x \varphi)$  for  $\mathcal{L}^2$ -formula  $\varphi$  and  $Q^i \in \{\forall^i, \exists^i\}$  are defined as in Definition 8.1.5. In addition, we define the following:

**Definition 9.1.4.** For  $Q^s \in \{\forall^s, \exists^s\}$ , a set variable X of  $\mathcal{L}^2$ , and an  $\mathcal{L}^2$ -formula  $\varphi$ :  $(Q^s X \varphi) = [(Q^s X \Phi)]_{\alpha} \text{ for some } \Phi \in \varphi.$ 

**Proposition 9.1.5.** Exactly one of the following holds for every  $\mathcal{L}^2$ -formula  $\varphi$ :

- $cp[\varphi] = 1$  and exactly one of the following holds:
  - $-\varphi = \{p(t_1, \dots, t_n)\}$  for some *n*-ary predicate symbol p of  $\mathcal{L}^2$ , and first-order  $\mathcal{L}^2$ -terms  $t_1, \dots, t_n$ .
  - $-\varphi = \{(t\varepsilon T)\}$  for some first-order  $\mathcal{L}^2$ -term t, and second-order  $\mathcal{L}^2$ -term T.
  - $-\varphi = \{\bot\}.$
- $\varphi = (\varphi_1 \diamond \varphi_2)$  for some  $\diamond \in \{\land, \lor, \supset\}$ , and unique  $\mathcal{L}^2$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ .
- For every individual variable  $x \notin Fv[\varphi]$ ,  $\varphi = (Q^i x \psi)$  for some  $Q^i \in \{\forall^i, \exists^i\}$ , and unique  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ .
- For every set variable  $X \notin Fv[\varphi]$ ,  $\varphi = (Q^s X \psi)$  for some  $Q^s \in \{\forall^s, \exists^s\}$ , and unique  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ .

Substitution operations are defined as follows:

**Definition 9.1.6.** Let t be a first-order  $\mathcal{L}^2$ -term, and x an individual variable of  $\mathcal{L}^2$ . Given an  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\varphi\{t/x\}$  is inductively defined by:

$$\varphi\{t/x\} = \begin{cases} \{p(t_1\{t/x\}, \dots, t_n\{t/x\})\} & \varphi = \{p(t_1, \dots, t_n)\} \\ \{(t'\{t/x\}\varepsilon T)\} & \varphi = \{(t'\varepsilon T)\} \end{cases} \\ \varphi & \varphi = \{\bot\} \\ (\varphi_1\{t/x\} \diamond \varphi_2\{t/x\}) & \varphi = (\varphi_1 \diamond \varphi_2) \\ (Q^i y \psi\{t/x\}) & \varphi = (Q^i y \psi) \text{ for } y \notin Fv[t] \cup \{x\} \\ (Q^s Y \psi\{t/x\}) & \varphi = (Q^s Y \psi) \end{cases}$$

**Definition 9.1.7.** Let T be a second-order  $\mathcal{L}^2$ -term, and X a set variable of  $\mathcal{L}^2$ . Given an  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\varphi\{^T/X\}$  is inductively defined by:

$$\varphi\{T/X\} = \begin{cases} \varphi & \varphi = \{p(t_1, \dots, t_n)\} \\ \varphi & \varphi = \{(t\varepsilon T')\} \text{ for } T' \neq X \\ \{(t\varepsilon T)\} & \varphi = \{(t\varepsilon X)\} \end{cases}$$

$$\varphi = \{(t \varepsilon X)\}$$

$$\varphi = \{\bot\}$$

$$(\varphi_1\{T/X\} \diamond \varphi_2\{T/X\}) \quad \varphi = (\varphi_1 \diamond \varphi_2)$$

$$(Q^i y \psi\{T/X\}) \quad \varphi = (Q^i y \psi)$$

$$(Q^s Y \psi\{T/X\}) \quad \varphi = (Q^s Y \psi) \text{ for } Y \notin Fv[T] \cup \{X\}$$

Note that the above substitution operations are well-defined. In particular, the choice of the variables y and Y is immaterial.

### 9.2 Henkin-style Second-Order Gödel Logic

In this section we precisely define Henkin-style second-order Gödel logic, via a semantic presentation. These definitions naturally extend the usual definitions of Henkin-style second-order classical logic, by replacing the usual two truth values True and False by any bounded complete linearly ordered set of truth values. From a different angle, these definitions naturally extend (standard) first-order Gödel logic (presented in Chapter 8) by adding an additional collection of fuzzy sets, over which the set quantifiers range.

#### **Definition 9.2.1.** A domain $\mathcal{D}$ for a Gödel set $\mathcal{V}$ consists of:

- A non-empty set, called *individuals domain* and denoted by  $\mathcal{D}_i$ .
- A non-empty collection of fuzzy subsets of  $\mathcal{D}_i$  over  $\mathcal{V}$  (see Definition 8.2.3), called sets domain and denoted by  $\mathcal{D}_s$ .

Given a domain  $\mathcal{D}$ , an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -interpretation I consists of:

- A function assigning an element in  $\mathcal{D}_i$  to every individual constant symbol of  $\mathcal{L}^2$ , and a function in  $\mathcal{D}_i^n \to \mathcal{D}_i$  to every *n*-ary function symbol of  $\mathcal{L}^2$ . We call this function *individuals interpretation*, and denote it by  $I_i$ .
- A function assigning a fuzzy subset in  $\mathcal{D}_s$  to every set constant symbol of  $\mathcal{L}^2$ . We call this function sets interpretation, and denote it by  $I_s$ .

Note that  $\mathcal{D}_i$  is a (first-order) domain, and  $I_i$  is an  $\langle \mathcal{L}^1, \mathcal{D} \rangle$ -interpretation (see Definition 8.2.2). Next, we define  $\mathcal{L}^2$ -structures exactly like we defined  $\mathcal{L}^1$ -structures (Definition 8.2.4), based on the new notions of domain and interpretation.

#### **Definition 9.2.2.** An $\mathcal{L}^2$ -structure is a triple $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$ , where:

- 1.  $\mathcal{V}$  is a Gödel set.
- 2.  $\mathcal{D}$  is a domain for  $\mathcal{V}$ .
- 3. I is an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -interpretation.
- 4. P is a function assigning a fuzzy subset of  $\mathcal{D}_i^n$  over  $\mathcal{V}$  to every n-ary predicate symbol of  $\mathcal{L}^2$ .

Assignments are also defined as their first-order counterparts (Definition 8.2.5):

#### **Definition 9.2.3.** Let $\mathcal{D}$ be a domain.

- 1. An  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment is a function assigning an element of  $\mathcal{D}_i$  to every individual variable of  $\mathcal{L}^2$ , and an element of  $\mathcal{D}_s$  to every set variable of  $\mathcal{L}^2$ .
- 2. Given an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -interpretation I and an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\sigma^I$  is the function assigning elements of  $\mathcal{D}_i$  and  $\mathcal{D}_s$  to  $\mathcal{L}^2$ -terms, recursively defined by:
  - $\sigma^{I}[c] = I_{i}[c]$  for every individual constant symbol c of  $\mathcal{L}^{2}$ .
  - $\sigma^I[x] = \sigma[x]$  for every individual variable x of  $\mathcal{L}^2$ .
  - $\sigma^I[f(t_1, ..., t_n)] = I_i[f](\sigma^I[t_1], ..., \sigma^I[t_n])$  for every *n*-ary function symbol f of  $\mathcal{L}^2$  and n first-order  $\mathcal{L}^2$ -terms  $t_1, ..., t_n$ .
  - $\sigma^{I}[C] = I_s[C]$  for every set constant symbol C of  $\mathcal{L}^2$ .
  - $\sigma^I[X] = \sigma[X]$  for every set variable X of  $\mathcal{L}^2$ .
- 3. Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment. Given an individual variable x of  $\mathcal{L}^2$  and  $d \in \mathcal{D}_i$ , we denote by  $\sigma_{x:=d}$  the  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment that is identical to  $\sigma$  except for  $\sigma_{x:=d}[x] = d$ . Similarly, given a set variable X of  $\mathcal{L}^2$ , and  $D \in \mathcal{D}_s$ , we denote by  $\sigma_{X:=D}$  the  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment that is identical to  $\sigma$  except for  $\sigma_{X:=D}[X] = D$ . These notations are naturally extended to several distinct variables (e.g.  $\sigma_{\nu_1:=d_1,\nu_2:=d_2,\chi_1:=D}$ ).

Next, we generalize Definition 8.2.7 for  $\mathcal{L}^2$ -structures.

**Definition 9.2.4.** Let  $W = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure. For every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $W[\varphi, \sigma]$  is the element of  $\mathcal{V}$  inductively defined as follows:

$$\mathcal{W}[\varphi, \sigma] \text{ is the element of } \mathcal{V} \text{ inductively defined as for } \begin{cases} P[p][\sigma^I[t_1], \dots, \sigma^I[t_n]] & \varphi = \{p(t_1, \dots, t_n)\} \\ \sigma^I[T][\sigma^I[t]] & \varphi = \{(t\varepsilon T)\} \\ 0 & \varphi = \{\bot\} \end{cases} \\ \min\{\mathcal{W}[\varphi_1, \sigma], \mathcal{W}[\varphi_2, \sigma]\} & \varphi = (\varphi_1 \land \varphi_2) \\ \max\{\mathcal{W}[\varphi_1, \sigma], \mathcal{W}[\varphi_2, \sigma]\} & \varphi = (\varphi_1 \lor \varphi_2) \\ \mathcal{W}[\varphi_1, \sigma] \to \mathcal{W}[\varphi_2, \sigma] & \varphi = (\varphi_1 \supset \varphi_2) \\ \inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi, \sigma_{x:=d}] & \varphi = (\forall^i x \psi) \\ \sup_{d \in \mathcal{D}_i} \mathcal{W}[\psi, \sigma_{x:=d}] & \varphi = (\exists^i x \psi) \\ \sup_{D \in \mathcal{D}_s} \mathcal{W}[\psi, \sigma_{x:=D}] & \varphi = (\exists^s x \psi) \end{cases}$$

Again, it can be verified that the choice of x and X in the last definition is immaterial. Note that the last definition establishes the connection between the predicate symbol  $\varepsilon$ , and the (fuzzy) set inclusion. The truth value assigned to a formula of the form  $\{(t\varepsilon T)\}$  with respect to an assignment  $\sigma$  is equal to the membership degree of  $\sigma^I[t]$  in the fuzzy subset  $\sigma^I[T]$ .

The following usual lemma will be needed below (the proof is similar to the proof of Lemma 8.2.8).

**Lemma 9.2.5.** Let  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure.

- 1. Let x be an individual variable of  $\mathcal{L}^2$  and d an element of  $\mathcal{D}_i$ . For every  $\mathcal{L}^2$ -formula  $\varphi$  such that  $x \notin Fv[\varphi]$ , and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ :  $\mathcal{W}[\varphi, \sigma_{x:=d}] = \mathcal{W}[\varphi, \sigma]$ .
- 2. Let X be a set variable of  $\mathcal{L}^2$  and D an element of  $\mathcal{D}_s$ . For every  $\mathcal{L}^2$ -formula  $\varphi$  such that  $X \notin Fv[\varphi]$ , and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ :  $\mathcal{W}[\varphi, \sigma_{X:=D}] = \mathcal{W}[\varphi, \sigma]$ .

Next, we define Henkin-style second-order Gödel logic. This amounts to the set of tautologies induced by the structures defined above with the additional restriction of *comprehension*. Thus, as done in Henkin-style classical second-order logic, we require that all (fuzzy) subsets of the universe that can be captured by some formula, are indeed included in the domain of (fuzzy) subsets. Structures satisfying this property (namely, admit the comprehension axiom) are called *comprehensive*.

**Definition 9.2.6.** Let  $W = \langle V, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure. Given an  $\mathcal{L}^2$ -formula  $\varphi$ , an individual variable x of  $\mathcal{L}^2$ , and an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ , we denote by  $W[\varphi, \sigma, x]$  the

fuzzy subset of  $\mathcal{D}_i$  over  $\mathcal{V}$  defined by  $\lambda d \in \mathcal{D}_i.\mathcal{W}[\varphi, \sigma_{x:=d}]$ .  $\mathcal{W}$  is called *comprehensive* if  $\mathcal{W}[\varphi, \sigma, x] \in \mathcal{D}_s$  for every  $\varphi$ , x, and  $\sigma$ .

**Definition 9.2.7.** For an  $\mathcal{L}^2$ -formula  $\varphi$ , we write  $\Vdash^{\mathbf{G}\ddot{\mathbf{o}}_{\mathcal{L}^2}} \varphi$  if  $\mathcal{W}[\varphi, \sigma] = 1$  for every comprehensive  $\mathcal{L}^2$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ .  $\mathbf{G}\ddot{\mathbf{o}}_{\mathcal{L}^2}$  is the logic consisting of all formulas  $\varphi$  such that  $\Vdash^{\mathbf{G}\ddot{\mathbf{o}}_{\mathcal{L}^2}} \varphi$ .

**Example 9.2.8.** It is easy to see that the comprehension axiom scheme is valid in  $G\ddot{o}_{\mathcal{L}^2}$ , i.e.

$$\Vdash^{\mathbf{G}\ddot{\mathbf{o}}_{\mathcal{L}^{2}}}(\exists^{s}X(\forall^{i}x((\varphi\supset\{(x\varepsilon X)\})\wedge(\{(x\varepsilon X)\}\supset\varphi))))$$

for every  $\mathcal{L}^2$ -formula  $\varphi$ , set variable  $X \notin Fv[\varphi]$ , and individual variable x. Indeed, let  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be a comprehensive  $\mathcal{L}^2$ -structure, and  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment. By definition,  $\mathcal{W}[\varphi, \sigma, x] \in \mathcal{D}_s$ . Thus

$$\mathcal{W}[(\exists^{s}X(\forall^{i}x((\varphi\supset\{(x\varepsilon X)\})\land(\{(x\varepsilon X)\}\supset\varphi)))),\sigma]\geq \\ \mathcal{W}[(\forall^{i}x((\varphi\supset\{(x\varepsilon X)\})\land(\{(x\varepsilon X)\}\supset\varphi))),\sigma_{X:=\mathcal{W}[\varphi,\sigma,x]}].$$

By definition, for every  $d \in \mathcal{D}_i$  we have:

$$\mathcal{W}[\{(x\varepsilon X)\}), \sigma_{X:=\mathcal{W}[\varphi,\sigma,x],x:=d}] = \mathcal{W}[\varphi,\sigma,x][d] = \mathcal{W}[\varphi,\sigma_{x:=d}].$$

Since  $X \notin Fv[\varphi]$ ,

$$\mathcal{W}[(\varphi \supset \{(x \in X)\}), \sigma_{X:=\mathcal{W}[\varphi,\sigma,x],x:=d}] = \mathcal{W}[\varphi,\sigma_{x:=d}] \to \mathcal{W}[\varphi,\sigma_{x:=d}] = 1,$$

and similarly,

$$\mathcal{W}[(\{(x \in X)\} \supset \varphi), \sigma_{X:=\mathcal{W}[\varphi,\sigma,x],x:=d}] = 1.$$

It follows that

$$\mathcal{W}[((\varphi \supset \{(x \in X)\}) \land (\{(x \in X)\} \supset \varphi)), \sigma_{X:=\mathcal{W}[\varphi,\sigma,x],x:=d}] = \min\{1,1\} = 1.$$

Since this holds for every  $d \in \mathcal{D}_i$ , we have:

$$\inf_{d \in \mathcal{D}_i} \mathcal{W}[((\varphi \supset \{(x \in X)\}) \land (\{(x \in X)\} \supset \varphi)), \sigma_{X := \mathcal{W}[\varphi, \sigma, x], x := d}] = 1.$$

Consequently,

$$\mathcal{W}[(\exists^s X(\forall^i x((\varphi \supset \{(x \in X)\}) \land (\{(x \in X)\} \supset \varphi)))), \sigma] = 1.$$

### 9.3 The Hypersequent Calculus HIF<sup>2</sup>

In this section we present a hypersequent calculus  $\mathbf{HIF}^2$  for  $\mathbf{G\ddot{o}_{\mathcal{L}^2}}$ .  $\mathbf{HIF}^2$  is obtained by augmenting the hypersequent calculus  $\mathbf{HIF}$  for standard first-order Gödel logic (presented in Section 8.3) with rules for second-order quantifiers. These are the hypersequent versions of the sequent rules used for classical logic (see the calculus  $\mathbf{L}^2\mathbf{K}$  in [58]). They have the same structure of the rules for individual quantifiers, where instead of using first-order terms in  $(\mathbf{f}:\forall)$  and  $(\mathbf{t}:\exists)$ , one uses abstraction terms (abstracts for short). Abstracts

are syntactic objects of the form  $\{x \mid \varphi\}$  that intuitively represent sets of individuals. Note that abstracts are just a syntactic tool for formulating the rules of the set quantifiers. Derivations in the calculus still consists solely of hypersequents, and no abstracts are mentioned in them. As we did for formulas, we first define *concrete abstracts*, and abstracts are defined as alpha-equivalence classes of concrete ones.

**Definition 9.3.1.** A concrete  $\mathcal{L}^2$ -abstract is an expression of the form  $\{x \mid \Phi\}$ , where x is an individual variable of  $\mathcal{L}^2$ , and  $\Phi$  is a concrete  $\mathcal{L}^2$ -formula. Alpha-equivalence between concrete  $\mathcal{L}^2$ -abstracts is defined as usual (where x is considered bound in  $\{x \mid \Phi\}$ ), and  $[\{x \mid \Phi\}]_{\alpha}$  is standing for the set of all concrete  $\mathcal{L}^2$ -abstracts which are alpha-equivalent to  $\{x \mid \Phi\}$ . An  $\mathcal{L}^2$ -abstract is an equivalence class of concrete  $\mathcal{L}^2$ -abstracts under alpha-equivalence. We mainly use  $\tau$  as a metavariable for  $\mathcal{L}^2$ -abstracts. The set of free variables of an  $\mathcal{L}^2$ -abstract is defined using representatives, i.e. for an  $\mathcal{L}^2$ -abstract  $\tau$ ,  $Fv[\tau] = Fv[\{x \mid \Phi\}]$  for some  $\{x \mid \Phi\} \in \tau$ .

**Definition 9.3.2.** Given an individual variable x of  $\mathcal{L}^2$  and an  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\langle x \mid \varphi \rangle$  is the  $\mathcal{L}^2$ -abstract  $[\langle x \mid \Phi \rangle]_{\alpha}$  for some  $\Phi \in \varphi$ .

**Proposition 9.3.3.** For every  $\mathcal{L}^2$ -abstract  $\tau$  and individual variable  $x \notin Fv[\tau]$ , there exists a unique  $\mathcal{L}^2$ -formula  $\varphi$ , such that  $\tau = \langle x \mid \varphi \rangle$ .

**Definition 9.3.4.** Let  $\tau$  be an  $\mathcal{L}^2$ -abstract, and t a first-order  $\mathcal{L}^2$ -term.  $\tau[t]$  is defined to be the  $\mathcal{L}^2$ -formula  $\varphi\{t/x\}$  for some individual variable x and  $\mathcal{L}^2$ -formula  $\varphi$ , such that  $\tau = \langle x \mid \varphi \rangle$ .

It is easy to see that  $\tau[t]$  is well-defined, as it does not depend on the choice of x.

**Definition 9.3.5.** Let  $\tau$  be an  $\mathcal{L}^2$ -abstract and X a set variable of  $\mathcal{L}^2$ . Given an  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\varphi\{\tau/X\}$  is inductively defined by:

$$\varphi\{\tau/X\} = \begin{cases} \varphi & \varphi = \{p(t_1, \dots, t_n)\}, \varphi = \{(t\varepsilon T)\} \text{ for } T \neq X, \varphi = \{\bot\} \\ \tau[t] & \varphi = \{(t\varepsilon X)\} \\ (\varphi_1\{\tau/X\} \diamond \varphi_2\{\tau/X\}) & \varphi = (\varphi_1 \diamond \varphi_2) \\ (Q^i y \psi\{\tau/X\}) & \varphi = (Q^i y \psi) \text{ for } y \notin Fv[\tau] \\ (Q^s Y \psi\{\tau/X\}) & \varphi = (Q^s Y \psi) \text{ for } Y \notin Fv[\tau] \cup \{X\} \end{cases}$$

Note that the this substitution operation is well-defined. In particular, the choice of the variables y and Y is immaterial.

**Example 9.3.6.** For  $\varphi = (\forall^i \nu_1(\{(\nu_1 \varepsilon \chi_1)\} \supset (\exists^s \chi_2 \{(\nu_1 \varepsilon \chi_2)\})))$ , and  $\tau = \{\nu_2 \mid \{p(\nu_2, \nu_2)\}\}$ , we have  $\varphi \{\tau/\chi_1\} = (\forall^i \nu_1(\{p(\nu_1, \nu_1)\} \supset (\exists^s \chi_2 \{(\nu_1 \varepsilon \chi_2)\})))$ .

The following lemmas will be useful in the sequel.

Notation 9.3.7. For a second-order  $\mathcal{L}^2$ -term T, the  $\mathcal{L}^2$ -abstract  $\{\nu_1 \mid \{(\nu_1 \varepsilon T)\}\}$  is denoted by  $T_{abs}$ .

**Lemma 9.3.8.** Let T be a second-order  $\mathcal{L}^2$ -term. For every  $\mathcal{L}^2$ -formula  $\varphi$  and set variable X of  $\mathcal{L}^2$ ,  $\varphi\{T_{abs}/X\} = \varphi\{T/X\}$ .

*Proof.* By usual induction on the complexity of  $\varphi$ .

**Lemma 9.3.9.** Let  $\tau$  be an  $\mathcal{L}^2$ -abstract, t, t' first-order  $\mathcal{L}^2$ -terms, and x an individual variable such that  $x \notin Fv[\tau]$ . Then,  $\tau[t']\{t/x\} = \tau[t'\{t/x\}]$ .

*Proof.* It is straightforward to prove that  $\varphi\{t'/y\}\{t/x\} = \varphi\{t'\{t/x\}/y\}$  for every  $\mathcal{L}^2$ -formula  $\varphi$ , first-order  $\mathcal{L}^2$ -terms t and t', and individual variables x and y, such that  $x \notin Fv[\varphi]$ . The claim then easily follows from our definitions.

Using abstracts, the rule schemes for the second-order quantifiers in  ${\bf HIF^2}$  are given by:

$$(\mathbf{f}:\forall^s) \quad \frac{H \mid \Gamma, \varphi\{\tau/X\} \Rightarrow E}{H \mid \Gamma, (\forall^s X \varphi) \Rightarrow E} \qquad (\mathbf{t}:\forall^s) \quad \frac{H \mid \Gamma \Rightarrow \varphi}{H \mid \Gamma \Rightarrow (\forall^s X \varphi)}$$

$$(\mathbf{f}:\exists^s) \quad \frac{H \mid \Gamma, \varphi \Rightarrow E}{H \mid \Gamma, (\exists^s X \varphi) \Rightarrow E} \qquad (\mathbf{t}:\exists^s) \quad \frac{H \mid \Gamma \Rightarrow \varphi\{\tau/X\}}{H \mid \Gamma \Rightarrow (\exists^s X \varphi)}$$

where X must not be a free variable in the lower hypersequent in applications of the rules  $(t:\forall^s)$  and  $(f:\exists^s)$ .

Below, we write  $\vdash_{\mathbf{HIF}^2} H$  to denote that an  $\mathcal{L}^2$ -hypersequent H is provable in  $\mathbf{HIF}^2$ , and  $\vdash_{\mathbf{HIF}^2}^{cf} H$  to denote that H is provable in  $\mathbf{HIF}^2$  without applying (cut).

Since formulas are equivalence classes, the rules  $(t:\forall^s)$ , and  $(f:\exists^s)$  could be written as well as:

$$(\mathbf{t}{:}\forall^s) \quad \frac{H \mid \Gamma \Rightarrow \varphi\{^Y/X\}}{H \mid \Gamma \Rightarrow (\forall^s X \varphi)} \qquad (\mathbf{f}{:}\exists^s) \quad \frac{H \mid \Gamma, \varphi\{^Y/X\} \Rightarrow E}{H \mid \Gamma, (\exists^s X \varphi) \Rightarrow E}$$

where Y must not be a free variable in the lower hypersequent.

**Remark 9.3.10.** Note that rules given by the schemes

$$\frac{H \mid \Gamma, \varphi\{T/X\} \Rightarrow E}{H \mid \Gamma, (\forall^s X \varphi) \Rightarrow E} \qquad \frac{H \mid \Gamma \Rightarrow \varphi\{T/X\}}{H \mid \Gamma \Rightarrow (\exists^s X \varphi)},$$

where T is a second-order  $\mathcal{L}^2$ -term, are particular instances of  $(\mathbf{f}:\forall^s)$  and  $(\mathbf{t}:\exists^s)$ , obtained by choosing  $\tau = T_{abs}$  (see Lemma 9.3.8).

### 9.4 Soundness

In this section we prove the soundness of  $HIF^2$  for  $G\ddot{o}_{\mathcal{L}^2}$ . Definition 8.4.1 (defining when a  $\mathcal{L}^1$ -structure is a model of an  $\mathcal{L}^1$ -hypersequent etc.) is adopted for  $\mathcal{L}^2$ -structures as is.

**Theorem 9.4.1.** Let H be an  $\mathcal{L}^2$ -hypersequent. If  $\vdash_{\mathbf{HIF}^2} H$ , then every comprehensive  $\mathcal{L}^2$ -structure is a model of H.

Soundness for  $G\ddot{o}_{\ell^2}$  is an obvious corollary (see Corollary 8.4.3):

Corollary 9.4.2. For every  $\mathcal{L}^2$ -formula  $\varphi$ , if  $\vdash_{\mathbf{HIF}^2} \{\{\mathbf{t}:\varphi\}\}$ , then  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^2}} \varphi$ .

Theorem 9.4.1 is proved in the usual way, by induction on the length of the derivation in **HIF**<sup>2</sup>. We use the following technical lemmas:

**Lemma 9.4.3.** Let  $W = \langle V, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure, t a first-order  $\mathcal{L}^2$ -term, and x an individual variable of  $\mathcal{L}^2$ . For every  $\mathcal{L}^2$ -formula  $\varphi$ , and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ :

$$\mathcal{W}[\varphi, \sigma_{x:=\sigma^I[t]}] = \mathcal{W}[\varphi\{t/x\}, \sigma].$$

Proof. The claim is proved by induction on the complexity of  $\varphi$ , similarly to the proof of Lemma 8.2.9 for the first-order case. We do here the case  $\varphi = \{(t'\varepsilon T)\}$ . Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment. Then  $\varphi\{t/x\} = \{(t'\{t/x\}\varepsilon T)\}$ . Thus  $\mathcal{W}[\varphi\{t/x\}, \sigma] = \sigma^I[T][\sigma^I[t'\{t/x\}]]$ . By Lemma 8.2.6 (adapted to second-order languages),  $\sigma^I[t'\{t/x\}] = \sigma_{x:=\sigma^I[t]}[t']$ . Now  $\sigma^I[T] = \sigma^I_{x:=\sigma[t]}[T]$ , and so  $\sigma^I[T][\sigma^I[t'\{t/x\}]] = \sigma^I_{x:=\sigma[t]}[T][\sigma^I_{x:=\sigma[t]}[t']]$ . By definition, this is equal to  $\mathcal{W}[\varphi, \sigma_{x:=\sigma[t]}]$ .

**Lemma 9.4.4.** Let  $W = \langle \mathcal{V}, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure,  $\tau$  an  $\mathcal{L}^2$ -abstract,  $x \notin Fv[\tau]$  an individual variable, and X a set variable. For every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ : if  $W[\tau[x], \sigma, x] \in \mathcal{D}_s$  then  $W[\varphi\{\tau/x\}, \sigma] = W[\varphi, \sigma_{X:=W[\tau[x], \sigma, x]}]$ .

Proof. If  $X \notin Fv[\varphi]$ , then  $\varphi\{\tau/x\} = \varphi$  and in this case the claim follows by Lemma 9.2.5. Suppose otherwise. We prove the claim by induction on the complexity of  $\varphi$ . Suppose that  $cp[\varphi] = 1$ . Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment, and let  $D_0 = \mathcal{W}[\tau[x], \sigma, x]$ . Suppose that  $D_0 \in \mathcal{D}_s$ . Since  $X \in Fv[\varphi]$ ,  $\varphi = \{(t\varepsilon X)\}$  for some first-order  $\mathcal{L}^2$ -term t. In this case,  $\varphi\{\tau/x\} = \tau[t]$ . By Lemma 9.3.9,  $\tau[t] = \tau[x\{t/x\}] = \tau[x]\{t/x\}$ . Thus  $\mathcal{W}[\varphi\{\tau/x\}, \sigma] = \mathcal{W}[\tau[x]\{t/x\}, \sigma]$ . By Lemma 9.4.3,  $\mathcal{W}[\tau[x]\{t/x\}, \sigma] = \mathcal{W}[\tau[x], \sigma_{x:=\sigma^I[t]}]$ . By definition,  $\mathcal{W}[\tau[x], \sigma_{x:=\sigma^I[t]}] = D_0[\sigma^I[t]] = \mathcal{W}[\varphi, \sigma_{X:=D_0}]$ .

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^2$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment, and again let  $D_0 = \mathcal{W}[\tau[x], \sigma, x]$ . Suppose that  $D_0 \in \mathcal{D}_s$ . Exactly one of the following holds:

9.4. Soundness 167

•  $\varphi = (\varphi_1 \diamond \varphi_2)$  for some  $\diamond \in \{\land, \lor, \supset\}$ , and  $\mathcal{L}^2$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . By definition,  $\varphi\{\tau/x\} = (\varphi_1\{\tau/x\} \diamond \varphi_2\{\tau/x\})$ . We continue with  $\diamond = \supset$  (the proof is similar for  $\land$  and  $\lor$ ). Thus,

$$\mathcal{W}[\varphi\{\tau/X\},\sigma] = \mathcal{W}[\varphi_1\{\tau/X\},\sigma] \to [\varphi_2\{\tau/X\},\sigma].$$

By the induction hypothesis (and the case in which  $X \notin Fv[\varphi]$ ), we have both  $\mathcal{W}[\varphi_1\{\tau/X\}, \sigma] = \mathcal{W}[\varphi_1, \sigma_{X:=D_0}]$  and  $\mathcal{W}[\varphi_2\{\tau/X\}, \sigma] = \mathcal{W}[\varphi_2, \sigma_{X:=D_0}]$ . By definition,  $\mathcal{W}[\varphi_1, \sigma_{X:=D_0}] \to \mathcal{W}[\varphi_2, \sigma_{X:=D_0}] = \mathcal{W}[\varphi, \sigma_{X:=D_0}]$ .

•  $\varphi = (Q^i y \psi)$  for some  $Q^i \in \{\forall^i, \exists^i\}$ , individual variable  $y \notin \{x\} \cup Fv[\tau]$  of  $\mathcal{L}^2$ , and  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . By definition,  $\varphi\{\tau/x\} = (Q^i y \psi\{\tau/x\})$ . We continue with  $Q^i = \forall^i$  (the proof is similar for  $\exists^i$ ). Thus,

$$\mathcal{W}[\varphi\{\tau/X\},\sigma] = \inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi\{\tau/X\},\sigma_{y:=d}].$$

Now, using Lemma 9.2.5, we have  $D_0 = \mathcal{W}[\tau[x], \sigma_{y:=d}, x]$  for every  $d \in \mathcal{D}_i$  (since  $y \notin Fv[\tau[x]]$ ). Therefore,  $\inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi\{\tau/X\}, \sigma_{y:=d}] = \inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi, \sigma_{y:=d,X:=D_0}]$  by the induction hypothesis. By definition,  $\inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi, \sigma_{y:=d,X:=D_0}] = \mathcal{W}[\varphi, \sigma_{X:=D_0}]$ .

•  $\varphi = (Q^s Y \psi)$  for some  $Q^s \in \{\forall^s, \exists^s\}$ , set variable  $Y \notin \{X\} \cup Fv[\tau]$  of  $\mathcal{L}^2$ , and  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . By definition,  $\varphi\{\tau/X\} = (Q^s Y \psi\{\tau/X\})$ . We continue with  $Q^s = \forall^s$  (the proof is similar for  $\exists^s$ ). Thus,

$$\mathcal{W}[\varphi\{\tau/X\},\sigma] = \inf_{D \in \mathcal{D}_s} \mathcal{W}[\psi\{\tau/X\},\sigma_{Y:=D}].$$

Now, using Lemma 9.2.5, we have  $D_0 = \mathcal{W}[\tau[x], \sigma_{Y:=D}, x]$  for every  $D \in \mathcal{D}_s$  (since  $Y \notin Fv[\tau[x]]$ ). Therefore,  $\inf_{D \in \mathcal{D}_s} \mathcal{W}[\psi\{\tau/x\}, \sigma_{Y:=D}] = \inf_{D \in \mathcal{D}_s} \mathcal{W}[\psi, \sigma_{Y:=D,X:=D_0}]$  by the induction hypothesis (note that  $Y \neq X$ ). By definition,

$$\inf_{D \in \mathcal{D}_s} \mathcal{W}[\psi, \sigma_{Y:=D,X:=D_0}] = \mathcal{W}[\varphi, \sigma_{X:=D_0}].$$

Proof of Theorem 9.4.1. Let  $W = \langle V, \mathcal{D}, I, P \rangle$  be an  $\mathcal{L}^2$ -structure, where  $V = \langle V, \leq \rangle$ . It suffices to prove soundness of each possible application of a rule of  $\mathbf{HIF}^2$ . For the rules of  $\mathbf{HIF}$ , this is done as in the proof of Theorem 8.4.2. We prove here the soundness of  $(\mathbf{f}: \forall^s)$ , and leave the other three new rules to the reader:

Suppose that  $H = H' \mid s \cup \{f:(\forall^s X \varphi)\}$  is derived from  $H' \mid s \cup \{f:\{\tau/x\}\}\}$  using  $(f:\forall^s)$ . Assume that  $\mathcal{W}, \sigma \not\models H$  for some  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ . Hence,  $\mathcal{W}, \sigma \not\models s'$  for every  $s' \in H'$ , and  $\mathcal{W}, \sigma \not\models s \cup \{f:(\forall^s X \varphi)\}$ . Let  $u = \mathcal{W}^{\mathsf{t}}[s, \sigma]$ . The assumption that  $\mathcal{W}, \sigma \not\models s \cup \{f:(\forall^s X \varphi)\}$  entails that  $\mathcal{W}^{\mathsf{f}}[s, \sigma] > u$ , and  $\mathcal{W}[(\forall^s X \varphi), \sigma] > u$ . By definition,  $\mathcal{W}[(\forall^s X \varphi), \sigma] = \inf_{D \in \mathcal{D}_s} \mathcal{W}[\varphi, \sigma_{X:=D}]$ . Thus  $\mathcal{W}[\varphi, \sigma_{X:=D}] > u$  for every  $D \in \mathcal{D}_s$ . Let x be an individual variable such that  $x \notin Fv[\tau]$ , and let  $D_0 = \mathcal{W}[\tau[x], \sigma, x]$ . Since  $\mathcal{W}$  is comprehensive,  $D_0 \in \mathcal{D}_s$ , and in particular,  $\mathcal{W}[\varphi, \sigma_{X:=D_0}] > u$ . Lemma 9.4.4 implies that  $\mathcal{W}[\varphi\{\tau/x\}, \sigma] > u$ . It follows that  $\mathcal{W}, \sigma \not\models s \cup \{f: \varphi\{\tau/x\}\}$ . Consequently,  $\mathcal{W}$  is not a model of  $H' \mid s \cup \{f: \varphi\{\tau/x\}\}$ .

### 9.5 Complete Non-deterministic Semantics

In this section we present a non-deterministic semantics for which the cut-free fragment of HIF<sup>2</sup> is complete. As for HIF in the previous chapter, this semantics will be used in the next section, where we show that ordinary counter-models can be extracted out of non-deterministic ones.

#### **Definition 9.5.1.** A quasi-domain $\mathcal{D}$ consists of:

- A non-empty set, called *individuals domain* and denoted by  $\mathcal{D}_i$ .
- A non-empty set, called sets domain and denoted by  $\mathcal{D}_s$ .

Given a quasi-domain  $\mathcal{D}$ , an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -interpretation I consists of:

- A function assigning an element in  $\mathcal{D}_i$  to every individual constant symbol of  $\mathcal{L}^2$ , and a function in  $\mathcal{D}_i^n \to \mathcal{D}_i$  to every *n*-ary function symbol of  $\mathcal{L}^2$ . We call this function *individuals interpretation*, and denote it by  $I_i$ .
- A function assigning an element of  $\mathcal{D}_s$  to every set constant symbol of  $\mathcal{L}^2$ . We call this function sets interpretation, and denote it by  $I_s$ .

 $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignments are defined for quasi-domains exactly as for domains (see Definition 9.2.3). Note that the elements of  $\mathcal{D}_s$  in quasi-domains may not be fuzzy subsets. This allows us to compose  $\mathcal{D}_s$  out of abstracts (as done in the completeness proof). Instead, as defined below, the interpretation function P of a quasi- $\mathcal{L}^2$ -structure assigns a (quasi) fuzzy subset of  $\mathcal{D}_i$  over  $\mathcal{V}$  to every element of  $\mathcal{D}_s$ .

### **Definition 9.5.2.** A quasi- $\mathcal{L}^2$ -structure is a tuple $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$ , where:

- 1.  $\mathcal{V}$  is a Gödel set.
- 2.  $\mathcal{D}$  is a quasi-domain.
- 3. I is an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -interpretation.
- 4. P is a function assigning a quasi fuzzy subset of  $\mathcal{D}_i^n$  over  $\mathcal{V}$  to every n-ary predicate symbol of  $\mathcal{L}^2$ , and a quasi fuzzy subset of  $\mathcal{D}_i$  over  $\mathcal{V}$  to every element of  $\mathcal{D}_s$  (see Definition 8.5.1).
- 5. v is a function assigning a pair in  $\{\langle u^{\mathbf{f}}, u^{\mathbf{t}} \rangle \in \mathcal{V} \times \mathcal{V} \mid u^{\mathbf{f}} \leq u^{\mathbf{t}} \}$  to every ordered pair of the form  $\langle \varphi, \sigma \rangle$ , where  $\varphi$  is an  $\mathcal{L}^2$ -formula and  $\sigma$  is an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment, such that the following hold:
  - (a) For every two individual variables x, y such that  $y \notin Fv[\varphi]$ , and every  $d \in \mathcal{D}_i$ :  $v[\varphi, \sigma_{x:=d}] = v[\varphi\{y/x\}, \sigma_{y:=d}].$
  - (b) For every two set variables X, Y such that  $Y \notin Fv[\varphi]$ , and every  $D \in \mathcal{D}_s$ :  $v[\varphi, \sigma_{X:=D}] = v[\varphi\{^Y/X\}, \sigma_{Y:=D}].$

Next, we define which quasi- $\mathcal{L}^2$ -structures are considered *legal*.

**Definition 9.5.3.** Let  $Q = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  be a quasi- $\mathcal{L}^2$ -structure. For every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma]$  is the pair defined as follows:

$$\varphi \text{ and } \langle \mathcal{L}^2, \mathcal{D} \rangle \text{-assignment } \sigma, \underline{v}_{\mathcal{Q}}[\varphi, \sigma] \text{ is the pair defined as follows:}$$

$$\begin{cases} P[p][\sigma^I[t_1], \dots, \sigma^I[t_n]] & \varphi = \{p(t_1, \dots, t_n)\} \\ P[\sigma^I[T]][\sigma^I[t]] & \varphi = \{(t\varepsilon T)\} \\ \langle 0, 0 \rangle & \varphi = \{\bot\} \end{cases} \\ \langle \min\{v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{f}}[\varphi_2, \sigma]\}, \min\{v^{\mathbf{t}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma]\} \rangle & \varphi = (\varphi_1 \wedge \varphi_2) \\ \langle \max\{v^{\mathbf{f}}[\varphi_1, \sigma], v^{\mathbf{f}}[\varphi_2, \sigma]\}, \max\{v^{\mathbf{t}}[\varphi_1, \sigma], v^{\mathbf{t}}[\varphi_2, \sigma]\} \rangle & \varphi = (\varphi_1 \vee \varphi_2) \\ \langle v^{\mathbf{t}}[\varphi_1, \sigma] \to v^{\mathbf{f}}[\varphi_2, \sigma], v^{\mathbf{f}}[\varphi_1, \sigma] \to v^{\mathbf{t}}[\varphi_2, \sigma] \rangle & \varphi = (\varphi_1 \vee \varphi_2) \\ \langle \inf_{d \in \mathcal{D}_i} v^{\mathbf{f}}[\psi, \sigma_{x:=d}], \inf_{d \in \mathcal{D}_i} v^{\mathbf{t}}[\psi, \sigma_{x:=d}] \rangle & \varphi = (\forall^i x \psi) \\ \langle \sup_{d \in \mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{x:=d}], \sup_{d \in \mathcal{D}_s} v^{\mathbf{t}}[\psi, \sigma_{x:=d}] \rangle & \varphi = (\exists^i x \psi) \\ \langle \sup_{D \in \mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{x:=D}], \sup_{D \in \mathcal{D}_s} v^{\mathbf{t}}[\psi, \sigma_{x:=D}] \rangle & \varphi = (\exists^s X \psi) \end{cases}$$

Conditions (a) and (b) in Definition 9.5.2 ensure that  $\mathcal{Q}$  is well-defined, namely that the choice of x and X is immaterial. It is straightforward to verify that  $\underline{v}_{\mathcal{Q}}^{\mathbf{f}}[\varphi, \sigma] \leq \underline{v}_{\mathcal{Q}}^{\mathbf{t}}[\varphi, \sigma]$  for every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ .

**Definition 9.5.4.** A quasi- $\mathcal{L}^2$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  is called *legal* if we have  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] \subseteq v[\varphi, \sigma]$  for every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ .

We adapt the definition of comprehensive structures to quasi-structures, keeping in mind that the function P interprets the elements of  $\mathcal{D}_s$  as quasi fuzzy sets.

**Definition 9.5.5.** A quasi- $\mathcal{L}^2$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  is called *comprehensive* if for every  $\mathcal{L}^2$ -formula  $\varphi$ , individual variable x, and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ , there exist some  $D \in \mathcal{D}_s$  such that  $P[D] = \lambda d \in \mathcal{D}_i.v[\varphi, \sigma_{x:=d}]$ .

The notion of model for quasi- $\mathcal{L}^2$ -structures is defined exactly as in the first-order case (Definition 8.5.8). In turn, the main completeness theorem is given by:

**Theorem 9.5.6.** Suppose that  $\not\vdash_{\mathbf{HIF}^2}^{cf} H_0$ , for some  $\mathcal{L}^2$ -hypersequent  $H_0$ . Then there exists a legal comprehensive quasi- $\mathcal{L}^2$ -structure which is not a model of  $H_0$ .

The rest of this section is devoted to prove this theorem.

### The Herbrand Quasi-Domain

A main ingredient in the completeness proof below is the *Herbrand quasi-domain*. It is defined as follows:

**Definition 9.5.7.** The Herbrand quasi-domain for  $\mathcal{L}^2$ , denoted by  $\mathcal{D}^{\mathcal{L}^2}$ , is the quasi-domain whose individuals domain  $\mathcal{D}_i^{\mathcal{L}^2}$  consists of all first-order  $\mathcal{L}^2$ -terms, and sets domain  $\mathcal{D}_s^{\mathcal{L}^2}$  consists of all  $\mathcal{L}^2$ -abstracts. The Herbrand interpretation for  $\mathcal{L}^2$ , denoted by  $I_{\mathcal{L}^2}$ , is the  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -interpretation defined by:  $I_i[c] = c$  for every individual constant symbol c of  $\mathcal{L}^2$ ,  $I_i[f] = \lambda t_1, \ldots, t_n \in \mathcal{D}_i.f(t_1, \ldots, t_n)$  for every n-ary function symbol f of  $\mathcal{L}^2$ , and  $I_s[C] = C_{abs} = \{ \nu_1 \mid \{(\nu_1 \varepsilon C)\} \}$  for every set constant symbol C of  $\mathcal{L}^2$ .

Below, given an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$  and a first-order (second-order)  $\mathcal{L}^2$ -term t (T), we write  $\sigma[t]$   $(\sigma[T])$  instead of  $\sigma^{I_{\mathcal{L}^2}}[t]$   $(\sigma^{I_{\mathcal{L}^2}}[T])$ . In addition,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignments are extended to apply on  $\mathcal{L}$ -formulas. Roughly speaking, every occurrence of a free variable x or X in a formula  $\varphi$  is replaced in  $\sigma[\varphi]$  by  $\sigma[x]$  or  $\sigma[X]$ . Formally, this is defined as follows.

**Definition 9.5.8.** Let  $\varphi$  be an  $\mathcal{L}^2$ -formula, and  $\sigma$  an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment. The set of free variables of the pair  $\langle \varphi, \sigma \rangle$  (denoted by  $Fv[\langle \varphi, \sigma \rangle]$ ) consists of the variables of  $\sigma[x]$  for every individual variable  $x \in Fv[\varphi]$ , and the free variables of  $\sigma[X]$  for every set variable  $X \in Fv[\varphi]$ .

**Definition 9.5.9.**  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignments are extended to  $\mathcal{L}^2$ -formulas, according to the following inductive definition:

aductive definition: 
$$\sigma[\varphi] = \begin{cases} \{p(\sigma[t_1], \dots, \sigma[t_n])\} & \varphi = \{p(t_1, \dots, t_n)\} \\ \sigma[T][\sigma[t]] & \varphi = \{(t\varepsilon T)\} \\ \{\bot\} & \varphi = \{\bot\} \\ (\sigma[\varphi_1] \diamond \sigma[\varphi_2]) & \varphi = (\varphi_1 \diamond \varphi_2) \\ (Q^i x \sigma_{x:=x}[\psi]) & \varphi = (Q^i x \psi) \text{ for } x \not\in Fv[\langle \varphi, \sigma \rangle] \\ (Q^s X \sigma_{X:=X_{abs}}[\psi]) & \varphi = (Q^s X \psi) \text{ for } X \not\in Fv[\langle \varphi, \sigma \rangle] \end{cases}$$

Note that the choice of x and X in the last definition is immaterial, and thus  $\sigma[\varphi]$  is well-defined. The following properties of the Herbrand quasi-domain are needed in the completeness proof.

**Lemma 9.5.10.** Let t be a first-order  $\mathcal{L}^2$ -term and  $\tau$  an  $\mathcal{L}^2$ -abstract.

1. For every  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and individual variables x, y such that  $y \notin Fv[\varphi]$ ,  $\sigma_{x:=t}[\varphi] = \sigma_{y:=t}[\varphi\{y/x\}]$ .

2. For every  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and set variables X, Y such that  $Y \notin Fv[\varphi]$ ,  $\sigma_{X:=\tau}[\varphi] = \sigma_{Y:=\tau}[\varphi\{^Y/X\}]$ .

*Proof.* First, as in Lemma 8.5.16, it is straightforward to show that for every first-order  $\mathcal{L}^2$ -terms t' and t,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and individual variables x, y such that  $y \notin Fv[t']$ ,  $\sigma_{x:=t}[t'] = \sigma_{y:=t}[t'\{y/x\}]$ .

Next, we prove the first claim in the lemma by induction on the complexity of  $\varphi$ . We include here only the cases that are special for second-order language (the other cases are handled as in the proof of Lemma 8.5.16). Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and x, y individual variables such that  $y \notin Fv[\varphi]$ . Consider the following cases:

- $\varphi = \{(t'\varepsilon T)\}$  for some first-order  $\mathcal{L}^2$ -term t', and second-order  $\mathcal{L}^2$ -term T. Then,  $\sigma_{x:=t}[\varphi] = \sigma_{x:=t}[T][\sigma_{x:=t}[t']]$ . Since  $y \notin Fv[t']$ , this equals  $\sigma_{x:=t}[T][\sigma_{y:=t}[t'\{y/x\}]]$ . Since x and y does not occur in T, this is equal to  $\sigma_{y:=t}[T][\sigma_{y:=t}[t'\{y/x\}]]$ . By definition, this formula is equal to  $\sigma_{y:=t}[\{(t'\{y/x\}\varepsilon T)\}]$ , which is  $\sigma_{y:=t}[\{(t'\varepsilon T)\}\{y/x\}]$ .
- $\varphi = (Q^s X \psi)$  for  $Q^s \in \{\forall^s, \exists^s\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and set variable  $X \notin Fv[\sigma[\varphi]]$ . Then,  $\sigma_{x:=t}[\varphi] = (Q^s X \sigma_{x:=t,X:=X_{abs}}[\psi])$ . By the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^s X \sigma_{y:=t,X:=X_{abs}}[\psi\{y/x\}])$ . And this is (by definition) equal to  $\sigma_{y:=t}[(Q^s X \psi\{y/x\})]$ , which in turn equals  $\sigma_{y:=t}[(\varphi\{y/x\})]$ .

Next, we prove the second claim in the lemma. Suppose that  $X \in Fv[\varphi]$  (otherwise, we have  $\sigma_{X:=\tau}[\varphi] = \sigma[\varphi] = \sigma_{Y:=\tau}[\varphi\{^Y/X\}]$ ). We use induction on the complexity of  $\varphi$ . First, suppose that  $cp[\varphi] = 1$ . Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and X, Y set variables such that  $Y \notin Fv[\varphi]$ . Assume that  $X \in Fv[\varphi]$ . Thus we have  $\varphi = \{(t\varepsilon X)\}$  for some first-order  $\mathcal{L}^2$ -term t. Then,  $\sigma_{X:=\tau}[\varphi] = \tau[\sigma_{X:=\tau}[t]]$ . Since X and Y do not occur in t, this is equal to  $\tau[\sigma_{Y:=\tau}[t]]$ , which in turn equals  $\sigma_{Y:=\tau}[\varphi\{^Y/X\}]$ . Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^2$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and X, Y set variables such that  $Y \notin Fv[\varphi]$ . Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for  $\diamond \in \{\land, \lor, \supset\}$  and  $\mathcal{L}^2$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . Then,  $\sigma_{X:=\tau}[\varphi] = (\sigma_{X:=\tau}[\varphi_1] \diamond \sigma_{X:=\tau}[\varphi_2])$ . By the induction hypothesis (and the case in which  $X \notin Fv[\varphi]$ ), this  $\mathcal{L}^2$ -formula is equal to  $(\sigma_{Y:=\tau}[\varphi_1\{Y/X\}]\diamond \sigma_{Y:=\tau}[\varphi_2\{Y/X\}])$ . And, by definition, this is equal to  $\sigma_{Y:=\tau}[\varphi\{Y/X\}]$ .
- $\varphi = (Q^i x \psi)$  for  $Q^i \in \{\forall^i, \exists^i\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and individual variable  $x \notin Fv[\tau] \cup Fv[\sigma[\varphi]]$ . Then,  $\sigma_{X:=\tau}[\varphi] = (Q^i x \sigma_{X:=\tau,x:=x}[\psi])$ . By the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^i x \sigma_{Y:=\tau,x:=x}[\psi\{Y/x\}])$ . And this is (by definition) equal to  $\sigma_{Y:=\tau}[\varphi\{Y/x\}]$ .
- $\varphi = (Q^s Z \psi)$  for  $Q^s \in \{\forall^s, \exists^s\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and set variable  $Z \notin Fv[\tau] \cup \{X,Y\} \cup Fv[\sigma[\varphi]]$ . Then,  $\sigma_{X:=\tau}[\varphi] = (Q^s Z \sigma_{X:=\tau,Z:=Z_{abs}}[\psi])$ . By

the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^s Z \sigma_{Y:=\tau,Z:=Z_{abs}} [\psi\{Y/X\}])$ . And this is (by definition) equal to  $\sigma_{Y:=\tau} [\varphi\{Y/X\}]$ .

#### **Lemma 9.5.11.** Let t be a first-order $\mathcal{L}^2$ -term and $\tau$ an $\mathcal{L}^2$ -abstract.

- 1. For every  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and individual variables x, z such that  $z \notin Fv[\sigma[\varphi]]$ ,  $\sigma_{x:=z}[\varphi]\{t/z\} = \sigma_{x:=t}[\varphi]$ .
- 2. For every  $\mathcal{L}^2$ -formula  $\varphi$ ,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and set variable  $X \notin Fv[\sigma[\varphi]]$ ,  $\sigma_{X:=X_{abs}}[\varphi]\{\tau/X\} = \sigma_{X:=\tau}[\varphi]$ .

*Proof.* First, as in Lemma 8.5.17, it is straightforward to show that for every first-order  $\mathcal{L}^2$ -terms t' and t,  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment  $\sigma$ , and individual variables x, z such that  $z \notin Fv[\sigma[t']], \sigma_{x:=z}[t']\{t/z\} = \sigma_{x:=t}[t']$ .

Next, we prove the first claim in the lemma by induction on the complexity of  $\varphi$ . We include here only the cases that are special for second-order language (the other cases are handled as in the proof of Lemma 8.5.17). Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and x, z individual variables such that  $z \notin Fv[\sigma[\varphi]]$ . Consider the following cases:

- $\varphi = \{(t'\varepsilon T)\}$  for some first-order  $\mathcal{L}^2$ -term t', and second-order  $\mathcal{L}^2$ -term T. Then,  $\sigma_{x:=z}[\varphi]\{t'z\} = \sigma_{x:=z}[T][\sigma_{x:=z}[t']]\{t'z\} = \sigma[T][\sigma_{x:=z}[t']]\{t'z\}$ . By Lemma 9.3.9, since  $z \notin Fv[\sigma[T]]$ , this formula equals  $\sigma[T][\sigma_{x:=z}[t']\{t'z\}]$ . Since  $z \notin Fv[\sigma[t']]$ , the proof above for terms entails that this formula equals  $\sigma[T][\sigma_{x:=t}[t']]$ . Since x does not occur in T, this is equal to  $\sigma_{x:=t}[T][\sigma_{x:=t}[t']]$ , which is, by definition,  $\sigma_{x:=t}[\varphi]$ .
- $\varphi = (Q^s X \psi)$  for  $Q^s \in \{\forall^s, \exists^s\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and set variable  $X \notin Fv[\sigma[\varphi]]$ . Then,

$$\sigma_{x:=z}[\varphi]\{t/z\} = (Q^s X \sigma_{x:=z,X:=X_{abs}}[\psi])\{t/z\} = (Q^i X \sigma_{x:=z,X:=X_{abs}}[\psi]\{t/z\}).$$

By the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^i y \sigma_{x:=t,X:=X_{abs}}[\psi])$ . And this is (by definition) equal to  $\sigma_{x:=t}[\varphi]$ .

Next, we prove the second claim in the lemma. First, if we have  $X \notin Fv[\varphi]$ , then  $\sigma_{X:=X_{abs}}[\varphi] = \sigma[\varphi] = \sigma_{X:=\tau}[\varphi]$ . Since  $X \notin Fv[\sigma[\varphi]]$ , we also have  $\sigma[\varphi]\{\tau/X\} = \sigma[\varphi]$  as well. Suppose now that  $X \in Fv[\varphi]$ . We use induction on the complexity of  $\varphi$ . First, suppose that  $cp[\varphi] = 1$ . Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and let  $X \notin Fv[\sigma[\varphi]]$ . Since  $X \in Fv[\varphi]$ , we must have  $\varphi = \{(t \in X)\}$  for some first-order  $\mathcal{L}^2$ -term t. Then,

$$\sigma_{X:=X_{abs}}[\varphi]\{\tau/\mathbf{X}\} = X_{abs}[\sigma[t]]\{\tau/\mathbf{X}\} = \{(\sigma[t]\varepsilon X)\}\{\tau/\mathbf{X}\} = \tau[\sigma[t]] = \sigma_{X:=\tau}[\varphi].$$

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^2$ -formulas of lower complexity. Let  $\sigma$  be an  $\langle \mathcal{L}^2, \mathcal{D}^{\mathcal{L}^2} \rangle$ -assignment, and let  $X \notin Fv[\sigma[\varphi]]$ . Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for  $\diamond \in \{\land, \lor, \supset\}$  and  $\mathcal{L}^2$ -formulas  $\varphi_1, \varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . Then,  $\sigma_{X:=X_{abs}}[\varphi]\{\tau/X\} = (\sigma_{X:=X_{abs}}[\varphi_1]\{\tau/X\} \diamond \sigma_{X:=X_{abs}}[\varphi_2]\{\tau/X\})$ . By the induction hypothesis (and the case in which  $X \notin Fv[\varphi]$ ), this  $\mathcal{L}^2$ -formula is equal to  $(\sigma_{X:=\tau}[\varphi_1] \diamond \sigma_{X:=\tau}[\varphi_2])$ . And, by definition, this is equal to  $\sigma_{X:=\tau}[\varphi]$ .
- $\varphi = (Q^i x \psi)$  for  $Q^i \in \{\forall^i, \exists^i\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and individual variable  $x \notin Fv[\tau] \cup Fv[\sigma[\varphi]]$ . Then,

$$\sigma_{X:=X_{abs}}[\varphi]\{\tau/X\} = (Q^i x \sigma_{X:=X_{abs},x:=x}[\psi])\{\tau/X\} = (Q^i x \sigma_{X:=X_{abs},x:=x}[\psi]\{\tau/X\}).$$

By the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^i x \sigma_{X:=\tau,x:=x}[\psi])$ . And this is (by definition) equal to  $\sigma_{X:=\tau}[\varphi]$ .

•  $\varphi = (Q^s Y \psi)$  for  $Q^s \in \{\forall^s, \exists^s\}$ ,  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ , and set variable  $Y \notin Fv[\tau] \cup \{X\} \cup Fv[\sigma[\varphi]]$ . Then,

$$\sigma_{X:=X_{abs}}[\varphi]\{\tau/X\} = (Q^s Y \sigma_{X:=X_{abs},Y:=Y_{abs}}[\psi])\{\tau/X\} = (Q^i Y \sigma_{X:=X_{abs},Y:=Y_{abs}}[\psi]\{\tau/X\}).$$
 By the induction hypothesis, this  $\mathcal{L}^2$ -formula is equal to  $(Q^i Y \sigma_{X:=\tau,Y:=Y_{abs}}[\psi])$ . And this is (by definition) equal to  $\sigma_{X:=\tau}[\varphi]$ .

#### Proof of Theorem 9.5.6

Suppose that  $\mathcal{V}_{\mathbf{HIF}^2}^{cf}$   $H_0$ . The availability of external and internal weakenings ensures that  $H_0$  is unprovable. As in Lemma 8.5.12, it is possible to extend  $H_0$  to a maximal extended  $\mathcal{L}^2$ -hypersequent  $\Omega^*$  such that  $H_0 \sqsubseteq \Omega^*$ .\(^1\) We use  $\Omega^*$  to construct a countermodel for  $H_0$  in the form of a quasi- $\mathcal{L}^2$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$ . First, the notations  $L[\varphi], R[\varphi], \Omega^*[\varphi]$  are defined exactly in the completeness proof for **HIF** (Theorem 8.5.9). Similarly, the bounded linearly ordered set  $\mathcal{V}_0$ , the Gödel set  $\mathcal{V}$  are constructed using  $\Omega^*$  exactly as in the first-order proof. Now, let  $\mathcal{D} = \mathcal{D}^{\mathcal{L}^2}$  be the Herbrand quasi-domain for  $\mathcal{L}^2$ ,  $I = I^{\mathcal{L}^2}$  the Herbrand interpretation for  $\mathcal{L}^2$ , and define P and v as follows:

- For every n-ary predicate symbol p of  $\mathcal{L}^2$ ,  $P[p] = \lambda t_1, \dots, t_n \in \mathcal{D}_i.\Omega^*[\{p(t_1, \dots, t_n)\}].$
- For every  $\mathcal{L}^2$ -abstract  $\tau \in \mathcal{D}_s$ ,  $P[\tau] = \lambda t \in \mathcal{D}_i . \Omega^*[\tau[t]]$ .
- For every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ ,  $v[\varphi, \sigma] = \Omega^*[\sigma[\varphi]]$ .

It is easy to verify, using Lemma 9.5.10, that conditions (a) and (b) from Definition 9.5.2 hold.

For every  $\mathcal{L}^2$ -formula  $\varphi$  and set variable X of  $\mathcal{L}^2$ :

 $<sup>^1</sup>$ A maximal extended  $\mathcal{L}^2$ -hypersequent is defined just like a maximal extended  $\mathcal{L}^1$ -hypersequent, with the following natural additional requirements in "the witness property":

<sup>1.</sup> If  $t:(\forall^s X\varphi) \in \mu$ , then  $t:\varphi\{Y/X\} \in \mu$  for some set variable Y of  $\mathcal{L}^2$ .

<sup>2.</sup> If  $\mathbf{f}:(\exists^s X\varphi) \in \mu$ , then  $\mathbf{f}:\varphi\{Y/X\} \in \mu$  for some set variable Y of  $\mathcal{L}^2$ .

We show that  $\mathcal{Q}$  is not a model of  $H_0$ . Consider the  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma_{id}$  defined by  $\sigma_{id}[x] = x$  for every individual variable x of  $\mathcal{L}^2$ , and  $\sigma_{id}[X] = X_{abs}$  for every set variable X of  $\mathcal{L}^2$ . Let  $s \in H_0$ . Since  $H_0 \sqsubseteq \Omega^*$ , there exists some  $\mu \in \Omega^*$ , such that  $s \subseteq \mu$ . We claim that  $\mu \in v^{\mathfrak{f}}[\varphi, \sigma_{id}]$  whenever  $\mathfrak{f}: \varphi \in s$ , and  $\mu \notin v^{\mathfrak{t}}[\varphi, \sigma_{id}]$  whenever  $\mathfrak{t}: \varphi \in s$ . To see this, it suffices to note that  $\sigma_{id}[\varphi] = \varphi$  for every  $\mathcal{L}^2$ -formula  $\varphi$ . This fact follows from the definition of  $\sigma_{id}[\varphi]$ . Consequently,  $v^{\mathfrak{f}}[\varphi, \sigma_{id}] \not\subseteq v^{\mathfrak{t}}[\varphi, \sigma_{id}]$ , and so  $\mathcal{Q}, \sigma_{id} \not\models s$ .

It remains to prove that  $\mathcal{Q}$  is legal and comprehensive. We first show that it is legal, namely that  $\underline{v}_{\mathcal{Q}}[\varphi,\sigma]\subseteq v[\varphi,\sigma]$  for every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\langle \mathcal{L}^2,\mathcal{D}\rangle$ -assignment  $\sigma$ . Let  $\varphi$  be an  $\mathcal{L}^2$ -formula, and  $\sigma$  an  $\langle \mathcal{L}^2,\mathcal{D}\rangle$ -assignment. If  $\varphi$  is a first-order formula ( $\mathcal{L}^1$ -formula) the proof proceeds as in the proof of Theorem 8.5.9. We consider here only the new possible cases:

- $\varphi = \{(t\varepsilon T)\}$  for some first-order  $\mathcal{L}^2$ -term t, and second-order  $\mathcal{L}^2$ -term T. Then  $\underline{v}_{\mathcal{Q}}[\varphi, \sigma] = P[\sigma[T]][\sigma[t]] = \Omega^*[\sigma[T][\sigma[t]]] = \Omega^*[\sigma[\varphi]] = v[\varphi, \sigma].$
- $\varphi = (\forall^s X \psi)$  for some set variable  $X \notin Fv[\sigma[\varphi]]$  and  $\mathcal{L}^2$ -formula  $\psi$ . Then:

$$\underline{v}_{\mathcal{Q}}[\varphi, \sigma] = \langle \inf_{\tau \in \mathcal{D}_s} v^{\mathsf{t}}[\psi, \sigma_{X:=\tau}], \inf_{\tau \in \mathcal{D}_s} v^{\mathsf{f}}[\psi, \sigma_{X:=\tau}] \rangle.$$

We first prove that  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq \inf_{\tau \in \mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{X:=\tau}]$ . Thus we show that for every  $\tau \in \mathcal{D}_s$  we have  $v^{\mathbf{f}}[\varphi, \sigma] \subseteq v^{\mathbf{f}}[\psi, \sigma_{X:=\tau}]$ . Suppose that  $\mu \notin v^{\mathbf{f}}[\psi, \sigma_{X:=\tau}]$  for some  $\tau \in \mathcal{D}_s$ . By definition, we have  $\mathbf{f}:\sigma_{X:=\tau}[\psi] \notin \mu$ . By Lemma 9.5.11, we have  $\sigma_{X:=\tau}[\psi] = \sigma_{X:=X_{abs}}[\psi]\{\tau/x\}$ . The maximality of  $\Omega^*$  ensures that there exist an  $\mathcal{L}^2$ -hypersequent  $H \sqsubseteq \Omega^*$ , and  $\mathcal{L}^2$ -sequents  $s_1, \ldots, s_n \subseteq \mu$ , such that

$$\vdash_{\mathbf{HIF}^2}^{cf} H \mid s_1 \cup \{\mathbf{f}: \sigma_{X:=X_{obs}}[\psi] \{\tau/X\}\} \mid \dots \mid s_n \cup \{\mathbf{f}: \sigma_{X:=X_{obs}}[\psi] \{\tau/X\}\}.$$

By n consecutive applications of  $(f:\forall^s)$ , we obtain that

$$\vdash_{\mathbf{HIF}^2}^{cf} H \mid s_1 \cup \{ \mathbf{f} : (\forall^s X \sigma_{X:=X_{abs}}[\psi]) \} \mid \dots \mid s_n \cup \{ \mathbf{f} : (\forall^s X \sigma_{X:=X_{abs}}[\psi]) \}.$$

Since  $\Omega^*$  is unprovable, we must have  $\mathbf{f}:(\forall^s X \sigma_{X:=X_{abs}}[\psi]) \notin \mu$ . By definition,  $(\forall^s X \sigma_{X:=X_{abs}}[\psi]) = \sigma[\varphi]$ . It follows that  $\mu \notin v^{\mathbf{f}}[\varphi, \sigma]$ .

Next, we prove that  $\inf_{\tau \in \mathcal{D}_s} v^{\mathbf{t}}[\psi, \sigma_{X:=\tau}] \subseteq v^{\mathbf{t}}[\varphi, \sigma]$ . Suppose that  $\mu \notin v^{\mathbf{t}}[\varphi, \sigma]$ . Thus  $\mathbf{t} : \sigma[\varphi] \in \mu$ . By definition,  $\sigma[\varphi] = (\forall^s X \sigma_{X:=X_{abs}}[\psi])$ . Since  $\Omega^*$  admits the witness property, there exists a set variable Y of  $\mathcal{L}^2$ , such that  $\mathbf{t} : \sigma_{X:=X_{abs}}[\psi]\{Y/X\} \in \mu$ . By Lemma 9.3.8,  $\sigma_{X:=X_{abs}}[\psi]\{Y/X\} = \sigma_{X:=X_{abs}}[\psi]\{Y_{abs}/X\}$ . By Lemma 9.5.11, we have  $\sigma_{X:=X_{abs}}[\psi]\{Y_{abs}/X\} = \sigma_{X:=Y_{abs}}[\psi]$ . Thus,  $\mathbf{t} : \sigma_{X:=Y_{abs}}[\psi] \in \mu$ . It follows that  $\mu \notin v^{\mathbf{t}}[\psi, \sigma_{X:=\tau}]$ , for  $\tau = Y_{abs} \in \mathcal{D}_s$ , and therefore  $\mu \notin \bigcap_{\tau \in \mathcal{D}_s} v^{\mathbf{t}}[\psi, \sigma_{X:=\tau}]$ .

• The case  $\varphi = (\exists^s X \psi)$  is handled similarly.

Finally, we show that  $\mathcal{Q}$  is comprehensive. Let  $\varphi$  be an  $\mathcal{L}^2$ -formula, x an individual variable, and  $\sigma$  an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment. Let  $y \notin Fv[\varphi] \cup Fv[\sigma[\varphi]]$  be an in-

dividual variable, and let  $\tau = \{y \mid \sigma_{y:=y}[\varphi\{y/x\}]\}$ . Then  $\tau \in \mathcal{D}_s$ . We show that  $P[\tau] = \lambda t \in \mathcal{D}_i.v[\varphi,\sigma_{x:=t}]$ . Let  $t \in \mathcal{D}_i$ . Then,  $P[\tau][t] = \Omega^*[\tau[t]] = \Omega^*[\sigma_{y:=y}[\varphi\{y/x\}]\{t/y\}]$ . By Lemma 9.5.10,  $\sigma_{y:=y}[\varphi\{y/x\}] = \sigma_{x:=y}[\varphi]$ . By Lemma 9.5.11,  $\sigma_{x:=y}[\varphi]\{t/y\} = \sigma_{x:=t}[\varphi]$ . Thus,  $P[\tau][t] = \Omega^*[\sigma_{x:=t}[\varphi]] = v[\varphi,\sigma_{x:=t}]$ .

## 9.6 Completeness for the Ordinary Semantics

In this section we use the complete semantics of quasi-structures to prove the completeness of  $\mathbf{HIF}^2$  for the (ordinary) structures of Henkin-style second-order Gödel logic. To do so, we show that from every legal quasi-structure which is a counter-model of some hypersequent H, it is possible to extract an (ordinary) structure, which is also not a model of H, without losing comprehension.

**Definition 9.6.1.** Let  $\mathcal{D}$  be a quasi-domain, and  $\mathcal{D}'$  a domain for  $\mathcal{L}^2$ , such that  $\mathcal{D}_i = \mathcal{D}'_i$ . Let  $\delta$  be a function from  $\mathcal{D}_s$  to  $2^{\mathcal{D}'_s} \setminus \{\emptyset\}$ . A pair  $\langle \sigma, \sigma' \rangle$  of an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment and an  $\langle \mathcal{L}^2, \mathcal{D}' \rangle$ -assignment (respectively) is called a  $\delta$ -pair if (i)  $\sigma[x] = \sigma'[x]$  for every individual variable; and (ii)  $\sigma'[X] \in \delta[\sigma[X]]$  for every set variable.

**Theorem 9.6.2.** Let  $Q = \langle \mathcal{V}, \mathcal{D}, I, P, v \rangle$  be a legal and comprehensive quasi- $\mathcal{L}^2$ -structure. Then there exists a comprehensive  $\mathcal{L}^2$ -structure  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}', I', P' \rangle$ , where  $\mathcal{D}'_i = \mathcal{D}_i$  and  $I'_i = I_i$ , and a function  $\delta : \mathcal{D}_s \to 2^{\mathcal{D}'_s} \setminus \{\emptyset\}$ , such that  $\mathcal{W}[\varphi, \sigma'] \in v[\varphi, \sigma]$  for every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\delta$ -pair  $\langle \sigma, \sigma' \rangle$  (of an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment and an  $\langle \mathcal{L}^2, \mathcal{D}' \rangle$ -assignment).

Proof. First, we define  $\mathcal{D}'_s$ . For every  $D \in \mathcal{D}_s$ , denote by  $F_D$  the set of fuzzy subsets D' of  $\mathcal{D}_i$  over  $\mathcal{V}$  (i.e.  $D': \mathcal{D}_i \to \mathcal{V}$ ), such that  $D'[d] \in P[D][d]$  for every  $d \in \mathcal{D}_i$ . Note that for every  $D \in \mathcal{D}_s$ ,  $F_D$  is non-empty, since P[D][d] is non-empty for every  $d \in \mathcal{D}_i$ . Define  $\mathcal{D}'_s$  to be  $\bigcup_{D \in \mathcal{D}_s} F_D$ . Next,  $I'_s$  and P' are defined as follows:

- For every set constant symbol C of  $\mathcal{L}^2$ ,  $I'_s[C]$  is an arbitrary element in  $F_{I_s[C]}$ .
- For every predicate symbol p of  $\mathcal{L}^2$ ,  $P'[p] = P[p]^{\mathbf{f}}$ .

Let  $\delta: \mathcal{D}_s \to 2^{\mathcal{D}'_s} \setminus \{\emptyset\}$  be defined by  $\delta = \lambda D \in \mathcal{D}_s.F_D$ . We prove that  $\mathcal{W}$  and  $\delta$  satisfy the requirement in the theorem:  $\mathcal{W}[\varphi, \sigma'] \in v[\varphi, \sigma]$  for every  $\mathcal{L}^2$ -formula  $\varphi$  and  $\delta$ -pair  $\langle \sigma, \sigma' \rangle$ . Let  $\mathcal{V} = \langle V, \leq \rangle$ . We use induction on the complexity of  $\varphi$ . Note that since  $\mathcal{Q}$  is legal, it suffices to show that  $\mathcal{W}[\varphi, \sigma'] \in \underline{v}_{\mathcal{O}}[\varphi, \sigma]$  for every  $\delta$ -pair  $\langle \sigma, \sigma' \rangle$ .

First, suppose that  $cp[\varphi] = 1$ , and let  $\langle \sigma, \sigma' \rangle$  be a  $\delta$ -pair of an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment and an  $\langle \mathcal{L}^2, \mathcal{D}' \rangle$ -assignment. Exactly one of the following holds:

•  $\varphi = \{p(t_1, \dots, t_n)\}\$  or  $\varphi = \{\bot\}$ . These cases are handled as in the proof of Theorem 8.6.1.

•  $\varphi = \{(t\varepsilon T)\}$  for some first-order  $\mathcal{L}^2$ -term t, and second-order  $\mathcal{L}^2$ -term T. By definition,  $\mathcal{W}[\varphi, \sigma'] = {\sigma'}^{I'}[T][{\sigma'}^{I'}[t]]$ . Clearly, we have  ${\sigma'}^{I'}[t] = {\sigma}^{I}[t]$ . In addition,  ${\sigma'}^{I'}[T] \in F_{\sigma^I[T]}$  (in case T is a variable, this holds since  $\langle \sigma, \sigma' \rangle$  is a  $\delta$ -pair, and if T is a constant then it holds by definition). Therefore,

$$\sigma'^{I'}[T][\sigma'^{I'}[t]] = \sigma'^{I'}[T][\sigma^I[t]] \in P[\sigma^I[T]][\sigma[t]] = \underline{v}_{\mathcal{O}}[\varphi, \sigma].$$

Next, suppose that  $cp[\varphi] > 1$ , and that the claim holds for  $\mathcal{L}^2$ -formulas of lower complexity. Let  $\langle \sigma, \sigma' \rangle$  be a  $\delta$ -pair. Exactly one of the following holds:

- $\varphi = (\varphi_1 \diamond \varphi_2)$  for  $\diamond \in \{\land, \lor, \supset\}$  and  $\mathcal{L}^2$ -formulas  $\varphi_1$  and  $\varphi_2$  such that  $cp[\varphi_1] < cp[\varphi]$  and  $cp[\varphi_2] < cp[\varphi]$ . This case is handled as in the proof of Theorem 8.6.1.
- $\varphi = (Q^i x \psi)$  for some  $Q^i \in \{\forall^i, \exists^i\}$ , individual variable x of  $\mathcal{L}^2$ , and  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . We continue with  $Q^i = \forall^i$  (the proof is similar for  $\exists^i$ ). Clearly, for every  $d \in \mathcal{D}_i$ ,  $\langle \sigma_{x:=d}, \sigma'_{x:=d} \rangle$  is a  $\delta$ -pair. Thus by the induction hypothesis, for every  $d \in \mathcal{D}_i$ ,  $\mathcal{W}[\psi, \sigma'_{x:=d}] \in v[\psi, \sigma_{x:=d}]$ . Hence,

$$\mathcal{W}[\varphi, \sigma'] = \inf_{d \in \mathcal{D}_i} \mathcal{W}[\psi, \sigma'_{x:=d}] \in \langle \inf_{d \in \mathcal{D}_i} v^{\mathbf{f}}[\psi, \sigma_{x:=d}], \inf_{d \in \mathcal{D}_i} v^{\mathbf{t}}[\psi, \sigma_{x:=d}] \rangle = \underline{v}_{\mathcal{Q}}[\varphi, \sigma].$$

•  $\varphi = (Q^s X \psi)$  for some  $Q^s \in \{\forall^i, \exists^i\}$ , set variable X of  $\mathcal{L}^2$ , and  $\mathcal{L}^2$ -formula  $\psi$  such that  $cp[\psi] < cp[\varphi]$ . We continue with  $Q^s = \forall^s$  (the proof is similar for  $\exists^s$ ). In this case, we should prove that:

$$\inf_{D' \in \mathcal{D}_s'} \mathcal{W}[\psi, \sigma_{X:=D'}'] \in \langle \inf_{D \in \mathcal{D}_s} v^{\mathsf{f}}[\psi, \sigma_{X:=D}], \inf_{D \in \mathcal{D}_s} v^{\mathsf{t}}[\psi, \sigma_{X:=D}] \rangle.$$

First, we show that  $\inf_{D\in\mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{X:=D}] \leq \inf_{D'\in\mathcal{D}_s'} \mathcal{W}[\psi, \sigma'_{X:=D'}]$ , by showing that  $\inf_{D\in\mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{X:=D}] \leq \mathcal{W}[\psi, \sigma'_{X:=D'}]$  for every  $D' \in \mathcal{D}_s'$ . Let  $D' \in \mathcal{D}_s'$ , and let D be an arbitrary element in  $\mathcal{D}_s$  such that  $D' \in F_D$ . Then  $\langle \sigma_{X:=D}, \sigma'_{X:=D'} \rangle$  is a  $\delta$ -pair. By the induction hypothesis,  $v^{\mathbf{f}}[\psi, \sigma_{X:=D}] \leq \mathcal{W}[\psi, \sigma'_{X:=D'}]$ . Thus,  $\inf_{D\in\mathcal{D}_s} v^{\mathbf{f}}[\psi, \sigma_{X:=D}] \leq \mathcal{W}[\psi, \sigma'_{X:=D'}]$ .

Next, we show that  $\inf_{D' \in \mathcal{D}'_s} \mathcal{W}[\psi, \sigma'_{X:=D'}] \leq \inf_{D \in \mathcal{D}_s} v^{\mathsf{t}}[\psi, \sigma_{X:=D}]$ , by proving that  $\inf_{D' \in \mathcal{D}'_s} \mathcal{W}[\psi, \sigma'_{X:=D'}] \leq v^{\mathsf{t}}[\psi, \sigma_{X:=D}]$  for every  $D \in \mathcal{D}_s$ . Let  $D \in \mathcal{D}_s$ . Take some  $D' \in F_D$ . Then  $D' \in \mathcal{D}'_s$ , and  $\langle \sigma_{X:=D}, \sigma'_{X:=D'} \rangle$  is a  $\delta$ -pair. By the induction hypothesis,  $\mathcal{W}[\psi, \sigma'_{X:=D'}] \leq v^{\mathsf{t}}[\psi, \sigma_{X:=D}]$ . Thus,  $\inf_{D' \in \mathcal{D}'_s} \mathcal{W}[\psi, \sigma'_{X:=D'}] \leq v^{\mathsf{t}}[\psi, \sigma_{X:=D}]$ .

Finally, we show that  $\mathcal{W}$  is comprehensive. Let  $\varphi$  be an  $\mathcal{L}^2$ -formula, x an individual variable, and  $\sigma'$  an  $\langle \mathcal{L}^2, \mathcal{D}' \rangle$ -assignment. We show that  $\mathcal{W}[\varphi, \sigma', x] \in \mathcal{D}'_s$ . Define an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$  as follows: (i)  $\sigma[x] = \sigma'[x]$  for every individual variable x; and (ii) for every set variable X,  $\sigma[X]$  is an (arbitrary) element of  $\mathcal{D}_s$  such that  $\sigma'[X] \in F_{\sigma[X]}$ . Since  $\mathcal{Q}$  is comprehensive, there exists some  $D \in \mathcal{D}_s$  such that  $P[D] = \lambda d \in \mathcal{D}_i.v[\varphi, \sigma_{x:=d}]$ . We claim that  $\mathcal{W}[\varphi, \sigma', x] \in F_D$  (and so,  $\mathcal{W}[\varphi, \sigma', x] \in \mathcal{D}'_s$ ). By definition, we should show that  $\mathcal{W}[\varphi, \sigma', x][d] \in P[D][d]$  for every  $d \in \mathcal{D}_i$ . Let

 $d \in \mathcal{D}_i$ . Obviously,  $\langle \sigma_{x:=d}, \sigma'_{x:=d} \rangle$  is a  $\delta$ -pair, and thus by the claim proved above, we have  $\mathcal{W}[\varphi, \sigma', x][d] = \mathcal{W}[\varphi, \sigma'_{x:=d}] \in v[\varphi, \sigma_{x:=d}] = P[D][d]$ .

Corollary 9.6.3. If  $\not\vdash_{\mathbf{HIF}^2}^{cf} H$ , then there exists a comprehensive  $\mathcal{L}^2$ -structure which is not a model of H.

Proof. Suppose that  $\not\vdash_{\mathbf{HIF}^2}^{cf} H$ . Then, by Theorem 9.5.6, there exists a legal and comprehensive quasi- $\mathcal{L}^2$ -structure  $\mathcal{Q} = \langle \mathcal{V}, \mathcal{D}, P, v \rangle$ , which is not a model of H. This implies that there exists an  $\langle \mathcal{L}^2, \mathcal{D} \rangle$ -assignment  $\sigma$ , such that  $\mathcal{Q}, \sigma \not\models s$  for every  $s \in H$ . Let  $\mathcal{W} = \langle \mathcal{V}, \mathcal{D}', P' \rangle$  be a comprehensive  $\mathcal{L}^2$ -structure and  $\delta$  a function, satisfying the requirement in Theorem 9.6.2. Let  $\sigma'$  be an  $\langle \mathcal{L}^2, \mathcal{D}' \rangle$ -assignment such that  $\langle \sigma, \sigma' \rangle$  is a  $\delta$ -pair (there exists such an assignment since the range of  $\delta$  does not include the empty set). We show that  $\mathcal{W}, \sigma' \not\models H$ . Let  $s \in H$ . Since  $\mathcal{Q}, \sigma \not\models s$ , we have  $\mathcal{Q}^{\mathbf{f}}[s, \sigma] > \mathcal{Q}^{\mathbf{t}}[s, \sigma]$ . The fact that  $\mathcal{W}[\varphi, \sigma'] \in v[\varphi, \sigma]$  for every  $\varphi$  entails that  $\mathcal{W}^{\mathbf{f}}[s, \sigma'] > \mathcal{W}^{\mathbf{t}}[s, \sigma']$ , and so  $\mathcal{W}, \sigma' \not\models s$ .

Just like in the first-order case, we immediately obtain the following:

Corollary 9.6.4. For every  $\mathcal{L}^2$ -formula  $\varphi$ , if  $\Vdash^{\mathbf{G}\ddot{o}_{\mathcal{L}^2}} \varphi$  then  $\vdash_{\mathbf{HIF}^2} \{\{\mathbf{t}:\varphi\}\}$ .

Corollary 9.6.5. If an  $\mathcal{L}^2$ -hypersequent is provable in  $HIF^2$ , then it is provable in  $HIF^2$  without applying (cut).

**Remark 9.6.6.** Note that cut-admissibility for **HIF** (the original system for first-order Gödel logic) is obtained as a corollary, for the reason that second-order quantifiers cannot be involved in a cut-free proof of a first-order formula.

## Chapter 10

## Summary and Further Work

In this thesis, we studied Gentzen-type calculi from a semantic point of view. This study encompassed several general abstract families of sequent calculi: pure calculi, canonical calculi, quasi-canonical calculi, and basic calculi, as well as the family of canonical Gödel hypersequent calculi. For each of these families, a corresponding denotational semantic framework was identified, based on certain (possibly non-deterministic) semantic structures: many-valued valuations for pure calculi (including canonical and quasi-canonical ones); Kripke valuations for basic calculi; and Gödel valuations for canonical Gödel calculi. It was shown that each calculus in these families induces a set of semantic structures for which it is sound and complete. Moreover, we provided general, modular and uniform methods to obtain such a set for a given calculus, of which many important soundness and completeness theorems for known calculi are particular instances. In the case of canonical calculi (both canonical sequent calculi, and canonical Gödel hypersequent calculi), the resulting semantics was proven to be effective, leading to a semantic decision procedure for each such calculus. In addition, for each of the families of proof systems mentioned above, we derived semantic characterizations of analyticity, cut-admissibility and axiom-expansion. This provides a "semantic toolbox", intended to complement the usual proof-theoretic methods. Indeed, as was demonstrated in many examples, the proofs (or refutations) of these properties based on the semantic characterizations is straightforward in many cases, and less tedious and error-prone than the usual inductive syntactic arguments.

Finally, we demonstrated these methods on **HIF** and **HIF**<sup>2</sup>, two hypersequent calculi for first-order and second-order languages. Based on the ideas from the semantic analysis of canonical Gödel propositional calculi in Chapter 7, we showed that these calculi are sound and (cut-free) complete for standard first-order Gödel logic, and Henkin-style second-order Gödel logic (respectively). To the best of our knowledge, **HIF**<sup>2</sup> is the first (cut-free) proof system introduced for Henkin-style second-order Gödel logic.

We believe that the semantic approach in the analysis of proof systems has demonstrated its potential in this thesis, as a useful methodology, that may be applied whenever a new Gentzen-type system (of one of the families mentioned above) is encountered. Evidently, this methodology is still in its early stages of its development, and there are various open questions, and possible promising extensions. The main directions for further research include the following:

Substructural Calculi All sequent and hypersequent calculi studied in this thesis are fully-structural, as they include (both internal and external, in the case of hypersequent calculi) the exchange rules, weakening rules, contraction rules, and expansion rules (recall that we defined sequents and hypersequents as sets). However, many important logics have only substructural sequent systems (in particular, contraction-free or weakening-free calculi), that cannot be treated in our framework. Extending some of the results above for families of substructural systems is an important future goal. This will require the development of new frameworks of non-deterministic semantics. In particular, an interesting question arises as to whether the semantic proof technique of cut-admissibility is applicable for substructural calculi. Two particularly important cases are the (sub-structural) hypersequent systems for the fundamental propositional fuzzy logics – Łukasiewicz's logic, and product logic (see [76]). In addition, the current work deals only with multiple-conclusion systems, while it can be useful to derive similar results for single-conclusion ones. For two-sided canonical single-conclusion systems, this was done in [14].

First-Order and Higher-Order Calculi All general families of sequent and hypersequent calculi studied above handle only propositional logics. Extending the methods and results of this thesis to languages that include quantifiers is evidently an important future goal. We believe that such an extension should be possible. Indeed, the original three-valued non-deterministic semantics for the (cut)-free fragment of LK applies also to the first-order quantifiers [58], as well as the three-valued non-deterministic semantics for its (id)-fragment [64]. Our results for HIF<sup>2</sup> in Chapter 9 demonstrate that similar ideas can be applied in different higher-order calculi. Note that families of canonical sequent calculi with first-order quantifiers were studied in [18, 98].

**Extending the Realm of PNmatrices** In Chapters 4 and 5 the semantic framework of finite PNmatrices has shown to be adequate for canonical sequent calculi and quasi-canonical ones. Besides the fact that this framework provides intuitive semantics, its main attractive property is its *effectiveness*. Thus every calculus that has

a sound and complete PNmatrix is decidable. A natural question is whether this semantic framework suffices for more than canonical and quasi-canonical calculi. In particular, consider the following schemes:

$$\frac{\Gamma \Rightarrow \neg^{i}\varphi, \Delta}{\Gamma, \neg^{i}\neg\varphi \Rightarrow \Delta} \qquad \frac{\Gamma \Rightarrow \varphi, \Delta \quad \Gamma \Rightarrow \neg\varphi, \Delta}{\Gamma, \neg(\varphi \land \neg\varphi) \Rightarrow \Delta}$$

The first one is an example of a family of rules used in proof systems that combine classical and paraconsistent negations [66], and the second is taken from [12] for handling certain paraconsistent logics. Note that both schemes (for i > 1 in the first one) cannot be presented as canonical or quasi-canonical rules, and thus our semantic methods do not apply to them. Nevertheless, we believe that the ideas in Chapter 5 can be extended to handle also rules as the first one above, in which more than one unary connective may precede the formulas in the premises and the conclusion. For the second rule above, [12] suggests an infinite-valued non-deterministic matrix, and it remains open whether other systems with rules of this kind can be characterized by non-deterministic matrices, and in what cases this infinite-valued semantics is effective.

General Canonical Hypersequent Calculi. In Chapter 7, we studied a family of canonical hypersequent calculi that are based on the communication rule. It is interesting to study canonical hypersequent calculi that employ other structural hypersequential rules. For example, we believe that a similar methodology, using Kripke semantics of directed frames, should work for canonical calculi based on the hypersequent rule (lq). This rule is used in systems for logics of weak excluded middle [42]. Developing a general theory of structural hypersequential rules in canonical hypersequent calculi is another interesting direction for a future work.

Different Proof-Theoretic Formalisms In this thesis we investigated sequent and hypersequent systems. It is interesting to pursue similar investigations in other proof-theoretic frameworks, that handle different families of non-classical logics. This includes: semantic and prefixed tableaux systems [52], display calculi [33], nested sequents calculi [53, 60, 36], labelled calculi as in [54, 77], and sequents of relations [23].

**Second-Order Gödel Logic** In [72], we proved that **HIF** (a calculus for *first-order* Gödel logic) enjoys cut-admissibility. In fact, for **HIF** we proved slightly stronger properties than those shown in this paper for **HIF**<sup>2</sup>. Obtaining these stronger results for **HIF**<sup>2</sup> seems to be straightforward. This includes:

• In [72] we considered also derivations from non-empty sets of hypersequents,

- and proved strong cut-admissibility.
- For applications, it is sometimes useful to enrich Gödel logic with a globalization connective (also known as Baaz Delta connective, see [22]). [72] studies the extension of **HIF** with rules for this connective, and the same can be done for **HIF**<sup>2</sup>.

In addition, the following extensions of the current result are left for a future work:

- It is interesting to consider equality, both between first-order terms and second-order ones. In this case, rules for extensionality should be added.
- Extending the calculus for richer second order signatures and also for full type theory seem to be possible. In the case of classical logic, cut-free completeness for the extended system was proved shortly after Tait's proof for the second-order one by Takahashi and Prawitz, [88, 82]. This extension is necessary in order to obtain a proof system for (the Gödel fragment) of fuzzy set theory (see [37]).

- [1] Alan R. Anderson and Nuel D. Belnap. *Entailment: The Logic of Relevance and Neccessity, Vol. I.* Princeton University Press, 1975.
- [2] Alessandro Avellone, Mauro Ferrari, and Pierangelo Miglioli. Duplication-free tableau calculi and related cut-free sequent calculi for the interpolable propositional intermediate logics. *Logic Journal of IGPL*, 7(4):447–480, 1999.
- [3] Arnon Avron. On modal systems having arithmetical interpretations. *The Journal of Symbolic Logic*, 49(3):935–942, 1984.
- [4] Arnon Avron. Hypersequents, logical consequence and intermediate logics for concurrency. Annals of Mathematics and Artificial Intelligence, 4:225–248, 1991.
- [5] Arnon Avron. Simple consequence relations. *Information and Computation*, 92(1):105 139, 1991.
- [6] Arnon Avron. Gentzen-type systems, resolution and tableaux. *Journal of Automated Reasoning*, 10(2):265–281, 1993.
- [7] Arnon Avron. The method of hypersequents in the proof theory of propositional non-classical logics. In Wilfrid Hodges, Martin Hyland, Charles Steinhorn, and John Truss, editors, *Logic: from foundations to applications*, pages 1–32, New York, NY, USA, 1996. Clarendon Press.
- [8] Arnon Avron. Non-deterministic Semantics for Families of Paraconsistent Logics. In J. Y. Beziau, W. A. Carnielli, and D. M. Gabbay, editors, *Handbook of Paraconsistency*, volume 9 of *Studies in Logic*, pages 285–320. College Publications, 2007.
- [9] Arnon Avron. Multi-valued semantics: Why and how. *Studia Logica*, 92:163–182, 2009.
- [10] Arnon Avron. A simple proof of completeness and cut-admissibility for propositional Gödel logic. *Journal of Logic and Computation*, 21(5):813–821, 2011.

[11] Arnon Avron, Agata Ciabattoni, and Anna Zamansky. Canonical calculi: Invertibility, Axiom-Expansion and (Non)-determinism. In *Proceedings of the 4th Computer Science Symposium in Russia*, *LNCS 5675*, pages 26–37. Springer, 2009.

- [12] Arnon Avron, Beata Konikowska, and Anna Zamansky. Modular construction of cut-free sequent calculi for paraconsistent logics. In *Logic in Computer Science* (LICS), 2012 27th Annual IEEE Symposium on, pages 85–94, 2012.
- [13] Arnon Avron, Beata Konikowska, and Anna Zamansky. Cut-free sequent calculi for C-systems with generalized finite-valued semantics. *Journal of Logic and Computa*tion, 23(3):517–540, 2013.
- [14] Arnon Avron and Ori Lahav. On constructive connectives and systems. *Logical Methods in Computer Science*, 6(4), 2010.
- [15] Arnon Avron and Ori Lahav. Kripke semantics for basic sequent systems. In *Proceedings of the 20th international conference on Automated reasoning with analytic tableaux and related methods*, TABLEAUX'11, pages 43–57, Berlin, Heidelberg, 2011. Springer-Verlag.
- [16] Arnon Avron and Ori Lahav. A multiple-conclusion calculus for first-order Gödel logic. In Alexander Kulikov and Nikolay Vereshchagin, editors, Computer Science Theory and Applications, volume 6651 of Lecture Notes in Computer Science, pages 456–469. Springer Berlin Heidelberg, 2011.
- [17] Arnon Avron and Iddo Lev. Non-deterministic multiple-valued structures. *Journal of Logic and Computation*, 15(3):241–261, 2005.
- [18] Arnon Avron and Anna Zamansky. Canonical calculi with (n,k)-ary quantifiers. Logical Methods in Computer Science, 4(3), 2008.
- [19] Arnon Avron and Anna Zamansky. Canonical signed calculi, non-deterministic matrices and cut-elimination. In Proceedings of the 2009 International Symposium on Logical Foundations of Computer Science, LFCS '09, pages 31–45, Berlin, Heidelberg, 2009. Springer-Verlag.
- [20] Arnon Avron and Anna Zamansky. Canonical signed calculi, non-deterministic matrices and cut-elimination. In Sergei Artemov and Anil Nerode, editors, *Logical Foundations of Computer Science*, volume 5407 of *Lecture Notes in Computer Science*, pages 31–45. Springer Berlin Heidelberg, 2009.

BIBLIOGRAPHY

[21] Arnon Avron and Anna Zamansky. Non-deterministic semantics for logical systems. In Dov M. Gabbay and Franz Guenthner, editors, *Handbook of Philosophical Logic*, volume 16, pages 227–304. Springer Netherlands, 2011.

- [22] Matthias Baaz, Agata Ciabattoni, and Christian G. Fermüller. Hypersequent calculi for Gödel logics a survey. *Journal of Logic and Computation*, 13(6):835–861, 2003.
- [23] Matthias Baaz, Agata Ciabattoni, and Christian G. Fermüller. Sequent of relations calculi: A framework for analytic deduction in many-valued logics. In Melvin Fitting and Ewa Orłowska, editors, Beyond Two: Theory and Applications of Multiple-Valued Logic, volume 114 of Studies in Fuzziness and Soft Computing, pages 157–180. Physica-Verlag HD, 2003.
- [24] Matthias Baaz and Christian G. Fermüller. Intuitionistic counterparts of finitely-valued logics. In *Multiple-Valued Logic*, 1996. Proceedings., 26th International Symposium on, pages 136–141, 1996.
- [25] Matthias Baaz and Christian G. Fermüller. Analytic calculi for projective logics. In Neil V. Murray, editor, Automated Reasoning with Analytic Tableaux and Related Methods, volume 1617 of Lecture Notes in Computer Science, pages 36–51. Springer Berlin / Heidelberg, 1999.
- [26] Matthias Baaz, Christian G. Fermüller, Gernot Salzer, and Richard Zach. Labeled calculi and finite-valued logics. *Studia Logica*, 61(1):7–33, 1998.
- [27] Matthias Baaz, Christian G. Fermüller, and Richard Zach. Dual systems of sequents and tableaux for many-valued logics. *Bulletin of the EATCS*, 49:192–197, 1993.
- [28] Matthias Baaz, Ori Lahav, and Anna Zamansky. Effective finite-valued semantics for labelled calculi. In Bernhard Gramlich, Dale Miller, and Uli Sattler, editors, *Automated Reasoning*, volume 7364 of *Lecture Notes in Computer Science*, pages 52–66. Springer Berlin / Heidelberg, 2012.
- [29] Matthias Baaz, Ori Lahav, and Anna Zamansky. Finite-valued semantics for canonical labelled calculi. *Journal of Automated Reasoning*, pages 1–30, 2013.
- [30] Matthias Baaz and Richard Zach. Hypersequent and the proof theory of intuitionistic fuzzy logic. In *Proceedings of the 14th Annual Conference of the EACSL on* Computer Science Logic, pages 187–201. Springer-Verlag, 2000.
- [31] Lev Beklemishev and Yuri Gurevich. Propositional primal logic with disjunction. Journal of Logic and Computation, 2012.

[32] Francesco Belardinelli, Peter Jipsen, and Hiroakira Ono. Algebraic aspects of cut elimination. *Studia Logica: An International Journal for Symbolic Logic*, 77(2):209–240, 2004.

- [33] Nuel D. Belnap. Display logic. Journal of philosophical logic, 11(4):375–417, 1982.
- [34] Jean-Yves Béziau. Sequents and bivaluations. *Logique et Analyse*, 44(176):373–394, 2001.
- [35] Patrick Blackburn, Maarten de Rijke, and Yde Venema. *Modal Logic*. Cambridge Tracts in Theoretical Computer Science. Cambridge University Press, 2002.
- [36] Kai Brünnler. Nested sequents. Habilitation Thesis. http://arxiv.org/pdf/1004. 1845, 2010.
- [37] Libor Běhounek and Petr Cintula. Fuzzy class theory. Fuzzy Sets and Systems, 154(1):34–55, 2005.
- [38] Libor Běhounek and Petr Cintula. General logical formalism for fuzzy mathematics: Methodology and apparatus. In Fuzzy Logic, Soft Computing and Computational Intelligence: Eleventh International Fuzzy Systems Association World Congress. Tsinghua. University Press Springer, 2005.
- [39] Carlos Caleiro, Walter Carnielli, Marcelo Coniglio, and João Marcos. Two's company: the humbug of many logical values. In Jean-Yves Beziau, editor, *Logica Universalis*, pages 169–189. Birkhuser Basel, 2005.
- [40] Walter A. Carnielli, Marcelo E. Coniglio, and João Marcos. Logics of formal inconsistency. In Dov M. Gabbay and Franz Guenthner, editors, *Handbook of Philosophical Logic*, volume 14, pages 15–107. Springer, 2007. Second edition.
- [41] Alexander Chagrov and Michael Zakharyaschev. *Modal Logic*. Oxford Logic Guides. Oxford University Press, USA, 1997.
- [42] Agata Ciabattoni, Dov M. Gabbay, and Nicola Olivetti. Cut-free proof systems for logics of weak excluded middle. *Soft Computing*, 2(4):147–156, 1999.
- [43] Agata Ciabattoni, Ori Lahav, and Anna Zamansky. Basic constructive connectives, determinism and matrix-based semantics. In Kai Brünnler and George Metcalfe, editors, Automated Reasoning with Analytic Tableaux and Related Methods, volume 6793 of Lecture Notes in Computer Science, pages 119–133. Springer Berlin / Heidelberg, 2011.

[44] Agata Ciabattoni and Kazushige Terui. Towards a semantic characterization of cut-elimination. *Studia Logica*, 82:95–119, 2006.

- [45] Petr Cintula, Petr Hájek, and Carles Noguera. *Handbook of Mathematical Fuzzy Logic volume 1*. Number 37 in Studies in Logic, Mathematical Logic and Foundations. College Publications, London, 2011.
- [46] Giovanna Corsi. Semantic trees for Dummett's logic. Studia Logica, 45:199–206, 1986.
- [47] Newton C. A. da Costa and E. H. Alves. A semantical analysis of the calculi  $C_n$ . Notre Dame Journal of Formal Logic, 18(4):621–630, 1977.
- [48] Michael Dummett. A propositional calculus with denumerable matrix. *The Journal of Symbolic Logic*, 24(2):97–106, 1959.
- [49] Roy Dyckhoff. Contraction-free sequent calculi for intuitionistic logic. *The Journal of Symbolic Logic*, 57(3):pp. 795–807, 1992.
- [50] Roy Dyckhoff. A deterministic terminating sequent calculus for Gödel-Dummett logic. Logic Journal of IGPL, 7(3):319–326, 1999.
- [51] Roy Dyckhoff and Sara Negri. Decision methods for linearly ordered Heyting algebras. *Archive for Mathematical Logic*, 45:411–422, 2006.
- [52] Melvin Fitting. Modal proof theory. In Johan Van Benthem Patrick Blackburn and Frank Wolter, editors, *Handbook of Modal Logic*, volume 3 of *Studies in Logic and Practical Reasoning*, pages 85 138. Elsevier, 2007.
- [53] Melvin Fitting. Prefixed tableaus and nested sequents. Ann. Pure Appl. Logic, 163(3):291–313, 2012.
- [54] Dov M. Gabbay. *Labelled Deductive Systems, Vol. 1.* Oxford Logic Guides. Clarendon Press, 1996.
- [55] Nikolaos Galatos, Peter Jipsen, Tomasz Kowalski, and Hiroakira Ono. Residuated Lattices: An Algebraic Glimpse at Substructural Logics: An Algebraic Glimpse at Substructural Logics. Studies in Logic and the Foundations of Mathematics. Elsevier Science, 2007.
- [56] Gerhard Gentzen. Investigations into logical deduction. American Philosophical Quarterly, 1(4):288–306, 1964.

[57] Jean-Yves Girard. Linear logic. Theoretical Computer Science, 50(1):1 – 101, 1987.

- [58] Jean-Yves Girard. *Proof theory and logical complexity*. Studies in proof theory. Bibliopolis, 1987.
- [59] Rajeev Goré and Linda Postniece. Combining derivations and refutations for cut-free completeness in bi-intuitionistic logic. *Journal of Logic and Computation*, 20(1):233– 260, 2010.
- [60] Rajeev Goré and Revantha Ramanayake. Labelled tree sequents, tree hypersequents and nested (deep) sequents. In Thomas Bolander, Torben Braüner, Silvio Ghilardi, and Lawrence Moss, editors, Advances in Modal Logic, volume 9, pages 279–299. College Publications, 2012.
- [61] Siegfried Gottwald. A Treatise on Many-Valued Logics. Studies in Logic and Computation. Research Studies Press, 2001.
- [62] Yuri Gurevich and Itay Neeman. Logic of infons: The propositional case. *ACM Trans. Comput. Logic*, 12(2):9:1–9:28, January 2011.
- [63] Peter Hájek. Metamathematics of Fuzzy Logic. Trends in Logic. Springer, 2001.
- [64] Brigitte Hösli and Gerhard Jäger. About some symmetries of negation. *The Journal of Symbolic Logic*, 59(2):473–485, 1994.
- [65] Ryo Ishigaki and Ryo Kashima. Sequent calculi for some strict implication logics. Logic Journal of IGPL, 16(2):155–174, 2008.
- [66] Norihiro Kamide. Proof systems combining classical and paraconsistent negations. Studia Logica, 91(2):217–238, 2009.
- [67] Beata Konikowska. Rasiowa-Sikorski deduction systems in computer science applications. *Theoretical Computer Science*, 286(2):323–366, 2002.
- [68] Ori Lahav. Non-deterministic matrices for semi-canonical deduction systems. In 42nd IEEE International Symposium on Multiple-Valued Logic (ISMVL), pages 79– 84, May 2012.
- [69] Ori Lahav. Semantic investigation of canonical Gödel hypersequent systems. *Journal of Logic and Computation*, 2013 (to appear).
- [70] Ori Lahav. Studying sequent systems via non-deterministic multiple-valued matrices. *Journal of Multiple-Valued Logic and Soft Computing*, 2013 (to appear).

[71] Ori Lahav and Arnon Avron. Non-deterministic connectives in propositional Gödel logic. In 7th conference of the European Society for Fuzzy Logic and Technology (EUSFLAT), pages 175–182, July 2011.

- [72] Ori Lahav and Arnon Avron. A semantic proof of strong cut-admissibility for first-order Gödel logic. *Journal of Logic and Computation*, 23(1):59–86, 2013.
- [73] Ori Lahav and Arnon Avron. A unified semantic framework for fully-structural propositional sequent systems. ACM Transactions on Computational Logic, 2013 (to appear).
- [74] René Lavendhomme and Thierry Lucas. Sequent calculi and decision procedures for weak modal systems. *Studia Logica*, 66:121–145, 2000.
- [75] Daniel Leivant. On the proof theory of the modal logic for arithmetic provability. The Journal of Symbolic Logic, 46(3):531–538, 1981.
- [76] George Metcalfe, Nicola Olivetti, and Dov M. Gabbay. Proof Theory for Fuzzy Logics. Applied Logic Series. Springer, 2009.
- [77] Sara Negri. Proof analysis in modal logic. Journal of Philosophical Logic, 34(5-6):507-544, 2005.
- [78] Hiroakira Ono. On some intuitionistic modal logics. Publications of the Research Institute for Mathematical Sciences, 13(3):687–722, 1977.
- [79] Hiroakira Ono. Proof-theoretic methods in nonclassical logic—an introduction. *Theories of types and proofs*, 2:207–254, 1998.
- [80] Luís Pinto and Tarmo Uustalu. Proof search and counter-model construction for biintuitionistic propositional logic with labelled sequents. In Martin Giese and Arild Waaler, editors, Automated Reasoning with Analytic Tableaux and Related Methods, volume 5607 of Lecture Notes in Computer Science, pages 295–309. Springer Berlin Heidelberg, 2009.
- [81] Francesca Poggiolesi. Gentzen Calculi for Modal Propositional Logic. Trends in Logic: Studia Logica Library. Springer, 2010.
- [82] Dag Prawitz. Hauptsatz for higher order logic. The Journal of Symbolic Logic, 33(3):452–457, 1968.
- [83] Arthur N. Prior. The runabout inference ticket. Analysis, 21(2):38–39, 1960.

[84] Giovanni Sambin and Silvio Valentini. The modal logic of provability. the sequential approach. *Journal of Philosophical Logic*, 11(3):311–342, 1982.

- [85] Kurt Schütte. Syntactical and semantical properties of simple type theory. *The Journal of Symbolic Logic*, 25(4):305–326, 1960.
- [86] Osamu Sonobe. A Gentzen-type formulation of some intermediate propositional logics. *Journal of Tsuda College*, 7:7–114, 1975.
- [87] William W. Tait. A nonconstructive proof of Gentzen's Hauptsatz for second order predicate logic. *Bulletin of the American Mathematical Society*, 72:980–983, 1966.
- [88] Moto-o Takahashi. A proof of cut-elimination theorem in simple type-theory. *Journal* of the Mathematical Society of Japan, 19(4):399–410, 1967.
- [89] Gaisi Takeuti. On a generalized logic calculus. *Japanese Journal of Mathematics*, 23:39–96, 1953.
- [90] Gaisi Takeuti. *Proof Theory*. Studies in Logic and the Foundations of Mathematics. North-Holland Publishing Company, 1975.
- [91] Kazushige Terui. Which structural rules admit cut elimination? an algebraic criterion. *The Journal of Symbolic Logic*, 72(3):pp. 738–754, 2007.
- [92] Anne S. Troelstra and Helmut Schwichtenberg. *Basic proof theory*. Cambridge University Press, New York, NY, USA, 1996.
- [93] Alasdair Urquhart. Basic many-valued logic. In Dov M. Gabbay and Franz Guenthner, editors, *Handbook of Philosophical Logic*, volume 2, pages 249–295. Springer Netherlands, 2001.
- [94] Rineke (L.C.) Verbrugge. Provability logic. In Edward N. Zalta, editor, *The Stanford Encyclopedia of Philosophy*. Winter 2010 edition, 2010.
- [95] Vilém Novák. On fuzzy type theory. Fuzzy Sets and Systems, 149(2):235 273, 2005.
- [96] Heinrich Wansing. Sequent systems for modal logics. In Dov M. Gabbay and Franz Guenthner, editors, *Handbook of Philosophical Logic*, 2nd edition, volume 8, pages 61–145. Springer, 2002.
- [97] Richard Zach. Proof theory of finite-valued logics. Technical Report TUW-E185.2-Z.1-93, Institut Für Computersprachen, Technische Universität Wien, 1993.

[98] Anna Zamansky and Arnon Avron. Canonical signed calculi with multi-ary quantifiers. Annals of Pure and Applied Logic, 163(7):951 – 960, 2012.